AS FILED WITH THE SECURITIES AND EXCHANGE COMMISSION ON APRIL 16, 2001

REGISTRATION STATEMENT NO. 333-

SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM S-4 REGISTRATION STATEMENT UNDER THE SECURITIES ACT OF 1933

SPX CORPORATION

(EXACT NAME OF REGISTRANT AS SPECIFIED IN ITS CHARTER)

DELAWARE

3429 (STATE OR OTHER JURISDICTION OF (PRIMARY STANDARD INDUSTRIAL INCORPORATION OR ORGANIZATION) CLASSIFICATION CODE NUMBER)

38-1016240 (I.R.S. EMPLOYER IDENTIFICATION NO.)

700 TERRACE POINT DRIVE MUSKEGON, MICHIGAN 49443-3301 (231) 724-5000

(ADDRESS, INCLUDING ZIP CODE, AND TELEPHONE NUMBER, INCLUDING AREA CODE, OF REGISTRANT'S PRINCIPAL EXECUTIVE OFFICES)

CHRISTOPHER J. KEARNEY, ESQ. VICE PRESIDENT, SECRETARY AND GENERAL COUNSEL SPX CORPORATION 700 TERRACE POINT DRIVE MUSKEGON, MICHIGAN 49443-3301

(231) 724-5000 (NAME, ADDRESS, INCLUDING ZIP CODE, AND TELEPHONE, INCLUDING AREA CODE, OF AGENT FOR SERVICE)

WITH COPIES TO:

GEORGE C. MCKANN, ESQ. TROY M. CALKINS, ESQ. GARDNER, CARTON & DOUGLAS 321 NORTH CLARK STREET, SUITE 2900 CHICAGO, ILLINOIS 60610-4795 (312) 245-8853

THOMAS G. APPLEMAN, ESQ. MILLER, CANFIELD, PADDOCK AND STONE, P.L.C. 840 WEST LONG LAKE ROAD, SUITE 200 TROY, MICHIGAN 48098-6358 (248) 879-2000

Approximate date of commencement of proposed sale of the securities to the public: As soon as practicable after this Registration Statement becomes effective and all other conditions to the merger of VSI Holdings, Inc. with and into SPX Corporation pursuant to the Agreement and Plan of Merger, dated as of March 24, 2001, between those two parties, described in the enclosed proxy

statement-prospectus, have been satisfied or waived.

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, please check the following box. []

If this form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective

registration statement for the same offering. []

If this form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. []

CALCULATION OF REGISTRATION FEE

AMOUNT MAXIMUM PROPOSED MAXIMUM AMOUNT OF TITLE OF EACH CLASS OF SECURITIES TO BE OFFERING PRICE AGGREGATE REGISTRATIC BEING REGISTERED PER UNIT OFFERING PRICE FEE REGISTRATION Common Stock, \$10.00 par value(1)... 1,443,286(2) \$96.19(3) \$138,821,122(3) \$34,706(3)

- (1) Associated with the common stock of SPX are rights to purchase Series A Junior Participating Preferred Stock of SPX that will not be exercisable or evidenced separately from the common stock of SPX prior to the occurrence of certain events.
- (2) Represents the maximum number of SPX shares issuable assuming that all VSI

Holdings shareholders elect 100% stock consideration and based on a conversion ratio of 0.043 and assuming the exercise of all VSI Holdings options with an exercise price below \$4.35.

(3) Pursuant to Rules 457(f)(1) and 457(c) of the Securities Act of 1933, as amended, and solely for purposes of calculating the registration fee, the registration fee was computed on the basis of a proposed maximum aggregate offering price equal to the average of the high and low prices of VSI Holdings common stock as reported on the American Stock Exchange, Inc. on April 11, 2001 multiplied by the sum of the number of VSI Holdings shares outstanding on that date and the number of VSI Holdings shares issuable upon exercise of options with an exercise price below \$4.35.

THE REGISTRANT HEREBY AMENDS THIS REGISTRATION STATEMENT ON SUCH DATE OR DATES AS MAY BE NECESSARY TO DELAY ITS EFFECTIVE DATE UNTIL THE REGISTRANT SHALL FILE A FURTHER AMENDMENT WHICH SPECIFICALLY STATES THAT THIS REGISTRATION STATEMENT SHALL THEREAFTER BECOME EFFECTIVE IN ACCORDANCE WITH SECTION 8(A) OF THE SECURITIES ACT OF 1933 OR UNTIL THIS REGISTRATION STATEMENT SHALL BECOME EFFECTIVE ON SUCH DATE AS THE COMMISSION, ACTING PURSUANT TO SAID SECTION 8(A), MAY DETERMINE.

- ------

THE INFORMATION IN THIS PROXY STATEMENT-PROSPECTUS IS NOT COMPLETE AND MAY BE CHANGED. SPX MAY NOT SELL THESE SECURITIES UNTIL THE REGISTRATION STATEMENT FILED WITH THE SECURITIES AND EXCHANGE COMMISSION IS EFFECTIVE. THIS PROXY STATEMENT-PROSPECTUS IS NOT AN OFFER TO SELL THESE SECURITIES AND IT IS NOT SOLICITING AN OFFER TO BUY THESE SECURITIES IN ANY STATE WHERE THE OFFER OR SALE IS NOT PERMITTED.

SUBJECT TO COMPLETION -- DATED APRIL 16, 2001

[VSI HOLDINGS LOGO]

PROXY STATEMENT PROSPECTUS

Dear Fellow VSI Holdings Shareholder:

You are cordially invited to attend a special meeting of the shareholders of VSI Holdings, Inc. to be held on , 2001 at [10:00] a.m., Eastern Daylight Savings time, at its Vision Center at 1664 Star Batt Drive, Rochester Hills, Michigan 48309. At this special meeting, you will be asked to consider and vote upon an Agreement and Plan of Merger, dated March 24, 2001, by and between VSI Holdings and SPX Corporation. If the merger is completed, then, subject to the limitations and proration procedures described in the proxy statement-prospectus, you will receive for all of your shares of VSI Holdings common stock your choice of either (1) 0.043 of a share of SPX common stock per share of VSI Holdings common stock, or (2) a combination of a cash payment of \$4.35 per share of VSI Holdings common stock for 45% of your shares and 0.043 of a share of SPX common stock per share of VSI Holdings common stock for 55% of your shares.

SPX common stock is traded on the New York Stock Exchange and the Pacific Exchange under the symbol "SPW". As of April 12, 2001, SPX's stock price was \$96.31. SPX has registered 1,443,286 shares for issuance in connection with the merger and may issue some or all of those shares.

The merger cannot be completed unless VSI Holdings' shareholders approve the merger agreement. We have scheduled a special meeting of VSI Holdings shareholders to vote on the merger agreement. If you were a shareholder of record on , 2001, you may vote at the meeting. Whether or not you plan to attend, please take the time to vote by completing and mailing the enclosed proxy card to us.

This proxy statement-prospectus provides you with detailed information about the merger. This document also serves as the prospectus of SPX for any common stock that SPX issues in the merger. We encourage you to read this entire document carefully. The board of directors of VSI Holdings believes that the merger is in the best interests of VSI Holdings' shareholders, unanimously adopted the merger and recommends that you vote in favor of the proposal to approve the merger agreement.

Included with this proxy statement-prospectus are copies of VSI Holdings' annual report on Form 10-K for the year ended September 30, 2000 and quarterly report on Form 10-Q for the quarter ended December 31, 2000.

Sincerely yours,

Steve Toth, Jr., Chairman and Chief Executive Officer VSI Holdings, Inc.

PLEASE SEE THE SECTION ENTITLED RISK FACTORS BEGINNING ON PAGE 14 FOR A DISCUSSION OF POTENTIAL RISKS INVOLVED IN THE MERGER AND IN OWNING SPX COMMON STOCK.

NEITHER THE SECURITIES AND EXCHANGE COMMISSION NOR ANY OTHER REGULATORY BODY HAS APPROVED OR DISAPPROVED OF THESE SECURITIES OR PASSED UPON THE ACCURACY OR ADEQUACY OF THIS PROSPECTUS. ANY REPRESENTATION TO THE CONTRARY IS A CRIMINAL OFFENSE.

This proxy statement-prospectus is dated $$\rm , 2001$$ and was first mailed to shareholders on or about $$\rm , 2001.$

SUBJECT TO COMPLETION -- DATED APRIL 16, 2001

VSI HOLDINGS, INC. 41000 WOODWARD AVENUE BLOOMFIELD HILLS, MICHIGAN 48304-2263

NOTICE OF SPECIAL MEETING OF SHAREHOLDERS TO BE HELD , 2001

A special meeting of the shareholders of VSI Holdings, Inc., a Georgia corporation, will be held on , 2001 at [10:00] a.m., Eastern Daylight Savings time, at its Vision Center at 1664 Star Batt Drive, Rochester Hills, Michigan 48309 to consider and act upon:

- (1) A proposal to approve and adopt an Agreement and Plan of Merger dated as of March 24, 2001 by and between SPX Corporation, a Delaware corporation, and VSI Holdings, pursuant to which VSI Holdings will merge with and into SPX.
- (2) Any other matter that is properly brought before the special meeting or any adjournment or postponement of the special meeting.

The attached proxy statement-prospectus contains a more complete description of the merger agreement and the transactions that it contemplates. We urge you to read this document carefully.

Under the Georgia Business Corporation Code, VSI Holdings shareholders are not entitled to assert dissenters' rights in connection with the merger.

Only the shareholders of record at the close of business on , 2001 are entitled to notice of, and to vote at, the special meeting and any adjournment or postponement of the meeting. A list of these shareholders will be available for inspection by any shareholders (or shareholder's agent or attorney) at the meeting.

We cannot complete the merger unless our shareholders approve the merger agreement. The affirmative vote of holders of a majority of the shares of VSI Holdings outstanding and entitled to vote is required to approve the merger agreement. It is important that your shares be represented at the meeting, regardless of the number of shares you hold. Please complete and return your proxy card whether or not you plan to attend the meeting.

By order of the Board of Directors

Thomas W. Marquis Treasurer and Secretary

Bloomfield Hills, Michigan , 2001

THE VSI HOLDINGS BOARD OF DIRECTORS HAS UNANIMOUSLY ADOPTED THE MERGER AGREEMENT AND RECOMMENDS THAT THE SHAREHOLDERS OF VSI HOLDINGS VOTE FOR ITS APPROVAL.

TABLE OF CONTENTS

	PAGE
Questions and Answers About the Merger	1
Special Note Regarding Forward-Looking Statements Summary	4 4
Comparative Market Prices	7
Selected Historical Consolidated Financial Data of SPX	8
Selected Historical Consolidated Financial Data of VSI	Ū
Holdings	10
Summary Unaudited Pro Forma Combined Condensed Financial	
Data For	
SPX and United Dominion	11
Comparative Historical and Pro Forma Per Share Data Risk Factors	12 14
Risk Factors Relating to the Merger	14
Risk Factors Relating to SPX's Business	16
The Special Meeting of VSI Holdings Shareholders	21
Date, Time and Place	21
Purpose of the Meeting	21
Record Date; Quorum	21
Required Vote	21
Revocation of Proxies	21
Solicitation of Proxies	22
The MergerBackground of the Merger	23 23
Reasons for the Merger	23 26
Interests of Directors and Officers in the Merger	28
VSI Holdings Board Approval and Recommendation of the	
Merger	30
The Merger; Effective Time	30
Merger Consideration	31
Election of Form of Consideration; Conversion of Shares	31
Exchange Agent; Procedures for Exchange of Certificates;	
Fractional Shares	32
Treatment of VSI Holdings Stock Options; Restricted	00
Stock	33
Governance Following the MergerRepresentations and Warranties	33 33
VSI Holdings' Conduct of the Business Before Completion of	33
the Merger	34
Conditions to Completion of the Merger	35
Regulatory Approvals	35
Termination of the Merger	36
Effect of Termination	36
Indemnification and Insurance	37
Indemnity and Restriction Agreement	38
Opinion of VSI Holdings' Financial Advisor	38
No Dissenters' Rights	42 42
Certain Federal Income Tax Consequences	43
Resale of SPX Common Stock Issued in the Merger;	43
Affiliates	45
Asset Purchase Agreement	45
Advanced Animations Merger Agreement	46
Comparative Per Share Prices	47
Dividend Policy	47
Information Concerning SPX	48
Information Concerning VSI Holdings	50
Comparison of Rights of SPX Stockholders and VSI Holdings Shareholders	50
SHOLEHOTHELD	30

	PAGE
Legal Matters	
Where You Can Find More Information	
Appendix A Merger Agreement	
Appendix B Indemnity and Restriction Agreement Appendix C Opinion of VST Holdings' Financial Adviser	

REFERENCES TO ADDITIONAL INFORMATION

THIS DOCUMENT INCORPORATES IMPORTANT BUSINESS AND FINANCIAL INFORMATION ABOUT VSI HOLDINGS THAT IS INCLUDED IN VSI HOLDINGS' ANNUAL REPORT ON FORM 10-K FOR THE YEAR ENDED SEPTEMBER 30, 2000 AND QUARTERLY REPORT ON FORM 10-Q FOR THE QUARTER ENDED DECEMBER 31, 2000 THAT ARE BEING DELIVERED TO YOU WITH THIS DOCUMENT. IT ALSO INCORPORATES IMPORTANT BUSINESS AND FINANCIAL INFORMATION ABOUT SPX THAT IS NOT INCLUDED IN OR DELIVERED WITH THIS DOCUMENT. YOU CAN OBTAIN COPIES OF THE SPX DOCUMENTS INCORPORATED BY REFERENCE INTO THIS PROXY STATEMENT-PROSPECTUS, AS WELL AS COPIES OF THE VSI HOLDINGS DOCUMENTS INCLUDED WITH OR INCORPORATED BY REFERENCE INTO THIS PROXY STATEMENT-PROSPECTUS, BY ACCESSING THE SECURITIES AND EXCHANGE COMMISSION'S WEBSITE MAINTAINED AT HTTP://www.sec.gov. you can also obtain copies of those documents by requesting THEM IN WRITING OR BY TELEPHONE FROM THE APPROPRIATE PARTY AT THE FOLLOWING ADDRESSES:

SPX CORPORATION
700 TERRACE POINT DRIVE
MUSKEGON, MICHIGAN 49443
ATTENTION: INVESTOR RELATIONS
(231) 724-5000

VSI HOLDINGS, INC. 41000 WOODWARD AVENUE BLOOMFIELD HILLS, MICHIGAN 48304 ATTENTION: INVESTOR RELATIONS (248) 644-0500

IN ORDER TO RECEIVE DELIVERY OF THE DOCUMENTS IN ADVANCE OF THE SPECIAL MEETING, YOU SHOULD MAKE YOUR REQUEST NO LATER THAN , 2001.

See Where You Can Find More Information on pages 61 through 63.

QUESTIONS AND ANSWERS ABOUT THE MERGER

WHAT IS THE PROPOSED MERGER?

The boards of directors of SPX and VSI Holdings have voted to merge VSI Holdings into SPX. The reasons for the merger are discussed below under The Merger -- Reasons for the Merger. The merger agreement is attached to this proxy statement-prospectus as Appendix A. We encourage you to read it carefully.

WHO MUST APPROVE THE MERGER?

A majority of the holders of outstanding shares of VSI Holdings common stock must approve the merger agreement. SPX stockholder approval of the merger agreement is not required.

WHAT WILL HAPPEN TO MY VSI HOLDINGS STOCK IN THE MERGER?

You can elect to receive either all SPX common stock or a combination of SPX common stock and cash in exchange for your VSI Holdings common stock. The amount of stock or cash that you receive will be based on an exchange ratio of 0.043 shares of SPX common stock or \$4.35 per share of VSI Holdings common stock. The ratio of 0.043 shares of SPX common stock per share of VSI Holdings common stock was determined at the time of the signing of the merger agreement and was based on a price of \$101.17 per share of SPX common stock, which represented the average of the trading price of the SPX common stock over a period of 20 trading days ending on the fifth trading day prior to the signing. The last reported sale price of SPX common stock on April 12, 2001 was \$96.31 per share.

If you hold VSI Holdings stock options when the merger becomes effective, you will automatically receive options exercisable for shares of SPX common stock. Your replacement options will have substantially the same terms and conditions as your VSI Holdings options, except that the per share exercise price and the number of shares issuable to you upon the exercise of the options will be adjusted based on the exchange ratio of 0.043 shares of SPX stock per VSI Holdings share. Unvested VSI Holdings options will become unvested SPX options that will vest in accordance with the original vesting schedule.

Any shares of restricted VSI Holdings stock you hold will become fully vested as of the effective time of the merger and will be treated the same as any other shares of VSI Holdings common stock in the merger.

IF I CHOOSE THE COMBINED CASH AND STOCK ELECTION, WILL I BE ASSURED OF RECEIVING 45% CASH AND 55% STOCK?

No. The amount of VSI Holdings common stock converted to cash in the merger is subject to two limitations. First, no more than 45% of the outstanding VSI Holdings shares may be converted to cash. Second, the amount of cash paid by SPX upon conversion of VSI Holdings common stock plus the amount of VSI Holdings affiliated debt assumed by SPX cannot exceed a limit imposed by the merger agreement. This limit is based upon guidelines established by the Internal Revenue Service for issuing letter rulings pertaining to whether a transaction qualifies as a reorganization for federal income tax purposes. If this limit is exceeded, the merger could be treated as a taxable event for VSI Holdings shareholders. Therefore, if VSI Holdings shareholders elect the combined stock and cash consideration for a number of shares that would result in this limit being exceeded, SPX will, on a pro rata basis, reduce the amount of cash consideration and increase the amount of stock consideration for those VSI Holdings shareholders electing the combination consideration.

IF I CHOOSE THE STOCK ELECTION, WILL I BE ASSURED OF RECEIVING ONLY STOCK?

Yes, other than cash in exchange for any fractional shares. SPX may pay the full amount of the merger consideration in stock. That means all VSI Holdings shareholders who elect to receive only SPX stock in exchange for their VSI Holdings stock will receive stock. Based on the exchange ratio, however, it is possible for a fractional share to remain after the exchange. SPX will not issue fractional shares.

Instead, you will receive a cash payment equal to \$101.17 times the fraction of an SPX share to which your would otherwise be entitled.

IF THE TRADING VALUE OF SPX SHARES INCREASES OR DECREASES WILL THE AMOUNT OF SPX COMMON STOCK I RECEIVE CHANGE?

No, the amount of SPX common stock that you will receive for your VSI Holdings shares that are converted into SPX shares will not change as a result of changes in the price of SPX shares between now and closing. The exchange ratio of 0.043 SPX shares per VSI Holdings share is fixed and was based on an SPX share price of \$101.17, which represented the average trading price of SPX shares for a period prior to the signing of the merger agreement. This exchange ratio will not be altered as a result of change in the price of SPX shares prior to closing. However, the price of SPX shares on the closing day could affect the proportions of cash and stock that you will receive if you elect the combination consideration of cash and SPX shares as described below under The Merger -- Election of Form of Consideration; Conversion of Shares.

WHEN AND HOW DO I MAKE MY ELECTION?

A form of election and letter of transmittal is enclosed with this proxy statement-prospectus. It provides instructions for electing the type of consideration you would like to receive in the merger and for transmitting your VSI Holdings stock certificates. If you became a shareholder of record after , 2001, the record date, you can get a form of election and letter of transmittal from EquiServe Trust Company, N.A., the exchange agent, by calling

You may only make your election by delivering the form of election and letter of transmittal and other required materials discussed in the form of election and letter of transmittal. The exchange agent must receive your election by the deadline of 5:00 p.m. (Eastern Daylight Savings time) on , 2001 (or, if the meeting of VSI Holdings shareholders to approve the merger is postponed, two business days before the date of the meeting). If your election is not received by the deadline and the merger is effected, you will automatically receive stock consideration in exchange for all of your VSI Holdings shares.

CAN I VOTE AGAINST THE MERGER AND STILL MAKE AN ELECTION?

Regardless of how you intend to vote with respect to the merger agreement, we urge you to complete and sign the form of election and letter of transmittal and return it along with the other required materials. If you do not make an election and the merger is completed, you will automatically receive stock consideration in exchange for all of your VSI Holdings shares.

CAN I CHANGE MY ELECTION?

Yes. You can change your election by giving written notice to the exchange agent prior to , 2001 or by withdrawing your VSI Holdings shares (or withdrawing your guarantee of delivery of your shares) prior to that deadline. After the deadline, you may not change your election as to the consideration you wish to receive in the merger.

WHEN WILL THE MERGER TAKE EFFECT?

We are working toward completing the merger as quickly as possible. In addition to your approval, we must obtain regulatory approvals, and the parties must comply with the closing conditions specified in the merger agreement. We expect to complete the merger immediately after the conclusion of the VSI Holdings shareholders meeting.

ARE THERE RISKS THAT I SHOULD CONSIDER IN DECIDING WHETHER TO VOTE FOR THE MEDGED?

Yes. In evaluating the merger, you should carefully consider the information discussed in the section entitled Risk Factors on page 14.

WHAT DO I NEED TO DO NOW IN ORDER TO VOTE ON THE MERGER?

After you have carefully read and considered the information in this proxy statement-prospectus, you should complete, sign and mail the enclosed proxy card in the enclosed return envelope as soon as possible so that your shares may be represented at the special meeting. If you do not mark your vote on the proxy card, your shares will be voted for approval of the merger.

IF MY BROKER HOLDS MY VSI HOLDINGS SHARES IN "STREET NAME", WILL MY BROKER AUTOMATICALLY VOTE MY SHARES FOR ME?

No. Your broker can only vote your shares if you provide instructions on how to vote. To instruct your broker to vote your shares you must follow the instructions that were provided by your broker and enclosed with this proxy statement-prospectus. You must also instruct your broker as to the making of an election pursuant to the form of election and letter of transmittal.

WHAT DO I DO IF I WANT TO CHANGE MY VOTE AFTER I HAVE MAILED IN A SIGNED PROXY CARD?

You can change your vote in one of the following ways at any time before your proxy is voted at the special meeting. First, you can revoke your proxy by written notice sent to the secretary of VSI Holdings at the address on the notice of the meeting and received prior to the meeting. Second, you can submit a new, later dated proxy card. Third, you can attend the meeting and vote in person. If you instructed a broker to vote your shares, you must follow your broker's directions for changing those instructions.

DO I HAVE DISSENTERS' APPRAISAL RIGHTS?

No. Under Georgia law, VSI Holdings shareholders are not entitled to dissenters' rights of appraisal in connection with the merger.

WHO CAN HELP ANSWER MY OUESTIONS?

If you have any questions about the merger or if you need additional copies of this proxy statement-prospectus, the enclosed proxy card or the form of election and letter of transmittal and related materials, you should contact: the exchange agent, EquiServe Trust Company, N.A. at

WHERE CAN I FIND MORE INFORMATION ABOUT SPX AND VSI HOLDINGS?

You can find information about SPX and VSI Holdings in addition to what is included in this proxy statement-prospectus from various sources described under Where You Can Find More Information.

SPECIAL NOTE REGARDING FORWARD-LOOKING STATEMENTS

This proxy statement-prospectus and the documents that are made part of this proxy statement-prospectus by reference to other documents filed with the SEC include various forward-looking statements about SPX and VSI Holdings that are subject to risks and uncertainties, including those described in Risk Factors beginning on page 14. Forward-looking statements include the information concerning future financial performance, business strategy, projected plans, reasons for the merger and objectives of SPX and VSI Holdings.

SUMMARY

The following summary highlights selected information from this proxy statement-prospectus and does not necessarily contain all of the information that is important to you. To understand the merger fully and for a more complete description of the legal terms of the merger, you should carefully read this entire document, including the appendices, and the documents to which we have referred you. See Where You Can Find More Information. References in this document to SPX and VSI Holdings include their subsidiaries unless we have indicated otherwise.

SPX CORPORATION

700 Terrace Point Drive, Muskegon, Michigan 49443, (231) 724-5000

SPX is a global provider of technical products and systems, industrial products and services, service solutions and vehicle components. Its products include networking and switching products, fire detection and building life-safety products, TV and radio broadcast antennas and towers, transformers, substations and industrial mixers and valves. SPX's products and services also include specialty service tools, diagnostic systems, service equipment, technical information services and vehicle components. With over 14,000 employees worldwide, SPX is a multinational corporation with operations in 19 countries. SPX common stock is traded on the NYSE and the Pacific Exchange under the symbol "SPW".

On March 12, 2001, SPX announced that it had entered into an agreement for its wholly owned subsidiary to acquire United Dominion Industries Limited in an all-stock transaction. SPX will issue approximately 9.2 million shares to acquire United Dominion. SPX will also assume approximately \$876 million in United Dominion debt. United Dominion's financial statements and pro forma SPX financial statements reflecting the impact of the acquisition of United Dominion on SPX are included in a Current Report on Form 8-K filed by SPX on April 13, 2001, which is incorporated by reference into this proxy statement-prospectus.

VSI HOLDINGS, INC.

41000 Woodward Avenue, Bloomfield Hills, Michigan 48304, (248) 644-0500

VSI Holdings is a leading full service provider of marketing services and customer relationship management, organizational training and development services, and entertaining and educational animatronic displays and other entertainment/edutainment products. VSI Holdings markets its services to the automotive and pharmaceuticals industries and markets its entertainment/edutainment products primarily to theme parks, casinos and museums.

THE MERGER AGREEMENT

The merger agreement is attached as Appendix A to this proxy statement-prospectus. We encourage you to read the merger agreement. It is the principal document governing the merger.

RECOMMENDATION OF THE BOARD

The VSI Holdings board of directors has unanimously determined that the merger is in the best interests of VSI Holdings and its shareholders and recommends that you vote FOR the approval of the merger agreement.

OPINION OF VSI HOLDINGS' FINANCIAL ADVISOR

McDonald Investments, as financial advisor to VSI Holdings' board of directors, has delivered to the board its written opinion dated March 30, 2001 stating that, based upon and subject to the various qualifications and assumptions described in the opinion, the consideration that VSI Holdings shareholders would receive in the merger was fair to the shareholders from a financial point of view. The full text of the opinion is attached as Appendix C to this proxy statement-prospectus, and we encourage you to read it carefully in its entirety. The opinion of McDonald Investments is directed to the VSI Holdings board of directors and does not constitute a recommendation to any shareholder as to how to vote with respect to the merger.

REASONS FOR THE MERGER

VSI Holdings' board of directors believes that VSI Holdings will become more competitive if it combines with SPX and becomes part of a larger, more significantly diversified and capitalized entity.

The VSI Holdings board and management anticipate that being part of such a larger entity will provide VSI Holdings with a global scope and the capabilities to more effectively meet customer demands in an industry where size is becoming an increasingly more significant factor. Indeed, the board and management expect the merger to provide VSI Holdings with the potential to undertake significantly larger projects not currently available to it because of its smaller capitalization. Consequently, VSI Holdings' board of directors believes that following the combination with SPX, VSI Holdings will have the potential to realize greater long-term positive operating and financial results than it would on a stand-alone basis. The merger also provides VSI Holdings shareholders with a premium over the price at which VSI Holdings common stock was trading prior to the announcement of the merger and with greater liquidity.

For SPX, the merger with VSI Holdings complements the strategy of its Service Solutions segment. A key component of this strategy is to provide a comprehensive portfolio of professional services to SPX's customer base. The acquisition of VSI Holdings will broaden the services offered by SPX's Service Solutions segment and allow the segment to market its services to a wider range of customers.

THE SPECIAL MEETING

The VSI Holdings special meeting will be held on , 2001 at [10:00] a.m., Eastern Daylight Savings time, at its Vision Center at 1664 Star Batt Drive, Rochester Hills, Michigan 48309. At the meeting, VSI Holdings shareholders will be asked to approve the merger agreement.

VOTING RIGHTS; APPROVAL OF THE MERGER

You are entitled to vote at the special meeting if you owned VSI Holdings shares as of the close of business on the record date of , 2001. On the record date, there were VSI Holdings common shares entitled to vote at the meeting. Each share entitles its holder to one vote.

A majority of the holders of outstanding VSI Holdings common stock must approve the merger agreement. If you do not vote or instruct your broker how to vote your shares, it will have the same effect as voting against the merger. SPX stockholders are not required to approve the merger.

SHARE OWNERSHIP OF MANAGEMENT

As of December 31, 2000, VSI Holdings directors, executive officers and their affiliates owned and were entitled to vote 29,886,937 VSI Holdings shares, representing approximately 90% of the outstanding voting power. The directors and executive officers have expressed an intention to vote in favor of the merger agreement.

INTERESTS OF DIRECTORS AND OFFICERS IN THE MERGER

When you consider the recommendation of VSI Holdings' board of directors that you vote in favor of the merger, you should be aware that a number of VSI Holdings directors and executive officers may have interests in the merger that are different from, or in addition to, your interests as a shareholder.

These interests may arise from stock options, employment arrangements, indemnity and insurance arrangements, and provisions in the merger agreement. They also may arise from the indemnity and restriction agreement, the asset purchase agreement and the Advanced Animations merger agreement. For a description of these interests, you should refer to The Merger -- Interests of Directors and Officers in the Merger, -- Indemnity and Restriction Agreement, -- Asset Purchase Agreement and -- Advanced Animations Merger Agreement below.

INDEMNITY AND RESTRICTION AGREEMENT

Steve Toth, Jr., who is VSI Holdings' Chairman and CEO and the indirect owner of a majority of VSI Holdings' outstanding shares, entered into an indemnity and restriction agreement when the parties executed the merger agreement. The agreement is attached to this proxy statement-prospectus as Appendix B, and we urge you to read it carefully. Under the agreement, Mr. Toth agreed to reimburse SPX for specific losses, fees and expenses that SPX or VSI Holdings may incur as a result of certain contingencies identified in the agreement. Mr. Toth's obligations will be secured by some of the SPX shares that he will receive in the merger. In addition, the agreement will restrict the ability of Mr. Toth, certain trusts for the benefit of Mr. Toth, his wife and their daughter and a related entity to dispose of their SPX shares received in the merger. Mr. Toth also agreed not to compete with SPX or solicit employees or customers of SPX for a period of five years.

ASSET PURCHASE AGREEMENT

Steve Toth, Jr. and PII Ventures, L.L.C., an entity owned and controlled (directly and indirectly) by Mr. Toth, also entered into an asset purchase agreement with SPX when the parties executed the merger agreement. Under the asset purchase agreement, PII Ventures agreed to purchase certain assets and assume certain liabilities from SPX following the closing of the merger. The assets are currently held directly or indirectly by VSI Holdings but, following the merger, will be held directly or indirectly by SPX. As consideration for the assets, PII Ventures will pay SPX \$3,690,000. SPX is selling the assets to PII Ventures on an "as is, where is" basis, without any representations, warranties or guaranties with respect to fitness, merchantability or otherwise.

ADVANCED ANIMATIONS MERGER AGREEMENT

In accordance with the terms of the asset purchase agreement, soon after the merger and simultaneously with the closing of the asset purchase agreement, SPX and PII Ventures will execute and deliver a merger agreement providing for the merger of Advanced Animations, Inc. with and into PII Ventures. PII Ventures is owned and controlled (directly or indirectly) by Mr. Toth. Advanced Animations is currently a wholly owned subsidiary of VSI Holdings, but following the merger will be wholly owned by SPX. Under the Advanced Animations merger agreement, Advanced Animations will be sold by SPX to, and merged with and into, PII Ventures, which will result in the ownership by PII Ventures of the assets and liabilities of Advanced Animations. As consideration for the merger, PII Ventures will pay SPX \$16,750,000, which amount was negotiated by SPX and Mr. Toth. SPX is selling Advanced Animations to PII Ventures on an "as is, where is" basis, without any representations, warranties or guaranties with respect to fitness, merchantability or otherwise.

COMPARTSON OF SHAREHOLDER RIGHTS

When the merger is completed, VSI Holdings shareholders will become stockholders of SPX. Unlike VSI Holdings, SPX is a Delaware corporation, and the certificate of incorporation and bylaws of SPX differ from those of VSI Holdings. As a result, you will have different rights as a SPX stockholder than you currently have as a VSI Holdings shareholder. For a comparison of the rights of VSI Holdings

shareholders under Georgia law and SPX stockholders under Delaware law, you should refer to Comparison of Rights of SPX Stockholders and VSI Holdings Shareholders below.

CONDITIONS TO COMPLETION OF THE MERGER; TERMINATION OF THE MERGER

In addition to approval of VSI Holdings shareholders, the consummation of the merger is conditioned upon the fulfillment of various other conditions described in the merger agreement. SPX and VSI Holdings may terminate the merger agreement by mutual consent before or after the approval of VSI Holdings' shareholders. In addition, either SPX or VSI Holdings may terminate the merger agreement if conditions specified in the merger agreement have not been satisfied. If the merger agreement is terminated under certain circumstances, or if VSI Holdings consummates or agrees to consummate various transactions specified in the merger agreement within six months of termination, VSI Holdings could be required to pay a \$9 million termination fee to SPX. You should refer to The Merger -- Conditions to Completion of the Merger and -- Termination of the Merger below for more information regarding the conditions that must be satisfied to complete the merger and the termination rights and fees.

FEDERAL INCOME TAX CONSEQUENCES

The parties have not sought, and do not intend to seek, a ruling from the Internal Revenue Service as to the anticipated federal income tax consequences of the merger. Consummation of the merger is conditioned upon SPX's receipt of an opinion of Gardner, Carton and Douglas, its counsel, and VSI Holdings' receipt of an opinion from Miller, Canfield, Paddock and Stone, P.L.C., its counsel, to the effect that, based upon certain facts, representations and assumptions, for federal income tax purposes the merger will constitute a reorganization under section 368(a) of the Internal Revenue Code of 1986, as

The tax consequences to you of the merger will depend on the facts of your particular situation. We urge you to consult your own tax advisor as to the specific tax consequences of the merger to you, including the applicable federal, state, local and foreign tax consequences. For more information, you should refer to The Merger -- Certain Federal Income Tax Consequences.

ACCOUNTING TREATMENT

SPX will account for the merger using the purchase method of accounting.

RESTRICTIONS ON RESALE OF SPX STOCK

You will be able to freely transfer the shares of SPX common stock that you receive in connection with the merger unless you are considered an affiliate of VSI Holdings under the Securities Act of 1933. VSI Holdings affiliates may only sell their SPX shares pursuant to a registration statement or Rule 145 or another exemption from the Securities Act. In addition, Mr. Toth's ability to dispose of his shares is restricted under the indemnity and restriction agreement discussed above. For more information you should refer to The Merger -- Resale of SPX Common Stock Issued in the Merger; Affiliates below.

COMPARATIVE MARKET PRICES

The following table presents the closing market prices for SPX and VSI Holdings common stock on March 23, 2001, the last full trading day prior to the announcement of the signing of the merger agreement, and as of April 12, 2001. The table also presents the equivalent value per share of VSI Holdings common stock, giving effect to the merger as of that date. The equivalent per share price of VSI Holdings common stock is based on a fixed exchange ratio of 0.043 SPX shares per VSI Holdings share. Because the market prices of SPX and VSI Holdings stock fluctuate from day to day, we urge you to obtain current market quotations to evaluate the merger.

	SPX	VSI HOLDINGS	VSI HOLDINGS EQUIVALENT VALUE
March 23, 2001	\$92.95	\$3.40	\$4.00
April 12, 2001	\$96.31	\$4.19	\$4.14

SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF SPX

The following selected historical consolidated historical financial data are derived from the audited consolidated financial statements of SPX contained in SPX's annual report on Form 10-K for the fiscal year ended December 31, 2000, which is incorporated by reference in this proxy statement-prospectus, and are qualified in their entirety by that document.

You should read the following data together with the financial information of SPX incorporated by reference in this proxy statement-prospectus. See Where You Can Find More Information.

		OF AND FOR TH			31,
	1996		1998	1999	2000
	(1	IN MILLIONS, E	XCEPT PER SH		5)
SUMMARY OF OPERATIONS(1)					
Revenues	\$2,065.0	\$1,954.6(9)	\$1,825.4	\$2,712.3	\$2,678.9
Operating income (loss)(2) Gain on issuance of Inrange stock	223.1	181.5	(39.5)	313.4	276.1
(3)					98.0
Other (expense) income, net (4)	20.8	72.7 11.8	(0.5) 40.2	64.3 34.7	22.2 34.3
Equity in earnings of EGS (5) Interest expense, net (6)	(21.5)	(13.2)	(45.1)	(117.6)	(95.0)
interest expense, het (0)	(21.3)	(13.2)	(43.1)	(117.0)	(93.0)
<pre>Income (loss) before income taxes</pre>	222.4	252.8	(44.9)	294.8	335.6
Income tax (expense) benefit	(89.0)	(121.8)	3.2		(137.3)
Income (loss) from continuing					
operations	133.4	131.0	(41.7)		198.3
Discontinued operation, net of tax		2.3			
Cumulative effect of accounting		(2.7)			
change (7) Extraordinary item, net of tax		(3.7)		(6.0)	(8.8)
Extraordinary Item, het or tax				(0.0)	(0.0)
Net income (loss)		\$ 129.6	\$ (41.7)	\$ 101.5	\$ 189.5
1100 11100 1100 (1000) 11111111111111111	=======	=======	=======	=======	=======
Income (loss) per share from continuing operations:					
Basic	\$ 6.41	\$ 6.23	\$ (1.94)		\$ 6.44
Diluted Weighted average number of common	6.25	6.22	(1.94)	3.46	6.25
shares outstanding:					
Basic	20.8	21.0	21.5	30.8	30.8
Diluted	21.9	21.1	21.5	31.1	31.8
Dividends paid OTHER FINANCIAL DATA:	47.6	51.7	820.7(8	3)	
Total assets	\$1,551.0	1,388.0	\$2,968.3	\$2,846.0	\$3,164.6
Total debt	206.9	216.4		1,114.7	1,295.6
Other long-term obligations	166.7	174.4	431.9	521.8	595.5
Shareholders' equity	743.8	629.7	390.5	552.3	608.2
Capital expenditures	59.3	56.5	69.2	102.0	123.3
Depreciation and amortization	60.2	65.2	60 /	105 /	110 0

69.2

65.3

69.4

105.4

110.9

Depreciation and amortization.....

⁽¹⁾ On October 6, 1998, SPX completed the merger of SPX and General Signal, which was accounted for as a reverse acquisition of SPX by General Signal. See Note 2 to the consolidated financial statements included in SPX's annual report on Form 10-K for the year ended December 31, 2000 for further discussion.

⁽²⁾ In 1996, SPX recorded a charge of \$13.7 for asset write-downs, lease termination costs, severance, warranty repairs, environmental matters, insurance recovery of destroyed assets, and a royalty settlement.

In 1997, SPX recorded \$27.9 of charges for asset valuations, restructuring charges, lease termination costs and other matters, offset by a \$10.0 gain on the settlement of patent litigation and the sale of related patents.

In the fourth quarter of 1998, SPX recorded special charges of \$101.7, which included \$69.3 of costs associated with closing the former General Signal corporate office and \$32.4 of restructuring costs related to General Signal operations. Additionally, SPX recorded \$108.2 of other one-time charges related to the General Signal merger and the restructuring.

In 1999, SPX recorded special charges of \$38.4 associated with restructuring actions initiated throughout the businesses.

SPX recorded special charges of \$90.9 in 2000 primarily associated with restructuring initiatives to consolidate manufacturing facilities, rationalize certain product lines and asset impairments. Additionally, SPX recorded a charge of \$12.3 against cost of goods sold for discontinued product lines associated with restructuring and other product changes primarily within the Service Solutions segment.

See Note 4 to the consolidated financial statements included in SPX's annual report on Form 10-K for the year ended December 31, 2000 for further discussion.

- (3) In 2000, Inrange Technologies, a subsidiary of SPX, issued 8,855,000 shares of its class B common stock for cash in an initial public offering. Accordingly, SPX recorded a \$98.0 pretax gain. See Note 5 to the consolidated financial statements included in SPX's annual report on Form 10-K for the year ended December 31, 2000 for further discussion.
- (4) In 1996, SPX recorded a \$29.0 gain on the sale of Kinney Vacuum Company.

In 1997, SPX recorded a \$63.7 gain on the sale of General Signal Power Group and a \$9.0 gain on the sale of an equity interest in a Mexican company.

In 1999, SPX recorded pretax gains of \$23.8 associated with the divestiture of Best Power and \$29.0 associated with the divestiture of Dual-Lite and an investment in a Japanese joint venture. Additionally, in 1999 SPX recorded a pretax gain of \$13.9 on the sale of marketable securities.

In 2000, SPX recorded a \$23.2 million pretax gain on the settlement of a patent infringement suit against American Power Conversion Corporation. See Note 15 to the consolidated financial statements included in SPX's annual report on Form 10-K for the year ended December 31, 2000 for further discussion.

- (5) These amounts represent SPX's share of the earnings of EGS Electrical Group LLC, a joint venture between SPX and Emerson Electric Co., formed during the third quarter of 1997. See Note 9 to the consolidated financial statements included in SPX's annual report on Form 10-K for the year ended December 31, 2000 for further discussion.
- (6) The increase in interest expense after 1997 relates to the General Signal merger. See Notes 2 and 13 to the consolidated financial statements included in SPX's annual report on Form 10-K for the year ended December 31, 2000 for further discussion.
- (7) In November 1997, the Emerging Issues Task Foce of the FASB issued consensus 97-13, "Accounting for costs Incurred in Connection with a Consulting Engagement or an Internal Project that Combines Business Process and Reengineering and Information Technology Transformation" (EITF 97-13). EITF 97-13 required all previously capitalized business process reengineering costs to be expensed as a cumulative effect of a change in accounting principle. SPX recorded a charge of \$3.7, net of tax, in connection with EITF 97-13 in the fourth quarter of 1997.
- (8) Includes \$784.2, which was the cash portion of the consideration paid in the General Signal merger in 1998 and which was accounted for as a special dividend.
- (9) During the third quarter of 1997, SPX sold General Signal Power Group and contributed substantially all of the assets of General Signal Electrical Group to EGS.

SELECTED HISTORICAL CONSOLIDATED FINANCIAL DATA OF VSI HOLDINGS

The following selected historical consolidated financial data are derived from the audited consolidated financial statements of VSI Holdings contained in VSI Holdings' annual report on Form 10-K for the fiscal year ended September 30, 2000 and from VSI Holdings' unaudited consolidated historical financial statements contained in its quarterly report on Form 10-Q for the period ended December 31, 2000, which are being delivered with and incorporated by reference in this proxy statement-prospectus, and are qualified in their entirety by those documents. VSI Holdings' operating results for the three months ended December 30, 2000 are not necessarily indicative of results for the full fiscal year ending September 30, 2001. VSI Holdings paid no dividends on its common stock during any of the periods presented and has no current plans to change that policy.

You should read the following data together with the financial information of VSI Holdings delivered with and incorporated by reference in this proxy statement-prospectus. See Where You Can Find More Information.

	AS (DF AND FOR THE Y	EAR ENDED	SEPTEMBER 30,		DECEMBE	DED
	1996(2)(4)	1997(2)(4)	1998(1)	1999(1)	2000(1)	1999	2000
		(IN MIL	LIONS, EXC	EPT PER SHARE	AMOUNTS)		
OPERATING DATA							
Net sales Income (loss) from continuing	\$128.2	\$130.5	\$163.2	\$143.4	\$187.3	\$ 38.1	\$ 43.3
operations	9.1	9.7	9.2	(0.5)(5)	5.6(5)	0.9	1.4
operations per share	0.18	0.20	0.28	(0.02)	0.17	0.03	0.04
Number of shares	32.8	32.8	32.9	32.8	33.1	32.8	33.2
BALANCE SHEET DATA							
Working capital	\$ 9.1	\$ (1.9)(3)	\$ 9.1	\$ 9.1	\$ 10.2	\$ 16.0	\$ 12.8
Total assets	66.9	77.0	89.6	98.3	116.1	97.9	110.4
Long-term debt	1.9	5.3	17.5	19.3	19.4	19.2	18.9
Redeemable common stock			2.0				
Total liabilities	36.8	65.4	69.4	76.0	88.6	72.7	83.2
Stockholders' equity	28.0	11.7(3)	20.2	22.3	27.6	25.2	27.2

- (1) September 30, 1998, 1999 and 2000 balance sheet data and operating data exclude BKNT Retail Stores, Inc. accounts.
- (2) September 30, 1996, and 1997 balance sheet data include the accounts of BKNT Retail Stores, Inc. The operating data excludes the operations of BKNT Retail Stores, Inc.
- (3) See Item 13 Declared Distributions to Stockholders in VSI Holdings' annual report on Form 10-K for the year ended September 30, 2000 regarding distributions of previously taxed undistributed earnings to the stockholders of acquired subsidiaries. This reduced working capital and stockholders' equity.
- (4) 1996 and 1997 operating data and earnings per share information are pro forma amounts and have been calculated as if the subsidiaries had been consolidated for both years.
- (5) The loss from continuing operations in 1999 includes charges to earnings of \$0.4 and \$2.2 related to the impairment of long-lived assets and a reduction in the value of goodwill, respectively. The income from continuing operations in 2000 includes a charge to earnings of \$0.5 related to the impairment of long-lived assets. See Notes 2 and 4 of Notes to consolidated financial statements included in VSI Holdings' annual report on Form 10-K for the year ended September 30, 2000 for further discussion.

SUMMARY UNAUDITED PRO FORMA COMBINED CONDENSED FINANCIAL DATA OF SPX AND UNITED DOMINION

The summary pro forma combined SPX and United Dominion financial data is presented as if this merger had occurred as of January 1, 2000 for statement of income data and on December 31, 2000 for balance sheet data. See the SPX current report on Form 8-K filed on April 13, 2001 Financial Statements, Pro Forma Financial Information and Exhibits for further information regarding the pro forma combined SPX and United Dominion financial data. The Form 8-K is incorporated by reference in this proxy statement-prospectus, which is qualified in its entirety by the Form 8-K.

SPX will account for the United Dominion acquisition using the purchase method of accounting. The estimated purchase price was allocated to the United Dominion assets and liabilities assumed based on SPX management's current estimate of their fair market values. The final allocation of the purchase price to the assets and liabilities assumed will not be completed until after that acquisition is completed and will include independent appraisals and the results of SPX's strategic review to determine fair values. SPX management believes the following preliminary allocations of the purchase price are reasonable based upon currently available information, but the allocations could change materially upon the completion of the appraisals and strategic review. United Dominion's financial position and results of operations will not be included in SPX's consolidated financial statements prior to the date that acquisition is completed.

The pro forma combined SPX and United Dominion financial data is intended for informational purposes only, and does not purport to represent what SPX's results of continuing operations or financial position would actually have been had the merger in fact occurred as of January 1, 2000 for statement of income and related data or December 31, 2000 for balance sheet data or to project the results for any future date or period. Upon consummation of the merger, the actual financial position and results of operations of SPX may differ, perhaps significantly, from the pro forma amounts reflected herein due to a variety of factors, including changes in operating results between the date of the pro forma financial information and the date of the United Dominion merger and thereafter, as well as the factors discussed under Risk Factors.

You should read the information presented in the following table in conjunction with the SPX historical financial statements and related notes, the United Dominion historical financial statements and related notes, and the Current Report on Form 8-K filed on April 13, 2001 by SPX that are incorporated by reference in this proxy statement-prospectus.

AS OF AND FOR THE
YEAR ENDED
DECEMBER 31, 2000

(UNAUDITED)
(IN MILLIONS, EXCEPT
PER SHARE AMOUNTS)

OPERATING RESULTS:

\$5,045.1 413.9 98.0 12.3 34.3 (191.5)
367.0 (165.0)
202.0
\$ 5.05 4.93
\$ 175.1 5,876.2 2,304.0 1,002.6 1,540.9 177.1 212.8

COMPARATIVE HISTORICAL AND UNAUDITED PRO FORMA PER SHARE DATA

The following table presents:

- Historical per share data for SPX and for VSI Holdings;
- Pro forma combined SPX/VSI Holdings per share data and related VSI Holdings equivalent per share data; and
- Pro forma combined SPX/United Dominion per share data and related VSI Holdings equivalent per share data.

The pro forma combined SPX/VSI Holdings per share data is presented as if the merger had occurred as of January 1, 2000 for statement of income data and on December 31, 2000 for the balance sheet data, assuming an exchange ratio of 0.043 shares of SPX's common stock for each share of VSI common stock and that 30% of the merger will be in cash at \$4.35 per share of common stock. This pro forma also excludes the operational results and assets associated with the asset purchase agreement and the Advanced Animations merger agreement, which will be sold by SPX at the completion of the merger.

The pro forma combined SPX/United Dominion per share data is presented as if this merger had occurred as of January 1, 2000 for statement of income data and on December 31, 2000 for balance sheet data. See the SPX current report on Form 8-K filed on April 13, 2001 Financial Statements, Pro Forma Financial Information and Exhibits for further information regarding the pro forma combined SPX/United Dominion per share data.

The pro forma combined SPX/VSI Holdings per share data, the pro forma combined SPX/United Dominion per share data, and the related VSI Holdings equivalents are intended for informational

purposes only, and do not purport to represent what SPX's results of continuing operations would actually have been had the mergers in fact occurred as of January 1, 2000 for statement of income data or December 31, 2000 for balance sheet data or to project the results for any future date or period. Upon consummation of these mergers, the actual financial position and results of operations of SPX may differ, perhaps significantly, from the pro forma amounts reflected herein due to a variety of factors, including changes in operating results between the date of the pro forma financial information and the respective dates of the two mergers and thereafter, as well as the factors discussed under Risk Factors.

You should read the information presented in the following tables in conjunction with the SPX historical financial statements and related notes, the VSI Holdings historical financial statements and related notes, and the Current Report on Form 8-K filed on April 13, 2001 by SPX that are delivered with or incorporated by reference in this proxy statement-prospectus.

			UNAUDITED PRO FORMA COMBINED SPX/VSI HOLDINGS		UNAUDITED PRO FORMA COMBINED SPX/UNITED DOMINION	
	HISTORICAL SPX	HISTORICAL VSI HOLDINGS	PRO FORMA	VSI HOLDINGS EQUIVALENT(2)	PRO FORMA	VSI HOLDINGS EQUIVALENT(2)
Fiscal year ended:(1) Income per share from continuing operations						
Basic	\$ 6.44	\$0.17	\$ 6.32	\$0.27	\$ 5.05	\$0.22
Diluted Dividends per	6.25	0.17	6.14	0.26	4.93	0.21
share(3)						
share(4)	\$20.06	\$0.83	\$22.48	\$0.97	\$38.98	\$1.68

- (1) Historical SPX per share information is as of December 31, 2000. Historical VSI Holdings per share information is as of September 30, 2000. Pro forma combined SPX/VSI Holdings per share information was computed using historical SPX information as of December 31, 2000 and historical VSI Holdings information as of September 30, 2000. Pro forma combined SPX/United Dominion per share information was computed using historical SPX information as of December 31, 2000 and historical United Dominion information as of December 31, 2000.
- (2) The table reflects the VSI Equivalent per share of VSI Holdings for each pro forma presented by multiplying the pro forma per share information by the exchange ratio of 0.043.
- (3) Neither SPX or VSI Holdings paid any dividends during the periods.
- (4) The book value per share is computed by dividing stockholders' equity by the number of shares of stock outstanding at fiscal year end. Historical SPX shares outstanding were 30,322,000 at December 31, 2000. Historical VSI Holdings shares outstanding were 33,180,000 at September 30, 2000. The pro forma combined SPX/VSI Holdings shares outstanding were 31,330,000 at December 31, 2000. The pro forma combined SPX/United Dominion shares outstanding were 39,530,000 at December 31, 2000.

RISK FACTORS

In addition to the other information included in this proxy statement-prospectus, including the matters addressed in Special Note Regarding Forward-Looking Statements, you should carefully consider the matters described below in determining whether you should vote in favor of the merger.

RISK FACTORS RELATING TO THE MERGER

THE VALUE OF THE CONSIDERATION THAT YOU RECEIVE IN THE MERGER MAY DEPEND ON THE PRICE OF SPX COMMON STOCK.

In recent years the stock market in general has experienced extreme price and volume fluctuations, which have often been unrelated to operating performance. These broad market fluctuations have in the past adversely affected, and may in the future adversely affect, the market price of SPX's common stock. Fluctuations in the price of SPX's common stock will affect the value of the consideration received by VSI Holdings shareholders in the merger.

Under the terms of the merger agreement, the stock consideration you will receive for your VSI Holdings shares is based on an exchange ratio of 0.043 shares of SPX common stock per share of VSI Holdings common stock. That exchange ratio was determined using a formula based, in part, on an average price for SPX common stock prior to the date the merger agreement was signed. There can be no assurance that the market price of SPX common stock on and after the effective time of the merger will be higher or lower than the price used in determining the exchange ratio or the price as of any other date. The SPX common stock price used for purposes of this formula was \$101.17. The last reported price of SPX common stock on April 12, 2001 was \$96.31. In addition, a decline in the price of SPX shares may have the effect of reducing the proportion of cash consideration paid by SPX for VSI Holdings shares. See The Merger -- Merger Consideration and -- Election of Form of Consideration; Conversion of Shares. VSI Holdings shareholders should obtain and consider recent trading prices of SPX common stock in determining whether to vote in favor of the merger agreement and the consummation of the merger as well as whether to elect all stock consideration or a combination of stock consideration and cash consideration.

YOU MAY NOT RECEIVE THE AMOUNT OF CASH THAT YOU ELECT TO RECEIVE IN THE MERGER.

The amount of VSI Holdings common stock converted to cash in the merger is subject to two limitations. First, no more than 45% of the outstanding VSI Holdings shares may be converted to cash. Second, the amount of cash paid by SPX upon conversion of VSI Holdings common stock plus the amount of VSI Holdings debt owed to affiliates assumed by SPX cannot exceed a limit imposed under federal tax law. If this limit is exceeded, the merger would be treated as a taxable event for VSI Holdings shareholders. Therefore, if VSI Holdings shareholders elect the combined stock and cash consideration for a number of shares that would result in the tax limit being exceeded, SPX will reduce the amount of cash consideration, and increase the amount of stock consideration for those VSI Holdings shareholders electing the combination consideration on a pro rata basis. See The Merger -- Election of Form of Consideration; Conversion of Shares. Therefore, it is possible that you may not receive the amount of cash that you request in your form of election in the merger. In addition, VSI Holdings shareholders who make no election as to the consideration they wish to receive on their form of election or who fail to make a valid and timely election in accordance with the designated procedures will receive only stock consideration under the terms of the merger agreement and will not be able to control to any extent the form of consideration they receive in the merger. However, SPX will pay cash in lieu of fractional shares of SPX common stock.

THE TERMINATION FEE COULD MAKE A COMPETING TAKEOVER PROPOSAL MORE DIFFICULT AND EXPENSIVE.

VSI Holdings must pay SPX a termination fee of \$9 million if the merger agreement terminates under specified circumstances. See The Merger -- Effect of Termination. The termination fee could discourage another company from making a competing takeover proposal that could be more advantageous

to VSI Holdings' shareholders by making the competing proposal more difficult or expensive, or it could deter VSI Holdings from entering into an alternative transaction.

THE INTEREST OF VSI HOLDINGS DIRECTORS AND OFFICERS IN THE MERGER MAY BE DIFFERENT FROM THE INTERESTS OF OTHER SHAREHOLDERS.

Some of the directors and officers of VSI Holdings have interests in the merger that are different from, and/or in addition to, the interests of VSI Holdings' shareholders. These interests may influence these directors and officers in making their recommendation that you vote in favor of the merger agreement. In that regard, see the discussion under The Merger -- Interests of Directors and Officers in the Merger, -- Indemnity and Restriction Agreement, -- Asset Purchase Agreement and -- Advanced Animations Merger Agreement.

THE PENDING MERGER COULD NEGATIVELY IMPACT VSI HOLDINGS' FUTURE BUSINESS OPERATIONS.

VSI Holdings customers may, in response to the pending merger, delay or defer purchasing decisions. Any delay or deferral in purchasing decisions by customers could have a material adverse effect on VSI Holdings' business, regardless of whether or not the merger is ultimately completed. Some current and prospective VSI Holdings employees may experience uncertainty about their future role within the combined companies. This may adversely affect VSI Holdings' ability to attract and retain key management, marketing, technical, sales and other personnel.

IF SPX AND VSI HOLDINGS FAIL TO COMPLETE THE MERGER, IT COULD NEGATIVELY IMPACT VSI HOLDINGS' STOCK PRICE AND FUTURE BUSINESS AND OPERATIONS.

If the merger is not completed for any reason, VSI Holdings may be subject to a number of material risks, including the following:

- VSI Holdings may, under certain circumstances, be required to pay SPX a termination fee of \$9 million;
- the price of VSI Holdings common stock may decline to the extent that the current market price of VSI Holdings common stock reflects an assumption that the merger will be completed; and
- VSI Holdings must pay its costs related to the merger, including legal and accounting fees and financial advisory expenses.

Further, if the merger is terminated and VSI Holdings' board of directors determines to seek another merger or business combination, it may not be able to find a partner willing to pay an equivalent or more attractive price than that which would have been paid in the merger.

IF MEMBERS OF VSI HOLDINGS' MANAGEMENT LEAVE AFTER THE MERGER, SPX MAY NOT REALIZE THE POTENTIAL BENEFITS OF THE MERGER.

Success of the merger will depend, in part, on the continued services of VSI Holdings' senior management personnel and of its key technical and sales personnel. If members of VSI Holdings' current management were to terminate their employment with VSI Holdings, the ability of SPX to manage VSI Holdings' business and workforce after the merger could be harmed and the operations of SPX's Service Solutions business could be disrupted. Although SPX has entered into employment agreements with several members of the VSI Holdings management team that will take effect upon completion of the merger, these agreements may not result in the retention of the members of the management of VSI Holdings for any significant period of time. A loss of any member of VSI Holdings' management could prevent or defer realization of the benefits that SPX anticipates from the merger.

THE MERGER COULD HARM KEY VSI HOLDINGS CUSTOMER AND THIRD-PARTY RELATIONSHIPS.

The proposed merger could harm relationships with customers and other third parties with whom VSI Holdings does business. For example, the continuance of VSI Holdings' customer relationships is generally based on continued contractual commitments. Customers might postpone or cancel sales orders for VSI Holdings' products and services if they perceive that customer service and support would decline as a result of the merger. Any changes in customer relationships could harm the combined company's business. VSI Holdings' customers and other third parties may delay or defer decisions concerning using VSI Holdings products. Any changes in strategic partner relationships could harm VSI Holdings' business and diminish the benefits that SPX anticipates from the merger.

THE PRICE OF SPX COMMON STOCK MAY BE AFFECTED BY FACTORS DIFFERENT FROM THOSE AFFECTING THE PRICE OF VSI HOLDINGS COMMON STOCK.

Upon completion of the merger, each holder of VSI Holdings common stock will become a holder of SPX common stock. SPX's business differs from that of VSI Holdings, and SPX's results of operations and the price of SPX common stock may be affected by factors different from those that affected VSI Holdings' results of operations and the price of VSI Holdings common stock before the merger. For a discussion of SPX's and VSI Holdings' businesses and the factors to consider in connection with those businesses, see SPX's annual report on Form 10-K for the fiscal year ended December 31, 2000 and other documents SPX has subsequently filed with the SEC, which are incorporated by reference into this proxy statement-prospectus, and VSI Holdings' annual report on Form 10-K for the fiscal year ended September 30, 2000 and other documents VSI Holdings has subsequently filed with the SEC, which are delivered with and incorporated by reference into this proxy statement-prospectus.

RISK FACTORS RELATING TO SPX'S BUSINESS

SPX'S LEVERAGE MAY AFFECT ITS BUSINESS AND MAY RESTRICT ITS OPERATING FLEXIBILITY.

As of the end of February 2001, SPX had approximately \$1.9 billion in total indebtedness. If SPX completes the acquisitions of United Dominion and VSI Holdings, SPX will be more leveraged than it is at present. In that event, SPX will be required to service the principal and interest obligations on total indebtedness of as much as \$3.0 billion. As such, SPX may be particularly susceptible to adverse changes in its industry, the economy and the financial markets generally. In addition, subject to certain restrictions set forth in the credit facility, SPX may incur additional indebtedness in the future. The level of SPX's indebtedness could:

- limit cash flow available for general corporate purposes, such as acquisitions and capital expenditures, due to the ongoing cash flow requirements for debt service;
- limit SPX's ability to obtain, or obtain on favorable terms, additional debt financing in the future for working capital, capital expenditures or acquisitions;
- limit SPX's flexibility in reacting to competitive and other changes in the industry and economic conditions generally;
- expose SPX to a risk that a substantial decrease in net operating cash flows could make it difficult to meet debt service requirements; and
- expose SPX to risks inherent in interest rate fluctuations because the
 existing borrowings are, and any new borrowings may be, at variable rates
 of interest, which could result in higher interest expense in the event
 of increases in interest rates.

Moreover, SPX's conduct of its business is restricted, and its ability to incur additional debt is limited, by restrictive covenants in its existing credit agreement. These restrictive covenants will continue in SPX's restated credit agreement. In particular, any debt incurrence restrictions may limit SPX's ability to service its existing debt obligations through additional debt financing if cash flow from operations is insufficient to service these obligations.

SPX's ability to make scheduled payments of principal of, to pay interest on, or to refinance, its indebtedness and to satisfy its other debt obligations will depend upon SPX's future operating performance, which may be affected by general economic, financial, competitive, legislative, regulatory, business and other factors beyond its control. In addition, there can be no assurance that future borrowings or equity financing will be available for the payment or refinancing of SPX's indebtedness. If SPX is unable to service its indebtedness, whether in the ordinary course of business or upon acceleration of the indebtedness, SPX may be forced to pursue one or more alternative strategies, such as restructuring or refinancing its indebtedness, selling assets, reducing or delaying capital expenditures or seeking additional equity capital. There can be no assurance that any of these strategies could be effected on satisfactory terms, if at all.

SPX'S FAILURE TO SUCCESSFULLY INTEGRATE UNITED DOMINION, VSI HOLDINGS AND OTHER RECENT ACQUISITIONS, AS WELL AS ANY FUTURE ACQUISITIONS, COULD HAVE A NEGATIVE EFFECT ON ITS OPERATIONS.

As part of SPX's business strategy, it reviews acquisitions in the ordinary course. In 2000, SPX made 21 acquisitions of businesses for an aggregate price of approximately \$226.8 million. SPX's past acquisitions and any potential future acquisitions involve a number of risks and present financial, managerial and operational challenges, including:

- adverse effects on SPX's reported operating results due to the amortization of goodwill associated with acquisitions;
- diversion of management attention from running SPX's existing businesses;
- difficulty with integration of personnel and financial and other systems;
- increased expenses, including compensation expenses resulting from newly hired employees:
- assumption of unknown liabilities; and
- potential disputes with the sellers of acquired businesses, technologies, services or products.

SPX may not be able to integrate successfully the technology, operations and personnel of any acquired business. Customer dissatisfaction or performance problems with an acquired business, technology, service or product could also have a material adverse effect on SPX's reputation and business. In addition, any acquired business, technology, service or product could underperform relative to SPX's expectations. SPX could also experience financial or other setbacks if any of the businesses that it has acquired or may acquire in the future have problems of which it is not aware. In addition, as a result of future acquisitions, SPX may further increase its leverage or, if SPX issues equity securities to pay for future acquisitions, it may dilute its existing stockholders.

SPX regularly engages in discussions with respect to potential acquisitions and joint ventures, some of which may be material. SPX cannot assure you that it will be able to consummate any transactions under negotiation or identify, acquire or make investments in any promising acquisition candidates on acceptable terms. Competition for acquisition or investment targets could result in increased acquisition or investment prices and a diminished pool of businesses, technologies, services or products available for acquisition or investment, which could materially affect SPX's growth rate.

In addition, on March 12, 2001, SPX announced that it had entered into an agreement for a wholly owned subsidiary to acquire United Dominion in an all-stock transaction. SPX will issue approximately 9.2 million shares to acquire United Dominion. SPX will also assume approximately \$876 million in United Dominion debt. SPX expects that the United Dominion acquisition will be dilutive to its earnings per share in the current fiscal year and will be accretive thereafter. This assumes that SPX will be able to take advantage of Synergies that SPX expects will be created by the acquisition of United Dominion.

The success of the business combination of SPX and United Dominion will in large part be dependent on the ability of SPX, following the completion of that merger, to realize cost savings and, to a lesser extent, to consolidate operations and integrate processes. The integration of businesses, moreover,

involves a number of risks, including the diversion of management's attention to the assimilation of the operations from other business concerns, delays or difficulties in the actual integration of operations or systems, and challenges in retaining customers and key personnel of the acquired company. We cannot assure you that future consolidated results will improve as a result of the business combination of SPX and United Dominion, or that the timing or extent of the cost savings and efficiencies that are anticipated by SPX will be achieved.

Following the closing of the United Dominion merger, SPX will be finalizing its strategic review of United Dominion's businesses and its plans to integrate the operations of United Dominion. Integration and rationalization of the operations of United Dominion may include certain costs that in turn would result in additional purchase consideration or in a charge to earnings of the combined company. Any additional purchase consideration or a charge to operations, which cannot now be quantified fully, may be material and would either adjust the purchase accounting or be recognized as a charge to earnings in the period in which such a charge occurs. These costs may include severance and related employee benefits costs, costs to close or consolidate facilities, integration costs, relocation and moving costs, training costs, and gains or losses on business divestitures, among others. Over the past five years, SPX has recorded several special charges to its results of operations associated with cost reductions, integrating acquisitions and achieving operating efficiencies. SPX believes that its actions have been required to improve its operations and, as described above, will, if necessary, record future charges as appropriate to address costs and operational efficiencies at the combined company.

THE UNITED DOMINION ACQUISITION MAY NOT BE COMPLETED, WHICH MAY ADVERSELY AFFECT SPX'S STOCK PRICE.

The completion of the United Dominion acquisition is subject to a number of conditions, including the approval of the United Dominion shareholders. It is possible that, as a result of the failure to meet one or more of the closing conditions or otherwise, SPX will be unable to complete the United Dominion acquisition. A failure by SPX to complete the United Dominion acquisition could result in a decline in the price of SPX's common stock.

THE LOSS OF KEY PERSONNEL AND ANY INABILITY TO ATTRACT AND RETAIN QUALIFIED EMPLOYEES COULD MATERIALLY ADVERSELY IMPACT SPX'S OPERATIONS.

SPX is dependent on the continued services of its management team, including its Chairman of the Board, President and Chief Executive Officer. The loss of any member of its management team without adequate replacement could have a material adverse effect on SPX. Additionally, SPX needs qualified managers and skilled employees with technical and manufacturing industry experience in order to operate its business successfully. From time to time there may be a shortage of skilled labor, which may make it more difficult and expensive for SPX to attract and retain qualified employees. If SPX is unable to attract and retain qualified individuals or its costs to do so increase significantly, SPX's operations could be materially adversely affected.

MANY OF THE INDUSTRIES IN WHICH SPX OPERATES ARE CYCLICAL AND, ACCORDINGLY, ITS BUSINESS IS SUBJECT TO CHANGES IN THE ECONOMY.

Many of the business areas in which SPX operates are subject to specific industry and general economic cycles. Certain businesses are subject to industry cycles, including the automotive industry, which influences SPX's Vehicle Components and Service Solutions segments, the process equipment and electric power markets, which influence its Industrial Products and Services segment and the telecommunications networks and building construction industries, which influence its Technical Products and Systems segment. Accordingly, any downturn in these or other markets in which SPX participates could materially adversely affect it. A decline in automotive sales and production may also affect not only sales of components, tools and services to vehicle manufacturers and their dealerships, but also sales of components, tools and services to aftermarket customers, and could result in a decline in SPX's results of operations or a deterioration in its financial condition. Similar cyclical changes could also affect aftermarket sales of products in SPX's other segments. If demand changes and SPX fails to respond

accordingly, its results of operations could be materially adversely affected in any given quarter. Negative changes in the business cycles of SPX's different operations may occur contemporaneously. Consequently, the effect of an economic downturn may have a magnified negative effect on SPX's business. In 2001, SPX's revenue and earnings growth may be adversely affected by a softer economy. There is also substantial and continuing pressure from the major original equipment manufacturers, particularly in the automotive industry, to reduce costs, including the cost of products and services purchased from outside suppliers such as SPX. If in the future SPX were unable to generate sufficient cost savings to offset price reductions, SPX's gross margins could be materially adversely affected.

IF FUTURE CASH FLOWS ARE INSUFFICIENT TO RECOVER THE CARRYING VALUE OF SPX'S GOODWILL, A MATERIAL NON-CASH CHARGE TO EARNINGS COULD RESULT.

SPX had goodwill and intangible assets of \$1.2 billion and shareholders' equity of \$0.6 billion at December 31, 2000. If SPX completes the acquisitions of United Dominion and VSI Holdings, goodwill and intangible assets could be approximately \$2.4 billion and shareholders' equity could be approximately \$1.6 billion. SPX expects to recover the carrying value of goodwill through its future cash flows. On an ongoing basis, SPX evaluates, based on projected undiscounted cash flows, whether it will be able to recover all or a portion of the carrying value of goodwill. If future cash flows are insufficient to recover the carrying value of SPX's goodwill, SPX must write off a portion of the unamortized balance of goodwill. There can be no assurance that circumstances will not change in the future that will affect the useful life or carrying value of SPX's goodwill.

SPX IS SUBJECT TO ENVIRONMENTAL LAWS AND POTENTIAL EXPOSURE TO ENVIRONMENTAL LIABILITIES.

SPX is subject to various federal, state and local environmental laws, ordinances and regulations, including those governing the remediation of soil and groundwater contaminated by petroleum products or hazardous substances or wastes, and the health and safety of its employees. Under certain of these laws, ordinances or regulations, a current or previous owner or operator of property may be liable for the costs of removal or remediation of certain hazardous substances or petroleum products on, under, or in its property, without regard to whether the owner or operator knew of, or caused, the presence of the contaminants, and regardless of whether the practices that resulted in the contamination were legal at the time they occurred. The presence of, or failure to remediate properly, these substances may materially adversely affect the ability to sell or rent the property or to borrow funds using the property as collateral. In connection with its acquisitions, SPX may assume significant environmental liabilities of which it is not aware. SPX is not aware of any issues relating to environmental matters that are reasonably likely to result in a material adverse effect on its results of operations, but it is possible that future developments related to new or existing environmental matters or changes in environmental laws or policies could lead to material costs for environmental compliance or cleanup. There can be no assurance that these costs, if any, will not have a material adverse effect on SPX's results of operations or financial position in the future.

SPX'S INRANGE SUBSIDIARY IS SUBJECT TO VARIOUS RISKS AND ANY MATERIAL ADVERSE EFFECT ON INRANGE COULD MATERIALLY ADVERSELY AFFECT SPX'S FINANCIAL RESULTS.

SPX owns approximately 89.5% of the total number of outstanding shares of common stock of Inrange Technologies Corporation. As of April 12, 2001, Inrange's market capitalization was approximately \$858.4 million. Unlike SPX's other businesses, Inrange is a technology company and is subject to additional and different risks. Any material adverse effect on Inrange could have a material adverse effect on SPX. In addition to the risks described herein for SPX's business as a whole, Inrange is subject to a number of other risks, including the following:

- Inrange's business will suffer if it fails to develop, successfully introduce and sell new and enhanced high-quality, technologically advanced cost-effective products that meet the changing needs of its customers on a timely basis. Inrange's competitors may develop new and more advanced products on a regular basis.

- Inrange relies on a sole manufacturer to produce a number of its key products and on sole sources of supply for some key components in its products. Any disruption in these relationships could increase product costs and reduce Inrange's ability to provide its products or develop new products on a timely basis.
- The price for Inrange's products may decrease in response to changes in product mix, competitive pricing pressures, maturing life cycles, new product introductions and other factors. Accordingly, Inrange's profitability may decline unless it can reduce its production and sales costs or develop new higher margin products.

SPX'S STOCK PRICE AND INRANGE'S STOCK PRICE COULD BE VOLATILE.

The market prices of SPX's common stock and Inrange's common stock have been, and could be, subject to wide fluctuations in response to quarterly fluctuations in operating results, acquisitions and divestitures, failure to achieve published estimates of, or changes in earnings estimates by, securities analysts, announcements of new products or services by competitors, sales of common stock by existing holders, loss of key personnel and market conditions in their industries.

DIFFICULTIES PRESENTED BY INTERNATIONAL ECONOMIC, POLITICAL, LEGAL, ACCOUNTING AND BUSINESS FACTORS COULD NEGATIVELY AFFECT SPX'S INTERESTS AND BUSINESS

Approximately 14% of SPX's 2000 sales were derived from SPX's international operations. In addition, approximately 40% of Inrange's 2000 sales were international, and SPX is seeking to increase its sales outside the United States. If SPX completes its acquisition of United Dominion, the proportion of SPX's sales derived from international operations will increase to approximately 23% of pro forma 2000 sales. SPX's international operations require it to comply with the legal requirements of foreign jurisdictions and expose it to the political consequences of operating in foreign jurisdictions. SPX's foreign business operations are also subject to the following risks:

- difficulty in managing, operating and marketing SPX's international operations because of distance, as well as language and cultural differences; and
- fluctuations in currency exchange rates that may make SPX's products less competitive in countries in which local currencies decline in value relative to the U.S. dollar.

PROVISIONS IN SPX'S CORPORATE DOCUMENTS AND DELAWARE LAW MAY DELAY OR PREVENT A CHANGE IN CONTROL OF THE COMPANY, AND SPX MAY NOT CONSUMMATE A TRANSACTION THAT ITS STOCKHOLDERS CONSIDER FAVORABLE.

Provisions of SPX's certificate of incorporation and bylaws may inhibit changes in its control not approved by its board of directors. SPX also has a rights plan designed to make it more costly and thus more difficult to gain control of SPX without the consent of its Board. SPX is also afforded the protections of Section 203 of the Delaware General Corporation Law, which could have similar effects. See Comparison of Rights of SPX Stockholders and VSI Holdings Shareholders -- Business Combinations with Substantial Shareholders and -- Shareholder Rights Plans.

THE SPECIAL MEETING OF VSI HOLDINGS SHAREHOLDERS

DATE, TIME AND PLACE

We are sending this proxy statement-prospectus to you as part of the solicitation of proxies by VSI Holdings' board of directors for use at the special meeting to be held on _____, 2001 at [10:00] a.m., Eastern Daylight Savings time, at its Vision Center at 1664 Star Batt Drive, Rochester Hills, Michigan 48309. We are first mailing this proxy statement-prospectus, the attached notice of special meeting of shareholders and the enclosed proxy card to you on or about ______, 2001.

PURPOSE OF THE MEETING

At the special meeting, VSI Holdings shareholders will consider and vote upon a proposal to approve the Agreement and Plan of Merger, dated as of March 24, 2001, by and between VSI Holdings and SPX. This agreement provides for the merger of VSI Holdings into SPX, with SPX as the surviving corporation.

We do not know of any other matter to be brought before the special meeting other than the merger. If any matter incident to the conduct of the special meeting is brought before the meeting, the individuals named in the proxy card will vote as to those matters in their discretion.

VSI Holdings' board of directors has unanimously approved the merger agreement and recommends that you vote FOR its approval.

RECORD DATE; QUORUM

The VSI Holdings board has fixed the close of business on as the record date for the special meeting. Only holders of VSI Holdings common stock on the record date will be entitled to vote at the meeting. On the record date, there were VSI Holdings common shares entitled to vote at the meeting. Each share entitles its holder to one vote. At least a majority of the issued and outstanding shares of VSI Holdings common stock must be represented at the meeting in person or by proxy to establish a quorum for the transaction of business at the meeting. Abstentions will be included in the determination of shares present at the special meeting for purposes of determining a quorum.

REOUIRED VOTE

According to Georgia law and VSI Holdings' articles of incorporation, a majority of the holders of outstanding VSI Holdings common stock must approve the merger agreement. All properly executed proxies delivered and not revoked will be voted at the special meeting as specified in the proxies. If you do not specify a choice, your shares represented by a signed proxy will be voted FOR the approval of the merger agreement. If you do not vote or instruct your broker how to vote your shares, it will have the same effect as voting against the merger.

On the record date, VSI Holdings directors, executive officers and their affiliates owned and were entitled to vote VSI Holdings shares, representing approximately % of the outstanding voting power. The directors and executive officers have expressed an intention to vote in favor of the merger agreement.

SPX's stockholders do not need to approve the merger agreement.

REVOCATION OF PROXIES

You can revoke your proxy in one of the following ways at any time before it is voted at the special meeting. First, you can revoke your proxy by written notice sent to the secretary of VSI Holdings at the address on the notice of the meeting and received prior to the meeting. Second, you can submit a new, later dated proxy card. Third, you can attend the meeting and vote in person. If you instructed a broker to vote your shares, you must follow your broker's directions for changing those instructions.

SOLICITATION OF PROXIES

SPX and VSI Holdings have each paid their expenses in connection with the preparation and filing with the SEC of the registration statement relating to this proxy statement-prospectus. VSI Holdings will pay the expenses incurred with printing this proxy statement-prospectus and distributing it to VSI Holdings' shareholders. Proxies will be solicited through the mail and directly by officers, directors and employees of VSI Holdings not specifically employed for that purpose. These individuals will not be specially compensated. VSI Holdings will reimburse banks, brokers, voting trustees and other nominees and fiduciaries for their reasonable expenses in forwarding these proxy materials to the beneficial owners of VSI Holdings common stock. In the Indemnity Agreement, Mr. Toth has agreed to pay all legal fees and expenses that VSI Holdings incurs in connection with the merger in excess of \$1.5 million.

THE MERGER

This section of the proxy statement-prospectus describes the merger agreement. While we believe that the description covers the material terms of the merger agreement, this summary may not contain all of the information that is important to you. The merger agreement is attached to this proxy statement-prospectus as Appendix A and is incorporated into this proxy statement-prospectus by reference. We urge you to read the merger agreement carefully.

BACKGROUND OF THE MERGER

As part of its strategic planning, VSI Holdings has continually reviewed market trends and strategic opportunities for its marketing services, organizational development and training services, and customer management services. It has also reviewed opportunities for expansion in areas related or complementary to its core businesses. In this vein, VSI Holdings has sought to explore the ways in which it could enhance shareholder value.

In early 2000, at the direction of VSI Holdings' board of directors, VSI Holdings' senior management began to examine and explore the company's ability and capacity to grow in a meaningful way in its core markets in light of current market trends. Among the trends that the board considered were:

- continued consolidation of suppliers within the automotive industry;
- increasing customer demand for large suppliers with a broad spectrum of resources, services and products, the ability able to provide services globally and the capacity to make significant investments in advanced technologies; and
- increasing competition from companies specializing in services such as customer relationship management, database management and e-business.

The VSI Holdings board considered various alternatives, including remaining independent, expanding and seeking a strategic affiliation, including a sale and merger. The board considered the potential risks and costs associated with these alternatives and determined that a strategic alliance, including a sale and merger, with another company would facilitate the company's growth strategy and accelerate the achievement of its strategic objectives. Accordingly, the board directed management to identify entities that might be interested in a strategic affiliation with VSI Holdings, or in acquiring VSI Holdings, and that fit the criteria outlined by the board. These criteria included:

- the ability to provide VSI Holdings with a significant asset base and global presence;
- the prospects for growth and perceived ability to enhance shareholder value; and
- the ability to provide VSI Holdings' shareholders with diversification and greater liquidity.

VSI Holdings' management identified eight entities that it believed satisfied these criteria and would provide an opportunity to implement VSI Holdings' growth strategy and accelerate the achievement of its strategic objectives. Management contacted these eight entities regarding the possibility of a strategic alliance with VSI Holdings. As part of this process, which occurred during the summer of 2000, members of VSI Holdings' management met with and made presentations to each of those potential partners.

On August 14, 2000, the VSI Holdings' board reviewed the status and results of those presentations. As part of that review, management gave a presentation to the board summarizing the status of the strategic affiliation process, its discussions with potential partners and its evaluation of the potential partners. It also presented a proposed timetable for the strategic affiliation process. After discussion, VSI Holdings' board of directors instructed management to continue the strategic affiliation process by conducting further discussions with the potential partners to better assess their interest in pursuing a strategic affiliation with VSI Holdings.

VSI Holdings then notified the potential partners of VSI Holdings' interest in receiving preliminary, non-binding indications of interest for a strategic affiliation. Further discussions between VSI Holdings'

management and the potential partners were held between September and November 2000 and, in connection with those discussions, the potential partners received additional material regarding VSI Holdings.

Following these discussions and initial due diligence by the potential partners, three of these potential partners, including SPX, responded with considered and defined non-binding indications of interest.

On December 14, 2000, VSI Holdings management reviewed the three indications of interest with the board of directors. Based on this review, the board instructed management to invite SPX to conduct detailed due diligence at VSI Holdings' headquarters. VSI Holdings' board of directors selected SPX for the reasons described under -- Reasons for the Merger and because:

- SPX's preliminary indication of value for VSI Holdings was higher than the value indicated in the other two indications of interest;
- SPX was willing to acquire the outstanding stock of VSI Holdings with SPX stock as part of the consideration in a transaction that could provide VSI Holdings shareholders with an opportunity for additional enhancements to shareholder value, while each of the other two potential partners expressed a strong preference for an asset purchase transaction using only cash as consideration and with less favorable tax consequences to VSI Holdings' shareholders;
- SPX expressed a strategic vision that was more compatible with VSI Holdings' strategic objectives and a stronger commitment to a strategic affiliation with VSI Holdings;
- SPX was larger and more diversified and could provide VSI Holdings with access to a more diversified customer base; and
- SPX could provide significant shareholder liquidity through the large public trading market for its stock.

Shortly after being notified of VSI Holdings' interest in pursuing a strategic affiliation with SPX, SPX's management received approval to commence further due diligence regarding a transaction with VSI Holdings. The due diligence, which commenced in December 2000 and continued until the signing of the merger agreement, included presentations by, and meetings with, VSI Holdings' senior management.

On January 10, 2001, Steve Toth, Jr., the Chairman and CEO of VSI Holdings, met with Tom Riordan, President, Service Solutions, of SPX, to discuss various aspects of a proposed transaction, though no definitive agreement was reached. In addition, in January, VSI Holdings retained McDonald Investments to provide a fairness opinion and investigate whether a business combination with SPX would be fair from a financial point of view to shareholders of VSI Holdings.

On January 19, 2001, Mr. Toth and three other executive officers of VSI Holdings traveled to SPX's headquarters in Muskegon, Michigan and met with John Blystone, the Chairman and CEO of SPX, to review and discuss SPX's strategic plan and the proposed transaction.

On February 12, 2001, at a special meeting of VSI Holdings' board of directors, Mr. Toth updated VSI Holdings' directors on the status of the discussions with SPX. Following discussions regarding the reasons for entering into a strategic affiliation with SPX, VSI Holdings' board of directors advised management to continue its negotiations with SPX. At this meeting, McDonald Investments made a presentation of the results of various financial analyses undertaken by it with respect to a business combination with SPX.

Between February 13 and February 24, 2001, representatives of VSI Holdings and SPX and their respective advisors continued negotiating toward a definitive agreement.

On February 25, 2001, Messrs. Toth and Riordan met to continue discussions on terms of the proposed transaction, including deal structure and value. At that time they also discussed the compatibility of VSI Holdings' interests in Advanced Animations, Inc., its indirect investment interests in OZ Entertainment Company and certain other assets and property with the strategic objectives of

combined entity following the proposed business combination of VSI Holdings and SPX. After Mr. Riordan suggested that Advanced Animations, the indirect investment interests in OZ Entertainment and the other property and interests did not constitute "core assets" and were unlikely to have any significant role in the combined entity's strategic plan, Mr. Toth expressed a desire to personally acquire those interests after the proposed business combination with SPX was completed if SPX ever decided to dispose of those interests. Mr. Toth was desirous of acquiring Advanced Animations, the indirect investment interests in OZ Entertainment and the other property and interests because of his previously existing significant personal investments in OZ Entertainment.

In early March, following further discussions regarding SPX's interest in divesting Advanced Animations, the indirect investment interests in OZ Entertainment and the related property and interests following the completion of the proposed business combination of VSI Holdings and SPX, Mr. Toth engaged his own legal advisors to represent him in connection with a proposal to acquire those interests.

On March 13, 2001, VSI Holdings' board of directors met to review the proposed transaction with SPX and to consider and vote upon the proposed transaction. Representatives of VSI Holdings' senior management and their advisors discussed with, and made presentations to, VSI Holdings' board of directors regarding the rationale for the proposed transaction, the proposed terms of the transaction, certain legal and tax considerations, and the terms and provisions contained in a proposed form of merger agreement that had been previously distributed to VSI Holdings' directors. In addition, at this meeting, McDonald Investments made a presentation of the results of its various financial analyses and advised VSI Holdings' board of directors that, as of that date, the aggregate consideration to be received by VSI Holdings shareholders in the proposed business combination with SPX was fair from a financial point of view. The VSI Holdings' board then discussed the presentations given by VSI Holdings' management, financial advisors and legal counsel at this and other of its meetings and unanimously (with all directors present and no directors abstaining):

- approved the proposed business combination with SPX;
- determined that the business combination with SPX was fair and in the best interests of VSI Holdings and its shareholders;
- recommended that VSI Holdings' shareholders vote FOR the business combination with SPX;
- instructed senior management, with the assistance of VSI Holdings' advisors, to promptly finish negotiating the transaction documents; and
- authorized VSI Holdings' officers to execute definitive transaction documents upon the conclusion of satisfactory negotiations.

In taking the foregoing actions, VSI Holdings' board of directors considered the factors described below under -- Reasons for the Merger. The board did not assign any relative or specific weights to any of those factors, and individual directors may have given differing weights to different factors.

During the period of March 13 to March 23, 2001, the management and advisors of VSI Holdings and SPX worked to finalize the transaction documents. On March 23 and 24, 2001, VSI Holdings' management discussed the proposed merger and transaction documents with the members of VSI Holdings' board of directors. Effective as of March 24, 2001, the members of VSI Holdings' board of directors executed a written consent resolution approving the proposed merger and transaction documents. On March 24, 2001, the parties executed the merger agreement. At the same time, as contemplated by the merger agreement, (1) SPX, Mr. Toth and certain persons and entities related to him entered into the indemnity and restriction agreement and (2) SPX, Mr. Toth and an entity controlled by him entered into an asset purchase and merger agreement providing for the sale of Advanced Animations, the indirect investment interests in OZ Entertainment, VSI Holdings' interest in its corporate jet and property and interests related to those entities. See below -- Interests of Directors and Officers in the Merger, -- Indemnity and Restriction Agreement, -- Asset Purchase Agreement and -- Advanced Animations Merger Agreement.

On March 26, 2001, before the opening of trading of SPX common stock and VSI Holdings common stock, the parties issued a joint news release announcing the signing of the merger agreement and the proposed merger of VSI Holdings and SPX

REASONS FOR THE MERGER

Certain matters discussed below are forward-looking statements involving substantial risks and uncertainties that could cause actual results to differ materially from targets or projected results. Factors that could cause actual results to differ materially include, among others, those factors described in Risk Factors. Many of these factors are beyond the ability of SPX or VSI Holdings to predict or control. We caution you not to put undue reliance on forward-looking statements. These statements have been made as of the date of this proxy statement-prospectus, and you should not infer that there has been no change in the affairs of SPX or VSI Holdings since that date that would warrant a modification of any forward-looking statement made in this proxy statement-prospectus. SPX and VSI Holdings disclaim any intent or obligation to update publicly these forward-looking statements, whether as a result of new information, future events or otherwise.

VSI Holdings' Reasons for the Merger

VSI Holdings' board of directors believes that VSI Holdings will become more competitive and grow faster if it combines with SPX and becomes part of a larger, more significantly diversified and capitalized entity.

VSI Holdings' board of directors and management anticipate that being part of a larger entity will provide VSI Holdings with a global scope and the capabilities to more effectively meet customer demands in an industry where size is becoming an increasingly more significant factor. Indeed, the board and management expect the merger to provide VSI Holdings with the potential to undertake significantly larger, global projects not currently available to it because of its smaller capitalization. Consequently, VSI Holdings' board of directors believes that, following the combination with SPX, VSI Holdings will have the potential to realize greater long-term positive operating and financial results than it would on a stand-alone basis.

During its deliberations, VSI Holdings' board of directors consulted with management, legal and financial advisors, and considered a number of factors, including those described in the preceding paragraph and the following:

- the premium over the current trading price for VSI Holdings common stock represented by the consideration to be paid to VSI Holdings shareholders on a per share basis;
- the increased liquidity to VSI Holdings' shareholders given (1) the fact that they may elect to receive a combination of cash and SPX common stock, and (2) the public ownership and significantly higher daily trading volume of SPX common stock;
- the opportunity for VSI Holdings shareholders, through their receipt of SPX shares, to participate in a larger, more diversified entity and to participate in the combined company's perceived greater potential for growth and further enhancements of shareholder value;
- that VSI Holdings had contacts with a number of potential bidders, including SPX, and that no other party expressed a greater commitment or was expected to yield a better acquisition value or more strategic benefits to VSI Holdings and its shareholders; and
- the financial presentation of McDonald Investments to VSI Holdings' board of directors and the oral opinion of McDonald Investments to the effect that, as of March 13, 2001 and based upon the qualifications and assumptions made and matters considered by McDonald Investments described in its written opinion dated March 30, 2001, the consideration to be paid in the consolidation to the holders of VSI Holdings common stock is fair to VSI Holdings shareholders from a financial point of view.

VSI Holdings' board of directors also considered the following factors:

- the creative and innovative forces, and financial discipline, associated with SPX's management and business practices and the perceived potential of those practices to create further enhancements of shareholder value;
- the role of the transaction in implementing and accelerating VSI
 Holdings' long-term external growth strategy by strengthening VSI
 Holdings' current market position within the automotive industry and the
 expectation that the merger would provide VSI Holdings with access to
 SPX's existing customer base and distribution network;
- the financial condition, cash flows and results of operations of VSI Holdings and SPX, both on a historical and prospective basis and the ability of SPX's operations to provide significant revenue diversification;
- the ability of VSI Holdings' senior managers to influence and participate in the management of VSI Holdings' operations in a meaningful way following the proposed combination with SPX; and
- that VSI Holdings shareholders are only expected to recognize gain for U.S. tax purposes to the extent that they receive cash consideration.

In the course of its deliberations, VSI Holdings' board of directors also considered a number of other items and factors, including the factors described under -- Background of the Merger and the following:

- the belief of management that a consolidation with SPX fit both VSI Holdings' long-term external growth strategy and the criteria for a strategic affiliation established by the board of directors;
- the terms of the merger agreement regarding VSI Holdings' right to consider and negotiate other transaction proposals, as well as the possible effects of the termination fees;
- the current and historical trading prices, values and volume of VSI Holdings common stock and SPX common stock, current and historical trading multiples of other comparable companies, and information regarding the value and prospects of VSI Holdings as an independent, stand-alone entity;
- VSI Holdings' knowledge and review of the financial condition, results of operations and business prospects of SPX;
- the terms and conditions of the merger and the representations, warranties, covenants and agreements of the parties and the conditions to their respective obligations, including the condition that VSI Holdings' shareholders approve the transaction; and
- the impact of the merger on VSI Holdings' shareholders and employees.

VSI Holdings' board of directors also identified and considered a variety of potentially negative factors in its deliberations, including the following:

- the risk that VSI Holdings shareholders might not realize the anticipated long-term and short-term benefits of the merger;
- the restrictions that the merger agreement placed on VSI Holdings during the period between signing the merger agreement and closing the merger, including the parties' agreement to a fixed-exchange ratio for the consideration to be paid by SPX;
- the difficulties and costs associated with undertaking a business combination and integrating the operations of two sizeable companies;
- as described under -- Interests of Directors and Officers in the Merger below, and by virtue of certain indemnification rights, change of control arrangements, stock options, stock grants and positions to be held in the combined company, certain members of VSI Holdings' board of

directors and management might have interests in the merger that are different than those of other VSI Holdings shareholders;

- the possible distraction of management from day-to-day operations and possible concerns of employees regarding their employment status; and
- the other risks described under Risk Factors above.

VSI Holdings' board of directors believed that the potential benefit of the merger outweighed these potentially negative factors and that the merger would enable VSI Holdings' shareholders to realize greater value than VSI Holdings could deliver to them alone. Based on the consideration of these and other relevant matters, at its meeting on March 13, 2001, VSI Holdings' board of directors unanimously (with all directors present and no directors abstaining):

- approved the merger with SPX;
- determined that the merger was fair and in the best interest of VSI Holdings and its shareholders;
- recommended that VSI Holdings' shareholders vote FOR the merger;
- instructed senior management, with the assistance of VSI Holdings' advisors, to promptly finish negotiating the transaction documents; and
- authorized VSI Holdings' officers to execute definitive transaction documents upon the conclusion of negotiations satisfactory to senior management.

The foregoing discussion of the factors considered by VSI Holdings' board is not intended to be exhaustive, but is believed to include all material factors that the board considered. In reaching its decision to approve the proposed consolidation with SPX, the board did not assign any relative or specific weights to any of these factors, and individual directors may have given differing weights to different factors.

SPX's Reasons for the Merger

The merger with VSI Holdings complements the strategy of SPX's Service Solutions segment. A key component of this strategy is to provide a comprehensive portfolio of professional services to Service Solutions' customer base. Historically, Service Solutions has focused on the service and maintenance function of the transportation industry. SPX anticipates that the merger with VSI Holdings will allow Service Solutions to expand its customer base by leveraging VSI Holdings' customer relationships. VSI Holdings' capabilities in marketing services, customer relationship management, organizational assessment and event planning also bolster Service Solutions' existing professional services capabilities, which are focused on information authoring, training development and information technology. Once the merger is finalized, SPX will cross-market the combined professional services of VSI Holdings and Service Solutions to new customers outside the automotive industry. Many of the competencies shared by Service Solutions and VSI Holdings can be applied to industries such as pharmaceuticals, healthcare and financial services. Finally, the emerging competencies of VSI Holdings in areas such as e-Learning, e-CRM and web development offer Service Solutions the opportunity to gain a presence in a rapidly emerging set of markets. SPX will also review VSI Holdings' business and identify areas to reduce structural cost, improve operating efficiencies and leverage technology.

INTERESTS OF DIRECTORS AND OFFICERS IN THE MERGER

When you consider the recommendation of VSI Holdings' board of directors that you vote in favor of the merger, you should be aware that a number of VSI Holdings directors and executive officers may have interests in the merger that are different from, or in addition to, your interests as a shareholder. These interests are described below. In each case, VSI Holdings' board of directors was aware of these interests and considered them, among other matters, in approving the merger.

Indemnification. Under the merger agreement, the indemnification and exculpation provisions contained in SPX's certificate of incorporation and bylaws must be at least as favorable to VSI Holdings' directors and executive officers as the indemnification and exculpation provisions contained in VSI Holdings' articles of incorporation and bylaws. Moreover, for a period of six years after the completion of the merger, those provisions may not be altered in any manner that would adversely affect the rights thereunder of VSI Holdings' directors and executive officers. In addition, SPX has agreed:

- to indemnify VSI Holdings' directors and executive officers against any liabilities arising out of the merger agreement or under the Securities Exchange Act; and
- to use its best efforts to obtain and maintain in effect for a period of six years after the completion of the merger VSI Holdings' current directors' and officers' liability insurance for the present directors and officers of VSI Holdings, subject to limitations described in the merger agreement.

Stock options. The approval of the merger by the shareholders of VSI Holdings will be a "change of control" for purposes of the stock options issued under VSI Holdings' 1997 Independent Directors Stock Option Plan. Consequently, all unvested options that are issued under that plan will vest and become exercisable prior to the merger. As of the date of this proxy statement-prospectus, there are a total of 60,000 unvested options issued and outstanding under that plan, each with an exercise price of \$3.25 per share.

The following table sets forth certain information concerning the unvested VSI Holdings stock options currently held by each of VSI Holdings' directors and named executive officers under the 1997 Independent Directors Stock Option Plan.

NAME	NUMBER OF OPTIONS SUBJECT TO ACCELERATED VESTING	VALUE OF IN-THE-MONEY OPTIONS SUBJECT TO ACCELERATED VESTING
Steve Toth, Jr	0	
Thomas W. Marquis	Θ	
Harold A. Poling	20,000*	\$22,000
Ralph Armijo	20,000*	22,000
William R. James	20,000*	22,000
Martin S. Suchik	0	
Terry Davis	0	
Steve Schultz	0	

* Absent accelerated vesting, these non-qualified stock options would be subject to vesting in annual increments of 50% on each of May 1, 2001 and May 1, 2002. These options would expire at 12:00 a.m. on May 23, 2005.

VSI Holdings has also granted options under its 1997 Incentive Stock Option Plan and 1997 Nonqualified Stock Option Plan. These plans do not provide for accelerated vesting of the options granted under them. Pursuant to the merger agreement, VSI Holdings has agreed to use its reasonable best efforts to obtain agreements from the holders of all VSI Holdings stock options for the cancellation of their options in exchange for cash and on terms specified by SPX. Upon completion of the merger, those VSI Holdings stock options that are not cancelled will be adjusted and will become options to purchase shares of SPX common stock. The options to purchase SPX common stock that replace any unvested VSI Holdings stock options that are neither cancelled nor subject to accelerated vesting in connection with the merger will be unvested and will vest in accordance with the terms of the unvested VSI Holdings stock option. See The Merger -- Treatment of VSI Holdings Stock Options; Restricted Stock below.

Stock Grants. As a result of the merger, all unvested stock grants made under the VSI Holdings restricted stock plan will vest and will be exchanged in the merger for either the all stock consideration or the combination consideration at the holder's election. The 66,438 shares of VSI Holdings common stock

that are the subject of those grants will be converted into merger consideration upon completion of the merger. None of these shares are held by directors or named executive officers of VSI Holdings.

Retention and Employment Agreements. In connection with the merger, SPX has entered into retention and employment agreements with 21 "key employees" of VSI Holdings, including Steve Schultz, VSI Holdings' Executive Vice President-Sales. Those agreements will only become effective upon the completion of the merger.

Under each agreement, the employee will agree to continue his or her employment as an at-will employee and will receive a stated base salary and an opportunity to earn a bonus equal to a stated percentage of his or her base salary in accordance with SPX's Economic Value Added Incentive Compensation Plan. Many of these key employees will also be eligible to receive SPX stock options under the SPX Stock Compensation Plan and to participate in the SPX Employee Stock Purchase Plan and the other SPX employee benefit programs.

Each retention and employment agreement provides that employment may be terminated at any time with or without cause. Each employee will also be subject to certain confidentiality, non-solicitation and non-competition provisions. Each VSI Holdings employee who signs a retention and employment agreement will also be agreeing to the cancellation of all of his or her vested and unvested VSI Holdings stock options.

Repayment of Indebtedness Owed to Executive Officers. Pursuant to the merger agreement, SPX has agreed to repay the indebtedness of VSI Holdings and its subsidiaries to certain of VSI Holdings' executive officers and their affiliates outstanding at the time the merger is completed. The following table sets forth certain information regarding that indebtedness as of the date of the merger agreement.

PAYEE	AMOUNT OUTSTANDING
Margaret A. Toth, Trustee U/A/D 7/9/82 F/B/O Margaret	
Toth	\$3,177,401.10
Terry Davis	877,293.07
Tom Marquis	52,637.58
Steve Toth, Jr., Trustee U/A/D 12/20/76 F/B/O Steve Toth,	
Jr	6,556,626.60

PRTNCTPAL

Other Matters. Steve Toth, Jr., VSI Holdings' Chairman and CEO, also has interests in the merger that are different from, or in addition to your interests as a shareholder because he is a party to, and his interest in the transactions contemplated by, the indemnity and restriction agreement, the asset purchase agreement and the Advanced Animations merger agreement. For a description of those agreements and Mr. Toth's interests therein, you should refer to -- Indemnity and Restriction Agreement, -- Asset Purchase Agreement and --- Advanced Animations Merger Agreement below.

VSI HOLDINGS BOARD APPROVAL AND RECOMMENDATION OF THE MERGER

VSI Holdings' board of directors believes that the merger agreement is fair to, and in the best interests of, VSI Holdings and its shareholders. It has unanimously adopted the merger and recommends that VSI Holdings shareholders vote for its approval.

THE MERGER; EFFECTIVE TIME

The merger agreement provides that upon completion of the merger, VSI Holdings will merge with and into SPX. SPX will be the surviving corporation in the merger and will continue its corporate existence under Delaware law. The corporate existence of VSI Holdings will cease as a result of the merger. On the closing date of the merger, SPX and VSI Holdings will file certificates of merger with the Delaware Secretary of State and the Georgia Secretary of State. The merger will be effective at the time of the filing of the certificates of merger.

MERGER CONSIDERATION

Upon completion of the merger and subject to the limitations and proration procedure as described below, the outstanding shares of VSI Holdings common stock held by each VSI Holdings shareholder will be automatically canceled and converted into the right to receive one of the following forms of consideration (determined as of the effective time of the merger):

- stock consideration equal to 0.043 shares of SPX common stock for each share of VSI Holdings common stock held by the VSI Holdings shareholder; or
- a combination of cash consideration in the form of a cash payment equal to \$4.35 per share for 45% of the shares of VSI Holdings common stock held by the VSI Holdings shareholder and stock consideration equal to 0.043 shares of SPX common stock for the remaining 55% of the shares of VSI Holdings common stock held by the shareholder.

ELECTION OF FORM OF CONSIDERATION; CONVERSION OF SHARES

Using the enclosed form of election and letter of transmittal, you may specify whether you would like to have your shares of VSI Holdings common stock converted into the right to receive the all stock consideration, or a combination of cash and stock consideration. The number of shares of VSI Holdings common stock converted to the right to receive cash consideration in accordance with the combination cash and stock election may be adjusted in order to preserve the intended tax-free nature of the transaction as described below.

THE FORM OF ELECTION AND LETTER OF TRANSMITTAL MUST BE RETURNED TO EQUISERVE TRUST COMPANY, N.A., THE EXCHANGE AGENT, BY THE ELECTION DEADLINE, WHICH WILL BE 5:00 P.M. EASTERN DAYLIGHT SAVINGS TIME , 2001 (OR, IF THE MEETING OF VSI HOLDINGS SHAREHOLDERS TO APPROVE THE MERGER IS POSTPONED, TWO BUSINESS DAYS BEFORE THE DATE OF THE MEETING). Any shares of VSI Holdings common stock held by VSI Holdings shareholders who did not return the election form before that time will automatically be converted into the right to receive the all stock consideration.

At the effective time of the merger, VSI Holdings common stock will automatically convert into the right to receive the merger consideration. If you elect the all stock consideration, each share of VSI Holdings common stock that you hold will be converted into the right to receive the stock consideration of 0.043 shares of SPX common stock. If you elect a combination of cash and stock consideration, each share of VSI Holdings common stock that you hold will be converted into a combination of cash and SPX common stock.

If you elect the combination consideration and there is no need to adjust the proportion of cash and SPX common stock in the combination consideration, then 45% of your VSI Holdings shares will be converted to cash at the rate of \$4.35 per share and 55% of your VSI Holdings shares will be converted to SPX common stock at the rate of 0.043 SPX shares for each VSI Holdings share. However, the merger agreement permits SPX to reduce the amount of cash and increase the amount of SPX stock paid to VSI Holdings shareholders electing the combination consideration if the sum of the cash to be paid by SPX upon conversion of VSI Holdings shares that VSI Holdings shareholders elect to convert to cash plus the amount of related party debt of VSI Holdings to be assumed by SPX is greater than the non-stock consideration permitted under Section 3.02 of Rev. Proc. 77-37, 1977-2 C.B. 568.

In Rev. Proc. 77-37, 1977-2, C.B. 568, the Internal Revenue Service has set forth its guidelines for determining whether the Internal Revenue Service will issue a ruling as to the tax consequences of a corporate reorganization. Section 3.02 of Rev. Proc. 77-37 states that the "continuity of interest" requirement, which must be satisfied in order for a merger to qualify as a reorganization for purposes of Section 368(a)(1) of the Internal Revenue Code of 1986, as amended, will be satisfied for purposes of issuing a ruling if, as of the effective date of a merger, at least 50% of the value of all of the formerly outstanding stock of a target company is exchanged for stock of the acquiring company. Thus, Section 3.02 of Rev. Proc. 77-37 permits no more than 50% of the total consideration paid by an acquiring corporation

in a merger to be in the form of cash consideration. For this purpose, the cash consideration paid by SPX in the merger will include:

- any cash paid by SPX upon conversion of VSI Holdings shares;
- any cash paid for fractional shares; and
- the amount of related party debt of VSI Holdings assumed by SPX.

The total consideration paid by SPX will equal the cash consideration plus the aggregate value, as of the closing date, of all SPX shares issued upon conversion of VSI Holdings shares.

If the cash consideration to be paid by SPX would exceed 50% of the total consideration to be paid by SPX, then SPX will, on a pro rata basis among VSI Holdings shareholders electing the combination consideration, reduce the proportion of VSI Holdings shares converted to cash and increase the proportion of VSI Holdings shares converted to SPX shares to the extent necessary to cause the cash consideration to equal 50% of the total consideration paid by SPX.

As of April 16, 2001, the amount of VSI Holdings related party debt to be assumed by SPX is \$11,805,396. Based on this amount and assuming a per share market price for SPX common stock on the closing date of \$92.00, if all VSI Holdings shareholders elect the combination consideration and if all holders of VSI Holdings options with an exercise price less than \$4.35 exercise those options prior to the effective time of the merger, the percentage of VSI Holdings shares that are actually converted to cash will be reduced from 45% to 43.394% and the percentage converted to SPX shares will be increased from 55% to 56.606%. To the extent that the price of SPX common stock on the closing date is higher or lower than \$92.00 and to the extent that the amount of VSI Holdings related party debt assumed by SPX is more or less than \$11,805,396, the adjustment required, if any, to the proportion between cash consideration and stock consideration paid to VSI Holdings shareholders electing the combination consideration will be affected.

EXCHANGE AGENT; PROCEDURES FOR EXCHANGE OF CERTIFICATES; FRACTIONAL SHARES

Immediately prior to the completion of the merger, SPX will deposit all of the merger consideration with the exchange agent, who will distribute it to VSI Holdings shareholders. Pursuant to the merger agreement, SPX has appointed EquiServe Trust Company, N.A. to act as exchange agent for the merger.

In addition to allowing you to elect between the all stock consideration and the combination cash and stock consideration, the form of election and letter of transmittal contains instructions on how to surrender VSI Holdings stock certificates to the exchange agent and receive the merger consideration. Upon completion of the merger, each certificate that previously represented shares of VSI Holdings common stock will represent only the right to receive the merger consideration into which the shares of VSI Holdings common stock were converted in the merger and any cash payable in lieu of fractional shares of SPX common stock. After a holder surrenders his or her certificates, any unpaid dividends and any cash in lieu of fractional shares of SPX common stock payable as described below will be paid to the holder without interest.

VSI Holdings shareholders will not receive interest on any cash consideration they receive. No dividend or distribution with respect to shares of SPX common stock will be payable on or with respect to any fractional share, and fractional share interests will not entitle their owners to vote or to any other rights of an SPX stockholder. In lieu of any fractional share of SPX common stock, SPX will pay to each former VSI Holdings shareholder who otherwise would be entitled to receive a fractional share of SPX common stock an amount in cash equal to \$101.17 multiplied by the fractional share of VSI Holdings common stock to which the holder would otherwise be entitled. SPX will not pay declared dividends or distributions to persons entitled to receive certificates for shares of SPX common stock until they surrender their VSI Holdings stock certificates for SPX common stock, at which time SPX will pay the dividends or distributions. In no event will the persons entitled to receive these dividends be entitled to interest on them.

TREATMENT OF VSI HOLDINGS STOCK OPTIONS; RESTRICTED STOCK

Pursuant to the merger agreement, VSI Holdings has agreed to use its reasonable best efforts to obtain agreements from the holders of VSI Holdings stock options for the cancellation of those options in exchange for cash and on terms specified by SPX. Upon completion of the merger, each outstanding and unexercised option to purchase VSI Holdings' common stock, whether vested or unvested, will be automatically converted into an option to purchase a number of shares of SPX common stock equal to the number of shares of VSI Holdings common stock covered under the VSI Holdings stock option multiplied by 0.043 at a per share price equal to the exercise price specified in the VSI Holdings stock option divided by 0.043. Each newly issued SPX stock option will contain terms that are substantially similar to those governing the original VSI Holdings stock option.

Any unvested restricted shares of VSI Holdings common stock awarded to employees, directors or consultants pursuant to any of VSI Holdings' plans or arrangements and outstanding at the effective time of the merger will become fully vested and treated as outstanding shares of VSI Holdings common stock. If you hold VSI Holdings restricted stock, it will be included in the election you make on the form of election.

SPX will file any required registration statements (on Form S-8 or other appropriate form) with respect to each VSI Holdings stock option converted into an SPX stock option as well as the shares of SPX common stock underlying the option. SPX will keep the registration statements current as long as the options remain outstanding.

GOVERNANCE FOLLOWING THE MERGER

After the merger is completed, the directors and officers of SPX will remain the directors and officers of SPX until their successors are duly elected, appointed or qualified or until their earlier death, resignation or removal in accordance with the certificate of incorporation and the bylaws of SPX.

Upon completion of the merger, SPX's certificate of incorporation and bylaws will remain unchanged and will constitute the certificate of incorporation and bylaws of the surviving corporation.

REPRESENTATIONS AND WARRANTIES

The merger agreement contains customary mutual representations and warranties by each of VSI Holdings and SPX. The most significant of these include:

- due incorporation and good standing;
- capitalization;
- authorization and enforceability;
- consents and approvals necessary to complete the merger;
- shareholder vote requirement:
- financial statements;
- changes in business or the occurrence of certain events since the end of its most recent fiscal year, and undisclosed liabilities;
- compliance with applicable laws; and
- required filings and reports with the Securities and Exchange Commission.

VSI Holdings made additional representations and warranties in the merger agreement regarding aspects of its business, financial condition, structure and other facts pertinent to the merger, including:

- corporate structure;
- permits and licenses necessary for the operation of business;

- pending or threatened litigation;
- material contracts;
- employees, labor and employee benefits matters;
- tangible assets;
- tax matters;
- fairness opinion;
- insurance;
- environmental matters;
- real property;
- notes and accounts receivable;
- intellectual property;
- guaranties; and
- investments.

SPX also made representations and warranties in the merger agreement regarding its financing of the transaction and the tax-free nature of the transaction

VSI HOLDINGS' CONDUCT OF THE BUSINESS BEFORE COMPLETION OF THE MERGER

VSI Holdings has agreed that, until the completion of the merger or unless SPX consents in writing, it will carry on its business in the ordinary course and will use commercially reasonable efforts to preserve intact its business organization, to keep available the services of its officers and key employees, and to maintain satisfactory relationships with all persons with whom it does business. VSI Holdings has also agreed that, until the completion of the merger or unless SPX consents in writing, it will conduct its business in compliance with specific restrictions relating to the following:

- the issuance and redemption of securities;
- employees and employee benefits and remuneration;
- the issuance of dividends or other distributions;
- modification of articles of incorporation or bylaws;
- the acquisition of assets or other entities;
- the disposition of assets;
- capital expenditures or other investments;
- the incurrence of indebtedness;
- the settlement of litigation;
- the entering into of contracts;
- maintenance of books, accounts and records; and
- maintenance of debt as of September 30, 2000.

CONDITIONS TO COMPLETION OF THE MERGER

The obligations of SPX and VSI Holdings to complete the merger are subject to the satisfaction or waiver of closing conditions. These conditions are provided in detail in the merger agreement and include the following:

- the merger agreement and the merger are approved by the VSI Holdings shareholders;
- no injunction or order preventing the completion of the merger is in effect;
- all necessary federal and state governmental and regulatory approvals are obtained, and the relevant waiting period imposed under the antitrust laws expires;
- the Form S-4 registration statement of SPX, of which this proxy statement-prospectus is a part, becomes effective and is not the subject of any stop order or proceeding seeking a stop order;
- the SPX common stock to be issued in the merger is authorized for listing on the NYSE;
- the representations and warranties made by the other party are true and correct in all material respects on the date of the merger as if they were made on that date, unless they were originally stated to be true as of a specific earlier date, in which case they must still have been true on that date, unless their failure to be true would not reasonably have a material adverse effect on the other party; and
- the other party has performed all obligations and complied with all covenants required by the merger agreement in all material respects.

VSI Holdings also has the right not to complete the merger if:

- VSI Holdings' tax counsel fails to deliver a bring-down opinion at closing, that, if the merger is completed in accordance with the terms of the merger agreement, the merger will constitute a reorganization for federal income tax purposes; or
- SPX has not repaid certain indebtedness of VSI Holdings to Steve Toth, Jr., the Chairman and Chief Executive Officer of VSI Holdings, and certain other VSI Holdings shareholders or to refinance VSI Holdings' bank debt.

SPX also has the right not to complete the merger if:

- SPX's tax counsel fails to deliver a bring-down opinion at closing, that, if the merger is completed in accordance with the terms of the merger agreement, the merger will constitute a reorganization for federal income tax purposes; or
- all conditions to the closing of the sale of assets by VSI Holdings to SPX described below under -- Asset Purchase Agreement have not been met.

REGULATORY APPROVALS

Under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, SPX and VSI Holdings may not complete the merger until:

- notifications are given to the Federal Trade Commission and the Antitrust Division of the United States Department of Justice;
- information is furnished to the Federal Trade Commission and the Antitrust Division; and
- specified waiting period requirements are satisfied or terminated.

In connection with the merger, on April 9, 2001, VSI Holdings filed and, on April 10, 2001, SPX filed a Pre-Merger Notification and Report Form under the Hart-Scott-Rodino Act with the Federal Trade Commission and the Antitrust Division of the United States Justice Department. At any time before or after the merger, the Antitrust Division or the Federal Trade Commission could, among other things, seek

to enjoin the completion of the merger or seek the divestiture of substantial assets of SPX or VSI Holdings. Although SPX and VSI Holdings believe that the merger is legal under the United States antitrust laws, there can be no assurance that a challenge to the merger on antitrust grounds will not be made or, if a challenge is made, that it would not be successful.

TERMINATION OF THE MERGER

SPX and VSI Holdings may agree to terminate the merger agreement at any time. In addition, either party may terminate the merger agreement, whether before or after approval by the VSI Holdings shareholders, if:

- the merger is not completed by June 30, 2001, subject to an extension by SPX or VSI Holdings:
 - (1) to a date not later than July 30, 2001 in order to cure a breach of a representation, warranty or covenant, obtain the opinion of tax counsel regarding the treatment of the merger for federal income tax purposes, or to call and hold the meeting of VSI Holdings shareholders, or
 - (2) to a date not later than August 31, 2001, in the event this prospectus-proxy statement has not been declared effective in time to hold the meeting of VSI Holdings shareholders on or prior to June 30, 2001;
- VSI Holdings shareholders do not approve the merger agreement at the meeting called to vote on the merger proposal;
- a law or regulation makes the merger illegal, or any order of injunction permanently prohibits the merger; or
- the other party breaches, in any material respect, its representations, warranties or covenants, the breach constitutes a material adverse effect on the other party and the breach is not reasonably cured by the applicable closing deadline.

VSI Holdings may terminate the merger agreement if, prior to approval of the merger agreement by VSI Holdings' shareholders, VSI Holdings' board of directors determines in good faith, with the advice of its independent financial advisors, that it has received a takeover proposal from another party that constitutes a superior transaction to the transaction with SPX.

SPX may terminate the merger agreement if VSI Holdings' board of directors:

- withdraws or modifies, or resolves to withdraw or modify, in a manner adverse to SPX, its approval of the merger agreement or recommendation that VSI Holdings shareholders vote in favor of the merger agreement;
- approves or recommends, or resolves to approve or recommend, to VSI Holdings shareholders a takeover proposal from another party that constitutes a superior transaction to the transaction with SPX; or
- does not call, or resolves not to call, the special meeting of VSI Holdings shareholders within a reasonable time after the SEC declares the registration statement effective.

EFFECT OF TERMINATION

VSI Holdings will be required to pay SPX a fee of 9 million in the event the merger agreement is terminated:

- due to VSI Holdings' failure to obtain shareholder approval at the meeting called for that purpose;
- by SPX if the only reason that the merger has not been completed is due to VSI Holdings' failure to obtain the previously mentioned opinion of counsel regarding the effect of the reorganization for federal income tax purposes where SPX has obtained a letter from nationally recognized tax counsel stating that it would execute and deliver the opinion on behalf of VSI Holdings, SPX has

delivered the opinion of its own tax counsel to VSI Holdings, and SPX and VSI Holdings have delivered the tax representations certificates required of them; or

- by SPX as a result of a breach by VSI Holdings in any material respect of any of its representations, warranties or covenants under the merger agreement, if (1) the breach constitutes a VSI Holdings material adverse effect, as defined in the merger agreement, and (2) SPX has given written notice to VSI Holdings of the breach promptly after learning of it and VSI Holdings has not reasonably cured the breach on or prior to the applicable closing deadline.

In addition, VSI Holdings will be required to pay the fee in the event that:

- after the date of the merger agreement and prior to the termination of the merger agreement a third party makes known to VSI Holdings a bona fide takeover proposal, as defined in the merger agreement, makes a takeover proposal directly to VSI Holdings' shareholders or publicly announces its intention (whether conditional or unconditional) to make a takeover proposal, and thereafter:
 - (1) SPX terminates the merger agreement after the VSI Holdings board of directors has withdrawn or modified in a manner adverse to SPX the board's approval or recommendation of the approval of the merger agreement by the VSI Holdings shareholders or approves or recommends a superior transaction (as defined in the merger agreement); or
 - (2) VSI Holdings terminates the merger agreement if, prior to the approval of the merger agreement by the VSI Holdings shareholders, the VSI Holdings board of directors determines in good faith by majority vote, with respect to any takeover proposal received by VSI Holdings after the date of the merger agreement that it constitutes a superior transaction (as defined in the merger agreement) and is in the best interest of VSI Holdings and its shareholders and the board of directors of VSI Holdings has received the advice of McDonald Investments as to whether the takeover proposal constitutes a superior transaction.

The fee is intended to compensate SPX for fees and expenses that it incurred prior to termination, the preparation of the agreement and the transaction contemplated by the agreement. In addition, if the merger agreement is terminated, SPX may not hire or solicit for employment any officer or key management employee of VSI Holdings.

VSI Holdings will not be liable for any breach or breaches of any representation, warranty or covenant unless SPX terminates the merger agreement as a result of the breach or breaches. Furthermore, any liability of VSI Holdings for breaches of representations and warranties will be limited to the lesser of the actual damages incurred by SPX as a direct result of the breaches or \$9 million. In the case of any breaches of covenants by VSI Holdings, VSI Holdings' liability will be limited to payment of the \$9 million termination fee.

SPX will not be liable for any breach or breaches of any representation, warranty or covenant unless VSI Holdings terminates the merger agreement as a result of the breach or breaches. Furthermore, any liability of SPX for breaches of representations, warranties or covenants will be limited to the lesser of the actual damages incurred by VSI Holdings as a direct result of the breaches or \$9 million, except in the case of willful breaches of material covenants that result in the merger not closing, in which case this limit will not apply.

INDEMNIFICATION AND INSURANCE

Under the merger agreement, SPX has agreed to indemnify the current and former officers, directors, agents and employees of VSI Holdings and its subsidiaries (other than Advanced Animations) for acts or omissions occurring at or prior to the effective time of the merger. SPX has further agreed that the indemnification and exculpation provisions in the SPX certificate of incorporation and bylaws will be at least as favorable to those persons as the provisions currently applicable to them and will not be amended, repealed or otherwise modified for a period of six years after the merger in any manner that would adversely affect the rights thereunder of these individuals.

The merger agreement also provides that, for six years after the effective time of the merger, SPX will obtain and maintain liability insurance covering acts or omissions occurring prior to the effective time of the merger for those persons who were covered by VSI Holdings' directors' and officers' liability insurance policy. These policies will continue under the same terms and in amounts no less favorable than those in effect on the date of the merger agreement. SPX, however, will not be required to pay more than 300% of the amount paid by VSI Holdings in 2000 to maintain that insurance, and, in the event that the annual premiums exceed 300%, SPX will obtain a policy with the greatest coverage available for a cost not exceeding that amount.

INDEMNITY AND RESTRICTION AGREEMENT

Contemporaneously with the execution of the merger agreement, Steve Toth, Jr., individually, as well as several entities owned and/or controlled by Mr. Toth, his wife and their daughter executed and delivered to SPX an indemnity and restriction agreement. The agreement is attached to this proxy statement-prospectus as Appendix B, and we urge you to read it carefully. Mr. Toth is the Chairman and CEO of VSI Holdings and is the beneficial owner, or is deemed to share control, of approximately 27,905,755 shares of VSI Holdings common stock. In the indemnity and restriction agreement, Mr. Toth agreed to pay or reimburse SPX for certain losses, fees and expenses related to certain specifically enumerated matters, as more particularly set forth in the agreement. In addition, Mr. Toth agreed to pay all legal fees and expenses (including investment banking fees) that VSI Holdings incurs in connection with the merger in excess of \$1.5 million. To secure Mr. Toth's indemnification obligations, Mr. Toth agreed to escrow 64,248 shares of SPX common stock that he will receive as stock consideration for the merger. After one year, the escrow agent will release to Mr. Toth the lesser of 32,124 shares and the amount obtained by subtracting the number of shares representing the value of contested claims, if any, plus any other fees and expenses to be reimbursed by Toth under the indemnity and restriction agreement from 32,124 shares. After two years, the escrow agent will release the balance of the pledged stock less any claimed amounts. Mr. Toth and the entities that signed the agreement also agreed that they would not, in any 90-day period and for a period of one year from the date of the agreement, sell, transfer or dispose of 25% or more of the aggregate shares of SPX common stock that they receive in the merger. Finally, Mr. Toth agreed not to compete with SPX or solicit employees or customers of SPX for a five-year period.

OPINION OF VSI HOLDINGS' FINANCIAL ADVISOR

Introduction

VSI Holdings asked McDonald Investments to render an opinion to VSI Holdings' board as to the fairness to VSI Holdings' shareholders of the merger consideration from a financial point of view. On March 13, 2001, McDonald Investments delivered its oral opinion that, as of that date and based upon and subject to the assumptions, limitations and qualifications contained in its opinion, the merger consideration was fair to VSI Holdings' shareholders from a financial point of view.

THE FULL TEXT OF THE WRITTEN OPINION OF MCDONALD INVESTMENTS DATED MARCH 30, 2001 IS ATTACHED TO THIS DOCUMENT AS APPENDIX C. YOU ARE URGED TO READ THAT OPINION CAREFULLY AND IN ITS ENTIRETY FOR ASSUMPTIONS MADE, PROCEDURES FOLLOWED, OTHER MATTERS CONSIDERED AND LIMITS OF THE REVIEW UNDERTAKEN IN ARRIVING AT THAT OPINION. THE DESCRIPTION OF MCDONALD INVESTMENTS' OPINION CONTAINED IN THIS DOCUMENT IS QUALIFIED IN ITS ENTIRETY BY REFERENCE TO THE FULL TEXT OF THE OPINION.

VSI Holdings retained McDonald Investments to serve as an advisor to its board and not as an advisor to or agent of any of VSI Holdings' shareholders. McDonald Investments prepared its opinion for VSI Holdings' board, and that opinion is directed only to the fairness of the merger consideration to VSI Holdings' shareholders from a financial point of view. McDonald Investments' opinion does not address the merits of VSI Holdings' decision to engage in the merger or other business strategies considered by VSI Holdings' board, nor does it address the board's decision to proceed with the merger. McDonald

Investments' opinion does not constitute a recommendation to any VSI Holdings shareholder as to how that shareholder should vote at VSI Holdings' special meeting of shareholders.

VSI Holdings retained McDonald Investments solely to render the fairness opinion. McDonald Investments did not participate in the negotiations between the parties regarding the merger consideration. VSI Holdings imposed no restrictions or limitations on McDonald Investments with respect to the investigations made or the procedures followed by McDonald Investments in rendering its opinion.

In rendering its written opinion, McDonald Investments reviewed, among other things:

- the merger agreement between VSI Holdings and SPX, including the exhibits and schedules;
- the asset purchase agreement among SPX, PII Ventures and Steve Toth, Jr.;
- the form of agreement and plan of merger between PII Ventures and Advanced Animations;
- Annual Reports on Form 10-K for the last four fiscal years, Quarterly Reports on Form 10-Q and other publicly available information about VSI Holdings and SPX;
- internal business and financial information, including projections, furnished to McDonald Investments by VSI Holdings' management;
- publicly available information concerning historical stock prices and trading volumes for VSI Holdings' common stock and SPX's common stock;
- publicly available information for certain other companies that McDonald Investments thought were comparable to VSI Holdings or SPX and the trading markets for certain of those other companies' securities; and
- publicly available information about the nature and terms of other business combination transactions that McDonald Investments considered relevant to its analysis of this transaction.

McDonald Investments also met with officers and employees of VSI Holdings to discuss VSI Holdings' business and prospects and other matters that McDonald Investments believed were relevant.

You should note that in rendering its opinion, McDonald Investments relied upon the accuracy and completeness of all of the financial and other information provided to it or publicly available. McDonald Investments also assumed and relied upon the accuracy of representations and warranties of VSI Holdings and SPX contained in the merger agreement. McDonald Investments was not engaged to verify, and did not independently attempt to verify, any of the information. McDonald Investments also relied upon VSI Holdings' management as to the reasonableness and achievability of the financial and operating projections, as well as the assumptions and bases for those projections, and assumed that they reflected the best currently available estimates and judgment of VSI Holdings' management. VSI Holdings did not engage McDonald Investments to assess the reasonableness or achievability of those projections or the assumptions underlying them and expressed no view on those matters. McDonald Investments did not conduct a physical inspection or appraisal of any of the assets, properties or facilities of either VSI Holdings or SPX, nor was it furnished with any evaluation or appraisal of those assets, properties or facilities. McDonald Investments also assumed that the conditions to the merger in the merger agreement would be satisfied and that the merger would be completed on a timely basis in the manner contemplated by the merger agreement.

McDonald Investments' opinion is based on economic and market conditions and other circumstances existing on the date of its opinion. McDonald Investments' opinion does not address any matters subsequent to its date, including the value of SPX's common stock when it is issued to VSI Holdings' shareholders. McDonald Investments' opinion is limited to the fairness, as of the date of the opinion, from a financial point of view of the merger consideration and does not address VSI Holdings' underlying business decision to effect the merger or any other terms of the merger.

You should understand that, although subsequent developments may affect its opinion, McDonald Investments does not have to update, revise or reaffirm its opinion.

McDonald Investments' Analysis

McDonald Investments developed several analyses to evaluate the merger consideration. These were a leveraged buyout analysis, a discounted cash flow analysis, a comparable public company analysis and a comparable merger and acquisition analysis. The following is a brief summary of the analyses performed by McDonald Investments to arrive at its opinion. The results determined from any particular analysis are not necessarily indicative of actual values or predictive of future results or values and are inherently subject to substantial uncertainty.

Leveraged Buyout Analysis. McDonald Investments performed a leveraged buyout analysis to determine the price a hypothetical financial buyer could afford to pay under prevailing market conditions. For purposes of this analysis, McDonald Investments used VSI Holdings' results for the twelve months ended December 31, 2000 and management's financial projections for fiscal years 2001 through 2005. McDonald Investments based this analysis on the following assumptions:

- a market rate of interest on revolving debt of 8.0%, a market rate of return on senior term debt of 8.0% and a market rate of return on subordinated debt of 12.0%;
- maximum senior debt determined by the current market conditions for asset based loans;
- expected internal rate of return of 29.8% for the equity investor; and
- expected internal rate of return of 20.0% for the subordinated debt

This methodology implied a valuation of between \$1.93\$ and \$2.52 per share, as compared to the \$4.35 per share merger consideration.

Discounted Cash Flow Analysis. McDonald Investments performed a discounted cash flow analysis to calculate VSI Holdings' implied present value per share based on management's projections for fiscal year 2001 through fiscal year 2005. Using this information, McDonald Investments calculated the free cash flows VSI Holdings could generate through fiscal year 2005. McDonald Investments also calculated an estimated terminal value for VSI Holdings at the end of 2005 based on a terminal multiple of 6.5 times earnings before interest, taxes, depreciation and amortization, or EBITDA. These future cash flows and terminal value were discounted using a discount rate of 15.1%. In deriving the discount rate, McDonald Investments calculated a weighted average cost of capital for VSI Holdings utilizing the capital asset pricing model. The sum of the present value of the free cash flows and terminal values less outstanding debt plus existing cash yielded implied prices per share ranging from \$2.67 per share to \$3.57 per share, as compared to the merger consideration of \$4.35 per share.

Inherent in any discounted cash flow valuation are the use of a number of assumptions, including the accuracy of management's projections, and the subjective determination of an appropriate terminal value and discount rate to apply to the projected cash flows of the entity under examination. Variations in any of these assumptions or judgments could significantly alter the results of a discounted cash flow analysis.

Comparable Public Company Analysis. McDonald Investments' comparable public company analysis involves an analysis of publicly traded companies that it considered to be comparable to VSI Holdings with regard to industry, operations, performance and/or markets served. This analysis is based on the theory that the market value of a company can be estimated by deriving market multiples from publicly traded companies that relate their stock prices to earnings, cash flows or other measures of the target company. No company used as a comparison in this analysis is identical to VSI Holdings.

McDonald Investments selected public companies that, in its estimation, were reasonably similar in scope of operations to VSI Holdings. The companies selected were TeleTech Holdings, Inc., APAC Customer Services, Inc., SITEL Corp., Aegis Communications Group, Boron, LePore & Associates, Inc., RMH Teleservices, Inc., Access Worldwide Communications, Inc. and Innotrac Corporation. This group includes companies with market values of between \$9.1 million and \$926.5 million.

The data and ratios McDonald Investments compared included, among other things:

- enterprise value to latest twelve months sales;
- enterprise value to latest twelve months EBITDA;
- enterprise value to latest twelve months EBIT;
- market value to book value;
- current stock price to earnings per share for the latest twelve months;
- current stock price to estimated earnings per share for fiscal 2001;
- current stock price to estimated earnings per share for fiscal 2002; and
- current stock price to estimated earnings growth for fiscal 2002.

Enterprise value is defined as market value, plus the book value of debt, less cash and cash equivalents. Market value is defined as current stock price multiplied by outstanding shares. McDonald Investments applied the valuation multiples, reflecting the average and median multiples of the comparable companies. The table below illustrates the range of valuation multiples, as well as the average and median multiples, for the comparable group resulting from this analysis. It also shows the valuation multiples for VSI Holdings implied by the merger consideration.

SUMMARY OF ANALYSIS OF PUBLIC COMPANY VALUATION MULTIPLES

LATEST TWELVE MONTHS	RANGE	RANGE AVERAGE MEDI		VSI HOLDINGS
Enterprise Value/Sales	0.2x-1.2x	0.6x	0.5x	0.9x
Enterprise Value/EBITDA	3.8x-23.7x	10.0x	8.2x	8.5x
Enterprise Value/EBIT	8.3x-32.2x	17.4x	15.9x	12.1x
Market Value/Book Value	0.2x-3.6x	1.7x	1.5x	4.1x
Price/Earnings	14.4x-28.8x	21.0x	20.5x	18.6x
2001 Price/Earnings	9.6x-27.3x	17.2x	16.0x	12.4x
2002 Price/Earnings	10.9x-14.6x	12.6x	12.3x	7.3x
2002 PEG Ratio	0.3x-0.3x	0.3x	0.3x	0.1x

This method implied a valuation of between \$2.67 and \$3.27 per share, as compared to the merger consideration of \$4.35 per share.

Comparable Merger and Acquisition Analysis. McDonald Investments analyzed information related to seven selected merger and acquisition transactions for which public information was available. They conducted this analysis to determine relevant valuation multiples for transactions considered similar to the merger. No transaction used as a comparison in this analysis is identical to the merger. McDonald Investments identified transactions of companies and industries related to VSI Holdings, including marketing and advertising companies that utilize in-person, telephonic and on-line means to provide services to their customers. After reviewing the transactions where financial information was available, McDonald Investments applied the median transaction multiples resulting from this analysis to VSI Holdings' latest twelve months sales, EBIT and EBITDA figures to derive an implied valuation for VSI Holdings. This methodology implied a valuation of between \$4.46 and \$5.65 per share, as compared to the merger consideration of \$4.35 per share.

SUMMARY OF ANALYSIS OF SELECTED ACQUISITION TRANSACTIONS

	RANGE	AVERAGE	MEDIAN
Enterprise Value/Revenue	0.1x-2.5x	1.5x	1.5x
Enterprise Value/EBIT	6.5x-19.1x	13.1x	14.6x
Enterprise Value/EBITDA	12.2x-5.9x	9.0x	9.5x

Conclusion

This summary describes the principal elements of McDonald Investments' presentation to VSI Holdings' board on March 13, 2001 and its written opinion dated March 30, 2001. The preparation of a fairness opinion involves various determinations as to the most appropriate and relevant methods of financial analysis and the application of these methods to the particular circumstances. A fairness opinion is not readily susceptible to summary description. Each of the analyses conducted by McDonald Investments was carried out to provide a different perspective on the transaction and add to the total mix of information available. McDonald Investments did not form a conclusion as to whether any individual analysis, considered in isolation, supported or failed to support an opinion as to fairness from a financial point of view. Rather, in reaching its conclusion, McDonald Investments considered the results of the analyses in light of each other and ultimately reached its opinion based upon the results of all analyses taken as a whole. McDonald Investments did not place particular reliance or weight on any individual analysis, but instead concluded that its analyses, taken as a whole, supported its determination. Accordingly, notwithstanding the separate factors summarized above, McDonald Investments believes that you must consider its analyses as a whole and that selecting portions of its analysis and the factors considered by it, without considering all analyses and factors, could create an incomplete or misleading view of the evaluation process underlying its opinion. In performing its analyses, McDonald Investments made numerous assumptions with respect to industry performance, business and economic conditions and other matters. The analyses performed by McDonald Investments are not necessarily indicative of actual values or future results, which may be significantly more or less favorable than suggested by those analyses.

Engagement Letter. Under the terms of an engagement letter dated January 16, 2001, VSI Holdings agreed to pay McDonald Investments a fee for its services in rendering its fairness opinion of \$350,000 upon oral delivery of a fairness opinion to VSI Holdings' board, provided, however, that if the merger agreement was not executed prior to March 15, 2001, the amount of McDonald Investments' fee for rendering its opinion would be increased by \$150,000, and if the merger agreement had not been executed prior to May 15, 2001, McDonald Investments' fee for rendering its opinion would have been increased by an additional \$150,000. VSI Holdings has also agreed to reimburse McDonald Investments promptly for all out-of-pocket expenses, including the reasonable fees and out-of-pocket expenses of counsel, incurred by McDonald Investments in connection with its engagement, whether or not the merger is consummated, and to indemnify McDonald Investments and related persons against liabilities in connection with its engagement, including liabilities under federal securities laws. The terms of the fee arrangement with McDonald Investments were negotiated at arm's-length between VSI Holdings and McDonald Investments.

In the ordinary course of business, McDonald Investments may actively trade the securities of VSI Holdings and SPX for its own account and for the accounts of its customers and, accordingly, may at any time hold a long or short position in those securities.

McDonald Investments, as part of its investment banking services, is regularly engaged in the valuation of businesses and securities in connection with mergers, acquisitions, underwritings, sales and distributions of listed and unlisted securities, private placements and valuations for corporate and other purposes. VSI Holdings selected McDonald Investments based on its experience in transactions similar to the merger and its reputation in the retail and investment communities.

NO DISSENTERS' RIGHTS

Under Georgia law, VSI Holdings shareholders are not entitled to dissenters' rights of appraisal in connection with the merger.

ACCOUNTING TREATMENT

 $\ensuremath{\mathsf{SPX}}$ will account for the merger using the purchase method of accounting.

CERTAIN FEDERAL INCOME TAX CONSEQUENCES

The following discussion of federal income tax consequences is for general information only and is based on existing law as of the date of this proxy statement-prospectus. We urge VSI Holdings shareholders to consult their tax advisors to determine the particular tax consequences to them of the merger, including the applicability and effect of federal, state, local, foreign and other tax laws.

The following summary, based upon current law, is a general discussion of certain material federal income tax consequences of the merger to SPX, VSI Holdings and VSI Holdings shareholders and assumes that the merger is consummated as described in this proxy statement-prospectus. This summary is based upon the Internal Revenue Code of 1986, as amended, applicable Treasury regulations promulgated thereunder, and administrative rulings and judicial authority, all as of the date of this proxy statement-prospectus, and all of which are subject to change, possibly with retroactive effect. Any change could affect the continuing validity of this summary. This summary applies to VSI Holdings shareholders who hold their VSI Holdings common stock as capital assets. It does not discuss all aspects of income taxation that a particular VSI Holdings shareholder may find relevant in light of his or her specific circumstances, or in light of special treatment that he or she might be subject to under the federal income tax laws (for example, foreign persons, dealers in securities, banks and other financial institutions, insurance companies, tax-exempt organizations, and shareholders who acquired their VSI Holdings common stock pursuant to the exercise of options or otherwise as compensation or through a tax-qualified retirement plan). In addition, this summary does not discuss any aspect of state, local, foreign or other tax laws.

Neither SPX nor VSI Holdings has sought, and neither SPX nor VSI Holdings intends to seek, a ruling from the Internal Revenue Service as to the anticipated tax consequences of the merger. Gardner, Carton & Douglas, counsel to SPX, has advised SPX, and Miller, Canfield, Paddock and Stone, P.L.C., counsel to VSI Holdings, has advised VSI Holdings, that, assuming the merger is consummated as contemplated in this proxy statement-prospectus, the material U.S. federal income tax consequences of the merger to SPX, VSI Holdings and VSI Holdings shareholders are described below.

The Merger. It is a condition to the consummation of the merger that SPX receive an opinion of its counsel, Gardner, Carton & Douglas, and that VSI Holdings receive an opinion of its counsel, Miller, Canfield, Paddock and Stone, P.L.C., that the merger will qualify as a reorganization for federal income tax purposes within the meaning of Section 368(a) of the Internal Revenue Code. The opinions of Gardner, Carton & Douglas and Miller, Canfield, Paddock and Stone, P.L.C. will be expressly based upon the accuracy of certain assumptions and the truth and accuracy of certain representations contained in the merger agreement or made to them by SPX and VSI Holdings. As a reorganization for federal income tax purposes within the meaning of Section 368(a) of the Internal Revenue Code, it is the opinion of counsel to VSI Holdings that the merger will result in the following general federal income tax consequences to VSI Holdings and its shareholders:

- 1. VSI Holdings will be a party to a reorganization within the meaning of Section 368(b) of the Internal Revenue Code.
- 2. VSI Holdings will recognize neither gain nor loss solely as a result of the consummation of the merger.
- 3. VSI Holdings shareholders who receive solely SPX common stock in exchange for their VSI Holdings common stock in the merger will not recognize any gain or loss pursuant to Section 354 of the Internal Revenue Code, except that those VSI Holdings shareholders may recognize gain to the extent cash is received in lieu of fractional share interests in SPX common stock.
- 4. VSI Holdings shareholders who receive SPX common stock and cash in exchange for their VSI Holdings common stock in the merger will recognize any gain that is realized by them under the applicable rules of the Internal Revenue Code to the extent of the cash received by them in the merger in accordance with the rules of Section 356 of the Internal Revenue Code.

- 5. The tax basis of the SPX common stock received by each VSI Holdings shareholder who exchanges his or her VSI Holdings common stock solely for SPX common stock in the merger will be the same as the tax basis of the shares of VSI Holdings common stock surrendered in exchange therefor (reduced by any amount allocable to a fractional share interest for which cash is received).
- 6. The tax basis of the SPX common stock received by each VSI Holdings shareholder who exchanges his or her VSI Holdings common stock for SPX common stock and cash in the merger will be the same as the tax basis of the shares of VSI Holdings common stock surrendered in exchange therefor, reduced by the amount of any cash received and increased by the amount of any gain recognized on the exchange.
- 7. Provided that the VSI Holdings common stock is held as a capital asset on the date of the exchange, the holding period of SPX common stock received by a VSI Holdings shareholder tax free in the merger will include the period during which the VSI Holdings shareholder held the VSI Holdings common stock exchanged therefor.
- 8. As to each VSI Holdings shareholder who exchanges his or her shares of VSI Holdings common stock solely for shares of SPX common stock in the merger, the cash received by that VSI Holdings shareholder in lieu of a fractional share interest of SPX common stock will be treated as having been received as a distribution in full payment in exchange for the fractional share interest of SPX common stock that the VSI Holdings shareholder would otherwise be entitled to receive and will qualify as capital gain or loss (assuming the shares of VSI Holdings common stock were a capital asset in that VSI Holdings shareholder's hands at the time the merger is completed), provided that the cash payment for fractional shares (a) will not constitute separately bargained for consideration, and (b) will not exceed 5% of the total consideration received by the VSI Holdings shareholders in the merger.

As a reorganization for federal income tax purposes within the meaning of Section 368(a) of the Internal Revenue Code, it is the opinion of counsel to SPX that the merger will result in the following general federal income tax consequences to SPX:

- 1. SPX will be a "party to a reorganization" within the meaning of Section 368(b) of the Internal Revenue Code;
- 2. No gain or loss will be recognized by SPX in the merger pursuant to Section 1032 of the Internal Revenue Code; and
- 3. SPX's basis in the assets acquired from VSI Holdings will be the same as the basis of the assets in the hands of VSI Holdings immediately before the merger increased by any gain recognized by VSI Holdings in the merger pursuant to Section 362(b) of the Internal Revenue Code, and the holding period of the assets acquired by SPX will include the holding period thereof when held by VSI Holdings immediately before the merger pursuant to Section 1223(2) of the Internal Revenue Code.

Other Federal Income Tax Consequences of the Merger. Generally, gain recognized by a shareholder of VSI Holdings as described in paragraph 4 above, will be capital gain, provided that the exchange of VSI Holdings common stock for SPX common stock does not have the effect of a distribution of a dividend under the rules of Section 356 of the Internal Revenue Code with respect to the shareholder. This capital gain would be long-term capital gain if the shareholder has held the stock for more than 12 months. However, because this is dependent upon individual circumstances, counsel has declined to give an opinion on this matter. Therefore, VSI Holdings shareholders who elect the combination consideration should consult their tax advisors to determine the character of any gain that they recognize as a result of the merger.

Although it is the opinion of counsel that neither VSI Holdings nor SPX will recognize gain or loss solely as a result of the consummation of the merger, other transactions described elsewhere in this proxy

statement-prospectus, including the transactions contemplated in the asset purchase agreement and the Advanced Animations merger agreement, will be taxable transactions.

Backup Withholding. To prevent "backup withholding" of federal income tax on any payments of cash to a VSI Holdings shareholder in the merger, a VSI Holdings shareholder must, unless an exception applies under the applicable law and regulations, provide the payor of the cash with his or her correct taxpayer identification number on a Substitute Form W-9 and certify under penalties of perjury that the number is correct and that he or she is not subject to backup withholding. The form of election and letter of transmittal mailed to VSI Holdings shareholders will include a Substitute Form W-9. If the VSI Holdings shareholder does not provide the correct taxpayer identification number and certifications, a \$50 penalty may be imposed on him or her by the Internal Revenue Service, and any cash received by him or her may be subject to backup withholding at a rate of 31%.

RESALE OF SPX COMMON STOCK ISSUED IN THE MERGER; AFFILIATES

This proxy statement-prospectus does not cover any resales of the SPX common stock that VSI Holdings shareholders will receive in the merger, and SPX has not authorized anyone to use this proxy statement-prospectus in connection with a resale.

VSI Holdings shareholders who are not deemed to be "affiliates" of VSI Holdings under the Securities Act at the time of VSI Holdings' special meeting can freely transfer any SPX shares that they receive in the merger. Affiliates of VSI Holdings may only sell their shares in transactions permitted by Rule 145 or another applicable exemption from the registration requirements of the Securities Act. VSI Holdings' affiliates for this purpose generally include individuals or entities that control, are controlled by, or are under common control with, VSI Holdings. These affiliates would typically not include shareholders who are not officers, directors or principal shareholders of VSI Holdings.

ASSET PURCHASE AGREEMENT

At the time of the execution of the merger agreement, Steve Toth, Jr. and PII Ventures, L.L.C., an entity owned and controlled (directly and indirectly) by Mr. Toth, entered into an asset purchase agreement with SPX. Immediately following the merger, PII Ventures will purchase certain assets of SPX acquired from VSI Holdings in accordance with the merger. Mr. Toth is the Chairman and CEO of VSI Holdings and is the beneficial owner, or is deemed to share control, of approximately 27,905,755 shares of VSI Holdings. Accordingly, he is familiar with the assets PII Ventures will purchase under the asset purchase agreement. The assets that SPX will sell to PII Ventures consist of: (1) all of its limited partnership interests in K.C. Investors, L.P.; (2) all of its membership interests in Corporate Eagle Five, L.L.C.; (3) all of its stock in the OZ Entertainment Company; and (4) the Agency of Record Marketing Agreement between OZ Entertainment and Visual Services, Inc. PII Ventures will pay SPX \$3,690,000 in cash. SPX is selling the assets to PII Ventures on an "as is, where is" basis, without any representations, warranties or guaranties with respect to fitness, merchantability or otherwise. PII Ventures will assumes all liabilities related to the assets it purchases from SPX.

In addition, SPX and PII Ventures have agreed to enter into a merger agreement providing for the merger of Advanced Animations, Inc., currently a wholly owned subsidiary of VSI Holdings, with and into PII Ventures. PII Ventures will pay SPX \$16,750,000 in cash for Advanced Animations. That merger would occur simultaneously with the closing of the asset purchase. See -- Advanced Animations Merger Agreement below.

The asset purchase agreement, which was the result of arm's-length negotiations between SPX and Mr. Toth, includes representations and warranties by SPX to PII Ventures including: corporate existence, corporate authorization and enforceability and no conflict. The asset purchase agreement also contains the following representations and warranties by PII Ventures to SPX: enforceability, no conflict and full disclosure. These representations and warranties will survive the closing of the sale of the assets indefinitely.

The obligations of SPX and PII Ventures to complete the sale of assets to PII Ventures are subject to the satisfaction or waiver of closing conditions, as provided in more detail in the asset purchase agreement. In particular, SPX and PII Ventures have the right not to consummate the sale if all of the covenants and conditions precedent to the merger agreement have not been satisfied.

SPX and PII Ventures may agree to terminate the asset purchase agreement at any time. Either SPX or PII Ventures may terminate the agreement if the other party fails to fulfill any of its covenants, or breaches any of its representations and warranties. In addition, either party may, under certain circumstances, terminate the asset purchase agreement if the sale of the assets is not completed within 10 days of the closing deadline for the merger agreement.

Finally, SPX agreed to indemnify PII Ventures, and PII Ventures and Mr. Toth agreed to indemnify SPX, for certain losses incurred in connection with certain excluded liabilities and assumed liabilities, respectively, and the breach or violation of any representation, warranty or covenant.

ADVANCED ANIMATIONS MERGER AGREEMENT

Simultaneously with the closing of the asset purchase agreement, Advanced Animations will be merged with and into PII Ventures. Advanced Animations is currently a wholly owned subsidiary of VSI Holdings, but following the merger will be wholly owned by SPX. As consideration for the merger, PII Ventures will pay SPX \$16,750,000 in cash. SPX is selling Advanced Animations to PII Ventures on an "as is, where is" basis, without any representations, warranties or guaranties with respect to fitness, merchantability or otherwise. The terms of this merger were the result of arm's-length negotiations between SPX and Mr.

COMPARATIVE PER SHARE PRICES

SPX common stock is listed and principally traded on the NYSE and is also listed on the Pacific Exchange under the symbol "SPW". On , the record date for the special meeting, there were registered holders of SPX common stock. SPX's fiscal year ends on December 31.

VSI Holdings common stock is listed on the AMEX under the symbol "VIS". As of the record date, VSI Holdings had registered holders. VSI Holdings' fiscal year ends on September 30.

The tables below show, for the periods indicated, the high and low trading prices per share of SPX common stock, as reported on the NYSE Composite Tape, and of VSI Holdings common stock, as reported on the AMEX.

SPX COMMON STOCK

	HIGH	LOW
Fiscal 1999:		
First quarter	\$ 71.75	\$ 48.75
Second quarter	87.13	50.69
Third quarter	94.00	80.19
Fourth quarter	92.75	73.00
Fiscal 2000:		
First quarter	\$122.00	\$ 74.00
Second quarter	124.50	89.88
Third quarter	186.00	118.31
Fourth quarter	147.00	90.50
Fiscal 2001:		
First quarter	\$118.75	\$ 88.00
Second quarter (through April 12, 2001)	96.31	84.99

VSI HOLDINGS COMMON STOCK

	HIGH		LOW	
Fiscal 1999:				
First quarter	\$	6.00	\$	3.19
Second quarter	•	7.25	-	4.75
Third quarter		5.88		4.13
Fourth quarter		4.69		3.13
Fiscal 2000:				
First quarter	\$	4.44	\$	3.00
Second quarter		4.00		2.75
Third quarter		3.38		2.50
Fourth quarter		3.81		2.13
Fiscal 2001:				
First quarter	\$	3.63	\$	2.13
Second quarter		4.19		2.63
Third quarter (through April 12, 2001)		4.19		3.85

DIVIDEND POLICY

Neither SPX nor VSI Holdings has declared or paid cash dividends in the past two fiscal years. SPX does not intend to pay dividends on its common stock. SPX has determined that for the foreseeable future any distribution of earnings will be in the form of open market purchases when deemed appropriate by management and the board. SPX's credit agreement contains restrictions that may limit the payment of dividends and other distributions. VSI Holdings' loan agreements prohibit the payment of cash dividends without prior bank approval.

INFORMATION CONCERNING SPX

SPX is a global multi-industry company that is focused on profitably growing its businesses, enabling it to continue to grow sales, earnings and cash flow. SPX's strategy is to create market advantages through product and technology leadership, by expanding its product and service offerings to full customer solutions and by building critical mass through strategic acquisitions.

OPERATING SEGMENTS

SPX is a global provider of technical products and systems, industrial products and services, service solutions and vehicle components. SPX's products include networking and switching products, fire detection and building life-safety products, TV and radio broadcast antennas and towers, transformers, substations and industrial mixers and valves. SPX products and services also include specialty service tools, diagnostic systems, service equipment, technical information services and vehicle components. With over 14,000 employees worldwide, SPX is a multinational corporation with operations in 19 countries.

 $\ensuremath{\mathsf{SPX's}}$ business has been operated through the following four business segments:

Technical Products and Systems: The operations in the Technical Products and Systems segment are focused on solving customer problems with complete technology-based systems. The emphasis is on growth through investment in new technology, new product introduction, alliances and acquisitions.

The Technical Products and Systems segment includes SPX's Inrange Technologies Corporation subsidiary that designs, manufactures, markets and services networking and switching products for storage, data and telecommunications networks, including fibre channel directors for storage area networks. SPX owns approximately 89.5% of the total number of outstanding shares of Inrange's common stock, which represents about 97.7% of the combined voting power of all classes of Inrange's common stock. The balance of Inrange's common stock is publicly held. Inrange serves Fortune 1000 businesses and other large enterprises that operate large-scale systems where reliability and continuous availability are critical. Inrange focuses on high-end, large-scale, fault tolerant products. Inrange's products have been installed at over 2,000 sites in 90 countries and are designed to be compatible with various vendors' products and multiple communication standards and protocols.

This segment also includes fire detection and integrated building life-safety systems, TV and radio broadcast antennas and towers, and automated fare collection systems.

Industrial Products and Services: The strategy of the Industrial Products and Services segment is to provide "Productivity Solutions for Industry." The business emphasis is on introducing new related services and products, as well as focusing on the replacement parts and service elements of the business and growing through acquisitions. This segment includes operations that design, manufacture and market power transformers, industrial valves and mixers, electric motors, laboratory and industrial ovens and freezers, hydraulic systems, material handling systems and electric motors for industrial chemical companies, pulp and paper manufacturers, laboratories and utilities.

Service Solutions: Service Solutions includes operations that design, manufacture and market a wide range of specialty service tools, hand-held diagnostic systems and service equipment and technical and training information, primarily to the motor vehicle industry in North America and Europe. Major customers are franchised dealers of motor vehicle manufacturers, aftermarket vehicle service facilities and independent distributors.

Vehicle Components: This segment includes operations that supply high-integrity aluminum and magnesium die-castings, forgings, automatic transmission and small engine filters and transmission kits for original equipment manufacturers. These operations also supply automatic transmission and small engine filters and transmission kits for aftermarket customers.

PENDING ACQUISITION

On March 12, 2001, SPX announced that it had entered into an agreement for a wholly owned subsidiary to acquire United Dominion Industries Limited. United Dominion is a diversified manufacturer of proprietary engineered products in four business segments -- flow technology, machinery, specialty engineered products, and test instrumentation. It has annual sales of \$2.4 billion and approximately 14,000 employees in 20 countries. The common stock of United Dominion is listed on the NYSE and the Toronto Stock Exchange under the symbol "UDT".

SPX will issue approximately 9.2 million shares in exchange for the shares of United Dominion in an all-stock transaction. SPX will also assume approximately \$876 million in United Dominion debt. United Dominion shareholders will receive SPX shares based on a fixed exchange ratio of 0.2353 of an SPX share per United Dominion share, which was valued at \$25 per United Dominion share based on the average closing price of a share of SPX common stock for the five-day period ended Friday, March 9, 2001, the day before the merger agreement between SPX and United Dominion was entered into. There is no collar on the fixed exchange ratio. The board of directors of both companies unanimously approved the agreement.

The combination of SPX and United Dominion will create a company with approximately \$5 billion in pro forma annual revenues and enhanced value-creation opportunities. SPX also will gain additional international reach as United Dominion derives approximately 33% of its revenues from sources outside the United States. United Dominion brings to SPX 18 businesses, many of which have leading market shares or product lines in key areas. SPX's business segments will be reconfigured to some extent, and United Dominion's business will be operated within the following four business segments: technical products and systems, industrial products and services, flow technology and service solutions.

The United Dominion acquisition will be a taxable transaction to the shareholders of United Dominion and is to be implemented as a court-approved plan of arrangement (similar to a merger) under Canadian corporate law, is subject to approval by United Dominion shareholders, approval of the Ontario Court, antitrust clearance and customary closing conditions, and is expected to close, assuming satisfaction of all of the foregoing, on or about May 24, 2001. If the merger agreement is terminated in certain circumstances, United Dominion will be required to pay SPX a break-up fee of \$40 million. United Dominion has scheduled a combined annual and special shareholders meeting for May 17, 2001 to vote on the merger agreement. For more information regarding the United Dominion acquisition, please refer to SPX's current reports on Form 8-K filed on March 12, April 12 and April 13, 2001.

Following the United Dominion acquisition, SPX will review each business unit and identify areas to reduce structural cost, improve operating efficiencies and leverage technology. SPX expects to achieve a minimum of \$30 million of annualized cost savings in the first full year following the acquisition. These include savings associated with the elimination of duplicate corporate office costs, consolidation of purchasing initiatives and rationalization of duplicate facilities, among others.

In connection with the United Dominion acquisition, SPX has obtained commitments for \$780 million in incremental term facilities consisting of a two-year term loan facility in an aggregate principal amount equal to \$250 million and an increase in SPX's existing tranche C term facility from \$300 million to \$830 million. The incremental term facilities will bear interest at either the ABR rate or the Eurodollar rate, plus an applicable margin. The proceeds of these borrowings will be used to refinance a like amount of United Dominion's outstanding indebtedness. SPX intends, immediately prior to the completion of the United Dominion acquisition, to enter into a restated credit agreement to reflect the addition of the \$780 million in incremental term facilities. The restated credit agreement will contain affirmative covenants, negative covenants, financial covenants, repayment provisions, representations and warranties, and events of default substantially similar to the terms contained in SPX's existing credit agreement.

Alternatively, SPX may, in connection with the United Dominion acquisition, issue notes in a public offering or a private placement for approximately \$250 million, in which case the \$780 million in incremental term facilities will be reduced by an equivalent amount.

INFORMATION CONCERNING VSI HOLDINGS

VSI Holdings is a leading full service provider of marketing services and customer relationship management, organizational training and development services, and entertaining and educational animatronic displays and other entertainment/edutainment products. VSI Holdings markets its services primarily to the automotive and pharmaceutical industries and markets its entertainment/edutainment products primarily to theme parks, casinos and museums. VSI Holdings' principal executive offices are located at 41000 Woodward Avenue, Bloomfield Hills, Michigan 48304; telephone (248) 644-0500.

MARKETING SERVICES

VSI Holdings helps companies directly market and sell their products using an extensive array of disciplines, including writing, art direction, print and graphic production, corporate website development for the Internet and intranet, electronic commerce and database management.

VSI Holdings also assists companies in establishing and maintaining customer relationships. It does so through a variety of interactive tools, including Internet, video and CD-ROM applications, video conferencing, inbound call centers and other telephony applications, ongoing customer film, slides, live theater, computer graphics and animation, print multimedia, and global satellite broadcasts.

VSI Holdings also provides companies with general marketing support, including administrative services, direct marketing services and event marketing services. Administrative services include data entry and processing and installment processing. Event marketing services include creating and operating product exhibitions, including managing the venues, media coverage, attendee logistical services and post-event reporting

ORGANIZATIONAL TRAINING AND DEVELOPMENT

VSI Holdings provides companies with a variety of organizational training, certification and development services that utilize contemporary learning theories and state-of-the-art technology, such as Interactive Distance Learning (a live, satellite based, two way interactive broadcast technology). VSI Holdings' training programs are generally designed to improve employee productivity in the areas of product knowledge, team building, sales skills, personal skills and behavioral development.

ENTERTAINMENT/EDUTAINMENT

VSI Holdings has leveraged its marketing and training expertise to develop traveling consumer attractions for museums and high-end animation figures for theme parks, casinos and retailers on a worldwide basis. VSI Holdings utilizes hydraulic motion technology to create lifelike movement in its 3-D animatronic (robotic) figures.

COMPARISON OF RIGHTS OF SPX STOCKHOLDERS AND VSI HOLDINGS SHAREHOLDERS

SPX is incorporated under the laws of Delaware, and VSI Holdings is incorporated under the laws of Georgia. The rights of VSI Holdings shareholders are governed by the Georgia Business Corporation Code and VSI Holdings' governing documents, which consist of its articles of incorporation and bylaws. If the merger is consummated, the holders of VSI Holdings common stock will become holders of SPX common stock (other than holders who elect, and receive, only cash for their shares). As SPX stockholders, their rights will be governed by the General Corporation Law of Delaware and SPX's governing documents, which consist of its certificate of incorporation and bylaws and shareholders rights plan. In certain respects the rights of SPX stockholders differ from those of VSI Holdings shareholders.

This summary is intended to highlight the material differences between the rights of SPX stockholders and VSI Holdings shareholders. It is not a complete statement of the rights of SPX stockholders under applicable Delaware laws or SPX's governing documents, or a comprehensive

comparison with the rights of VSI Holdings shareholders under applicable Georgia laws or VSI Holdings' governing documents. It is not intended to be a complete description of the specific provisions to which we refer. For complete information, you should refer to the Delaware law, the Georgia law and the governing documents of SPX and VSI Holdings. To find out how to get copies of the governing documents, you should refer to Where You Can Find More Information.

SHAREHOLDER PROPOSALS AND NOMINATION OF DIRECTORS

SPX's bylaws provide that for a matter to be properly brought before an annual meeting by a stockholder, the stockholder must give timely notice of the matter in writing to the secretary of SPX not less than 120 days nor more than 150 days prior to the anniversary date of the immediately preceding annual meeting. The bylaws also specify requirements for a stockholder's notice to be in proper form.

Under SPX's bylaws, nominations for the election of directors may be made by the SPX board of directors, a committee appointed by the board or any stockholder entitled to vote in the elections of directors. A stockholder may nominate someone for election as a director at a meeting only if he or she has given written notice of his or her intent to make a nomination to SPX's secretary. Stockholders must submit their nominations for an election to be held at an annual meeting by 120 days prior to the anniversary date of the immediately preceding annual meeting. Stockholder nominations for an election to be held at a special meeting of the stockholders must be made at the close of business on the tenth day following the date on which notice of the special meeting was first given to stockholders. The bylaws also specify requirements for a stockholder's nomination to be in proper written form.

Unlike SPX's governing documents, VSI Holdings' governing documents do not specify any advance notice of shareholder proposals or director nominations. In addition, neither the Delaware law nor the Georgia law contains requirements as to these issues.

DIRECTORS

Both the Delaware law and the Georgia law provide that a board of directors may have one or more directors. The exact number of directors is fixed by or in the manner provided in the bylaws, unless the certificate or articles of incorporation fixes the number. In that case, an amendment to the certificate or articles of incorporation is required to change the number of directors.

The SPX certificate of incorporation provides for the number of directors to be fixed by or pursuant to the bylaws. The SPX bylaws provide that, except as otherwise fixed pursuant to the certificate of incorporation relating to the rights of the holders of any preferred stock, the number of directors will be fixed from time to time by the SPX board, but cannot be less than three. At present, there are seven directors on SPX's board.

The VSI Holdings bylaws provide that its board may have up to nine directors. Currently, seven directors serve on VSI Holdings' board.

Under the Delaware law, the directors of a corporation may, by the certificate of incorporation, an initial bylaw or a bylaw adopted by a vote of the stockholders, be divided into one, two or three classes. The SPX certificate of incorporation provides that the directors, other than those who may be elected by the holders of preferred stock, will be classified into three classes, as nearly equal in number as possible. The Georgia law also provides for a classified board when this type of provision is in the articles of incorporation or in a bylaw adopted by the shareholders. VSI Holdings' governing documents do not provide for a classified board of directors.

Under the Delaware law and the Georgia law, cumulative voting in the election of directors is not available unless it is specifically provided for in the certificate or articles of incorporation. SPX's certificate of incorporation does not provide for cumulative voting, and VSI Holdings' articles of incorporation specifically disallow it.

DIRECTOR VACANCIES

Under the Delaware law, a majority of directors then on the board may fill vacancies and newly created directorships, unless otherwise provided in the certificate of incorporation or bylaws. The certificate of incorporation may direct that a particular class of stock is to elect the director, in which case any other directors elected by that class, or a sole remaining director so elected, may fill the vacancy. SPX's governing documents provide that a majority of the remaining directors then in office, even if less than a quorum, will fill any board vacancies, including those resulting from an increase in the number of directors.

Under the Georgia law, unless the articles of incorporation or a bylaw adopted by the shareholders provides otherwise, vacancies and newly created directorships may be filled by the shareholders, by the board of directors or, if the directors constitute less than a quorum, by a majority of the directors remaining in office. If, however, the director was elected by a voting group, the shareholders of the voting group or the remaining directors elected by the voting group will fill the vacancy. VSI Holdings' bylaws provide that VSI Holdings board vacancies, including those from any increase in the number of directors, are filled by the majority of the directors then in office, even if less than a quorum, or by the shareholders at a special meeting of shareholders called for that purpose.

REMOVAL OF DIRECTORS

Under the Delaware law, the holders of a majority of the shares then entitled to vote at an election of directors may remove any or all directors, with or without cause. If a corporation has a classified board, however, stockholders can remove directors only for cause unless the certificate of incorporation provides otherwise. In addition, under the Delaware law, when a corporation's certificate of incorporation provides that holders of a series or class, voting as a series or class, are entitled to elect one or more directors, then only the holders of that series or class may remove those directors without cause. The Delaware law does not permit directors to remove other directors.

SPX's governing documents provide that, subject to certain rights of holders of preferred stock, any director may be removed from office only for cause and only by the affirmative vote of the holders of 80% of the combined voting power of the outstanding shares of stock entitled to vote in the election of directors, voting together as a single class.

Under the Georgia law, shareholders may remove a director with or without cause, unless the articles of incorporation or a bylaw adopted by the shareholders only allow removal for cause. Under VSI Holdings' bylaws, a majority vote of the shareholders present or represented by proxy at a special meeting may remove a director with or without cause.

SPECIAL MEETINGS OF SHAREHOLDERS

Both the Delaware law and the Georgia law provide that the board or any other person authorized in the corporation's certificate or articles of incorporation or bylaws may call a special meeting of shareholders. In addition, the Georgia law provides that a special meeting of shareholders may also be called by the holders of at least 25%, or a greater or lesser percentage as the articles of incorporation or bylaws provide, of all votes entitled to be cast on any issue proposed to be considered at a special meeting.

Under the SPX governing documents, only the chairman, the president or a majority of the board may call special meetings of the stockholders. The VSI Holdings bylaws allow a special meeting to be called by the chairman of the board, the president, the board of directors or the holders of at least one-tenth of all the shares entitled to vote at the special meeting.

QUORUM FOR SHAREHOLDERS' MEETING

VSI Holdings' bylaws require the holders of a majority of the shares issued and outstanding to be present in person or by proxy to establish quorum. SPX's bylaws provide that the holders of one-third of the stock issued and outstanding and entitled to vote at the meeting must be present in person or by proxy to establish a quorum.

CORPORATE ACTION WITHOUT A SHAREHOLDERS' MEETING

Under the Delaware law and the Georgia law, shareholders can act by written consent in lieu of voting at a shareholders' meeting. The Delaware law permits a corporation to eliminate stockholder action by written consent in its certificate of incorporation. SPX's certificate of incorporation prohibits stockholders from acting by written consent in lieu of a meeting. VSI Holdings' bylaws allow for shareholder action by written consent without a meeting, without prior notice and without a vote.

PREFERRED STOCK

SPX's certificate of incorporation and VSI Holdings' articles of incorporation both authorize the issuance of preferred stock. SPX has authorized the issuance of 3,000,000 preferred shares and VSI Holdings has authorized the issuance of 2,000,000 preferred shares. Neither company has issued any preferred shares.

STOCK EXCHANGE LISTING

SPX common stock is listed on the NYSE and the Pacific Exchange under the symbol "SPW". VSI Holdings common stock is listed on the AMEX under the symbol "VIS".

FUNDAMENTAL CORPORATE CHANGES

The Delaware law requires that the holders of a majority of the outstanding voting shares of the acquiring and target corporations approve statutory mergers. It does not provide explicitly for share exchanges. The Delaware law does not require a stockholder vote of the surviving corporation in a merger, unless the certificate of incorporation provides otherwise, if:

- the merger agreement does not amend the surviving corporation's certificate of incorporation;
- each share of the surviving corporation outstanding before the merger is equal to an identical outstanding or treasury share of the surviving corporation after merger; and
- the number of shares to be issued by the surviving corporation in the merger does not exceed 20% of the shares outstanding immediately prior to the merger.

The Georgia law requires that holders of a majority of the voting shares of the acquiring and target corporations approve statutory mergers and share exchanges. It contains an exception to its voting requirements for mergers and share exchanges where:

- the articles of incorporation of the surviving corporation will not differ from its articles of incorporation before the merger;
- each shareholder of the surviving corporation whose shares were outstanding immediately before the merger will hold the same number and type of shares immediately after the merger; and
- the number and kind of shares outstanding immediately after the merger, plus the number and kind of shares issuable as a result of the merger, do not exceed the total number and kind of shares of the surviving corporation authorized by its articles of incorporation immediately before the merger.

Both the Delaware law and the Georgia law require that a sale of all or substantially all of a corporation's assets be approved by a majority of the outstanding voting shares of the corporation transferring the assets.

BUSINESS COMBINATIONS WITH SUBSTANTIAL SHAREHOLDERS

Unless specified conditions are met, business combinations by corporations with interested shareholders are subject to a moratorium of three years under the Delaware law and five years under the Georgia law. Prohibited business combinations include a merger with, disposition of assets to, or the issuance of stock to, the interested shareholder, or specified transactions that have the effect of increasing

the proportionate amount of the outstanding securities held by an interested shareholder. The law considers the beneficial owner of 10% or more of the voting power of a company an interested shareholder.

The Delaware law prohibits a corporation from engaging in a business combination with an interested stockholder for three years following the date that the stockholder became an interested stockholder, unless:

- prior to that date, the corporation's board approved either the business combination or the transaction that resulted in the stockholder becoming an interested stockholder;
- upon consummation of the transaction that resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the corporation's voting stock; or
- on or subsequent to that date, the board approves the business combination and the stockholders authorize it at an annual or special meeting, and not by written consent, by the affirmative vote of at least two-thirds of the outstanding voting stock not owned by the interested stockholder.

The Delaware law permits a corporation to opt-out of this business combination statute. SPX's governing documents do not exclude SPX from the restrictions imposed by the Delaware law.

Under the SPX certificate of incorporation, a substantial stockholder is any stockholder who directly or indirectly beneficially owns 10% or more of the voting power, an affiliate who within the previous two years directly or indirectly beneficially owned 10% of more of the voting power, or an assignee of these beneficial owners and affiliates. SPX's certificate of incorporation requires that a supermajority of SPX stockholders approve certain business combinations with substantial stockholders. Except as described below, 80% of the voting power of the SPX stockholders, voting together as a single class, must approve the following business combinations:

- a merger of SPX or any of its subsidiaries with either a substantial stockholder or a corporation that is, or after the merger would be, an affiliate or associate (each as defined in the Securities Exchange Act of 1934) of a substantial stockholder;
- a sale, lease, exchange, mortgage, pledge, transfer or other disposition to or with any substantial stockholder, its affiliate or associate, of assets or a subsidiary of SPX, in each case having an aggregate fair market value of \$10 million or more;
- the issuance or transfer by SPX or a subsidiary of any securities of SPX or any of its subsidiaries to a substantial stockholder, or its affiliate or associate, in exchange for cash, securities or other consideration having an aggregate fair market value of \$10 million or more;
- the adoption of any plan or proposal for the liquidation or dissolution of SPX proposed by or on behalf of any substantial stockholder, or its affiliate or associate; or
- any reclassification of securities (including any reverse stock split), recapitalization of SPX, a merger or consolidation of SPX with a subsidiary, or any other transaction (whether or not involving a substantial stockholder) that directly or indirectly increases the proportionate share of the outstanding shares of any class of equity or convertible securities of SPX or any of its subsidiaries that a substantial stockholder, or its affiliate or associate, directly or indirectly owns.

The 80% vote requirement does not apply:

- to business combinations where SPX stockholders do not receive any cash or other consideration, solely in their capacities as stockholders, and the business combination has been approved by two-thirds of the SPX directors who were in office before the substantial stockholder became a substantial stockholder and who are not affiliates or representatives of the substantial stockholder, or

- to all other business combinations that two-thirds of the continuing directors then in office have approved and that satisfy all of the following conditions:
 - -- The aggregate amount of cash and the fair market value as of the date of the consummation of the business combination of the consideration other than cash to be received per share by SPX common stockholders is at least equal to the highest of the following:
 - (1) the highest per share price (including any brokerage commissions, transfer taxes and soliciting dealers' fees) paid to acquire any SPX shares beneficially owned by the substantial stockholder that were acquired within the two-year period immediately prior to the first public announcement of the proposal of the business combination or in the transaction in which it became a substantial stockholder, whichever is higher, plus interest compounded annually from the date when the substantial stockholder became a substantial stockholder through the consummation date of the business combination at the prime rate of interest of Harris Trust and Savings Bank from time to time in effect in Chicago, Illinois, less the aggregate amount of any cash dividends paid, and the fair market value of any non-cash dividends paid, per share of SPX common stock from the date when the substantial stockholder became a substantial stockholder through the consummation date of the business combination in an amount up to but not exceeding the amount of the interest; or
 - (2) the fair market value per share of SPX common stock on the date the business combination was first publicly announced or on the date on which the substantial stockholder became a substantial stockholder, whichever is higher; or
 - (3) the higher of the two numbers referred to in clause (2) above, multiplied by a fraction, the numerator of which is the highest per share price (including any brokerage commissions, transfer taxes and soliciting dealers' fees) paid in order to acquire any shares of SPX common stock beneficially owned by the substantial stockholder that were acquired within the two-year period immediately prior to the date when the business combination was announced, and the denominator of which is the fair market value per share of SPX common stock on the first day in the two-year period on which the substantial stockholder beneficially owned any shares of SPX common stock.
 - -- The consideration that SPX stockholders will receive will be in cash or in the same form as the substantial stockholder has previously paid for shares of the same stock. If the substantial stockholder has used multiple forms of consideration, then the form of consideration for the business combination will be cash or the form used to acquire the largest number of shares that the substantial stockholder previously acquired.
 - -- Except as approved by at least two-thirds of the directors who were in office when the substantial stockholder became a substantial stockholder and who are not affiliates or representatives of the substantial stockholder, between the time that a substantial stockholder becomes a substantial stockholder and the consummation of the business combination, there has been no failure to declare and pay periodic dividends with respect to any preferred stock outstanding and no reduction in the annual rate of dividends paid on SPX common stock (with certain exceptions).
 - -- After becoming a substantial stockholder, the substantial stockholder must not have received any loans, advances, guarantees, pledges or other financial assistance or any tax credits or other tax advantages from SPX or any of its subsidiaries, whether in anticipation of or in connection with the business combination or otherwise.
 - -- A proxy or information statement describing the proposed business combination and complying with the requirements of the Exchange Act has been mailed to public stockholders of SPX at least 30 days prior to the consummation of the business

combination, whether or not the proxy or information statement is required to be mailed pursuant to the Exchange Act.

The provision of the Georgia law governing business combinations with substantial shareholders is similar to the Delaware law provision. Georgia corporations may adopt a provision requiring that certain business combinations be approved by a special vote of the board of directors and/or the shareholders unless certain fair pricing criteria are met. They may also adopt a provision requiring a supermajority vote for business combinations with interested shareholders. VSI Holdings' governing documents do not contain these types of provisions.

Under the Georgia law, an interested shareholder may avoid the prohibition against effecting significant transactions with the corporation if the board of directors, prior to the time the shareholder becomes an interested shareholder, approves the transaction or the transaction by which the shareholder becomes an interested shareholder.

INDEMNIFICATION

Delaware and Georgia have similar laws relating to indemnification by a corporation of its officers, directors, employees and other agents. Under the Delaware law, a corporation may indemnify directors, officers, employees and agents for actions taken in good faith and in a manner they reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe the person's conduct was unlawful. The Delaware law provides that a corporation may advance expenses of defense (upon receipt of a written undertaking from the person seeking the advance to reimburse the corporation if indemnification is not appropriate) and must reimburse a successful defendant for expenses, including attorneys' fees, actually and reasonably incurred. The law permits a corporation to have liability insurance for its directors and officers. Under the Delaware law, a corporation may not indemnify for any claim, issue or matter as to which a person has been adjudged by a court of competent jurisdiction, after exhaustion of all appeals, to be liable to the corporation, unless and only to the extent that a court determines that the person is entitled to indemnity for the expenses.

The SPX certificate of incorporation provides that SPX will indemnify directors and officers of SPX and those serving at the request of SPX as a director, officer, employee or agent of another corporation or entity to the fullest extent authorized by the Delaware law. This includes the right to be reimbursed by SPX for expenses incurred in defending any proceeding in advance of its final disposition. The SPX certificate of incorporation provides that the indemnification rights provided therein are not exclusive of any other right to which persons seeking indemnification may be entitled under any law, bylaw, agreement, vote of stockholders or disinterested directors or otherwise. The certificate of incorporation authorizes SPX to purchase and maintain insurance on behalf of its directors and officers.

The Georgia law provides that a corporation may indemnify a director if he or she acted in good faith and reasonably believed that his or her actions were in the corporations' best interests. The law also allows a corporation to advance expenses upon receiving written confirmation that the relevant standard of conduct (good faith and reasonable belief) has been met. Under the Georgia law, the corporation may purchase and maintain liability insurance for its directors and officers.

VSI Holdings' governing documents provide indemnification for directors, officers, agents or employees of the corporation or persons who served as directors, officers, agents, employees, partners or trustees of another corporation or enterprise at the corporation's request. VSI Holdings' governing documents provide that the right of indemnification is not exclusive of any other right to which persons seeking indemnification may be entitled under the articles of incorporation, any other bylaw, agreement, vote of shareholders or law.

VSI Holdings' articles of incorporation provide indemnification in a derivative suit if the person is successful on the merits or if he or she acted in good faith in the transaction that is the subject of the suit and under a reasonable belief that he or she was advancing the corporation's best interests. VSI Holdings

will pay the expenses reasonably incurred in defense of the derivative suit (including attorneys' fees but excluding amounts paid in settlement), except where the person has been found guilty of negligence or misconduct in the performance of his duty to VSI Holdings. A plea of guilty or nolo contendere will not be deemed to be guilt of negligence or misconduct.

VSI Holdings' articles of incorporation provide indemnification in a nonderivative suit if the person is successful on the merits or if he or she acted in good faith in the transaction that is the subject of the suit and under a reasonable belief that he or she was acting in the corporation's best interests. VSI Holdings will pay the expenses reasonably incurred by the person in defending the nonderivative suit, including attorneys' fees, amounts paid in settlement, judgements and fines.

A court may determine whether a person has met the good faith and reasonable belief standard or the determination may be made by a majority of the VSI Holdings directors who were not parties to the action, by independent legal counsel in a written opinion or by the VSI Holdings shareholders. A court, the board of directors or the VSI Holdings shareholders may also prorate indemnification amounts based on whether or not a person has met the good faith and reasonable belief standard for all matters.

VSI Holdings may advance any payment of indemnification expenses. The board of directors must authorize the specific advance payment, and the person receiving the payment must promise in writing that he or she will repay any expenses advanced unless it is ultimately determined that he or she is entitled to indemnification.

VSI Holdings may purchase and maintain insurance on behalf of any person who holds or who has held any position whether or not the corporation would have power to indemnify him against that liability. Insurance purchases and any type of indemnification payment must be reported in writing to the VSI Holdings shareholders with the next notice of the annual meeting or within six months, whichever is sooner.

LIMITATION OF LIABILITY

Through provisions in the certificate of incorporation, the Delaware law generally allows a corporation to limit a director's personal liability to the corporation or its stockholders for monetary damages for breach of his or her fiduciary duty. The provisions may not, however, eliminate or limit the liability of a director:

- for any breach of his or her duty of loyalty to the corporation or its stockholders;
- for acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of law;
- under Section 174 of the Delaware law concerning unlawful dividends and stock repurchases and redemptions; or
- for any transaction from which the director derived an improper personal

The provision does not eliminate or limit the liability of a director for any act or omission occurring prior to the date when the provision becomes effective. These limitation of liability provisions do not affect the availability of nonmonetary remedies like injunctive relief or rescission.

The SPX certificate of incorporation provides for the limiting of directors' liability. It also provides that any repeal or modification of this provision by the SPX stockholders will not adversely affect any right or protection of a director of SPX that exists at the time of the repeal or modification.

The Georgia law permits a corporation to limit director liability in its articles of incorporation. VSI Holdings' articles of incorporation do not contain this type of provision.

DIVIDENDS

The Delaware law permits a corporation to declare and pay dividends out of surplus. If there is no surplus, a corporation can pay dividends out of the net profits for the fiscal year in which the dividend is declared and/or for the preceding fiscal year as long as the amount of capital of the corporation following the declaration and payment of the dividend is not less than the aggregate amount of the capital represented by the issued and outstanding stock of all classes having a preference upon the distribution of assets. The SPX certificate of incorporation permits SPX to pay dividends to common stockholders only after it has paid dividends to preferred stockholders who are entitled to dividends.

Under the Georgia law, a corporation may make distributions to its shareholders subject to any restrictions imposed in its articles of incorporation, except that a corporation may not make a distribution if, after giving it effect:

- the corporation would not be able to pay its debts as they become due in the usual course of business, or
- the corporation's total assets would be less than the sum of its total liabilities plus, unless the articles of incorporation permit otherwise, the amount that would be needed, if the corporation were to be dissolved at the time of the distribution, to satisfy the preferential rights upon dissolution of shareholders whose preferential rights are superior to those receiving the distribution.

The VSI Holdings articles of incorporation permit the payment of dividends to common shareholders out of any funds of the corporation legally available for distribution as dividends, subject to the rights of any preferred shareholders.

REDEMPTION AND REPURCHASE OF STOCK

The Delaware law generally allows a corporation to redeem or repurchase its shares if the redemption or repurchase would not impair the capital of the corporation. A Delaware corporation may also redeem or repurchase shares having a preference upon the distribution of any of its assets, or shares of common stock, if there are no shares of preferred stock, if the shares will be retired upon acquisition, and provided that, after the reduction in capital made in connection with the retirement of shares, the corporation's remaining assets are sufficient to pay any debts not otherwise provided for.

A Georgia corporation may acquire its own shares. Reacquired shares are considered authorized but unissued shares, unless the articles of incorporation provide that the shares become treasury shares or prohibit the reissuance of reacquired shares. If the reissuance is prohibited, the number of authorized shares will be reduced by the number of shares reacquired.

The VSI Holdings and SPX governing documents do not address the redemption and repurchase of common stock.

NO DISSENTERS' RIGHTS

Under both the Delaware law and the Georgia law, a shareholder who participates in specified major corporate transactions may be entitled to dissenters' or appraisal rights pursuant to which the shareholder may receive cash in the amount of the fair value of his or her shares in lieu of the consideration he or she would otherwise receive in the transaction.

Under the Delaware law, appraisal rights are generally available to stockholders entitled to vote with respect to mergers and consolidations. Stockholders do not, however, have appraisal rights with respect to a merger or consolidation by a corporation when the corporation's shares are either listed on a national securities exchange or held of record by more than 2,000 stockholders so long as the stockholders are not required to accept anything other than shares of the surviving corporation, shares of any other corporation that are either listed on a national securities exchange or held of record by more than 2,000 holders, cash in lieu of fractional shares or a combination of the foregoing. Delaware law permits a corporation to expand these statutory rights pursuant to its certificate of incorporation, but SPX has not done so.

- a merger, if shareholder approval is required for the merger and the shareholder is entitled to vote on the merger, or if the corporation is a subsidiary that is merged with its parent;
- a share exchange in which the corporation's shares will be acquired, if the shareholder is entitled to vote on the share exchange;
- a sale or exchange of all or substantially all of a corporation's
 assets, if a shareholder vote is required, other than a sale pursuant to
 a court order or a sale for cash the proceeds of which will be
 distributed to the shareholders within one year;
- an amendment of the articles of incorporation that adversely affects rights relating to the shareholder's shares; or
- any corporate action taken pursuant to a shareholder vote to the extent the articles of incorporation, bylaws or a resolution of the board of directors provide that voting or non-voting shareholders are entitled to dissent and obtain payment for their shares.

The Georgia law provides that if a shareholder is entitled to exercise dissenters' rights, those rights constitute the shareholder's exclusive remedy in the absence of fraud or failure to comply with certain procedural requirements.

Dissenters' rights are not available under the Georgia law when the affected shares are listed on a national securities exchange or held of record by more than 2,000 shareholders unless:

- the articles of incorporation or a resolution of the board of directors approving the transaction provide otherwise; or
- in a plan of merger or share exchange, the shareholders are required to accept anything other than shares of the surviving corporation or another publicly held corporation that at the effective date of the merger or share exchange are either listed on a national securities exchange or held of record by more than 2,000 shareholders, except for payments in lieu of fractional shares.

SHAREHOLDER DERIVATIVE SUITS

Under the Delaware law and the Georgia law, a shareholder may only bring a derivative action on behalf of the corporation if he or she was a shareholder at the time of the transaction in question or if the stock was transferred to him or her by operation of law. In addition, the Georgia law requires that the shareholder fairly and adequately represent the corporation's interests in enforcing its rights.

DISSOLUTION

Under the Delaware law, a corporation may be dissolved in two ways. First, a majority of the board along with a majority of the voting stockholders can approve a dissolution. Second, the vote of 100% of the voting stockholders can approve a dissolution. A Delaware corporation's certificate of incorporation can impose a supermajority voting requirement for a dissolution initiated by the board of directors. The SPX certificate of incorporation does not impose any supermajority vote requirement for dissolution.

Dissolution under the Georgia law requires the approval of the board of directors, unless the board elects not to make a recommendation because of a conflict of interest or other special circumstances, and a majority of the stockholders entitled to vote. Under Georgia law, the articles of incorporation may require a greater vote or voting by groups for dissolution. The VSI Holdings articles of incorporation do not impose any supermajority vote requirement for dissolution.

PREEMPTIVE RIGHTS

Under both the Delaware law and the Georgia law, preemptive rights are not available to stockholders unless specifically authorized by the certificate or articles of incorporation. Both the SPX certificate of

incorporation and the VSI Holdings articles of incorporation provide that shareholders will not have any preemptive rights.

SHAREHOLDER RIGHTS PLANS

SPX entered into a rights agreement with The Bank of New York, as rights agent, on June 25, 1996, as amended October 22, 1997. Under the SPX rights agreement, the board declared a dividend of one right for each outstanding share of SPX common stock held of record on June 25, 1996. Each SPX right entitles its holder to purchase, upon the occurrence of certain specified events, one one-thousandth of a share of SPX series A junior participating preferred stock, at an exercise price of \$200, subject to adjustment. At no time do the holders of SPX rights have any voting rights.

The SPX rights will no longer be exercisable after the earlier of June 25, 2006, the redemption of the rights or the exchange of the rights for SPX common stock. The description and terms of the SPX rights are set forth in the SPX rights agreement. In general, pursuant to the rights agreement, upon the occurrence of specified triggering events, for example, the acquisition by any person (other than SPX or any of its subsidiaries) of the beneficial ownership of securities representing 20% or more of the outstanding SPX common stock without the prior approval of SPX's board, each holder of an SPX right will have the right to receive, upon exercise of the SPX right, that amount of SPX common stock having a market value equal to two times the exercise price of the SPX right.

The SPX rights agreement further provides that if SPX is acquired in a merger or other business combination or sells more than 50% of its assets and the transaction is not approved by SPX's board, the SPX stockholders will have the right to receive, with respect to each SPX right, common stock of the acquiring company having a value equal to two times the exercise price of the SPX right. Under certain circumstances, SPX may redeem the rights for a redemption price of \$.01 per right.

VSI Holdings has not adopted any shareholder rights plan.

INTERESTED DIRECTOR TRANSACTIONS

Under both the Delaware law and the Georgia law, contracts or transactions in which a corporation's director has an interest are not void or voidable because of that interest, so long as specified conditions are met, for example, obtaining required approvals and fulfilling the requirements of good faith and full disclosure. The conditions are substantially similar under the Delaware law and the Georgia law. The SPX certificate of incorporation requires disclosure of the interest to the board or board committee that will vote on the interested-director transaction. Similarly, the VSI Holdings bylaws require disclosure of the interest to the board of directors or shareholders voting on the interested-director transaction.

AMENDMENT OF CERTIFICATE OF INCORPORATION

Under the Delaware law, a corporation may amend its certificate of incorporation if, among other things, the amendment is approved by a majority of the outstanding stock entitled to vote approves the amendment. If, however, the provision of the certificate of incorporation requires an affirmative vote of directors or stockholders greater than that normally provided for by the Delaware law, that provision of the certificate of incorporation may not be amended or repealed without that greater vote.

The SPX certificate of incorporation provides that the holders of at least 80% of the voting shares of SPX must approve the amendment of Articles Eighth (number, classification and power of directors), Ninth (stockholder meetings) and Fifteenth (business combinations) of the certificate of incorporation. In addition, a majority of the voting shares can amend Article Fifteenth (business combinations) if at least two-thirds of the directors who are not affiliates or representatives of the substantial stockholder (as defined in the section above entitled Business Combinations with Substantial Stockholders) and who were directors before the substantial stockholder became a substantial stockholder approve the amendment.

Under the Georgia law, articles of incorporation may be amended if the board recommends the amendment and the shareholders approve the amendment by a majority vote of the shares eligible to vote. The VSI Holdings articles of incorporation do not alter these requirements.

AMENDMENT OF BYLAWS

Under the Delaware law, the stockholders have the power to adopt, amend or repeal bylaws unless the certificate of incorporation also confers that power upon the directors. SPX's certificate of incorporation, with limited exceptions, expressly authorizes the board of directors to adopt, amend and repeal the bylaws. The SPX bylaws provide that, subject to the provisions of the SPX certificate of incorporation, a majority of the board or a majority of the stockholders entitled to vote at a meeting may alter the bylaws. The SPX certificate of incorporation provides that at least 80% of the voting power of all SPX shares must amend or repeal specified sections of the SPX bylaws dealing with aspects of stockholder action and the election and removal of directors.

Under the Georgia law, the board of directors may generally amend or repeal bylaws, unless this power is reserved to shareholders in the articles of incorporation or if the bylaw involved provides for staggered terms for the directors, which only the shareholders may create or eliminate. VSI Holdings' governing documents provide that its board of directors has the power to alter, amend or repeal the bylaws, unless that power is modified or divested by a vote of a majority of the shareholders. VSI Holdings' shareholders have not modified or divested the board's authority to amend the VSI Holdings bylaws.

CONSTDERATION OF NON-STOCKHOLDER CONSTITUENCIES

Under the Delaware law, a board of directors may consider the impact of its decisions on constituencies other than stockholders. Non-stockholder constituencies may include creditors, customers, employees, and perhaps the community generally. The board may only consider the interests of other constituencies, however, if there are rationally related benefits to stockholders or there is some reasonable relationship to general stockholder interests. SPX's certificate of incorporation also provides that in determining whether an acquisition proposal is in the best interests of SPX and its stockholders, the SPX board must consider all factors it deems relevant, including:

- the consideration offered, not only in relation to the then current market price, but also in relation to the then current value of SPX in a freely negotiated transaction and in relation to the board's estimate of the future value of SPX as an independent entity; and
- the social, legal and economic effects on employees, suppliers, customers and the communities in which SPX is located, as well as on the long term business prospects of SPX.

 $\ensuremath{\mathsf{VSI}}$ Holdings' governing documents do not contain a non-stockholder constituencies provision.

LEGAL MATTERS

Gardner, Carton & Douglas, Chicago, Illinois, will pass on the validity of the shares of SPX common stock to be issued in connection with the merger.

WHERE YOU CAN FIND MORE INFORMATION

SPX and VSI Holdings file annual, quarterly and current reports, proxy statements and other information with the SEC. You may read and copy any reports, statements or other information that the companies file at the SEC's public reference rooms in Washington, D.C., New York, New York and Chicago, Illinois. Please call the SEC at 1-800-SEC-0330 for further information on the public reference rooms. SPX's and VSI Holdings' public filings are also available to the public from commercial document retrieval services and at the Internet site maintained by the SEC at http://www.sec.gov. You may also inspect reports, proxy statements and other information concerning SPX at the offices of the New York Stock Exchange, 20 Broad Street, New York, New York 10005 and at the offices of the Pacific Exchange

at 301 Pine Street, San Francisco, California 94104. Reports, proxy statements and other information concerning VSI Holdings are available for inspection at the offices of the American Stock Exchange, Inc., 86 Trinity Place, New York, New York 10006.

SPX has filed a registration statement on Form S-4 to register with the SEC the shares of SPX common stock to be issued to VSI Holdings shareholders in the merger. This proxy statement-prospectus is a part of that registration statement and constitutes a prospectus of SPX and a proxy statement of VSI Holdings for purposes of the VSI Holdings special meeting.

As allowed by SEC rules, this proxy statement-prospectus omits certain information contained in the registration statement or the exhibits to the registration statement. Any statements contained in this proxy statement-prospectus concerning the provisions of any other document are not necessarily complete, and, in each instance, reference is made to the copy of that document filed as an exhibit to the registration statement or otherwise filed with the SEC. Each statement is qualified in its entirety by that reference.

The SEC allows SPX and VSI Holdings to incorporate by reference information into this proxy statement-prospectus. This means that SPX can disclose important information by referring you to other documents that it has filed separately with the SEC. In addition, VSI Holdings can disclose important information by referring you to other documents that are being delivered to you with this proxy statement-prospectus. VSI Holdings has also filed these documents with the SEC. The information incorporated by reference is deemed to be part of this proxy statement-prospectus, except for any information superseded by information contained directly in the proxy statement-prospectus. This proxy statement-prospectus incorporates by reference the following documents:

SPX SEC FILINGS (FILE NO. 1-6948)

- annual report on Form 10-K for the fiscal year ended December 31, 2000 filed on March 9, 2001, as amended on Form 10-K/A filed on April 12, 2001;
- current reports on Form 8-K filed March 12, April 12 and April 13, 2001;
- definitive proxy statement on Schedule 14A dated March 27, 2001; and
- the description of SPX's capital stock (and the SPX preferred stock purchase rights) contained in SPX's registration statements pursuant to Section 12 of the Exchange Act and any amendments or reports filed for the purpose of updating any of those descriptions.

VSI HOLDINGS SEC FILINGS (FILE NO. 1-12942)

- annual report on Form 10-K for the fiscal year ended September 30, 2000;
- quarterly report on Form 10-Q for the quarter ended December 31, 2000;
- current report on Form 8-K filed March 30, 2001; and
- definitive proxy statement on Schedule 14A dated March 31, 2000.

Copies of VSI Holdings annual report on Form 10-K and quarterly report on Form 10-Q are being delivered to you with this proxy statement-prospectus.

SPX and VSI Holdings incorporate by reference additional documents that either company may file with the SEC under Sections 13(a), 13(c), 14 or 15(d) between the date of this proxy statement-prospectus and the date that the merger is consummated. These include periodic reports, such as Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q and Current Reports on Form 8-K, as well as proxy statements.

VSI Holdings has supplied all information contained or incorporated by reference into this proxy statement-prospectus relating to VSI Holdings, and SPX has supplied all information relating to SPX.

You can obtain without charge a copy of any SPX document or any VSI Holdings document incorporated by reference and not delivered with this proxy statement-prospectus, except for the exhibits to those documents, from the appropriate company. You may also obtain these documents from the SEC or through the SEC's Internet website described above. You may obtain documents incorporated by reference into but not delivered with this proxy statement-prospectus by requesting them in writing or by telephone from the appropriate company at the following addresses:

SPX Corporation 700 Terrace Point Drive Muskegon, Michigan 49443 Attention: Investor Relations (231) 724-5000 VSI Holdings, Inc. 41000 Woodward Avenue Bloomfield Hills, Michigan 48304 Attention: Investor Relations (248) 644-0500

If you would like to request documents from either company, please do so by , 2001 to receive them before the VSI Holdings special meeting. If you request any of these documents from us we will mail them to you by first-class mail or similar means.

YOU SHOULD RELY ONLY ON THE INFORMATION CONTAINED IN, DELIVERED WITH OR INCORPORATED BY REFERENCE INTO THIS PROXY STATEMENT-PROSPECTUS IN VOTING YOUR SHARES AT THE VSI HOLDINGS SPECIAL MEETING. VSI HOLDINGS AND SPX HAVE NOT AUTHORIZED ANYONE TO PROVIDE YOU WITH INFORMATION THAT IS DIFFERENT FROM WHAT IS CONTAINED IN THIS PROXY STATEMENT-PROSPECTUS. THIS PROXY STATEMENT-PROSPECTUS IS DATED , 2001. You should not assume that the information contained in the proxy statement-prospectus is accurate as of any other date, and neither the mailing of this proxy statement-prospectus to VSI Holdings' shareholders nor the issuance of SPX's securities in the merger will create any implication to the contrary.

EXPERTS

The consolidated balance sheets of SPX as of December 31, 2000 and 1999 and the related consolidated statements of income and comprehensive income, shareholders' equity and cash flows for each of the three years in the period ended December 31, 2000 incorporated in this proxy statement-prospectus by reference to the Annual Report on Form 10-K/A have been audited by Arthur Andersen LLP, independent public accountants, as indicated in their report dated February 9, 2001 with respect thereto, and are incorporated by reference herein in reliance upon the authority of said firm as experts in accounting and auditing.

The consolidated financial statements of VSI Holdings appearing in VSI Holdings' 2000 Form 10-K have been audited by Plante & Moran, LLP, independent auditors, as set forth in their report thereon included therein and incorporated herein by reference. These consolidated financial statements are incorporated herein in reliance upon that report given upon the authority of that firm as experts in accounting and auditing.

The consolidated financial statements of United Dominion incorporated in this proxy statement-prospectus by reference to the Annual Report on Form 40-F of United Dominion for the three years ended December 31, 2000 and Form 8-K of SPX dated April 13, 2001 have been audited by KPMG LLP, independent public accountants, as indicated in their report with respect thereto, and are incorporated by reference herein in reliance upon the authority of said firm as experts in accounting and auditing.

APPENDIX A

AGREEMENT AND PLAN OF MERGER
BY AND BETWEEN
SPX CORPORATION
AND
VSI HOLDINGS, INC.
Dated as of March 24, 2001

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (this "Agreement") is made and entered into as of March 24, 2001, by and between VSI Holdings, Inc., a Georgia corporation (the "Company") and SPX Corporation, a Delaware corporation ("Purchaser").

WITNESSETH:

WHEREAS, the respective Boards of Directors of Purchaser and the Company have adopted and approved this Agreement (which includes the "plan of merger" contemplated by the applicable provisions of the Georgia Business Corporation Code (the "GBCC")) and the merger (the "Merger") of the Company with and into Purchaser in accordance with the laws of the State of Georgia and the laws of the State of Delaware and the provisions of this Agreement; and

WHEREAS, for United States federal income tax purposes, it is intended that the Merger provided for herein shall qualify as a reorganization under Section 368(a) of the Internal Revenue Code of 1986, as amended, and the rules and regulations promulgated thereunder (the "Code"), and this Agreement is intended to be and is adopted as a plan of reorganization within the meaning of Section 368 of the Code; and

WHEREAS, the Company and Purchaser desire to make certain representations, warranties and agreements in connection with, and establish various conditions precedent to, the Merger.

WHEREAS, the Purchaser, Steve Toth, Jr. ("Toth") and certain other shareholders of the Company are entering into an Indemnity and Restriction Agreement contemporaneously herewith ("Indemnity and Restriction Agreement");

WHEREAS, PII Ventures, L.L.C., a Michigan limited liability company, the Purchaser and Toth are entering into an Asset Purchase Agreement contemporaneously herewith; and

NOW, THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements hereinafter set forth, and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE 1

TERMS OF THE MERGER

- 1.1 The Merger. Upon the terms and subject to the conditions of this Agreement, the Merger shall be consummated in accordance with the GBCC and the Delaware General Corporation Law (the "DGCL"). Except as provided below, at the Effective Time (as defined below), upon the terms and subject to the conditions of this Agreement, the Company shall be merged with and into Purchaser in accordance with the GBCC and the DGCL and the separate existence of the Company shall thereupon cease, and Purchaser, as the surviving corporation in the Merger (the "Surviving Corporation"), shall continue its corporate existence under the laws of the State of Delaware. The parties shall prepare and the Surviving Corporation shall execute and file (i) a mutually acceptable Certificate of Merger (the "Georgia Certificate of Merger") with the Secretary of State of Georgia in order to comply in all respects with the requirements of the GBCC and with the provisions of this Agreement, and (ii) a mutually acceptable Certificate of Merger (the "Delaware Certificate of Merger" and together with the Georgia Certificate of Merger, the "Certificates of Merger") with the Secretary of State of Delaware in order to comply in all respects with the requirements of the DGCL and with the provisions of this Agreement. Not later than the next business day after the filing the Georgia Certificate of Merger, the Surviving Corporation shall comply with Section 14-2-1105.2 of the GBCC.
- 1.2 The Closing; Effective Time. (a) The closing of the Merger (the "Closing") shall take place at the offices of Miller, Canfield, Paddock and Stone, P.L.C. in Troy, Michigan, at 10:00 a.m. local time on a date to be specified by the parties which shall be no later than the third business day after the date that all

of the closing conditions, except for conditions which, by their terms, are to be satisfied by deliveries made on the Closing Date, set forth in Article 6, have been satisfied or waived (if waivable), unless another time, date or place is agreed upon in writing by the parties hereto.

- (b) The Merger shall become effective at the time of the filing of the Georgia Certificate of Merger with the Secretary of State of the State of Georgia in accordance with the applicable provisions of the GBCC and the filing of the Delaware Certificate of Merger with the Secretary of State of the State of Delaware in accordance with the applicable provisions of the DGCL, or at such later time as may be specified in the Certificates of Merger. The time when the Merger shall become effective is herein referred to as the "Effective Time" and the date on which the Effective Time occurs is herein referred to as the "Closing Date."
- 1.3 Merger Consideration. (a) Subject to the provisions of this Agreement and any applicable backup or other withholding requirements, each of the issued and outstanding shares (the "Company Shares") of common stock, \$.01 par value, of the Company (the "Company Stock") outstanding immediately prior to the Effective Time (except for Company Shares to be cancelled as set forth in Section 1.3(d) shall be converted, by virtue of the Merger and without any action on the part of the holder thereof, into the right to receive the Purchaser Stock Consideration or the combination of Purchaser Stock Consideration and Cash Consideration, without any interest thereon, as specified in Section 1.5 hereof, subject to payment of cash in lieu of any fractional share as hereinafter provided (the "Merger Consideration"). For purposes hereof, the following terms have the following respective meanings:

"Cash Consideration" means a per Company Share amount in cash equal to \$4.35.

"Outstanding Company Shares" means the Company Stock outstanding immediately prior to the Effective Time, including, without limitation, restricted shares referred to in Section 1.10(b), but excluding Company Shares to be cancelled pursuant to Section 1.3(d).

"Purchaser Rights" means Rights to purchase Series A Junior Participation Preferred Stock of Purchaser distributed to holders of Purchaser Stock pursuant to the Rights Agreement dated June 25, 1996, as amended, between the Purchaser and Bank of New York, as Rights Agent.

"Purchaser Stock" means the common stock, \$10 per share, of Purchaser and associated Purchaser Rights. For clarification, each reference herein to Purchaser Stock shall include the associated Purchaser Rights.

"Purchaser Stock Consideration" means .043 of a share of Purchaser Stock. In the event that between the date of this Agreement and the Effective Time, the issued and outstanding shares of Purchaser Stock shall have been affected or changed into a different number of shares or a different class of shares as a result of a stock split, reverse stock split, stock dividend, spin-off, extraordinary dividend, recapitalization, reclassification, subdivision, combination of shares or other similar transaction, or there shall have been a record date declared for any such matter, the Purchaser Stock Consideration shall be appropriately adjusted as mutually agreed upon by Purchaser and the Company.

- (b) No fractional shares of Purchaser Stock shall be issued pursuant to the Merger nor will any fractional share interest involved entitle the holder thereof to vote, to receive dividends or to exercise any other rights of a shareholder of Purchaser. In lieu thereof, any holder of Company Stock who would otherwise be entitled to a fractional share of Purchaser Stock pursuant to the provisions hereof shall receive an amount in cash pursuant to Section 1.5(g) hereof.
- (c) Each share of Purchaser Stock and any and all shares of Preferred Stock of Purchaser, if any, outstanding immediately prior to the Merger shall remain issued and outstanding.
- (d) Any shares of Company Stock owned by Purchaser, or any other wholly owned subsidiaries of Purchaser or held in the treasury of the Company immediately prior to the Merger shall be cancelled and retired at the Effective Time and shall cease to exist and no Purchaser Stock or other consideration shall be delivered in exchange therefor.

- (e) On and after the Effective Time, holders of certificates representing shares of Company Stock (the "Certificates") immediately prior to the Effective Time shall cease to have any rights as stockholders of the Company, except the right to receive the Merger Consideration for each Company Share held by them.
- 1.4 Election Procedure. Each holder of Company Shares (other than holders of Company Shares to be cancelled as set forth in Section 1.3(d)) shall have the right to submit a request specifying either that such holder's Company Shares shall be converted into the Purchaser Stock Consideration or that such holder's Company Shares shall be converted into a combination of Cash Consideration and Purchaser Stock Consideration, without interest, in the Merger in accordance with the following procedure:
 - (a) Each holder of Company Shares may specify in a request made in accordance with the provisions of this Section 1.4 (herein called an "Election") to either: (i) convert one-hundred percent (100%) of the Company Shares owned by such holder into the right to receive the Purchaser Stock Consideration in the Merger (a "Stock Election"), or (ii) convert forty-five percent (45%) of the Company Shares owned by such holder into the right to receive the Cash Consideration in the Merger, and fifty-five percent (55%) of the Company Shares owned by each such holder into the right to receive the Purchaser Stock Consideration in the Merger (a "Cash/Stock Election"). A Form of Election shall be included with each copy of the Prospectus/Proxy Statement (as defined in Section 1.15) mailed to stockholders of the Company in connection with the meeting of stockholders called to consider the Company Proposals (as defined in Section 1.15). Purchaser and the Company shall each use its reasonable best efforts to mail or otherwise make available the Form of Election to all persons who become holders of Company Shares during the period between the record date for such stockholder meeting and the Election Deadline (as defined in Section 1.4(d)).
 - (b) Purchaser shall prepare a form (the "Form of Election"), which shall be in form and substance acceptable to the Company, pursuant to which each holder of Company Shares at the close of business on the Election Deadline may make an Election and which shall be mailed to the Company's stockholders in accordance with Section 1.4(a) so as to permit the Company's stockholders to exercise their right to make an Election prior to the Election Deadline.
 - (c) Holders of record of Company Shares who hold such shares as nominees, trustees, or in other representative capacities may submit multiple Forms of Election, provided that such representative certifies that each Form of Election covers all Company Shares held by such representative for a particular beneficial owner.
 - (d) Not later than the filing of the Prospectus/Proxy Statement (as defined below) with the SEC (as defined below), as contemplated in Section $\,$ 1.15(a) hereof, the Purchaser shall appoint a bank acceptable to the Company as the person to receive Forms of Election and to act as exchange agent under this Agreement, which bank shall be acceptable to the Company (the "Exchange Agent"). Any Company stockholder's Election shall have been made properly only if the Exchange Agent shall have received, by 5:00 p.m. local time in the city in which the principal office of such Exchange Agent is located, on the date of the Election Deadline, a Form of Election properly completed and signed and accompanied by certificates for the Company Shares to which such Form of Election relates (or by an appropriate guarantee of delivery of such certificates, as set forth in such Form of Election, from a member of any registered national securities exchange or of the National Association of Securities Dealers, Inc. or a commercial bank or trust company in the United States provided such certificates are in fact delivered to the Exchange Agent by the time required in such guarantee of delivery). Failure to deliver Company Shares covered by such a guarantee of delivery within the time set forth on such guarantee shall be deemed to invalidate any otherwise properly made Election. As used herein, "Election Deadline" means the date announced by Purchaser (which date shall be agreed upon by the Company), as the last day on which Forms of Election will be accepted; provided, that such date shall be a business day no earlier than ten (10) business days prior to the Effective Time and no later than the date on which the Effective Time occurs. In the event this Agreement shall have been terminated prior to the Effective Time, the Exchange Agent shall

immediately return all Election Forms and certificates for Company Shares to the appropriate Company stockholders.

- (e) Any Company stockholder may at any time prior to the Election Deadline change his Election by written notice received by the Exchange Agent prior to the Election Deadline accompanied by a revised Form of Election properly completed and signed.
- (f) Any Company stockholder may, at any time prior to the Election Deadline, revoke his Election by written notice received by the Exchange Agent prior to the Election Deadline or by withdrawal prior to the Election Deadline of his certificates for Company Shares, or of the guarantee of delivery of such certificates, previously deposited with the Exchange Agent. All Elections shall be revoked automatically if the Exchange Agent is notified in writing by Purchaser or the Company that this Agreement has been terminated. Any Company stockholder who shall have deposited certificates for Company Shares with the Exchange Agent shall have the right to withdraw such certificates by written notice received by the Exchange Agent prior to the Election Deadline and thereby revoke his Election as of the Election Deadline if the Merger shall not have been consummated prior thereto.
- (g) Purchaser shall have the right to make rules, not inconsistent with the terms of this Agreement, governing the validity of the Forms of Election, the manner and extent to which Elections are to be taken into account in making the determinations prescribed by Section 1.5, the issuance and delivery of certificates for Purchaser Stock into which Company Shares are converted in the Merger and the payment of cash for Company Shares converted into the right to receive the Cash Consideration in the Merger.
- 1.5 Issuance of Purchaser Stock Consideration and Payment of Cash Consideration; Proration. The manner in which each Company Share (except Company Shares to be cancelled as set forth in Section 1.3(d)) shall be converted into the Purchaser Stock Consideration or the right to receive a combination of Purchaser Stock Consideration and Cash Consideration at the Effective Time shall be as set forth in this Section 1.5.
 - (a) As is more fully set forth below, the number of Company Shares to be converted into the right to receive the Cash Consideration in the Merger pursuant to this Agreement shall not exceed the lesser of (i) 45% of all Outstanding Company Shares, or (ii) the maximum number of Company Shares permissible under Sec. 3.02 of Rev. Proc. 77-37, 1977-2, C.B. 568, including for this purpose only as Company Shares the amount of the debt being repaid pursuant to Section 5.10 (the "Maximum Cash Amount").
 - (b) If Stock Elections and Cash/Stock Elections received providing for the conversion of Company Shares in the Merger into Purchaser Stock Consideration is greater than the amount of the Outstanding Company Shares minus the Maximum Cash Amount (the "Minimum Stock Amount"), each Company Share as to which the holder has elected to have converted into Purchaser Stock Consideration shall be converted in the Merger into the Purchaser Stock Consideration.
 - (c) If Cash/Stock Elections are received providing for the conversion of Company Shares in the Merger into Cash Consideration in an amount less than the Maximum Cash Amount, each Company Share covered by a Cash/Stock Election that is to be converted in the Merger into Cash Consideration shall be converted in the Merger into the right to receive the Cash Consideration
 - (d) If Cash/Stock Elections are received providing for conversion of Company Shares into Cash Consideration in an amount that is more than the Maximum Cash Amount, each Non-Electing Company Share (as defined in Section 1.5(f)) and each Company Share for which a Stock Election has been received shall be converted in the Merger into the Purchaser Stock Consideration, and the Company Shares for which Cash/Stock Elections have been received shall be converted into the right to receive the Cash Consideration and Purchaser Stock Consideration in the following manner:
 - (1) The Exchange Agent will distribute to each holder of Company Shares as to which a Cash/Stock Election has been made the Cash Consideration for a fraction of such Company

Shares, the numerator of which fraction shall be the Maximum Cash Amount and the denominator of which shall be the aggregate number of Company Shares covered by Cash/Stock Elections that have elected to be converted into Cash Consideration.

- (2) Company Shares covered by a Cash/Stock Election and not fully converted into the right to receive the Cash Consideration as set forth in clause (1) above shall be converted in the Merger into the right to receive a number of shares of Purchaser Stock equal to the Purchaser Stock Consideration for each such Company Share to offset the reduction in Cash Consideration affected pursuant to clause (1) above.
- (e) If Non-Electing Company Shares are not converted under Section 1.5(d), the Exchange Agent shall convert each Non-Electing Company Share into the Purchaser Stock Consideration.
- (f) For the purposes of this Section 1.5, Outstanding Company Shares as to which an Election is not in effect at the Election Deadline shall be called "Non-Electing Company Shares." If Purchaser shall determine that any Election is not properly made with respect to any Company Shares, such Election shall be deemed to be not in effect, and the Company Shares covered by such Election shall, for purposes hereof, be deemed to be Non-Electing Company Shares. The Purchaser and the Exchange Agent shall have no obligation to notify any person of any defect in any Form of Election submitted to the Exchange Agent.
- (g) Notwithstanding anything to the contrary contained herein, no certificates or scrip representing fractional shares of Purchaser Stock shall be issued upon the surrender for exchange of Certificates, no dividend or distribution with respect to Company Shares shall be payable on or with respect to any fractional share and such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a stockholder of Purchaser. In lieu of any such fractional share of Purchaser Stock, Purchaser shall pay to each former stockholder of the Company who otherwise would be entitled to receive a fractional share of Purchaser Stock an amount in cash determined by multiplying (i) \$101.17 by (ii) the fractional interest in a share of Purchaser Stock to which such holder would otherwise be entitled. If more than one Certificate shall be surrendered for the account of the same holder, the number of shares of Purchaser Stock to be issued to such holder in exchange for the Certificates shall be computed on the basis of the aggregate number of shares represented by all Certificates surrendered for the account of such holder. Any payment received by a holder of Company Shares with respect to fractional share interest is merely intended to provide a mechanical rounding off of, and not separately bargained for, consideration.
- (h) The Exchange Agent shall make all computations contemplated by this Section 1.5 and all such computations shall be conclusive and binding on the holders of Company Shares absent manifest error.
- 1.6 Issuance of Purchaser Stock Consideration. (a) Immediately prior to the Effective Time, Purchaser shall (i) deliver to the Exchange Agent, in trust for the benefit of the holders of Company Shares, certificates representing an aggregate number of shares of Purchaser Stock as nearly as practicable equal to the number of shares to be converted into Purchaser Stock as determined in Section 1.5 and (ii) deposit with the Exchange Agent, in trust for the benefit of the holders of Company Shares, an amount in cash equal to the cash to be paid in lieu of fractional shares pursuant to Section 1.5(g).
 - (b) As soon as practicable on the day of the Closing (but after the Effective Time), each holder of Company Shares converted into Purchaser Stock Consideration pursuant to Section 1.3(a), upon surrender to the Exchange Agent with a properly completed Letter of Transmittal (to the extent not previously surrendered with a Form of Election) of one or more Certificates for such Company Shares for cancellation, shall be entitled to receive (and the Exchange Agent shall deliver) certificates representing the number of shares of Purchaser Stock into which such Company Shares shall have been converted in the Merger and a bank check in an amount equal to any cash in lieu or fractional shares pursuant to Section 1.5(g).

- (c) No dividends or distributions that have been declared, if any, will be paid to persons entitled to receive certificates for shares of Purchaser Stock until such persons surrender their certificates for Company Shares, at which time all such dividends and distributions shall be paid. In no event shall the persons entitled to receive such dividends be entitled to receive interest on such dividends. If any certificate for such Purchaser Stock is to be issued in a name other than that in which the certificate for Company Shares surrendered in exchange therefor is registered, it shall be a condition of such exchange that the person requesting such exchange shall pay to the Exchange Agent any transfer or other taxes required by reason of issuance of certificates for such Purchaser Stock in a name other than the registered holder of the certificate surrendered, or shall establish to the satisfaction of the $\,$ Exchange Agent that such tax has been paid or is not applicable. Notwithstanding the foregoing, neither the Exchange Agent nor any party hereto shall be liable to a holder of Company Shares for any Purchaser Stock or dividends thereon delivered to a public official pursuant to any applicable abandoned property, escheat or similar law.
- 1.7 Payment of Cash Consideration. Immediately prior to the Effective Time, Purchaser shall deposit with the Exchange Agent, in trust for the benefit of the holders of Company Shares, an amount in cash equal to the Cash Consideration to be paid to holders of Company Shares to be converted into the right to receive the Cash Consideration as determined in Section 1.5. As soon as practicable on the day of the Closing (but after the Effective Time), the Exchange Agent shall distribute to holders of Company Shares converted into the right to receive the Cash Consideration and determined in Section 1.5, upon surrender to the Exchange Agent (to the extent not previously surrendered with a Form of Election) of one or more Certificates for such Company Shares for cancellation, a bank check for an amount equal to the Cash Consideration times the number of Company Shares so converted. In no event shall the holder of any such surrendered certificates be entitled to receive interest on any of the Cash Consideration to be received in the Merger. If such check is to be issued in the name of a person other than the person in whose name the certificates for the Company Shares surrendered for exchange therefor are registered, it shall be a condition of the exchange that the person requesting such exchange shall pay to the Exchange Agent any transfer or other taxes required by reason of issuance of such check to a person other than the registered holder of the certificates surrendered, or shall establish to the satisfaction of the Exchange Agent that such tax has been paid or is not applicable. Notwithstanding the foregoing, neither the Exchange Agent nor any party hereto shall be liable to a holder of Company Shares for any amount paid to a public official pursuant to any applicable abandoned property, escheat or similar law.
- 1.8 Letters of Transmittal. Purchaser will instruct the Exchange Agent to mail to each holder of record of Certificates who has not previously surrendered such holder's Certificates with a validly executed Form of Election as soon as reasonably practicable after the Effective Time, (i) a Letter of Transmittal (which shall specify that delivery shall be effected, and risk of loss and title to such holder's Certificates shall pass, only upon proper delivery of the Certificates to the Exchange Agent and shall be in such form and have such other provisions as shall be agreed upon by the Company prior to the Effective Time) and (ii) instructions for use in effecting the surrender of Certificates in exchange for the Merger Consideration (the "Letter of Transmittal").
- 1.9 Missing Certificates. If any holder of Company Shares convertible into the right to receive the Merger Consideration is unable to deliver the certificate which represents such shares, the Exchange Agent shall deliver to such holder the Merger Consideration to which the holder is entitled for such shares upon presentation of the following: (a) evidence to the reasonable satisfaction of the Purchaser that any such certificate has been lost, wrongfully taken or destroyed; (b) such security or indemnity as may be reasonably requested by the Purchaser to indemnify and hold harmless the Purchaser and the Exchange Agent; and (c) evidence satisfactory to the Purchaser that such person is the owner of the shares theretofore represented by each certificate claimed to be lost, wrongfully taken or destroyed and that the holder is the person who would be entitled to present such certificate for payment pursuant to this Agreement.
 - 1.10 Options.

- (a) At the Effective Time, unless such options are otherwise cancelled pursuant to Section 1.10(d), Purchaser shall cause each holder of a then outstanding and unexercised option whether vested or unvested (the "Company Options"), to receive, by virtue of the Merger and without any action on the part of the holder thereof, options exercisable for shares of Purchaser Stock ("Purchaser Replacement Options") having substantially the same terms and conditions as the Company Options (including such terms and conditions as may be incorporated by reference into the agreements evidencing the Company Options pursuant to the plans or arrangements pursuant to which such Company Options were granted) except that the per share exercise price and the number of shares issuable upon exercise shall be divided and multiplied, respectively, by the Purchaser Stock Consideration, and rounded to the nearest whole cent or number, respectively. Each Company Option or portion thereof, that is not vested pursuant to its original terms will remain unvested and all Purchaser Replacement Options will vest in accordance with the original terms of the Company Option; provided, however, that it is acknowledged for clarification purposes only, that all options that are outstanding under the Company's 1997 Independent Directors Stock Option Plan will become vested in accordance with the provisions of the plan prior to the Effective Time. Purchaser shall use all reasonable efforts to ensure that any Company Options that qualified as incentive stock options under Section 422 of the Code prior to the Effective Time continue to so qualify after the Effective Time. Purchaser shall take all corporate action necessary to reserve for issuance a sufficient number of shares of Purchaser Stock for delivery upon the exercise of Purchaser Replacement Options after the Effective Time.
- (b) At the Effective Time, any unvested restricted Company Shares awarded to employees, directors or consultants pursuant to any of the Company's plans or arrangements and outstanding immediately prior to the Effective Time shall become fully vested and shall be treated as Outstanding Company Shares for purposes of this Article 1.
- (c) Promptly after the Effective Time, Purchaser shall file or cause to be filed all registration statements on Form S-8 or other appropriate form as may be necessary in connection with the purchase and sale of Purchaser Stock contemplated by the Purchaser Replacement Options subsequent to the Effective Time, and shall maintain the effectiveness of such registration statements (and maintain the current status of the prospectus or prospectuses contained therein) for so long as any of the Purchaser Replacement Options registered thereunder remain outstanding. As soon as practicable after the Effective Time, Purchaser shall qualify under applicable state securities laws the issuance of such shares of Purchaser Stock issuable upon exercise of Purchaser Replacement Options. Purchaser's Board of Directors shall take all actions necessary on the part of Purchaser to enable the acquisition of Purchaser Stock and Purchaser Replacement Options and subsequent transactions in Purchaser Stock after the Effective Time pursuant to Purchaser Replacement Options by persons subject to the reporting requirements of Section 16(a) of the Securities Exchange Act (as defined below) to be exempt from the application of Section 16(b) of the Securities Exchange Act, to the extent permitted
- (d) Prior to the Effective Time, the Company shall use its reasonable best efforts to obtain consents, releases and amendments to option agreements from holders of Company Options to the cancellation thereof in exchange for cash, all in accordance with terms to be specified by Purchaser. For the purpose of this Agreement, "Company Stock Option Plans" means and includes the Company's 1997 Incentive Stock Option Plan, the Company's 1997 Non-Qualified Stock Option Plan, and the Company's Independent Directors Stock Option Plan.
- 1.11 Certificate of Incorporation and Bylaws. At and after the Effective Time until the same have been duly amended, (i) subject to Section 5.9(a), the Certificate of Incorporation of the Surviving Corporation shall be identical to the Certificate of Incorporation of Purchaser in effect at the Effective Time and (ii) the Bylaws of the Surviving Corporation shall be identical to the Bylaws of Purchaser in effect at the Effective Time.

- 1.12 Directors and Officers. At and after the Effective Time, the directors of Purchaser immediately prior to the Effective Time shall be the directors of the Surviving Corporation, and the officers of the Purchaser immediately prior to the Effective Time shall be the officers of the Surviving Corporation, in each case until their successors are elected or appointed and qualified. If, at the Effective Time, a vacancy shall exist on the Board of Directors or in any office of the Surviving Corporation, such vacancy may thereafter be filled in the manner provided by Law.
- 1.13 Stock Transfer Books. At the Effective Time, the stock transfer books of the Company shall be closed and no transfer of Company Shares shall thereafter be made. If, after the Effective Time, Certificates are presented to the Surviving Corporation, they shall be cancelled and exchanged for the Merger Consideration pursuant to this Article I.
- 1.14 Other Effects of Merger. The Merger shall have all further effects as specified in the applicable provisions of the GBCC and the DGCL.
- ${\tt 1.15}$ Registration Statement Prospectus/Proxy Statement and Listing Application.
 - (a) For the purposes of (i) registering Purchaser Stock for issuance to holders of the Company Shares in connection with the Merger with the Securities and Exchange Commission ("SEC") under the Securities Act of 1933, as amended, and the rules and regulations thereunder (the "Securities Act"), and complying with applicable state securities laws, and (ii) holding the meeting of the Company's stockholders to vote upon the approval of this Agreement and the Merger and the other transactions contemplated hereby (collectively, the "Company Proposals"), Purchaser and the Company will cooperate in the preparation of a registration statement on Form S-4 (such registration statement, together with any and all amendments and supplements thereto, being herein referred to as the "Registration Statement"), including a prospectus/proxy statement satisfying all requirements of applicable state securities laws, the Securities Act and the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (the "Securities Exchange Act"). Such prospectus/proxy statement in the form mailed by the Company to its shareholders, together with any and all amendments or supplements thereto, is herein referred to as the "Prospectus/Proxy Statement."
 - (b) The Company will furnish Purchaser with such information concerning the Company and its subsidiaries as is necessary in order to cause the Prospectus/Proxy Statement, insofar as it relates to the Company and its subsidiaries, to comply with applicable Law. None of the information relating to the Company and its subsidiaries supplied by the Company for inclusion in the Prospectus/Proxy Statement will be false or misleading with respect to any material fact or will omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. The Company agrees promptly to advise Purchaser if, at any time prior to the meeting of the stockholders of the Company, referenced herein, any information provided by it in the Prospectus/Proxy Statement is or becomes incorrect or incomplete in any material respect and to provide Purchaser with the information needed to correct such inaccuracy or omission. The Company will furnish Purchaser with such supplemental information as may be necessary in order to cause the Prospectus/Proxy Statement, insofar as it relates to the Company and its subsidiaries, to comply with applicable Law after the mailing thereof to the stockholders of the Company.
 - (c) Purchaser will furnish the Company with such information concerning Purchaser and its subsidiaries as is necessary in order to cause the Prospectus/Proxy Statement, insofar as it relates to Purchaser and its subsidiaries, to comply with applicable Law. None of the information relating to Purchaser and its subsidiaries supplied by Purchaser for inclusion in the Prospectus/Proxy Statement will be false or misleading with respect to any material fact or will omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. Purchaser agrees promptly to advise the Company if, at any time prior to the meeting of stockholders of the Company referenced herein, any information provided by it in the Prospectus/Proxy Statement is or becomes incorrect or incomplete

in any material respect and to provide the Company with the information needed to correct such inaccuracy or omission. Purchaser will furnish the Company with such supplemental information as may be necessary in order to cause the Prospectus/Proxy Statement, insofar as it relates to Purchaser and its subsidiaries, to comply with applicable Law after the mailing thereof to the stockholders of the Company.

- (d) The Company and Purchaser agree to cooperate in making any preliminary filings of the Prospectus/Proxy Statement with the SEC, as promptly as practicable, under the Securities Exchange Act.
- (e) Purchaser will file the Registration Statement with the SEC and appropriate materials with applicable state securities agencies as promptly as practicable and will use all reasonable efforts to cause the Registration Statement to become effective under the Securities Act and all such state filed materials to comply with applicable state securities Laws. Purchaser shall provide the Company for its review a copy of the Registration Statement at least such amount of time prior to each filing thereof as is customary in transactions of the type contemplated hereby. The Company authorizes Purchaser to utilize in the Registration Statement and in all such state filed materials, the information concerning the Company and its subsidiaries provided to Purchaser in connection with, or contained in, the Prospectus/Proxy Statement. Purchaser promptly will advise the Company when the Registration Statement has become effective, and of any supplements or amendments thereto, and Purchaser will furnish the Company with copies of all documents. Except for the Prospectus/Proxy Statement or the preliminary prospectus/proxy statement, and except as otherwise required by law, neither Purchaser nor the Company shall distribute any written material that might constitute a "prospectus" relating to the Merger or the Company Proposals within the meaning of the Securities Act or any applicable state securities Law without the prior written consent of the other party.
- (f) Promptly after the execution of this Agreement, Purchaser shall prepare and file with the New York Stock Exchange, Inc. ("NYSE") a listing application covering the shares of Purchaser Stock issuable in the Merger and upon exercise of the Purchaser Replacement Options, and use its reasonable best efforts to obtain, prior to the Effective Time, approval or the listing of such shares of Purchaser Stock, subject only to official notice of issuance.
- 1.16 Tax-Free Reorganization. The parties intend that the Merger qualify as a reorganization within the meaning of Section 368(a) of the Code. None of the parties will knowingly take any action that would cause the Merger to fail to qualify as a reorganization within the meaning of Section 368(a) of the Code.
- 1.17 Additional Actions. If, at any time after the Effective Time, the Surviving Corporation shall consider or be advised that any deeds, bills of sale, assignments, assurances or any other actions or things are necessary or desirable to vest, perfect or confirm of record or otherwise in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of the Company or otherwise carry out this Agreement, the officers and directors of the Surviving Corporation shall be authorized to execute and deliver, in the name and on behalf of the Company, all such deeds, bills of sale, assignments and assurances and to take and do, in the name and on behalf of the Company, all such other actions and things as may be necessary or desirable to vest, perfect or confirm any and all right, title and interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out this Agreement.

ARTICLE 2

REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the disclosure schedule from the Company to Purchaser to be delivered upon the execution of this Agreement, which sets forth certain disclosures concerning the Company and its business (the "Company Disclosure Schedule"), the Company hereby represents and warrants to Purchaser as follows:

2.1 Due Incorporation and Good Standing. The Company and each subsidiary of the Company (the "Company Subsidiaries") is a corporation duly incorporated, validly existing and in good standing under the laws of the jurisdiction of its incorporation and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. The Company and each of the Company Subsidiaries is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except where the failure to be so duly qualified or licensed and in good standing would not be reasonably likely to have a Company Material Adverse Effect. For the purposes of this Agreement, any reference to a state of facts, event, change or effect having a "Company Material Adverse Effect" means such state of facts, event, change or effect that (a) has had, or is reasonably likely to have, a material adverse effect on the business, results of operation, prospects or financial condition of the Company and the Company Subsidiaries taken as a whole or (b) would reasonably be expected to have a material adverse effect on the legality, binding nature or enforceability of this Agreement against the Company or to prevent or substantially delay the consummation of the Merger; provided that the following state of facts, events, changes and the effects thereof shall be disregarded and shall in no event constitute a Company Material Adverse Effect: (i) general business or economic conditions, (ii) conditions generally affecting the industry in which the Company and the Company Subsidiaries compete, (iii) the taking of any action contemplated by this Agreement, and (iv) the announcement or pendency of the transactions contemplated in this Agreement. The Company has heretofore made available to Purchaser accurate and complete copies of the Articles of Incorporation and Bylaws, as currently in effect, of the Company.

2.2 Capitalization.

- (a) As of the date hereof, the authorized capital stock of the Company consists of 62,000,000 shares of capital stock, of which 2,000,000 shares are of a class designated as Preferred Stock, \$1.00 par value per share and 60,000,000 shares are of a class designated as Company Stock. As of February 16, 2001, 33,415,780 shares of Company Stock were issued and outstanding, including 66,438 unvested restricted Company Shares, that had been awarded by the Company under its Restricted Stock Plan. No other shares of capital stock of the Company are issued and outstanding. As of February 16, 2001, a total of 821,938 Company Shares (including no more than Company Options for 149,000 shares of Company Stock that have an exercise price of \$4.35 or less) are reserved for future issuance to employees and directors upon exercise of any Company Options, warrants or other rights to purchase or acquire any shares of capital stock of the Company. Since February 16, 2001, the Company has not issued or granted additional Company Options.
- (b) As of the Closing Date, the total number of shares of outstanding Company Shares, including shares issued upon the exercise of Company Options and Restricted Company Shares (whether vested or unvested), shall not exceed 33,602,218.
- (c) All issued and outstanding shares of the Company Stock are, and all shares which may be issued upon exercise of then outstanding Company Options will be when issued, duly authorized, validly issued, fully paid and non-assessable. Except as otherwise contemplated by this Agreement, as of the date hereof there are no outstanding rights, subscriptions, warrants, puts, calls, unsatisfied preemptive rights, options or other agreements of any kind relating to any of the outstanding, authorized but unissued or unauthorized shares of capital stock or any other security of the Company, and there is no authorized or issued security of any kind convertible into or exchangeable, for any such capital stock or other security.
- 2.3 Subsidiaries. Section 2.3 of the Company Disclosure Schedule sets forth the name and jurisdiction of incorporation or organization of each Company Subsidiary, each of which is wholly owned by the Company except as otherwise indicated in said Section 2.3 of the Company Disclosure Schedule. All of the capital stock and other interests of the Company Subsidiaries so held by the Company are owned by it or a Company Subsidiary as indicated in said Section 2.3 of the Company Disclosure Schedule, free and clear of any claim, lien, encumbrance, security interest or agreement with respect

thereto. All of the outstanding shares of capital stock in each of the Company Subsidiaries directly or indirectly held by the Company are duly authorized, validly issued, fully paid and non-assessable and were issued free of preemptive rights and in compliance with applicable Laws. No equity securities or other interests of any of the Company Subsidiaries are or may become required to be issued or purchased by reason of any options, warrants, rights to subscribe to, puts, calls or commitments of any character whatsoever relating to, or securities or rights convertible into or exchangeable for, shares of any capital stock of any Company Subsidiary, and there are no contracts, commitments, understandings or arrangements by which any Company Subsidiary is bound to issue additional shares of its capital stock, or options, warrants or rights to purchase or acquire any additional shares of its capital stock or securities convertible into or exchangeable for such shares, in each case, except as disclosed in Section 2.3 of the Company Disclosure Schedule.

- 2.4 Authorization; Binding Agreement. The Company has all requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, including, but not limited to, the Merger, have been duly and validly authorized by the Company's Board of Directors, and no other corporate proceedings on the part of the Company or any Company Subsidiary are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby (other than the requisite approval of this Agreement and the Merger by the stockholders of the Company in accordance with the GBCC). This Agreement has been duly and validly executed and delivered by the Company and constitutes the legal, valid and binding agreement of the Company, enforceable against the Company in accordance with its terms, except to the extent that enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by principles of equity regarding the availability of remedies ("Enforceability Exceptions").
- 2.5 Governmental Approvals. No consent, approval, waiver or authorization of, notice to or declaration or filing with ("Consent"), any nation or government, any state or other political subdivision thereof, any entity, authority or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, including, without limitation, any governmental or regulatory authority, agency, department, board, commission, administration or instrumentality, any court, tribunal or arbitrator or any self regulatory organization ("Governmental Authority") on the part of the Company or any of the Company Subsidiaries is required in connection with the execution or delivery by the Company of this Agreement or the consummation by the Company of the transactions contemplated hereby other than (i) the filing of the Certificate of Merger with the Secretary of State of the State of Georgia in accordance with the GBCC, (ii) filings with the Secretary of SEC, state securities laws administrators and the American Stock Exchange, Inc., (iii) filings under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder (the "HSR Act"), (iv) such filings as may be required in any jurisdiction where the Company is qualified or authorized to do business as a foreign corporation in order to maintain such qualification or authorization, (v) notices under Michigan Compiled Laws Sect. 211.27a and similar requirements of jurisdictions where the Company or the Company Subsidiaries own or lease real property, (vi) those Consents required under any foreign Law, and (vii) those Consents that, if they were not obtained or made, would not be reasonably likely to cause the Company to incur liabilities or obligations in excess of \$250,000.
- 2.6 No Violations. Except as set forth in Section 2.6 of the Company Disclosure Schedule, the execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance by the Company with any of the provisions hereof will not (i) conflict with or result in any breach of any provision of the Articles of Incorporation or Bylaws or other governing instruments of the Company or any of the Company Subsidiaries, (ii) except as set forth on Section 2.6 of the Company Disclosure Schedule, require any Consent under or result in a violation or breach of, or constitute (with or without due notice or lapse of time or both) a default (or give rise to any right of termination, cancellation or acceleration) under any of the terms, conditions or provisions of any agreement or other instrument to which the Company or any Company Subsidiaries are parties or by which their respective assets are

bound, (iii) result in the creation or imposition of any lien or encumbrance of any kind upon any of the assets of the Company or any Company Subsidiary or (iv) subject to obtaining the Consents from Governmental Authorities referred to in Section 2.5 hereof, contravene any applicable provision of any statute, law, rule or regulation or any order, decision, injunction, judgment, award or decree ("Law") to which the Company or any Company Subsidiary or its or any of their respective assets or properties are subject, except in the case of clauses (ii), (iii) and (iv) above, for any deviations from the foregoing which would not be reasonably likely to have a Company Material Adverse Effect.

- 2.7 Securities Filings. The Company has made available to Purchaser true and complete copies of (i) its Annual Reports on Form 10-K for the years ended September 30, 2000 and September 30, 1999 as filed with the SEC, (ii) its proxy statements relating to all of the meetings of stockholders (whether annual or special) of the Company since September 30, 1999, as filed with the SEC, and (iii) all other reports, statements and registration statements and amendments thereto (including, without limitation, Quarterly Reports on Form 10-Q and Current Reports on Form 8-K, as amended) filed by the Company with the SEC since September 30, 1999. Except as set forth in Section 2.7 of the Company Disclosure Schedule, the Company has timely filed (with extensions) with the SEC all forms, reports, schedules, statements or other documents required to be filed by it under the Securities Act or Exchange Act since September 30, 1997 (such documents, as supplemented or amended since the time of filing, the "Company Securities Filings"). Except as set forth in Section 2.7 of the Company Disclosure Schedule, as of their respective dates, and as of the date of the last amendment thereof, if amended after filing, none of the Company Securities Filings contained or, as to the Company Securities Filings subsequent to the date hereof, will contain, any untrue statement of a material fact or omitted or, as to the Company Securities Filings subsequent to the date hereof, will omit, to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Except as set forth in Section 2.7 of the Company Disclosure Schedule, each of the Company Securities Filings at the time of filing and as of the date of the last amendment thereof, if amended after filing, complied or, as to the Company Securities Filings subsequent to the date hereof, will comply in all material respects with the Securities Exchange Act or the Securities Act, as applicable.
- 2.8 Company Financial Statements. The audited consolidated financial statements and unaudited interim financial statements (including any related notes and schedules) of the Company included in the Company Securities Filings (the "Company Financial Statements") have been prepared or will be prepared in accordance with generally accepted accounting principles applied on a consistent basis (except as may be indicated therein or in the notes thereto) and present or will present fairly, in all material respects, the financial position of the Company and the Company Subsidiaries as at the dates thereof and the results of their operations and cash flows for the periods then ended, in each case in accordance with generally accepted accounting principles applied on a consistent basis, subject, in the case of the unaudited interim financial statements, to normal year-end audit adjustments, any other adjustments described therein and the fact that certain information and notes have been condensed or omitted in accordance with the Securities Exchange Act.
- 2.9 Absence of Certain Changes or Events; No Undisclosed Liabilities. Except as set forth in Section 2.9 of the Company Disclosure Schedule, since September 30, 2000, through the date of this Agreement, there has not been: (i) any event that has had or would reasonably be expected to have a Company Material Adverse Effect, (ii) any declaration, payment or setting aside for payment of any dividend or other distribution or any redemption or other acquisition of any shares of capital stock or securities of the Company by the Company or any Company Subsidiary, (iii) any material damage or loss to any material asset or property, whether or not covered by insurance, or (iv) any change by the Company in accounting principles or practices. Except as set forth in Section 2.9 of the Company Disclosure Schedule, since September 30, 2000, through the date of this Agreement, there has not been any action taken by the Company or any of the Company Subsidiaries that, if taken during the period from the date of this Agreement through the Effective Time, would constitute a breach of Section 4.1. Except as set forth in Section 2.9 of the Company Disclosure Schedule and except for those liabilities that are reflected or reserved against on the balance sheet of the Company included in its September 30, 2000

- Form 10-K, and liabilities incurred in the ordinary course of business since September 30, 2000, neither the Company nor any of the Company Subsidiaries has incurred any liability of any nature required under generally accepted accounting principles, applied on a consistent basis, to be reflected on a balance sheet of the Company that, either individually or in the aggregate, has had or would be reasonably likely to have a Company Material Adverse Effect.
- 2.10 Compliance with Laws. To the knowledge of the Company, the businesses of the Company and the Company Subsidiaries have been operated in compliance with all Laws applicable thereto, except for any instances of non-compliance which, individually or in the aggregate, would not be reasonably likely to have a Company Material Adverse Effect.
- 2.11 Permits. To the knowledge of the Company, (i) the Company and each of the Company Subsidiaries have all permits, certificates, licenses, approvals and other authorizations required in connection with the operation of their respective businesses (collectively, "Company Permits"), (ii) neither the Company nor any of the Company Subsidiaries is in material violation of any Company Permit and (iii) no proceedings are pending or, to the knowledge of the Company, threatened, to revoke or limit any Company Permit, except, in each case, those the absence, violation or revocation of which would not be reasonably likely to have a Company Material Adverse Effect.
- 2.12 Litigation. Except as disclosed in Section 2.12 of the Company Disclosure Schedule, there is no suit, action or proceeding ("Litigation") pending or, to the knowledge of the Company, threatened against the Company or any of the Company Subsidiaries, nor is there any judgment, decree, injunction, rule or order of any Governmental Authority outstanding against the Company or any of the Company Subsidiaries, in any case which, individually or in the aggregate, would be reasonably likely to have a Company Material Adverse Effect.
- 2.13 Material Contracts. Neither the Company nor any of the Company Subsidiaries is a party to any material contract described in Rule 601(b)(10) of Regulation S-K that was required to be filed as an exhibit to the Company's 2000 Form 10-K (or would be required to be filed as an exhibit to the Company's 2000 Form 10-K if it were being filed as of the date of this Agreement) ("Company Material Contract"), except as filed as an exhibit to the Company's 2000 Form 10-K or as disclosed on Section 2.13 of the Company's Disclosure Schedule. To the knowledge of the Company, all Company Material Contracts are valid and binding and are in full force and effect and enforceable against the Company or such Company Subsidiary in accordance with their respective terms, subject to the Enforceability Exceptions. Neither the Company nor any of the Company Subsidiaries is in violation or breach of or default under any Company Material Contract where such violation or breach would be reasonably likely to have a Company Material Adverse Effect.
- 2.14 Employee Benefit Plans. (a) Section 2.14 of the Company Disclosure Schedule contains a complete list of all material written bonus, vacation, deferred compensation, pension, retirement, profit-sharing, thrift, savings, employee stock ownership, stock bonus, stock purchase, restricted stock and stock option plans, employment or severance contracts, medical, dental, disability, health and life insurance plans, and other employee benefit and fringe benefit plans or other agreements maintained or contributed to by the Company or any of its subsidiaries for the benefit of officers, former officers, employees, former employees, directors, former directors, or the beneficiaries of any of the foregoing, or pursuant to which the Company or any of its subsidiaries may have any liability (collectively, the "Company Compensation and Benefit Plans") that are agreements with, or plans maintained primarily for the benefit of, individuals employed or rendering services and are not multiemployer plans within the meaning of Section 4001(a)(3) of ERISA (as defined in Section 3.8(c)) (the "Company Compensation and Benefit Plans").
 - (b) The Company has used its reasonable best efforts to have provided or made available to Purchaser copies of all Company Compensation and Benefit Plans listed on Section 2.14 of the Company Disclosure Schedule (excluding, however, multiemployer plans within the meaning of Section 4001(a)(3) of ERISA ("Multiemployer Plans") and Company Compensation and Benefit Plans which have been filed with or incorporated by reference into the Company's 1999 Form 10-K),

including, but not limited to, all amendments thereto, and all of such copies that have been delivered are true and correct.

- (c) To the knowledge of the Company, each of the Company Compensation and Benefit Plans has been and is being administered in accordance with the terms thereof and all applicable Law except where the failure to do so does not, individually or in the aggregate, have a Company Material Adverse Effect. Each "employee pension benefit plan" within the meaning of Section 3(2) of the Employee Retirement Income Security Act of 1974, as amended ("ERISA") (each such plan, a "Pension Plan") included in the Company Compensation and Benefit Plans (a "Company Pension Plan") which is intended to be qualified under Section 401(a) of the Code has received a favorable determination letter from the Internal Revenue Service, and the Company is not aware of any circumstances which could result in the revocation or denial of any such favorable determination letter. To the knowledge of the Company, no material "prohibited transaction," within the meaning of Section 4975 of the Code or Section 406 of ERISA, has occurred with respect to any Company Compensation and Benefit Plan. Except as disclosed in Section 2.12 of the Company's Disclosure Schedule, there is no pending or, to the Company's knowledge, threatened litigation relating to any of the Company Compensation and Benefit Plans.
- (d) To the knowledge of the Company, no material liability under Title IV of ERISA has been or, to the knowledge of the Company, is reasonably expected to be incurred by the Company or any of its Subsidiaries or any entity which is considered one employer with the Company under Section 4001(a)(15) of ERISA or Section 414 of the Code (any such entity, a "Company ERISA Affiliate"), other than such liabilities that have previously been satisfied. To the knowledge of the Company, no notice of a "reportable event," within the meaning of Section 4043 of ERISA, for which the 30-day reporting requirement has not been waived has been required to be filed for any Company Pension Plan or by any Company ERISA Affiliate within the past 12 months.
- (e) All contributions, premiums and payments required to be made under the terms of any Company Compensation and Benefit Plan have been made, except where the failure to do so does not, individually or in the aggregate, have a Company Material Adverse Effect. To the knowledge of the Company, neither any Company Pension Plan nor any single-employer plan of a Company ERISA Affiliate has an "accumulated funding deficiency" (whether or not waived) within the meaning of Section 412 of the Code or Section 302 of ERISA. Neither the Company nor any of its Subsidiaries has provided, or is required to provide, security to any Company Pension Plan or to any single-employer plan of a Company ERISA Affiliate pursuant to Section 401(a)(29) of the Code.
- (f) No Company Pension Plan is a "defined benefit plan", within the meaning of $\ensuremath{\mathsf{ERISA}}.$
- (g) Neither the Company nor any of its Subsidiaries contributes to or is required to contribute to any Multiemployer Plan. Neither the Company nor any of its Subsidiaries has incurred any material withdrawal liability (within the meaning of Section 4201 of ERISA) under any Multiemployer Plan within the past 5 years that has not been satisfied, nor, to the knowledge of the Company, could any such material withdrawal liability reasonably be expected to be incurred.
- (h) Except as set forth in the Company Compensation and Benefit Plans listed in Section 2.14 of the Company Disclosure Schedule or filed with or incorporated by reference into the Company's 2000 Form 10-K, the execution of, and performance of the transactions contemplated in, this Agreement will not (either alone or upon the occurrence of any additional or subsequent events) (i) constitute an event under any Company Compensation and Benefit Plan, trust or loan that will or may result in any payment (whether of severance pay or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any officers and directors of the Company or (ii) to the knowledge of the Company result in any payment or benefit that will or may be made by the Company, any of its Subsidiaries, Purchaser or any of their respective affiliates that will be characterized as an "excess parachute payment," within the meaning of Section 280G(b)(1) of the Code.

- (i) To the knowledge of the Company, the contributions of the Company and any of its Subsidiaries to any trust described in Section 501(c)(9) of the Code have complied with Section 419A of the Code.
- 2.15 Tangible Assets. The Company has good and marketable title to, or a valid leasehold interest in, the properties and assets shown on the latest of the Company Financial Statements as owned by it, except for properties and assets disposed of in the ordinary course of business, free and clear of all liens, claims and encumbrances except for Permitted Encumbrances, if any. The Company owns or leases pursuant to a Company Material Contract all buildings, machinery, equipment and other tangible assets material to the conduct of its business as presently conducted. To the Company's knowledge, the tangible assets are, in the aggregate, free from defects (patent and latent) other than defects that do not individually or in the aggregate materially impair their value or intended use, have been maintained in accordance with normal industry practice, are in good operating condition and repair (subject to normal wear and tear) and are suitable for the purposes for which they presently are used, except where failure to do so would not result in a Company Material Adverse Effect. Section 2.15 of the Company Disclosure Schedule contains a schedule of such tangible assets owned or leased by the Company that have a value in excess of \$50,000.
- 2.16 Taxes and Returns. (a) Except as set forth in Section 2.16 of the Company Disclosure Schedule, the Company and each Company Subsidiary has timely filed, or caused to be timely filed, all Tax Returns (as defined below) required to be filed by it, and such returns are true, correct, and complete in all material respects. Except as set forth in Section 2.16 of the Company Disclosure Schedule, the Company and each Company Subsidiary has paid, collected or withheld, or caused to be paid, collected or withheld, all Taxes (as defined below) required to be paid, collected or withheld, other than such Taxes for which adequate reserves in the Company Financial Statements have been established. There are no material claims or assessments pending against the Company or any of the Company Subsidiaries for any alleged deficiency in any Tax, and the Company has not been notified in writing of any proposed Tax claims or assessments against the Company or any of the Company Subsidiaries (other than, in each case, claims or assessments for which adequate reserves in the Company Financial Statements have been established or which are being contested in good faith or are immaterial in amount). Neither the Company nor any of the Company Subsidiaries has any outstanding waivers or extensions of any applicable statute of limitations to assess any material amount of Taxes. There are no outstanding requests by the Company or any of the Company Subsidiaries for any extension of time within which to file any Tax Return or within which to pay any Taxes shown to be due on any return. There are no liens for material amounts of Taxes on the assets of the Company or any of the Company Subsidiaries except for statutory liens for current Taxes not yet due and payable.
 - (b) None of the Company or any of the Company Subsidiaries has taken or agreed to take any action that would prevent the Merger from constituting a reorganization qualifying under the provisions of Section 368(a) of the Code.
 - (c) Except as set forth in Section 2.16 of the Company Disclosure Schedule, the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby (either alone or in combination with another event) will not result in any payment (whether of severance pay, unemployment compensation, golden parachute, bonus or otherwise), acceleration, forgiveness of indebtedness, vesting, distribution, increase in benefits or obligation to fund benefits with respect to any employee or director of the Company or any of the Company Subsidiaries.
 - (d) None of the Company or any of the Company Subsidiaries has been a member of any consolidated, combined, unitary or affiliated group of corporations for any Tax purposes other than a group of which the Company is or was the common parent corporation.
 - (e) As of the date of this Agreement, none of Company or any of the Company Subsidiaries is currently being audited by any taxing authority and none of the Company or any of the Company Subsidiaries has been notified by any tax authority that any such audit is contemplated or pending.

- (f) The Company is not a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code.
- (g) The Company and each Company Subsidiary will not as a result of the transaction contemplated by this Agreement make or become obligated to make any parachute payment as defined in Section 280(G) of the Code.
- (h) None of the Company or any Company Subsidiary has made an election under Section 341(f) of the Code.
- (i) None of the Company or any Company Subsidiary is a party to or has any obligation under any tax allocation, tax sharing, tax indemnity or other similar agreement or arrangement.
- (j) The Company has not applied for or received a tax ruling from the Internal Revenue Service or any foreign, state or local taxing authority and has not entered into a closing agreement pursuant to Section 7121 of the Code or similar provision of foreign, state or local law, which closing agreement is still in effect.
- (k) Except as set forth in Section 2.16 of the Company Disclosure Schedule, none of the Company or Company Subsidiaries has either distributed stock of a controlled corporation pursuant to Section 355 of the Code or had its stock distributed by another corporation pursuant to Section 355 of the Code.
- (1) Except as set forth in Section 2.16 of the Company Disclosure Schedule, no claim has been made or is expected to be made against or in respect of the Company or any Company Subsidiary by any Governmental Authority in a jurisdiction in which the Company or any Company Subsidiary does not file returns or pay or collect taxes in respect of a particular type of tax imposed by that jurisdiction that the Company or any Company Subsidiary is or may be subject to an obligation to file returns or collect or pay taxes in respect of such tax in that jurisdiction.
- (m) True, correct and complete copies of all income franchise property payroll sales and use tax returns and forms filed by the Company and any Company Subsidiary for the last 3 years have been furnished to Purchaser.
- (n) For purposes of this Agreement, the term "Tax" shall mean any tax, custom, duty, governmental fee or other like assessment or charge of any kind whatsoever, imposed by any Governmental Authority (including, but not limited to, any federal, state, local, foreign or provincial income, gross receipts, property, sales, use, license, excise, franchise, employment, payroll, alternative or added minimum, ad valorem, transfer or excise tax) together with any interest, addition or penalty imposed thereon. The term "Tax Return" shall mean a report, return or other information (including any attached schedules or any amendments to such report, return or other information) required to be supplied to or filed with a Governmental Authority with respect to any Tax, including an information return, claim for refund, amended return or declaration or estimated Tax.
- 2.17 Certain Labor Matters. As of the date of this Agreement, the Company and the Company Subsidiaries do not have any collective bargaining agreements with any persons employed by the Company or any of the Company Subsidiaries, nor is the Company or any of the Company Subsidiaries in the process of negotiating any such agreement. There is no labor strike, dispute or stoppage pending, or to the knowledge of the Company, threatened against the Company or any of the Company Subsidiaries. None of the Company or the Company Subsidiaries is the subject of a proceeding asserting that it has committed an unfair labor practice (within the meaning of the National Labor Relations Act) or seeking to compel it to bargain with any labor organization as to wages and conditions of employment which proceeding has had or is reasonably likely to have a Company Material Adverse Effect. To the knowledge of the Company, no executive, key employee or group of employees has any plans to terminate its or their employment with the Company which termination would have a Company Material Adverse Effect.
- 2.18 Finders and Investment Banker. Except as set forth in Section 2.18 of the Company Disclosure Schedule, neither the Company nor any of its officers or directors has employed any broker or

finder or otherwise incurred any liability for any brokerage fees, commissions or finders' fees in connection with the transactions contemplated hereby.

- 2.19 Fairness Opinion. The Company has received from McDonald Investments, its financial advisor, a verbal opinion to the effect that the consideration to be received in the Merger by the Company's shareholders is fair to the Company's shareholders from a financial point of view.
- 2.20 Insurance. Section 2.20 of the Company Disclosure Schedule sets forth a true and complete list of all insurance policies carried by, or covering the Company and the Company Subsidiaries with respect to their businesses, assets and properties and with respect to which records are maintained at the Company's principal executive offices, together with, in respect of each such policy, the amount of coverage and the deductible. The Company and the Company Subsidiaries maintain insurance policies against all risk of a character, including without limitation, business interruption insurance, and in such amounts as the Company deems appropriate. Each insurance policy set forth on Section 2.20 of the Company Disclosure Schedule is in full force and effect and all premiums due thereon have been paid in full.
- 2.21 Vote Required; Ownership of Purchaser Capital Stock; State Takeover Statutes. (a) The approval of the Company Proposals by a vote of a majority of the holders of the issued and outstanding Company Shares is the only vote of the holders of any class or series of the Company's capital stock necessary to approve the Merger and the transactions contemplated hereunder.
 - (b) Neither the Company nor any of the Company Subsidiaries beneficially owns, either directly or indirectly, any shares of Purchaser capital stock.
 - (c) The Company has taken all actions necessary under the GBCC to adopt and approve this Agreement, the Merger and the transactions contemplated by this Agreement except as required under Section 2.21(a). Sections 14-2-1111 and 14-2-1132 of the GBCC applicable to a "business combination" do not (by reason of the Company's participation therein) apply to the Merger. To the knowledge of the Company no other "fair price," "moratorium," or other similar anti-takeover statute or regulation prohibits (by reason of Company's participation therein) the Merger or the other transactions contemplated by this Agreement.
- 2.22 Environmental Matters. To the knowledge of the Company and except as otherwise disclosed in Section 2.22 of the Company Disclosure Schedule:
 - (a) the Company has not, and no third party has, generated, treated, stored, released or disposed of, or otherwise placed, deposited in or located on real property owned, leased or operated by the Company or any of the Company Subsidiaries (the "Company Real Property"), any toxic or hazardous substances or wastes, pollutants or contaminants (including, without limitations, asbestos, urea formaldehyde, the group of organic compounds known as polychlorinated biphenyls, petroleum products including gasoline, fuel oil, crude oil and various constituents of such products, and any hazardous substance as defined in the Comprehensive Environmental Response, Compensation and Liability Act of 1980 ("CERCLA"), 42 U.S.C. sec.sec. 9601-9657, as amended) (collectively, "Hazardous Substances") and no Hazardous Substances have been generated, treated, stored, released or disposed of, or otherwise placed, deposited in or located on the Company Real Property, nor has any activity been undertaken on the Company Real Property that would cause or contribute to (a) the Company Real Property becoming a treatment, storage or disposal facility in material violation of, the Resource Conservation and Recovery Act of 1976 ("RCRA"), 42 U.S.C. sec.sec. 6901 et seq., or any similar state law or local ordinance, (b) a release or threatened release of toxic or hazardous wastes or substances, pollutants or contaminants from the Company Real Property in violation of CERCLA or any similar state law or local ordinance, or (c) the discharge of pollutants or effluents into any water source or system, the dredging or filling of any waters or the discharge into the air of any emissions, for which the Company does not have all required permits under the Federal Water Act, 33 U.S.C. sec.sec. 1251 et seq., or the Clean Air Act, 42 U.S.C. sec.sec. 7401 et seq., or any similar state law or local ordinance, in

each case except for any such noncompliance, violations, or failures as would not be reasonably likely to have a Company Material Adverse Effect;

- (b) there are no substances or conditions in or on the Company Real Property that may support a claim or cause of action under RCRA, CERCLA or any other federal, state or local environmental statutes, regulations, ordinances or other environmental regulatory requirements, except for any such claims or causes of action as would not be reasonably likely to have a Company Material Adverse Effect; and
- (c) there are no above ground or underground tanks that have been located under, in or about the Company Real Property which have been subsequently removed or filled and to the extent storage tanks exist on or under the Company Real Property, such storage tanks have been duly registered with all appropriate regulatory and governmental bodies and are otherwise in compliance with applicable federal, state and local statutes, regulations, ordinances and other regulatory requirements.
- 2.23 Real Property. (a) Except as set forth in Section 2.23 of the Company Disclosure Schedule (the "Owned Real Property"), neither the Company nor any Company Subsidiaries own any real property. The Company and each of the Company Subsidiaries have good and insurable title to all Owned Real Property Except as disclosed in Section 2.23 of the Company Disclosure Schedule, to the knowledge of the Company, (i) the current use and operation of all Owned Real Property is in compliance with all Laws (including and with limitation, laws relating to parking, zoning and land use) and public and private covenants and restrictions, and neither the Company nor any of its Subsidiaries have received notice of non-compliance of any Owned Real Property with any Laws and (ii) the utilities, access and parking for each such item of Owned Real Property are reasonably adequate for the current use and operation thereof, except, with respect to this Section 2.23, in instances that, in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect. Except as disclosed in Section 2.23 of the Company Disclosure Schedule hereto, the Company does not have knowledge that there are any zoning, building code, occupancy restriction or other land-use regulation proceedings or any proposed changes in any Laws, which would materially and adversely affect the present use or operation of any Owned Real Property, nor has the Company or any Subsidiary received any notice of any special assessment proceedings affecting the Owned Real Property, or applied for any change to the zoning or land use status of the Owned Real Property. Except, with respect to this Section 2.23, in instances that, in the aggregate, would not be reasonably expected to have a Company Material Adverse Effect. Except as disclosed in Section 2.23 of the Company Disclosure Schedule hereto, to the knowledge of the Company, all water, sewer, gas, electric, telephone and drainage facilities and all other utilities required by all Laws or for the present use and operation of each item of Owned Real Property are reasonably adequate to service such item of Owned Real Property, as improved, and to permit compliance with all Laws in all material respects and present usage of each Owned Real Property. To the knowledge of the Company, the Company and its Subsidiaries have obtained all licenses, permits, approvals, easements and rights of way (and all such items are currently in full force and effect) required from any Governmental Authority or from private parties for the present use and operation of such item of Owned Real Property.
 - (b) Section 2.23 of the Company Disclosure Schedule lists and describes briefly all real property leased or subleased to the Company or the Company Subsidiaries. The Company has made available to the Purchaser correct and complete copies of each such lease and sublease. Except as set forth in Sections 2.23 and 2.26 of the Company Disclosure Schedule:
 - (i) to the knowledge of the Company, each such lease or sublease is legal, valid, binding, enforceable and in full force and effect subject to the Enforceability Exceptions;
 - (ii) the consummation of the transactions contemplated hereby will neither cause the termination of each such lease or sublease nor effect a change in any of its terms;
 - (iii) the Company is not, and, to the knowledge of the Company, no other party to such lease or sublease is, in breach or default, and no event has occurred which, with notice or lapse

of time, or both, would constitute a breach or default that would permit termination, modification or acceleration thereunder;

- (iv) neither the Company nor, to the knowledge of the Company, any other party to each such lease or sublease has repudiated or disputed any provision thereof;
- (v) to the knowledge of the Company, there are no oral agreements in effect as to each such lease or sublease; and
- (vi) the Company has not assigned, transferred, conveyed, mortgaged, deeded in trust or encumbered any interest in any leasehold or subleasehold except for Permitted Encumbrances, if any. For the purposes of this Agreement, "Permitted Encumbrances" means and includes those items set forth on Section 2.23 of the Company Disclosure Schedule.
- 2.24 Notes and Accounts Receivables. All notes and accounts receivable of the Company are reflected properly on the Company's books and records, are not, to the knowledge of the Company, subject to any setoff or counterclaim.
- 2.25 Bank Accounts and Powers of Attorney. Section 2.25 of the Company Disclosure Schedule sets forth a list of all accounts, borrowing resolutions and deposit boxes maintained by the Company at any bank or other financial institution and the names of the persons authorized to effect transactions in such accounts and pursuant to such resolutions and with access to such boxes. There are no outstanding powers of attorney executed on behalf of the Company.
- 2.26 Guaranties. Except as disclosed in Section 2.26 of the Company Disclosure Schedule, the Company is not a guarantor or otherwise is liable for any indebtedness, liability or other obligation of any other person or entity.
- 2.27 Intellectual Property. Section 2.27 of the Company Disclosure Schedule sets forth all of the following that are used in, useful, necessary, incidental or pertain to or associated with the operation of the Company and the Company Subsidiaries: (i) all trademarks, service marks, trade names, trade dress and the like, including all common law marks (collectively, together with the associated goodwill of each, "Trademarks"), together with information regarding all registrations and pending applications to register any such rights; (ii) all patents on and pending application to patent any technology or design; (iii) all copyrights and all registrations of and applications to register copyrights; (iv) all licenses of rights in software (including all source codes with respect to such software that are available to the Company and the Company Subsidiaries), Trademarks, patents, copyrights and other intellectual property, whether to or by the Company and the Company Subsidiaries; and (v) all trade secrets (as defined in the Michigan Uniform Trade Secret Act). The rights of the Company and the Company Subsidiaries required to be so identified and used in, useful, necessary or pertaining to the Company and the Company Subsidiaries herein collectively referred to as the "Intellectual Property." The Company and/or the Company Subsidiaries are the owner of the Intellectual Property, and all such Intellectual Property exists and has been maintained in good standing. To the Company's knowledge, no other firm, corporation, association or person claims the right to use in connection with similar or closely related goods and in the same geographic area any mark that is identical or confusingly similar to any of the Trademarks. To the Company's knowledge, no third party asserts ownership rights in any of the Intellectual Property (except to the extent that such Intellectual Property has been properly licensed to or by the Company and the Company Subsidiaries), except to the extent the same would not reasonably be expected to cause a Material Adverse IP Effect. To the Company's knowledge, the Company's and the Company Subsidiaries' use of the Intellectual Property does not infringe upon any right of any third party and no third party is infringing any of the Company's and the Company Subsidiaries' rights in any of the Intellectual Property, except to the extent the same would not reasonably be expected to cause a Material Adverse IP Effect. To the Company's knowledge, and without limitation of the foregoing, the Company and/or the Company Subsidiaries have the legal right (including, but not limited to, site licenses, as applicable) to use all copies of all computer software currently used by the Company and/or the Company Subsidiaries, except to the extent the same would not reasonably be expected to cause a Material Adverse IP Effect. For

purposes of this Section 2.27, a "Material Adverse IP Effect" means any claim for infringement which results in a payment of damages or royalties to a third-party in excess of \$1,000,000, or a loss of revenues in excess of \$1,000,000.

- 2.28 Investments. Except as disclosed in Section 2.28 of the Company Disclosure Schedule and as reflected on the Company Financial Statements, neither the Company nor the Company Subsidiaries (a) own any equity or debt of any other person, and (b) have any liabilities or obligations related to any equity or debt ownership of any other person.
- 2.29 No Other Representations or Warranties. The Company acknowledges and agrees that, in entering into this Agreement and in consummating the transactions contemplated hereby:
 - (i) It has relied and will rely solely upon its own investigation and analysis and the representations and warranties contained in Article 3 of this Agreement (and, without limiting the generality of the foregoing, not on any of the information learned or provided in connection with any presentation made by management or other Purchaser representatives, contained in any document provided to it, any projections or forecasts or otherwise).
 - (ii) None of the Purchaser nor any of the Purchaser representatives has made any statement, representation or warranty to the Company except for the representations and warranties made by the Purchaser as expressly set forth in Article 3 of this Agreement (including in any other document or any projections or forecasts).

ARTICLE 3

REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as set forth in the disclosure schedule from Purchaser to the Company to be delivered upon the execution of this Agreement, which sets forth certain disclosures concerning Purchaser and its business (the "Purchaser Disclosure Schedule"), Purchaser hereby represents and warrants to the Company as follows:

- 3.1 Due Incorporation and Good Standing. Each of Purchaser and each subsidiary of the Purchaser (the "Purchaser Subsidiaries") is a corporation duly incorporated, validly existing and in good standing under the laws of the jurisdiction of its incorporation or organization and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except where the failure to be so duly qualified or licensed and in good standing would not be reasonably likely to have a Purchaser Material Adverse Effect. For the purposes of this Agreement, any reference to a state of facts, event, change or effect having a "Purchaser Material Adverse Effect" means any such state of facts, event, change or effect that (a) has had, or is reasonably likely to have, a material adverse effect on the business, results of operation, prospects or financial condition of the Purchaser and the Purchaser Subsidiaries taken as a whole or (b) would reasonably be expected to have a material adverse effect on the legality, binding nature or enforceability of this Agreement against the Purchaser or to prevent or substantially delay the consummation of the Merger; provided that the following state of facts, events, changes and the effects thereof shall be disregarded and shall in no event constitute a Purchaser Material Adverse Effect: (i) general business or economic conditions, (ii) conditions generally affecting the industry in which the Purchaser competes, (iii) the taking of any action contemplated by this Agreement, (iv) the announcement or pendency of the transactions contemplated in this Agreement, and (v) any acquisition and related financing of Purchaser. Purchaser has heretofore made available to the Company accurate and complete copies of the Certificate of Incorporation and Bylaws, as currently in effect, of Purchaser.
- 3.2 Capitalization. As of the date hereof, the authorized capital stock of Purchaser consists of 103,000,000 shares of stock, of which 100,000,000 shares have been designated as Purchaser Stock and

3,000,000 shares have been designated as Preferred Stock, no par value per share. As of March 6, 2001, 31,399,172 shares of Purchaser Stock were issued and outstanding, no shares of Purchaser Preferred Stock were issued or outstanding. No other shares of capital stock of Purchaser were issued or outstanding. As of December 31, 2000, a total of approximately 8,247,000 shares of Purchaser Stock are reserved for future issuance to employees and directors upon exercise of any options, warrants or other rights to purchase or acquire any shares of capital stock of the Purchaser (including restricted stock, stock equivalents and stock units) ("Purchaser Options"). As of December 31, 2000, there were 6,147,000 Purchaser Options outstanding. Except as set forth in Section 3.2 of the Purchaser Disclosure Schedule and as otherwise contemplated by this Agreement, as of the date hereof there are no outstanding rights, subscriptions, warrants, puts, calls, unsatisfied preemptive rights, options or other agreements of any kind relating to any of the outstanding, authorized but unissued or unauthorized shares of capital stock or any other security of the Purchaser, and there is no authorized or issued security of any kind convertible into or exchangeable, for any such capital stock or other security. All issued and outstanding shares of Purchaser Stock are, and all shares of Purchaser Stock to be issued to Company stockholders in connection with the Merger will upon issuance be, duly authorized, validly issued, fully paid and non-assessable.

- 3.3 Authorization; Binding Agreement. Purchaser has all requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby, including, but not limited to, the Merger, have been duly and validly authorized by the Board of Directors of Purchaser, and no other corporate proceedings on the part of Purchaser are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Purchaser and constitutes the legal, valid and binding agreement of Purchaser, enforceable against Purchaser in accordance with its terms, subject to the Enforceability Exceptions.
- 3.4 Governmental Approvals. No Consent from or with any Governmental Authority on the part of Purchaser is required in connection with the execution or delivery by Purchaser of this Agreement or the consummation by Purchaser of the transactions contemplated hereby other than (i) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware in accordance with the DGCL; (ii) filings with the SEC, state securities laws administrators and the NYSE; (iii) filings under the HSR Act; (iv) such filings as may be required in any jurisdiction where Purchaser is qualified or authorized to do business as a foreign corporation in order to maintain such qualification or authorization; (v) those Consents required under any foreign Law; and (vi) those Consents that, if they were not obtained or made, would not be reasonably likely to have a Purchaser Material Adverse Effect.
- 3.5 No Violations. Except as set forth in Section 3.5 of the Purchaser Disclosure Schedule, the execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance by Purchaser with any of the provisions hereof will not (i) conflict with or result in any breach of any provision of the Certificate of Incorporation or Bylaws or other governing instruments of Purchaser, (ii) require any Consent under or result in a violation or breach of, or constitute (with or without due notice or lapse of time or both) a default (or give rise to any right of termination, cancellation or acceleration) under any of the terms, conditions or provisions of any agreement or other instrument to which Purchaser is a party or by which its assets are bound, (iii) result in the creation or imposition of any lien or encumbrance of any kind upon any of the assets of Purchaser or (iv) subject to obtaining the Consents from Governmental Authorities referred to in Section 3.4 hereof, contravene any Law to which Purchaser or its or any of its assets or properties are subject, except, in the case of clauses (ii), (iii) and (iv) above, for any deviations from the foregoing which would not be reasonably likely to have a Purchaser Material Adverse Effect.
- 3.6 Securities Filings. Purchaser has made available to the Company true and complete copies of (i) its Annual Report on Form 10-K for the year ended December 31, 1999, as filed with the SEC, (ii) its proxy statements relating to all of the meetings of stockholders (whether annual or special) of Purchaser since December 31, 1999, as filed with the SEC, and (iii) all other reports, statements and registration statements and amendments thereto (including, without limitation, Quarterly Reports on

Form 10-Q and Current Reports on Form 8-K, as amended) filed by Purchaser with the SEC since December 31, 1999. The reports and statements set forth in clauses (i) through (iii) above, and those subsequently provided or required to be provided pursuant to this Section 3.6, are referred to collectively herein as the "Purchaser Securities Filings." The Purchaser has timely filed with the SEC all of the Purchaser Securities Filings that have been filed prior to the date hereof. As of their respective dates, or as of the date of the last amendment thereof, if amended after filing, none of the Purchaser Securities Filings contained or, as to Purchaser Securities Filings subsequent to the date hereof, will contain, any untrue statement of a material fact or omitted or, as to Purchaser Securities Filings subsequent to the date hereof, will omit, to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Each of the Purchaser Securities Filings at the time of filing or as of the date of the last amendment thereof, if amended after filing, complied or, as to Purchaser Securities Filings subsequent to the date hereof, will comply in all material respects with the Securities Exchange Act or the Securities Act, as applicable.

- 3.7 Finders and Investment Bankers. Neither Purchaser nor any of its officers or directors has employed any broker or finder or otherwise incurred any liability for any brokerage fees, commissions or finders' fees in connection with the transactions contemplated hereby.
- 3.8 Vote Required; Ownership of Company Capital Stock; State Takeover Statutes. (a) The adoption and approval of this Agreement, the Merger and the transactions contemplated by this Agreement by the Board of Directors of Purchaser is the only corporate action necessary to approve the Merger and the transactions contemplated hereunder. No vote of any holder of any class or series of the Purchaser's capital stock is necessary to approve this Agreement, the Merger or the transactions contemplated hereunder.
 - (b) Neither the Purchaser nor any of the Purchaser Subsidiaries beneficially owns, either directly or indirectly, any shares of Company capital stock.
 - (c) Purchaser has taken all actions necessary under the GBCC and the DGCL to adopt and approve this Agreement, the Merger and the transactions contemplated by this Agreement. Immediately prior to the execution of this Agreement, Purchaser was not an "interested shareholder", as defined in Section 14-2-1110(11) of the GBCC with respect to the Merger. To the knowledge of the Purchaser, no other "fair price," moratorium," or other similar anti-takeover statute or regulation prohibits (by reason of the Purchaser's participation therein) the Merger or the other transactions contemplated by this Agreement.
- 3.9 Purchaser Financial Statements. The audited consolidated financial statements and unaudited interim financial statements (including any related notes and schedules) of Purchaser included or incorporated by reference in the Purchaser Securities Filings (the "Purchaser Financial Statements") have been prepared or will be prepared in accordance with generally accepted accounting principles applied on a consistent basis (except as may be indicated therein or in the notes thereto) and present or will present fairly, in all material respects, the financial position of Purchaser and the Purchaser Subsidiaries as at the dates thereof and the results of their operations and cash flows for the periods then ended, in each case in accordance with generally accepted accounting principles applied on a consistent basis, subject, in the case of the unaudited interim financial statements, to normal year-end audit adjustments, any other adjustments described therein and the fact that certain information and notes have been condensed or omitted in accordance with the Securities Exchange Act.
- 3.10 Absence of Certain Changes or Events; No Undisclosed Liabilities. Except as set forth in Section 3.10 of the Purchaser Disclosure Schedule, since December 31, 2000, through the date of this Agreement, there has not been (i) any event that has had or would reasonably be expected to have a Purchaser Material Adverse Effect or (ii) any declaration, payment or setting aside for payment any dividend or other distribution or redemption or other acquisition of any shares of capital stock of Purchaser by Purchaser.

- 3.11 Compliance with Laws. To the knowledge of Purchaser, the businesses of Purchaser and the Purchaser Subsidiaries have been operated in compliance with all Laws applicable thereto, except for any instances of non-compliance which would not be reasonably likely to have a Purchaser Material Adverse Effect.
- 3.12 Financing. Purchaser has sufficient funds and/or commitments for financing available and/or commitments for financing to pay (a) the aggregate Cash Consideration and cash in lieu of fractional shares to be paid in the Merger and (b) to repay the indebtedness of the Company to be repaid at the Closing pursuant to Section 5.11. No funds or other assets of the Company or any of the Company Subsidiaries will be used to fund any portion of the Merger Consideration. Except as set forth in Section 3.12 of the Purchase Disclosure Schedule, the Company and the Company Subsidiaries will not guarantee any financing for any portion of the Merger Consideration nor will any liens or encumbrances be imposed on any of the assets of the Company or any of the Company Subsidiaries in support of or to secure any such financing. Copies of the financing commitments have been furnished to the Company.
- 3.13 Tax-Free Reorganization. None of the Purchaser, or any of the Purchaser Subsidiaries has taken or agreed to take any action that would prevent the Merger from constituting a reorganization qualifying under the provisions of Section 368(a) of the Code.
- $3.14\,$ No Other Representations or Warranties. Purchaser acknowledges and agrees that, in entering into this Agreement and in consummating the transactions contemplated hereby:
 - (i) It has relied and will rely solely upon its own investigation and analysis and the representations and warranties contained in Article 2 of this Agreement. Without limiting the generality of the foregoing, it has not and will not rely on any of the information learned or provided in connection with any presentation made by management or other Company Representatives, contained in any document provided to it, any projections or forecasts or otherwise, including but not limited to, the undated Confidential Memorandum received from the Company's financial advisor (the "Confidential Memorandum").
 - (ii) None of the Company nor any of the Company representatives has made any statement, representation or warranty (including in the Confidential Memorandum or in any other document or any projections or forecasts) except for the representations and warranties made by the Company as expressly set forth in Article 2 of this Agreement.

ARTICLE 4

ADDITIONAL COVENANTS OF THE COMPANY

The Company covenants and agrees as follows:

- 4.1 Conduct of Business of the Company and the Company Subsidiaries. (a) Unless Purchaser shall otherwise agree in writing and except as expressly contemplated by this Agreement or as set forth on Section 4.1 of the Company Disclosure Schedule (the inclusion of any item having been consented to by Purchaser), during the period from the date of this Agreement to the Effective Time, (i) the Company shall conduct, and it shall cause each of the Company Subsidiaries to conduct, its or their businesses in the ordinary course, and the Company shall, and it shall cause each of the Company Subsidiaries to, use its or their commercially reasonable efforts to preserve intact its business organization, to keep available the services of its officers and key employees, and to maintain satisfactory relationships with all persons with whom it does business, and (ii) without limiting the generality of the foregoing, neither the Company nor any Company Subsidiary will:
 - (A) amend or propose to amend its Articles of Incorporation or Bylaws (or comparable governing instruments);
 - (B) authorize for issuance, issue, grant, sell, pledge or dispose of any shares of, or any options, warrants, commitments, subscriptions or rights of any kind to acquire or sell any shares

- of, the capital stock or other equity securities of the Company or any Company Subsidiary including, but not limited to, any securities convertible into or exchangeable for shares of stock of any class of the Company or any Company Subsidiary, except for the issuance of Company Shares pursuant to the exercise of Company Options outstanding on the date of this Agreement in accordance with their present terms;
- (C) split, combine or reclassify any shares of its capital stock or declare, pay or set aside any dividend or other distribution (whether in cash, stock or property or any combination thereof) in respect of its capital stock, other than dividends or distributions to the Company or any Company Subsidiary, or directly or indirectly redeem, purchase or otherwise acquire or offer to acquire any shares of its capital stock or other securities and other than pursuant to commitments outstanding on the date of this Agreement in accordance with their present terms as set forth on Schedule 4.1 of the Company Disclosure Schedule;
- (D) (i) create, incur, assume, forgive or make any changes to the terms or collateral of any debt for borrowed money, except incurrences that constitute refinancing of existing obligations on terms that are no less favorable to the Company or the Company Subsidiaries than the existing terms; (ii) assume, guarantee, endorse or otherwise become liable or responsible (whether directly, indirectly, contingently or otherwise) for the obligations of any person other than in the ordinary course of business; (iii) make any capital expenditures other than as set forth in Section 4.1 of the Company Disclosure Schedule; (iv) make any loans, advances or capital contributions to, or investments in, any other person (other than to a Company Subsidiary and customary travel, relocation or business advances to employees); (v) acquire the stock or assets of, or (subject to Section 7.1) merge or consolidate with, any other person; (vi) voluntarily incur any material liability or obligation (absolute, accrued, contingent or otherwise) other than in the ordinary course of business; or (vii) sell, transfer, mortgage, pledge, or otherwise dispose of, or encumber, or agree to sell, transfer, mortgage, pledge or otherwise dispose of or encumber, any assets or properties, real, personal or mixed material to the Company and the Company Subsidiaries taken as a whole other than to secure debt permitted under subclause (i) of this clause (D) or other than in the ordinary course of business;
- (E) increase in any manner the wages, salaries, bonus, compensation or other benefits of any of its officers or employees or enter into, establish, amend or terminate any employment, consulting, retention, change in control, collective bargaining, bonus or other incentive compensation, profit sharing, health or other welfare, stock option or other equity, pension, retirement, vacation, severance, termination, deferred compensation or other compensation or benefit plan, policy, agreement, trust, fund or arrangement with, for or in respect of, any shareholder, officer, director, other employee, agent, consultant or affiliate other than as required pursuant to the terms of agreements in effect on the date of this Agreement, or enter into or engage in any agreement, arrangement or transaction with any of its directors, officers, employees or affiliates;
- (F) (i) commence or settle any litigation or other proceedings with any Governmental Authority or other person, or (ii) make or rescind any election relating to Taxes, settle any claim, action, suit, litigation, proceeding, arbitration, investigation, audit or controversy relating to Taxes, change any method of accounting or make any other material change in its accounting or Tax policies or procedures;
- (G) knowingly commit or omit to do any act, which act or omission causes a breach of any covenant contained in this Agreement or is intended to cause any representation or warranty contained in this Agreement to become untrue in a material respect, as if each such representation and warranty were continuously made from and after the date hereof;

- (I) enter into any new line of business;
- (J) enter into any lease, contract or agreement pursuant to which the Company is obligated to pay or incur obligations of more than \$100,000 per year, other than the purchase of inventory in the ordinary course of business;
- (K) allow Company Debt to increase above \$21,392,928. "Company Debt" means the aggregate amount of obligations of the Company and its Subsidiaries as set forth on the attached Schedule 4.1(K);
- (L) allow the increase, if any, in Revolving Debt, from September 30, 2000, to be greater than the increase, if any, in Working Capital from September 30, 2000. "Revolving Debt" means the annual daily average loan balance under the Company's revolving credit facility, as set forth on Schedule 4.1(L). "Working Capital" means the Company's current assets less its current liabilities, as reflected on the Company Financial Statements; or
- (M) authorize any of, or agree to commit to do any of, the foregoing actions.
- (b) The Company shall, and the Company shall cause each of the Company Subsidiaries to, use its or their reasonable efforts to comply in all material respects with all Laws applicable to it or any of its properties, assets or business and maintain in full force and effect all the Company Permits necessary for, or otherwise material to, such business.
- 4.2 Notification of Certain Matters. The Company shall give prompt notice to Purchaser if any of the following occur after the date of this Agreement: (i) receipt of any written notice from any third party alleging that the Consent of such third party is or may be required in connection with the transactions contemplated by this Agreement, provided that such Consent would have been required to have been disclosed in this Agreement; (ii) receipt of any material written notice from any Governmental Authority (including, but not limited to, American Stock Exchange, Inc. ("AMEX") or any securities exchange) in connection with the transactions contemplated by this Agreement; (iii) the occurrence of an event which would be reasonably likely to have a Company Material Adverse Effect or which the Company believes would or would be reasonably likely to cause or constitute a material breach of any of its representations, warranties or covenants contained herein; or (iv) the commencement or threat of any Litigation involving the Company or any Company Subsidiary which, if pending on the date hereof, would have been required to have been disclosed in this Agreement or which relates to the consummation of the Merger.
- 4.3 Access and Information. Between the date of this Agreement and the Effective Time, the Company will give, and shall direct its accountants and legal counsel to give, Purchaser and its respective authorized representatives (including, without limitation, its financial advisors, accountants and legal counsel), at all reasonable times, access as reasonably requested to all offices and other facilities and to all contracts, agreements, commitments, books and records of or pertaining to the Company and the Company Subsidiaries, will permit the foregoing to make such reasonable inspections as they may require and will cause its officers to furnish Purchaser with (a) such financial and operating data and other information with respect to the business and properties of the Company and the Company Subsidiaries as Purchaser may from time to time reasonably request, and (b) a copy of each material report, schedule and other document filed or received by the Company or any Company Subsidiary pursuant to the requirements of applicable securities laws or AMEX.
- 4.4 Shareholder Approval. As soon as practicable after the registration statement has been declared effective, the Company shall call, give notice of, convene and hold a meeting of its shareholders for the purpose of approving the Company Proposals and for such other purposes as may be necessary or desirable in connection with effectuating the transactions contemplated hereby. Except as otherwise contemplated by this Agreement, the Company will use reasonable efforts to obtain any necessary approval by the Company's shareholders of the Company Proposals. Notwithstanding the foregoing, if the Board of Directors of the Company determines that to do so would result in a breach of the fiduciary duties of the Company's Board of Directors under applicable law, the Company, acting through its Board of Directors,

may refuse to call, give notice, convene and/or hold such a meeting of its shareholders for such purpose and/or may adjourn or terminate any such meeting previously called.

- 4.5 Reasonable Best Efforts. Subject to the terms and conditions herein provided, the Company agrees to use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective as promptly as practicable the Merger and the other transactions contemplated by this Agreement, including, but not limited to, (i) obtaining all Consents from Governmental Authorities and other third parties required for the consummation of the Merger and the transactions contemplated hereby and (ii) timely making all necessary filings under the HSR Act. Upon the terms and subject to the conditions hereof, the Company agrees to use reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary to satisfy the other conditions to Closing set forth herein.
- 4.6 Public Announcements. So long as this Agreement is in effect, the Company shall not, and shall cause its affiliates not to, (a) issue or cause the publication of any press release or any other announcement or communication with respect to the Merger or the other transactions contemplated hereby without the written consent of Purchaser, or (b) discuss with the press or the media this Agreement, the Merger or the transactions contemplated hereby (and will refer any and all questions and inquiries to Purchaser), except in any case under (a) or (b) where such release or announcement is required by applicable Law or pursuant to any applicable listing agreement with, or rules or regulations of, AMEX, in which case the Company, prior to making such announcement, will consult with Purchaser regarding the same.
- 4.7 Compliance. In consummating the Merger and the other transactions contemplated hereby, the Company shall comply in all material respects with the provisions of the Securities Exchange Act and shall comply, and/or cause the Company Subsidiaries to comply or to be in compliance, in all material respects, with all other applicable Laws.
- 4.8 Tax Opinion Certificate. The Company shall execute and deliver a certificate, in form reasonably satisfactory to Purchaser dated on or about the date that is two business days prior to the date the Prospectus/Proxy Statement is mailed and again dated as of the Closing Date (the "Company Tax Opinion Certificate") signed by an officer of the Company setting forth factual representations and covenants that will serve as a basis for the tax opinions required pursuant to Section 6.2(d) and Section 6.3(d) of this Agreement.
- $4.9\,$ Other Tax-related Certificates. The Company shall provide such certificates as may be required so that no withholding of taxes is required pursuant to Section 1445 of the Code.
- 4.10 SEC and Shareholder Filings. The Company shall send to Purchaser a copy of all material public reports and materials as and when it sends the same to its shareholders, the SEC or any state or foreign securities commission.
- 4.11 Accountant's Comfort Letter. The Company shall use reasonable efforts to cause to be delivered to the Purchaser a letter from its independent accountant, dated a date within two business days before the effective date of the Registration Statement, in a form reasonably satisfactory to the Purchaser and customary in scope for comfort letters delivered by independent accountants in connection with registration statements on Form S-4 under the Securities Act.

ARTICLE 5

ADDITIONAL COVENANTS OF PURCHASER

Purchaser covenants and agrees as follows:

5.1 Access and Information. Between the date of this Agreement and the Effective Time, Purchaser will give, and shall direct its accountants and legal counsel to give, the Company and its authorized representatives (including, without limitation, its financial advisors, accountants and legal

counsel), at all reasonable times upon reasonable prior notice by the Company, access as reasonably requested to all contracts, agreements, commitments, books and records of or pertaining to Purchaser and the Purchaser Subsidiaries, will permit the foregoing to make such reasonable inspections as they may require and will cause its officers promptly to furnish the Company with (a) such financial and operating data and other information with respect to the business and properties of Purchaser and its subsidiaries as the Company may from time to time reasonably request and (b) a copy of each material report, schedule and other document filed or received by Purchaser or any of its subsidiaries pursuant to the requirements of applicable securities laws or the NYSE, all only to the extent reasonably required by the Company to facilitate and consummate the transactions contemplated by this Agreement.

5.2 Notification of Certain Matters. Purchaser shall give prompt notice to the Company if any of the following occur after the date of this Agreement: (i) receipt of any notice or other communication in writing from any third party alleging that the Consent of such third party is or may be required in connection with the transactions contemplated by this Agreement, provided that such Consent would have been required to have been disclosed in this Agreement; (ii) receipt of any material notice or other communication from any Governmental Authority (including, but not limited to, the NYSE or any securities exchange) in connection with the transactions contemplated by this Agreement; (iii) the occurrence of an event which would be reasonably likely to have a Purchaser Material Adverse Effect or (iv) the commencement or threat of any Litigation involving or affecting Purchaser or any of its subsidiaries, or any of their respective properties or assets, or, to its knowledge, any employee, agent, director or officer, in his or her capacity as such, of Purchaser or any of its subsidiaries which, if pending on the date hereof, would have been required to have been disclosed in this Agreement or which relates to the consummation of the Merger.

5.3 [Reserved.]

- 5.4 Reasonable Best Efforts. Subject to the terms and conditions herein provided, Purchaser agrees to use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective as promptly as practicable the Merger and the other transactions contemplated by this Agreement, including, but not limited to, (i) obtaining all Consents from Governmental Authorities and other third parties required for the consummation of the Merger and the other transactions contemplated hereby and (ii) timely making all necessary filings under the HSR Act. Upon the terms and subject to the conditions hereof, Purchaser agrees to use reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary to satisfy the other conditions to Closing set forth herein.
- 5.5 Public Announcements. So long as this Agreement is in effect, the Purchaser shall not, and shall cause its affiliates not to, (a) issue or cause the publication of any press release or any other announcement or communication with respect to the Merger or the other transactions contemplated hereby without the written consent of the Company, or (b) discuss with the press or the media this Agreement, the Merger or the transactions contemplated hereby (and will refer any and all questions and inquiries to Purchaser), except in any case under (a) or (b) where such release or announcement is required by applicable Law or pursuant to any applicable listing agreement with, or rules or regulations of, NYSE, in which case the Purchaser, prior to making such announcement, will consult with the Company regarding the same.
- 5.6 Compliance. In consummating the Merger and the other transactions contemplated hereby, Purchaser shall comply in all material respects with the provisions of the Securities Exchange Act and the Securities Act and shall comply, and/or cause its subsidiaries to comply or to be in compliance, in all material respects, with all applicable Laws.
- 5.7 SEC and Shareholder Filings. Purchaser shall send to the Company a copy of all material public reports and materials as and when it sends the same to its shareholders, the SEC or any state or foreign securities commission.

- 5.8 Tax Opinion Certificate. Purchaser shall execute and deliver a certificate, in form reasonably satisfactory to Company dated on or about the date that is two business days prior to the date the Prospectus/Proxy Statement is mailed and again dated as of the Closing Date (the "Purchaser Tax Opinion Certificate"), signed by an officer of Purchaser setting forth factual representations and covenants that will serve as a basis for the tax opinions required pursuant to Section 6.2(d) and Section 6.3(d) of this Agreement.
- 5.9 Indemnification and Insurance. (a) As of the Effective Time, the indemnification and exculpation provisions contained in the Bylaws and the Certificate of Incorporation of the Surviving Corporation shall be at least as favorable to individuals who immediately prior to the Closing Date were directors, officers, agents or employees of the Company or its Subsidiaries or otherwise entitled to indemnification under the Company's or its Subsidiaries' Bylaws or Articles of Incorporation as those contained in the Bylaws and the Articles of Incorporation of the Company or its Subsidiaries, respectively, and shall not be amended, repealed or otherwise modified for a period of six years after the Closing Date in any manner that would adversely affect the rights thereunder of any Indemnified Party. The Company hereby covenants that it shall, to the fullest extent permitted under Georgia law and regardless of whether the Merger becomes effective, indemnify, defend and hold harmless, and after the Effective Time, Purchaser shall, to the fullest extent permitted under applicable law, indemnify, defend and hold harmless, each of the individuals who are, or at any time have been, an officer, director, employee or agent of the Company or any of the Company Subsidiaries (but excluding Advanced Animations, Inc.), or otherwise entitled to indemnification under the Company's or Company Subsidiary's (but excluding Advanced Animations, Inc.), articles of incorporation or bylaws (each, an "Indemnified Party"), against any costs or expenses (including reasonable attorneys' fees), judgments, fines, losses, claims, damages, liabilities and amounts paid in settlement in connection with any claim, action, suit, proceeding or investigation, including, without limitation, liabilities arising out of this Agreement or under the Securities Exchange Act, occurring through the Closing Date, and in the event of any such claim, action, suit, proceeding or investigation (whether arising before or after the Effective Time), (i) the Company or Purchaser shall pay the reasonable fees and expenses of counsel selected by the Indemnified Parties, which counsel shall be reasonably satisfactory to the Company or the Surviving Corporation, promptly as statements therefor are received, and (ii) the Company and the Surviving Corporation will cooperate in the defense of any such matter; provided, however, that neither the Company nor the Surviving Corporation shall be obliged pursuant to this Section 5.9 to pay the fees and disbursements of more than one counsel for all Indemnified Parties in any single action except to the extent that, in the opinion of counsel for the Indemnified Parties, two or more of such Indemnified Parties have conflicting interests in the outcome of such action. Purchaser shall reimburse all expenses, including reasonable attorney's fees and expenses, incurred by any person to enforce the obligations of Purchaser under this Section 5.9. To the fullest extent permitted by Law, Purchaser shall advance expenses.
 - (b) If the Surviving Corporation or any of its successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any person, then and in each such case, proper provision shall be made so that the successors and assigns of the Surviving Corporation assume the obligations set forth in this Section 5.9.
 - (c) Purchaser shall use its reasonable best efforts to obtain and maintain in effect for six years from the Effective Time, the Company's current directors' and officers' liability insurance covering those persons who are currently covered by the Company's directors' and officers' liability insurance policy (a copy of which has heretofore been delivered to Purchaser); provided, however, that in no event shall Purchaser be required to expend in any year an amount in excess of 300% of the annual premiums currently paid by the Company for such insurance, and, provided, further, that if the annual premiums of such insurance coverage exceed such amounts, Purchaser shall obtain a policy with the greatest coverage available for a cost not exceeding such amount.

- 5.10 Repayment of Certain Company Indebtedness. At the Closing, Purchaser shall repay in full the outstanding principal and accrued interest of the indebtedness of the Company described in Section 5.10 of the Company Disclosure Schedule.
- 5.11 Refinancing. Purchaser shall refinance and replace the indebtedness of the Company identified in Section 5.11 of the Company Disclosure Schedule. Purchaser acknowledges and agrees that all such indebtedness must be refinanced and replaced as a result of the Merger, unless other accommodations are made with respect thereto between Purchaser and the lender thereunder.

ARTICLE 6

CONDITIONS

- 6.1 Conditions to Each Party's Obligations. The respective obligations of each party to effect the Merger shall be subject to the fulfillment or waiver at or prior to the Effective Time of the following conditions:
 - (a) Shareholder Approval. The Company Proposals shall have been approved at or prior to the Effective Time by the requisite vote of the shareholders of the Company required under the GBCC.
 - (b) No Injunction or Action. No order, statute, rule, regulation, executive order, stay, decree, judgment or injunction shall have been enacted, entered, promulgated or enforced by any court or other Governmental Authority since the date of this Agreement which prohibits or prevents the consummation of the Merger which has not been vacated, dismissed or withdrawn prior to the Effective Time. The Company and Purchaser shall use their commercially reasonable efforts to have any of the foregoing vacated, dismissed or withdrawn by the Effective Time.
 - (c) HSR Act. Any waiting period applicable to the Merger under the HSR Act shall have expired or early termination thereof shall have been granted.
 - (d) Registration Statement. The Registration Statement shall have been declared effective and no stop order suspending the effectiveness of the Registration Statement shall have been issued and no action, suit, proceeding or investigation for that purpose shall have been initiated or threatened by any Governmental Authority.
 - (e) Blue Sky. Purchaser shall have received all state securities law authorizations necessary to consummate the transactions contemplated hereby.
 - (f) Listing of Purchaser Stock. The shares of Purchaser Stock comprising the Merger Consideration shall have been approved for listing on the NYSE, subject only to official notice of issuance.
 - (g) Governmental Approval. All Consents of any Governmental Authority required for the consummation of the Merger and the transactions contemplated by this Agreement shall have been obtained, except those Consents the failure of which to obtain will not be reasonably likely to have a Company Material Adverse Effect or a Purchaser Material Advise Effect.
- 6.2 Conditions to Obligations of the Company. The obligation of the Company to effect the Merger shall be subject to the fulfillment at or prior to the Effective Time of the following additional conditions, any one or more of which may be waived by the Company:
 - (a) Purchaser Representations and Warranties. The representations and warranties of Purchaser set forth in this Agreement shall be true and correct in all material respects as of date hereof and as of the Closing Date as if made on and as of the Closing Date, except those representations and warranties that speak of an earlier date, which shall be true and correct in all material respects as of such earlier date (it being understood that, for purposes of determining the accuracy of such representations and warranties, any update of or modification to the Purchaser Disclosure Schedule

made or purported to have been made after the date of this Agreement shall be disregarded). Notwithstanding anything to the contrary in this Agreement, this Section 6.2(a) will be deemed to have been satisfied even if such representations or warranties are not true and correct in all material respects unless the failure of any of the representations or warranties to be so true and correct shall have had, or shall be reasonably likely to have, a Purchaser Material Adverse Effect.

- (b) Performance by Purchaser. Purchaser shall have performed and complied with all the covenants and agreements in all material respects and satisfied in all material respects all the conditions required by this Agreement to be performed or complied with or satisfied by Purchaser at or prior to the Effective Time. Notwithstanding anything to the contrary in this Agreement, this Section 6.2(b) will be deemed to have been satisfied even if Purchaser has not performed or complied with all agreements, covenants and obligations required by this Agreement to be so performed or complied with unless the failure to perform or comply with such agreements, obligations and covenants shall have had, or shall be reasonably likely to have, a Purchaser Material Adverse Effect.
- (c) Certificates and Other Deliveries. Purchaser shall have delivered, or caused to be delivered, to the Company: (i) a certificate executed on its behalf by its President or another authorized officer to the effect that the conditions set forth in Sections 6.2(a) and (b) hereof have been satisfied; (ii) a certificate of good standing from the Secretary of State of the State of Delaware stating that Purchaser is a validly existing corporation in good standing; (iii) duly adopted resolutions of the Board of Directors of Purchaser adopting and approving the Merger and adopting and approving the execution, delivery and performance of this Agreement and the instruments contemplated hereby, each certified by its Secretary; (iv) a true and complete copy of the Certificate of Incorporation of the Purchaser certified by the Secretary of State of Delaware and a true and complete copy of the Bylaws of the Purchaser certified by the Secretary of Purchaser; (v) the duly executed Purchaser Tax Opinion Certificate and the Purchaser's Tax Opinion.
- (d) Tax Opinion. The Company shall have received an opinion from its tax counsel dated on or about the date that is two business days prior to the date the Prospectus/Proxy Statement is first mailed to stockholders of the Company, substantially in the form of Exhibit 6.2(d) attached hereto, which opinion shall not have been withdrawn or modified in any material respect, and again dated as of the Closing Date. For purposes of rendering its opinion, the Company's tax counsel may rely on the statements and representations set forth in the Company Tax Opinion Certificate and the Purchaser Tax Opinion Certificate, without regard to any qualification as to knowledge and belief.
- (e) Payment of Certain Indebtedness. Purchaser shall have repaid the indebtedness of the Company described in Section 5.10 and Section 5.11 (if necessary) of the Company Disclosure Schedule.
- (f) Fairness Opinion. McDonald Investments shall have confirmed in writing addressed to the Board of the Company for inclusion in the Prospectus/Proxy Statement its verbal fairness opinion described in Section 2.19 (the "Fairness Opinion") within five (5) business days after the date hereof.
- 6.3 Conditions to Obligations of Purchaser. The obligations of Purchaser to effect the Merger shall be subject to the fulfillment at or prior to the Effective Time of the following additional conditions, any one or more of which may be waived by Purchaser:
 - (a) Company Representations and Warranties. The representations and warranties of the Company set forth in this Agreement shall be true and correct in all material respects as of date hereof and as of the Closing Date as if made on and as of the Closing Date, except those representations and warranties that speak of an earlier date, which shall be true and correct in all material respects as of such earlier date (it being understood that, for purposes of determining the accuracy of such representations and warranties, any update of or modification to the Company Disclosure Schedule made or purported to have been made after the date of this Agreement shall be disregarded). Notwithstanding anything to the contrary in this Agreement, this Section 6.3(a) will be

deemed to have been satisfied even if such representations or warranties are not true and correct in all material respects unless the failure of any of the representations or warranties to be so true and correct shall have had, or shall be reasonably likely to have, a Company Material Adverse Effect.

- (b) Performance by the Company. The Company shall have performed and complied with all the covenants and agreements in all material respects and satisfied in all material respects all the conditions required by this Agreement to be performed or complied with or satisfied by the Company at or prior to the Effective Time. Notwithstanding anything to the contrary in this Agreement, this Section 6.3(b) will be deemed to have been satisfied even if the Company has not performed or complied with all agreements, covenants and obligations required by this Agreement to be so performed or complied with unless the failure to perform or comply with such agreements, obligations and covenants shall have had, or shall be reasonably likely to have, a Company Material Adverse Effect.
- (c) Certificates and Other Deliveries. The Company shall have delivered, or caused to be delivered, to Purchaser (i) a certificate executed on its behalf by its President or another duly authorized officer to the effect that the conditions set forth in Sections 6.3(a) and (b) hereof have been satisfied; (ii) a certificate of good standing from the Secretary of State of the State of Georgia stating that the Company is a validly existing corporation in good standing; (iii) duly adopted resolutions of its Board of Directors adopting and approving the Merger and adopting and approving the execution, delivery and performance of this Agreement and the instruments contemplated hereby, and of the Company's shareholders approving the Company Proposals, each certified by the Secretary of the Company; (iv) a true and complete copy of the Articles of Incorporation of the Company certified by the Secretary of State of the State of Georgia and a true and complete copy of the Bylaws of the Company certified by the Secretary thereof; (v) the duly executed Company Tax Opinion Certificate and the Tax Opinion; and (vi) a copy of the Fairness Opinion addressed to the Company.
- (d) Tax Opinion. Purchaser shall have received an opinion from its tax counsel dated on or about the date that is two business days prior to the date the Prospectus/Proxy Statement is first mailed to stockholders of the Company, substantially in the form of Exhibit 6.3(d) attached hereto, which opinion shall not have been withdrawn or modified in any material respect, and again dated as of the Closing Date. For purposes of rendering its opinion, the Purchaser's tax counsel may rely on the statements and representations set forth in the Company Tax Opinion Certificate and the Purchaser Tax Opinion Certificate, without regard to any qualification as to knowledge and belief.
- (e) Required Consents. The required Consents of the persons to the Merger or the transactions contemplated hereby that are identified in Section 6.3(b) of the Company Disclosure Schedule shall have been obtained and shall be in full force and effect.
- (f) Asset Purchase Agreement. All conditions to the closing of that certain Asset Purchase Agreement, dated as of the date hereof, by and among PII Ventures, L.L.C., Toth and Purchaser, shall have been fulfilled, except as they relate to the actual closing thereof.
- 6.4 Frustration of Conditions. Neither Purchaser nor the Company may rely on the failure of any condition set forth in this Article VI to be satisfied if such failure was caused by such party's failure to comply with or perform any of its covenants or obligations set forth in this Agreement.

ARTICLE 7

TERMINATION AND ABANDONMENT

- 7.1 Termination. This Agreement may be terminated at any time prior to the Effective Time, whether before or after approval of the stockholders of the Company:
 - (a) By mutual written consent of Purchaser and the Company.
 - (b) By either Purchaser or the Company if:
 - (i) the Merger shall not have been consummated on or prior to June 30, 2001 (the "Initial Closing Deadline"), provided that the Company or Purchaser, as applicable, may elect to extend the Initial Closing Deadline by notice to the Purchaser or Company, as applicable, for such period as may be appropriate, but not later than July 30, 2001, in connection with (1) Company's or Purchaser's efforts to cure any breach of any representation, warranty or covenant under this Agreement as to which it has been given notice by Purchaser or Company, as applicable, pursuant to this Agreement (including but not limited to Section 7.1(c)), (2) obtaining the tax opinion referred to in Section 6.2(d), (3) obtaining the tax opinion referred to in Section 6.3(d), or (4) calling and holding the Company's stockholder meeting to consider the Company Proposals and to schedule and consummate the closing; and provided further that the Company or Purchaser, as applicable, may elect to extend the Initial Closing Deadline by notice to Purchaser or the Company, as applicable, for such period as may be appropriate, but not later than August 31, 2001, if the Registration Statement has not been declared effective at a time sufficient to allow the Company's stockholders meeting to consider the Company Proposals prior to the Initial Closing Deadline (the Initial Closing Deadline, as so extended by the Company or Purchaser, is referred to herein as the "Closing Deadline"), provided, however, that the right to terminate this Agreement pursuant to this Section 7.1(b)(i) shall not be available to any party whose failure to perform any of its obligations under this Agreement results in the failure of the Merger to be consummated by such time;
 - (ii) after the Company has called, given notice of, duly convened and held a meeting as required pursuant to Section 4.4, the vote of the Company's stockholders taken at such meeting or at any adjournment or postponement thereof, shall be insufficient to approve the Company Proposals; or
 - (iii) any Governmental Authority shall have issued an order, decree or ruling or taken any other action permanently enjoining, restraining or otherwise prohibiting the consummation of the Merger and such order, decree or ruling or other action shall have become final and nonappealable.
 - (c) By Purchaser if the Company shall have breached in any material respect any of its representations, warranties or covenants under this Agreement, such breach shall constitute a Company Material Adverse Effect, Purchaser shall have given written notice to the Company of such breach promptly after learning of the same, and such breach shall not be reasonably cured on or prior to the Closing Deadline.
 - (d) By the Company if the Purchaser shall have breached in any material respect any of its representations, warranties or covenants under this Agreement, and such breach shall constitute a Purchaser Material Adverse Effect, the Company shall have given written notice to the Purchaser of such breach promptly after learning of the same, and such breach shall not be reasonably cured on or prior to the Closing Deadline.
 - (e) By Purchaser if (i) the Board of Directors of the Company shall have withdrawn or modified in a manner adverse to Purchaser its approval or recommendation of the Company Proposals, or approved or recommended any Company Superior Transaction (as defined below) or failed within a reasonable time after the Registration Statement is declared effective by the SEC, to call the Company stockholder meeting to consider the Company Proposals or to mail the Prospectus/

Proxy Statement to its stockholders, or (ii) the Board of Directors of the Company shall have resolved to take any of the foregoing actions.

(f) By the Company if, prior to the approval of the Company Proposals by the stockholders of the Company, the Board of Directors of the Company determines in good faith by majority vote, with respect to any Company Takeover Proposal received by the Company after the date hereof, that the Company Takeover Proposal constitutes a Company Superior Transaction and is in the best interest of the Company and its stockholders and the Board of Directors of the Company has received the advice of its independent financial advisors as to whether the Company Takeover Proposal constitutes a Company Superior Transaction. For purposes hereof, a "Company Superior Transaction" means a Company Takeover Proposal which the Board of Directors of the Company determines in good faith, by majority vote, is more favorable to the Company and its stockholders than the Merger. Nothing in this Agreement shall prevent or limit the Company from providing information to, or discussing or negotiating with, any third person concerning any Company Takeover Proposal. For purposes hereof, a "Company Takeover Proposal" means any proposal with respect to any recapitalization, merger, consolidation, or other business combination involving the Company, or the acquisition of 50% or more of the outstanding capital stock of the Company or any Significant Subsidiary (as defined in Regulation 12b-2 promulgated under the Securities Exchange Act) of the Company or the acquisition of 50% or more of the assets of the Company and the Company Subsidiaries, taken as a whole, in a single transaction or series of related transactions.

The party desiring to terminate this Agreement pursuant to the preceding paragraphs shall give written notice of such termination to the other party in accordance with Section 8.5 hereof.

- 7.2 Effect of Termination and Abandonment. (a) In the event of termination of this Agreement and the abandonment of the Merger pursuant to this Article 7, this Agreement (other than Sections 7.2, 8.1, 8.4, 8.5, 8.6, 8.7, 8.8, 8.9, 8.10, 8.11, 8.12, 8.13, 8.14, 8.15 and 8.16) shall become void and of no effect with no liability on the part of any party hereto (or of any of its directors, officers, employees, agents, legal or financial advisors or other representatives); provided, however, that no such termination shall relieve any party hereto from any liability for any breach of this Agreement prior to termination. Notwithstanding the foregoing, (a) in no event shall the Company be liable or responsible for any breach of any representation, warranty or covenant unless the Purchaser terminates the Agreement on the basis of such breach and (i) in the case of breaches of any representation or warranty the Company's liability shall be limited to the lesser of the actual damages incurred by the Purchaser as a direct result of such breaches and \$9,000,000 and (ii) in the case of a breach of a covenant, the Company's liability shall be limited to payment of the Termination Fee as provided in Section 7.2(b); and (b) in no event shall the Purchaser be liable or responsible for any breach of any representation, warranty or covenant unless the Company terminates the Agreement on the basis of such breach and (i) in the case of breaches of any representation, warranty or covenant the Purchaser's liability shall be limited to the lesser of the actual damages incurred by the Company as a direct result of such breaches and \$9,000,000; provided, however, that the foregoing limitation shall in no event apply in connection with any willful breach of a material covenant in this Agreement resulting in the failure of the closing of the Merger. If this Agreement is terminated as provided herein, each party shall use its commercially reasonable efforts to redeliver all documents, work papers and other material (including any copies thereof) of any other party relating to the transactions contemplated hereby, whether obtained before or after the execution hereof, to the party furnishing the same.
 - (b) In the event that this Agreement is terminated pursuant to any of the following:
 - (i) by Purchaser pursuant to Section 7.1(b)(i), but only if all of the following conditions are fully satisfied: (w) the only reason that the Effective Time has not occurred by the Closing Deadline is the failure of the Company's tax counsel to deliver the tax opinion contemplated in Section 6.2(d), (x) Purchaser has delivered a written confirmation from a nationally recognized tax counsel that it would execute and deliver the tax opinion contemplated in Section 6.2(d), (y) the Purchaser Tax Opinion Certificate and the Company Tax Opinion Certificate have been

delivered in substantially the forms agreed upon at the date of this Agreement, and (z) Purchaser has delivered to the Company the tax opinion of its tax counsel contemplated in Section 6.3(d); or

- (ii) pursuant to Section 7.1(b)(ii); or
- (iii) by Purchaser pursuant to Section 7.1(c) as a result of a breach by the Company of a covenant of the Company in this Agreement; or
- (iv) after the date of this Agreement and prior to termination of this Agreement a bona fide Company Takeover Proposal shall have been made known to the Company or has been made directly to its stockholders generally or any person shall have publicly announced an intention (whether or not conditional) to make a bona fide Company Takeover Proposal (a "Competing Company Takeover Proposal"), and thereafter this Agreement is (x) terminated pursuant to Section 7.1(e)(i) and within six months of such termination the Company or any of the Company Subsidiaries enters into a definitive agreement as to such Competing Company Takeover Proposal, (y) terminated by Purchaser pursuant to Section 7.1(e), or (z) terminated by the Company pursuant to Section 7.1(f),

then the Company shall promptly pay Purchaser a fee equal to \$9,000,000 (the "Purchaser Termination Fee"), payable by wire transfer of same day funds. The Company acknowledges that the agreements contained in this Section 7.2(b) are an integral part of the transactions contemplated by this Agreement and that, without these agreements, Purchaser would not enter into this Agreement. Notwithstanding the foregoing, no fee shall be paid pursuant to this Section 7.2(b) if Purchaser shall be in material breach of its obligations hereunder. Purchaser acknowledges that payments made under this Section 7.2(b) shall constitute its exclusive remedy with respect to any termination of this Agreement that gives rise to such payment obligation. The parties further acknowledge and agree that if the Termination Fee shall be payable by the Company to the Purchaser as provided in this Agreement that the Company shall have no other or further liability or responsibility to Purchaser for any breach of this Agreement or otherwise and the Purchaser's sole remedy shall be payment of the Termination Fee.

(c) In the event of any termination of this Agreement pursuant to this Article 7, Purchaser shall not hire or solicit for employment any officer or key management employee of the Company or any of the Company Subsidiaries. For the purposes of this Agreement, the term "key management employee" has the meaning set forth in Section 7.2(c) of the Company Disclosure Schedule.

ARTICLE 8

MISCELLANEOUS

8.1 Confidentiality. Unless (i) otherwise expressly provided in this Agreement, (ii) required by applicable Law or any listing agreement with, or the rules and regulations of, any applicable securities exchange, (iii) necessary to secure any required Consents as to which the other party has been advised or (iv) consented to in writing by Purchaser and the Company, any information or documents furnished in connection herewith shall be kept strictly confidential by the Company, Purchaser and their respective officers, directors, employees and agents. Prior to any disclosure pursuant to the preceding sentence, the party intending to make such disclosure shall consult with the other party regarding the nature and extent of the disclosure. Nothing contained herein shall preclude disclosures to the extent necessary to comply with accounting, SEC and other disclosure obligations imposed by applicable Law. To the extent required by such disclosure obligations, Purchaser or the Company, after consultation with the other party, may file with the SEC a Report on Form 8-K pursuant to the Securities Exchange Act with respect to the Merger. In connection with any filing with the SEC of a proxy statement or amendment thereto under the Securities Exchange Act, the Company or Purchaser, after consultation with the other party, may include any information required to be included therein with respect to the Merger with respect to the other party, and thereafter distribute said proxy statement. Purchaser and the Company shall cooperate with the other

and provide such information and documents as may be required in connection with any such filings. In the event the Merger is not consummated, each party shall return to the other any documents furnished by the other and all copies thereof any of them may have made and will hold in absolute confidence any information obtained from the other party except to the extent (i) such party is required to disclose such information by Law or such disclosure is necessary or desirable in connection with the pursuit or defense of a claim, (ii) such information was known by such party prior to such disclosure or was thereafter developed or obtained by such party independent of such disclosure or (iii) such information becomes generally available to the public other than by breach of this Section 8.1. Prior to any disclosure of information pursuant to the exception in clause (i) of the preceding sentence, the party intending to disclose the same shall so notify the party which provided the name in order that such party may seek a protective order or other appropriate remedy should it choose to do so.

- $8.2\,$ Amendment and Modification. This Agreement may be amended, modified or supplemented only by a written agreement between the Company and Purchaser.
- 8.3 Waiver of Compliance; Consents. Any failure of a party or Purchaser to comply with any obligation, covenant, agreement or condition herein may be waived by the other party, only by a written instrument signed by the party granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure. Whenever this Agreement requires or permits consent by or on behalf of any party hereto, such consent shall be given in writing in a manner consistent with the requirements for a waiver of compliance as set forth in this Section 8.3.
- 8.4 Survival. The respective representations, warranties, covenants and agreements of the Company and Purchaser contained herein or in any certificates or other documents delivered prior to or at the Closing shall survive the execution and delivery of this Agreement, notwithstanding any investigation made or information obtained by the other party, but shall terminate at the Effective Time, except for those covenants contained in Article 1, Section 5.9 and this Article 8, which shall survive beyond the Effective Time in accordance with their terms.
- 8.5 Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given when delivered in person, by facsimile, receipt confirmed, or on the next business day when sent by overnight courier or on the second succeeding business day when sent by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified by like notice):
 - (i) if to the Company, to:
 VSI Holdings, Inc.
 41000 Woodward Avenue
 Bloomfield Hills, Michigan 48304-2263
 Attention: Steve Toth, Jr.
 Telecopy: (248) 647-8540

with a copy to (but which shall not constitute notice to the Company):
Miller, Canfield, Paddock and Stone, P.L.C.
840 West Long Lake Road
Suite 200
Troy, Michigan 48098
Attention: Thomas G. Appleman, Esq.
Telecopy: (248) 879-2001

and

(ii) if to Purchaser to:
 SPX Corporation
 700 Terrace Point Drive
 Muskegon, Michigan 49443-3301
 Attention: Christopher J. Kearney, Esq.
 Telecopy: (231) 724-5940

with a copy to:

SPX Corporation 28635 Mound Road Warren, Michigan 48092 Attention: Fabrizio A. Rasetti, Esq. Telecopy: (810) 578-7470

and an additional copy to (but which shall not constitute notice to Purchaser):

Gardner, Carton & Douglas Quaker Tower 321 North Clark Street Chicago, Illinois 60610-4795 Attention: Stephen A. Tsoris, Esq. Telecopy: (313) 644-3381

- 8.6 Binding Effect; Assignment. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto prior to the Effective Time without the prior written consent of the other parties hereto.
- 8.7 Expenses. Whether or not the Merger is consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such costs or expenses.
- 8.8 Governing Law. This Agreement shall be deemed to be made in, and shall be interpreted, construed and governed by and in accordance with the internal laws of, the Laws of the State of Michigan, except that Georgia Law and Delaware Law shall apply to the Merger, in each case, without regard to principles of conflicts of law thereof. Each of the Company and Purchaser hereby irrevocably and unconditionally consents to submit to the jurisdiction of the federal and state courts located in Michigan for any litigation arising out of or relating to this Agreement and the transactions contemplated hereby (and agrees not to commence any litigation relating thereto except in such courts), waives any objection to the laying of venue of any such litigation in such courts and agrees not to plead or claim in any such court that such litigation brought therein has been brought in an inconvenient forum.
- 8.9 Counterparts. This Agreement may be executed in one or more counterparts, each of which together be deemed an original, but all of which together shall constitute one and the same instrument.
- 8.10 Interpretation. The article and section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the parties and shall not in any way affect the meaning or interpretation of this Agreement. As used in this Agreement, (i) the term "person" shall mean and include an individual, a partnership, a joint venture, a corporation, a limited liability company, a trust, an association, an unincorporated organization, a Governmental Authority and any other entity, (ii) unless otherwise specified herein, the term "affiliate," with respect to any person, shall mean and include any person controlling, controlled by or under common control with such person, (iii) the term "subsidiary" of any specified person shall mean any corporation majority or more of the outstanding voting power of which, or any partnership, joint venture, limited liability company or other entity majority or more of the total equity interest of which, is directly or indirectly owned by such specified person, (iv) the term

"knowledge," when used with respect to the Company, shall mean the knowledge of the executive officers of the Company, and when used with respect to Purchaser, shall mean the knowledge of the executive officers of Purchaser, and (v) the term "including" shall mean "including, without limitation". This Agreement is being entered into by and among competent and sophisticated parties who are experienced in business matters and represented by counsel and other advisors, and have been reviewed by the parties and their counsel and other advisors. Therefore, any ambiguous language in this Agreement will not necessarily be construed against any particular party as the drafter of the language.

- 8.11 Entire Agreement. This Agreement and the documents or instruments referred to herein including, but not limited to, the Exhibit(s) attached hereto and the Disclosure Schedules referred to herein, which Exhibit(s) and Disclosure Schedules are incorporated herein by reference, embody the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants, or undertakings, other than those expressly set forth or referred to herein. This Agreement supersedes all prior agreements and the understandings between the parties with respect to such subject matter.
- 8.12 Severability. In case any provision in this Agreement shall be held invalid, illegal or unenforceable in a jurisdiction, such provision shall be modified or deleted, as to the jurisdiction involved, only to the extent necessary to render the same valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby nor shall the validity, legality or enforceability of such provision be affected thereby in any other jurisdiction.
- 8.13 Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, the parties further agree that each party shall be entitled to an injunction or restraining order to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court of the United States or any state having jurisdiction, this being in addition to any other right or remedy to which such party may be entitled under this Agreement, at law or in equity.
- 8.14 Third Parties. Nothing contained in this Agreement or in any instrument or document executed by any party in connection with the transactions contemplated hereby shall create any rights in, or be deemed to have been executed for the benefit of, any person or entity that is not a party hereto or thereto or a successor or permitted assign of such a party; provided however, that the parties hereto specifically acknowledge that (a) the provisions of this Agreement are for the benefit of, and shall be enforceable by, the stockholders and optionholders of the Company and (b) the provisions of Section 5.9 hereof are intended to be for the benefit of, and shall be enforceable by, the current or former employees, officers and directors of the Company and/or the Company Subsidiaries affected thereby and their heirs and representatives.
- 8.15 Disclosure Schedules. The Company and Purchaser acknowledge that the Company Disclosure Schedule and the Purchaser Disclosure Schedule (i) relate to certain matters concerning the disclosures required and transactions contemplated by this Agreement, (ii) are qualified in their entirety by reference to specific provisions of this Agreement and (iii) are not intended to constitute and shall not be construed as indicating that such matter is required to be disclosed, nor shall such disclosure be construed as an admission that such information is material with respect to the Company or Purchaser.

[BALANCE OF PAGE INTENTIONALLY LEFT BLANK.]

IN WITNESS WHEREOF, each of the parties hereto have caused this Agreement and Plan of Merger to be signed and delivered by their respective duly authorized officers as of the date first above written.

VSI HOLDINGS, INC.

By: /s/ Thomas W. Marquis Name: Thomas W. Marquis Title: Senior Vice President/CFO

SPX CORPORATION

By: /s/ Thomas J. Riordan Name: Thomas J. Riordan Title: Vice President

A-38

CERTIFICATE OF SECRETARY OF SPX CORPORATION

I, Christopher J. Kearney, Secretary of SPX Corporation, a corporation organized under the laws of the State of Delaware (the "Purchaser"), hereby certify, as such Secretary, that the Agreement and Plan of Merger to which this Certificate is attached, after having been first duly signed on behalf of the Corporation and having been signed on behalf of VSI Holdings, Inc., a corporation organized under the laws of the State of Georgia, was duly adopted pursuant to subsection (f) of Section 251 of Title 8 of the Delaware Code without any vote of the stockholders of the Purchaser; and that (i) the Agreement and Plan of Merger does not amend in any respect the Certificate of Incorporation of the Purchaser, (ii) each share of stock of the Purchaser outstanding immediately prior to the effective date of the merger is to be an identical outstanding share of the Purchaser after the effective date of the merger and (iii) the authorized, unissued shares of common stock of the Purchaser to be issued under the Agreement and Plan of Merger plus those initially issuable upon conversion of any other shares, securities or obligations to be issued or delivered under such plan do not exceed 20% of the shares of common stock of the Purchaser outstanding immediately prior to the effective date of the merger; and that the Agreement and Plan of Merger was adopted by action of the Board of Directors of the Purchaser and is the duly adopted agreement and act of the Purchaser.

Witness my hand on this 24th day of March, 2001.

/s/ Christopher J. Kearney Christopher J. Kearney, Secretary

A-39

INDEMNITY AND RESTRICTION AGREEMENT

This Indemnity and Restriction Agreement (the "Agreement") is made and entered into as of March 24, 2001 by each of Steve Toth, Jr., an individual and resident of the State of Michigan ("Toth"), CLT Associates, L.P., a Michigan partnership ("CLT"), Toth as trustee under Trust Agreement dated December 20, 1976 f/b/o Toth (the "Toth Trust I"), Toth as trustee under Trust Agreement dated July 9, 1983 f/b/o Toth (the "Toth Trust II"), Margaret Ann Toth ("Mrs. Toth") as trustee under Trust Agreement dated September 1, 1976 f/b/o Margaret Joan Toth ("Child Trust") and, Mrs. Toth as trustee under Trust Agreement dated July 9, 1982 f/b/o Mrs. Toth ("Spouse Trust", and together with each of Toth, CLT, Toth Trust I, Toth Trust II and Child Trust, individually, a "Holder," and collectively, the "Holders") to and for the benefit of SPX Corporation, a Delaware corporation ("Purchaser"). Terms used herein but not defined herein shall have the meanings set forth in the Merger Agreement (defined below).

WITNESSETH:

WHEREAS, The Holders own, in the aggregate, 27,905,755 shares of common stock, \$0.01 par value per share, of VSI Holdings, Inc., a Georgia corporation (the "Company");

WHEREAS, Toth, through CLT, Toth Trust I and Toth Trust II, is deemed to have the sole right to vote and dispose of 14,281,909 shares of common stock, \$0.01 par value per share, of the Company;

WHEREAS, Mrs. Toth is deemed to control, and Toth is deemed to share, through Child Trust and Spouse Trust, voting and dispositive powers, of 12,722,996 shares of common stock, \$0.01 par value per share, of the Company;

WHEREAS, pursuant to the terms of that certain Agreement and Plan of Merger by and among the Company and Purchaser dated as of the date hereof (the "Merger Agreement"), the Company shall be merged with and into Purchaser (the "Merger");

WHEREAS, pursuant to the terms of the Merger Agreement, each of the Holders shall be issued shares of Purchaser Stock in exchange for their respective shares of the Company, in the amounts as determined as of the Effective Time;

WHEREAS, as an inducement to Purchaser to consummate the Merger, and as a condition to such merger, the Holders are entering into this Agreement.

NOW THEREFORE, in consideration of the foregoing and the mutual covenants and agreements set forth below, the parties hereto agree as follows:

1. INDEMNIFICATION.

(a) Effective only if the Effective Time shall occur, Toth will pay or reimburse each of Purchaser and its respective officers, directors, employees, agents, successors and assigns ("Purchaser Group") from and against and in respect of any Reimbursable Expenses, related to or arising out of any matter set forth on Schedule 1 attached hereto (a "Liability"); provided, however, that any such Reimbursable Expense shall be paid and reimbursed solely out of the assets held by the Escrow Agent pursuant to the Stock Pledge and Escrow Agreement (the "Escrow Fund"); and provided further that Toth shall, except for his obligations hereunder, have no personal liability or obligation of any kind in respect of the Reimbursable Expenses and the Purchaser Group's sole recourse and remedy in respect of the Reimbursable Expenses shall be to the Escrow Fund. For purposes hereof, "Reimbursable Expenses" shall mean and include only the amount of any damages (including incidental and consequential damages), losses, deficiencies, liabilities, costs, claims, penalties and expenses (including attorneys' and accountants' fees and the costs of investigation and defense and of enforcing or preserving their respective rights hereunder or with respect to a Liability).

- (b) Notwithstanding the provisions of Section 1(a) to the contrary, the aggregate amount of any Reimbursable Expenses that may be claimed by the Purchaser Group pursuant to the terms of Section 1(a) shall be limited to \$6,500,000 (the "Limit"); provided, however, that the foregoing limitation shall be reduced on the first anniversary of the Effective Date so that any remaining responsibilities of Toth is no greater than (the "Lower Limit"): (a) \$3,250,000 or (b) the aggregate amount claimed against the Purchaser Group with respect to any Liabilities then being contested plus the amount of any claims by the Purchaser Group for Reimbursable Expenses that may then be pending against the Escrow Fund (the "First Anniversary Pending Claims").
- (c) In the event that any claim or demand not pending on the date hereof and related to a Liability is asserted against or sought to be collected from the Purchaser Group by a third party (a "New Claim"), the Purchaser Group shall promptly notify Toth of the New Claim, specifying the nature of such claim or demand and the amount or the estimated amount thereof to the extent then feasible (which estimate shall not be conclusive of the final amount of such claim and demand) (the "Claim Notice"); provided that the failure to provide such Claim Notice will not affect the payment and reimbursement obligations of Toth unless Toth is prejudiced thereby. Toth's payment and reimbursement obligation as to any New Claim shall expire on the second anniversary of this Agreement; provided, however, that the expiration of such obligation shall have no effect on Toth's continuing obligation to pay or reimburse for Reimbursable Expenses relating to claims pending as of such second anniversary date which shall continue to be subject to the reimbursement or payment obligations set forth in this Agreement.
- (d) In determining the amount of any Reimbursable Expense which Purchaser Group is entitled to reimbursement or payment pursuant to Section 1(a), there shall be subtracted an amount equal to (i) any net tax benefit (federal, State, local or foreign) realized by the Purchaser Group in connection with or by reason of such Reimbursable Expense, and (ii) all insurance proceeds actually received by the Purchaser Group by reason of such Reimbursable Expense (net of any retrospective or prospective insurance premium adjustments related to such claim charged to the Purchaser Group).
- (e) Payment of Reimbursable Expenses shall be due and payable on the date Purchaser Group incurs such Reimbursable Expense. Toth shall pay to the Purchaser, within thirty (30) days after delivery of the Claim Notice, the amount of any claim for payment or reimbursement made pursuant to a Claim Notice.
- (f) Purchaser shall have the right to control the conduct of any litigation, action, suit, claim, proceeding or investigation related to a Liability; provided, however, no such claim shall be settled, adjusted or compromised, or the defense thereof terminated, without the prior consent of Toth, which consent shall not be unreasonably withheld. The Purchaser Group shall defend any claim, demand, action, lawsuit or proceeding in a commercially reasonable manner.
- (g) Toth shall pay, or reimburse the Purchaser Group, for any and all legal, consulting and investment banking fees, expenses and other fees incurred by the Company (including, without limitation, the fees of Deloitte & Touche and McDonald Investments) related to the negotiation of the Merger Agreement and the consummation of the Merger in excess of \$1,500,000.
- 2. ESCROW OF PURCHASER STOCK. In order to secure the payment of Toth's obligations under Section 1, Toth shall, on the Effective Date, pledge and deposit that number of shares of Purchaser Stock (the "Pledged Stock") equal to the Limit (based upon a per share price of the Purchaser Stock equal to \$101.17) with an escrow agent to be selected by Purchaser that is reasonably acceptable to Toth (the "Escrow Agent"), to be governed in accordance with the Stock Pledge and Escrow Agreement substantially in the form attached hereto as Exhibit A ("Stock Pledge and Escrow Agreement"). The Escrow Agent shall, on the first anniversary of the Effective Date (or the first business day thereafter if such date is not a business day), tabulate the First Anniversary Pending Claims and, if after subtracting the number of shares representing the value of the First Year Claims (based upon a per share price (the "Applicable Price") of the Purchaser Stock equal to \$101.17) from the Pledged Stock (valued at the

Applicable Price), the amount remains is greater than the Lower Limit, the Escrow Agent shall release that amount of the remaining Pledged Stock as exceeds the Lower Limit. The Pledged Stock held under the Escrow Agreement shall be the sole and exclusive source of repayment and/or reimbursement of Reimbursable Expenses.

3. RESTRICTION ON PURCHASER STOCK.

- (a) The Holders hereby agree that, for one (1) year from the Effective Date, they shall not sell, transfer or dispose of twenty-five percent (25%) or more of any Purchaser Stock received by each of them in the aggregate, during any ninety (90) day period.
- (b) Purchaser agrees that at all times during the one (1) year period after the Effective Date it will meet the information and reporting requirements of Rule 144(c) promulgated under the Securities Act of 1933, as amended.
- (c) Notwithstanding the foregoing limitations, any Holder may sell, transfer or dispose of any shares of Purchaser Stock to any other person or entity in any transaction other than on the open market; provided that the transferee agrees to be bound by the restrictions in Section 3(a) as to the transferred shares of Purchaser Stock.
- (d) The foregoing limitations shall automatically expire without notice or other action upon the first to occur of any of the following: (i) Purchaser shall effect or announce any intention to effect any "Rule 13e-3 transaction" (as defined in Rule 13e-3 promulgated under the Securities Exchange Act of 1934, as amended), (ii) Purchaser shall enter into any letter of intent or definitive agreement with respect to any business combination or similar transaction in which the Purchaser shall not be the surviving corporation, (iii) Purchaser shall effect or announce any intention to effect a dissolution and liquidation of Purchaser, or (iv) any person or entity shall commence any tender offer, exchange offer or comparable transaction for all or any portion of the outstanding shares of Purchaser Stock.

4. NONCOMPETITION AND NONSOLICITATION.

- (a) Toth agrees that for a period of five (5) years from the date hereof he will not, without the express written authorization of Purchaser, enter employment with, or directly or indirectly own an interest in, or manage, operate, control, or participate in the business of, any organization doing business anywhere in the world, whose business is in competition with that of the Purchaser or its respective subsidiaries or affiliates including without limitation the business of the Company as conducted immediately prior to the Merger; provided, however, that Toth shall be permitted to conduct the business of Advanced Animations, Inc. as it is conducted as of the date hereof, and to conduct the business associated with assets and liabilities acquired pursuant to that certain Asset Purchase Agreement by and among Purchaser, PII Ventures, L.L.C. and Toth dated as of the date hereof, as such business is conducted as of the date hereof. It is understood that while such authorization may be withheld for any reason in the sole discretion of Purchaser, such authorization shall not be unreasonably withheld.
- (b) Toth further agrees that for a period of five (5) years from the date hereof he will not, without the express written authorization of Purchaser, directly or indirectly, (i) solicit, divert or attempt to divert from Purchaser or its subsidiaries or affiliates, any customer or prospective customer of the Purchaser or its subsidiaries or affiliates (including without limitation any customer or prospective customer of the Company immediately prior to the Merger); or (ii) solicit, divert or attempt to, divert from the Purchaser or its subsidiaries or affiliates, or attempt to hire or hire on his behalf on or behalf of any other person or entity whatsoever, any employee or prospective employee of the Purchaser or its subsidiaries or affiliates who was an employee or prospective employee as of the time of the expiration of his obligations hereunder or was engaged by the Company, Purchaser or their respective subsidiaries or affiliates within six (6) months of such expiration.

- (c) Toth acknowledges and agrees that the above restrictions on his ability to compete with the Purchaser and its subsidiaries and affiliates, are reasonable in light of the Company's and Purchaser's significant business interests. Toth acknowledges and agrees that he has received sufficient consideration to support his obligations under this Agreement. Toth further acknowledges that his agreement to be bound by this Section 4 constitutes a material inducement to Purchaser to enter into the Merger Agreement and this Agreement.
- 5. BINDING ON SUCCESSORS. This Agreement is binding upon and shall inure to the benefit of the respective successors and permitted assigns of the parties, whether so expressed or not.
- 6. NO ASSIGNMENT. No Party may assign this Agreement or any of the rights or obligations hereunder without the express written consent of the other parties.
- 7. NOTICE. All notices, requests, consents or other communications required pursuant to this Agreement shall be in writing and shall be delivered personally, mailed by certified or registered mail (return receipt requested), or sent by facsimile addressed as follows:

if to the Purchaser:
SPX Corporation
28635 Mound Road
Warren, Michigan 48092-3499
Fax: (810) 578-7470
Attention: Fabrizio F. Rasetti, Esq.
with a copy to:
Gardner, Carton & Douglas
321 North Clark Street
Suite 3400
Chicago, Illinois 60610
Fax: (312) 644-3381
Attention: Stephen A. Tsoris

if to any Holder, at the address of such Holder shown in the stock transfer records of Purchaser, or, to such other address as may be given pursuant to this Section 7.

- 8. GOVERNING LAW. This Agreement shall be governed and construed in accordance with the laws of the State of Michigan (irrespective of its choice of law principles).
- 9. AMENDMENT. This Agreement may not be amended without the written consent of the Purchaser and each of the Holders.
- 10. TERMINATION. This Agreement shall remain in full force and effect until all obligations for Reimbursable Expenses made by the Purchaser Group or any member thereof are paid in full or the Pledged Stock is exhausted. In the event that the Merger is not consummated, this Agreement shall become null and void and of no further force or effect, and no Holder (or any of their respective affiliates, directors, officers, agents or representatives) shall have any liability or obligation hereunder.
- 11. UNENFORCEABLE PROVISION. If any provision of this Agreement is held to be illegal, invalid, or unenforceable, such provision shall be severed from this Agreement and shall not in any manner affect or render illegal, invalid or unenforceable any other provision of this Agreement.
- 12. COUNTERPARTS. This Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original and all of which together shall constitute one and the same instrument.
- 13. GUARANTY OF PAYMENT. All payments made by Toth hereunder shall be deemed to be a guaranty of payment as opposed to a guaranty of collections.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, the parties have executed this Indemnity and Restriction Agreement as of the date first written above.

> /s/ Steve Toth, Jr. Steve Toth, Jr. CLT Associates, L.P.

By: /s/ Steve Toth, Jr. Name: Steve Toth, Jr. Its: General Partner Trust established under Trust Agreement dated December 20, 1976 f/b/o Steve Toth, Jr.

By: /s/ Steve Toth, Jr. Name: Steve Toth, Jr. Its: Trustee Trust established under Trust Agreement dated July 9, 1983 f/b/o Steve Toth,

By: /s/ Steve Toth, Jr. Name: Steve Toth, Jr. Its: Trustee Trust established under Trust Agreement dated September 1, 1976 f/b/o Margaret Joan Toth

By: /s/ Margaret Ann Toth Name: Margaret Ann Toth Its: Trustee Trust established under Trust Agreement dated July 9, 1982 f/b/o Margaret Ann Toth

By: /s/ Margaret Ann Toth Name: Margaret Ann Toth Its: Trustee

Accepted and agreed to as of the date set forth above: SPX CORPORATION

By: /s/ Thomas J. Riordan Name: Thomas J. Riordan Title: Vice President

APPENDIX C

[LETTERHEAD OF MCDONALD INVESTMENTS]

MARCH 30, 2001

PERSONAL AND CONFIDENTIAL

Board of Directors VSI Holdings, Inc. 41000 Woodward Avenue Bloomfield Hills, MI 48304

Ladies and Gentlemen:

You have requested our opinion as to the fairness, from a financial point of view, to the shareholders of VSI Holdings, Inc. (the "Company") of the consideration to be received by the holders of the issued and outstanding shares of the Company's Common Stock, \$0.01 par value (the "Common Stock"), pursuant to the Agreement and Plan of Merger dated as of March 24, 2001 (the "Agreement") by and between the Company and SPX Corporation ("SPX").

You have advised us that, under the terms of the Agreement, and subject to the terms and conditions set forth therein, at the effective time, the Company will be merged with SPX (the "Merger") and the separate corporate existence of the Company will cease. Holders of the Company's Common Stock issued and outstanding immediately prior to the effective time of the Merger will have the right to receive, at the election of each holder, either (i) .043 of a share of SPX's common stock per share of the Company's Common Stock held by that holder (the "Purchaser Stock Consideration") or (ii) a combination of (a) the Purchaser Stock Consideration for 55% of such holder's shares of the Company's Common Stock and (b) \$4.35 in cash, without any interest thereon, per share of the Company's Common Stock (the "Cash/Stock Consideration"). We refer to the Purchaser Stock Consideration and the Cash/Stock Consideration as the "Merger Consideration." The Agreement further provides that the aggregate number of shares of the Company's Common Stock to be converted into cash may not exceed 45% of the Company's Common Stock.

McDonald Investments Inc., as part of its investment banking business, is customarily engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, negotiated underwritings, competitive biddings, secondary distributions of listed and unlisted securities, private placements and valuations for estate, corporate and other purposes.

In connection with rendering this opinion, we have reviewed and analyzed, among other things, the following: (i) the Agreement, including the exhibits and schedules thereto; (ii) the Asset Purchase Agreement dated March 24, 2001 by and among SPX Corporation, PII Ventures, L.L.C. and Steve Toth, Jr. (the "Asset Purchase Agreement"); (iii) the Agreement and Plan of Merger dated March 24, 2001 by and between PII Ventures L.L.C. and Advanced Animations, Inc. ("Advanced Animations Merger Agreement"); (iv) certain publicly available information concerning the Company, including the Annual Reports on Form 10-K of the Company for each of the years in the four year period ended September 30, 2000 and the Quarterly Reports on Form 10-Q of the Company for the quarters ended December 31, 1999, March 31, 2000, June 30, 2000, and December 31, 2000; (v) certain other internal information, primarily financial in nature, including projections, concerning the business and operations of the Company furnished to us by the Company for purposes of our analysis; (vi) certain publicly available information concerning the trading of, and the trading market for, the Company's Common Stock and SPX's common stock; (vii) certain publicly available information concerning SPX, including the Annual Reports on Form 10-K of the Company for each of the years in the four year period ended December 31, 2000 and the Quarterly Reports on Form 10-Q of SPX for the quarters ended March 31, 2000, June 30, 2000, and September 30, 2000; (viii) certain publicly available information with respect to certain other companies that we believe to be comparable to the Company or to SPX and the trading markets for

certain of such other companies' securities; and (ix) certain publicly available information concerning the nature and terms of certain other transactions that we consider relevant to our inquiry. We have also met with certain officers and employees of the Company to discuss the business and prospects of the Company, as well as other matters we believe relevant to our inquiry.

In our review and analysis and in arriving at our opinion, we have assumed and relied upon the accuracy and completeness of all of the financial and other information provided us or publicly available and have assumed and relied upon the representations and warranties of the Company and SPX contained in the Agreement. We have not been engaged to, and have not independently attempted to, verify any of such information. We have also relied upon the management of the Company as to the reasonableness and achievability of the financial and operating projections (and the assumptions and bases therefor) provided to us and, with your consent, we have assumed that such projections reflect the best currently available estimates and judgments of such respective management of the Company. We have not been engaged to assess the reasonableness or achievability of such projections or the assumptions on which they were based and express no view as to such projections or assumptions. In addition, we have not conducted a physical inspection or appraisal of any of the assets, properties or facilities of either the Company or SPX nor have we been furnished with any such evaluation or appraisal. We have also assumed that the conditions to the Merger as set forth in the Agreement would be satisfied and that the Merger would be consummated on a timely basis in the manner contemplated by the Agreement. We have had no independent discussions with SPX's management and have not had access to any non-public information about SPX.

It should be noted that this opinion is based on economic and market conditions and other circumstances existing on, and information made available as of, the date hereof and does not address any matters subsequent to such date including the value of SPX's common stock at the time of issuance thereof to the holders of the Company's Common Stock. In addition, our opinion is, in any event, limited to the fairness, as of the date hereof, from a financial point of view, to the Company's stockholders, of the Merger Consideration to be received pursuant to the Merger, and does not address the Company's underlying business decision to effect the Merger or any other terms of the Merger or any of the other transactions contemplated thereby (including the Asset Purchase Agreement and the Advanced Animations Merger Agreement). It should be noted that although subsequent developments may affect this opinion, we do not have any obligation to update, revise or reaffirm our opinion.

In the ordinary course of our business, we may actively trade securities of both the Company and SPX for our own account and for the accounts of customers and, accordingly, may at any time hold a long or short position in such securities.

It is understood that this opinion is directed to the Board of Directors and senior management of the Company and may not be disclosed, summarized, excerpted from or otherwise publicly referred to without our prior written consent. Our opinion does not constitute a recommendation to any stockholder of the Company as to how such stockholder should vote at the stockholders' meeting held in connection with the Merger.

We have received a fee for our services in rendering this Opinion, and the Company has agreed to indemnify us against certain liabilities, including liabilities arising under the federal securities laws.

Based upon and subject to the foregoing and such other matters as we consider relevant, it is our opinion that as of the date hereof, the Merger Consideration is fair, from a financial point of view, to the stockholders of the Company.

Very truly yours, /s/ McDonald Investments Inc.

McDONALD INVESTMENTS INC.

PART II INFORMATION NOTE REQUIRED IN PROSPECTUS

ITEM 20. INDEMNIFICATION OF DIRECTORS AND OFFICERS.

Section 145 of the General Corporation Law of the State of Delaware authorizes and empowers SPX to indemnify the directors, officers, employees and agents of SPX against liabilities incurred in connection with, and related expenses resulting from, any claim, action or suit brought against any such person as a result of his or her relationship with SPX, provided that those persons acted in good faith and in a manner that person reasonably believed to be in, and not opposed to, the best interests of SPX in connection with the acts or events on which the claim, action or suit is based. The finding of either civil or criminal liability on the part of those persons in connection with the acts or events is not necessarily determinative of the question of whether those persons have met the required standard of conduct and are, accordingly, entitled to be indemnified. The foregoing statements are subject to the detailed provisions of Section 145 of the General Corporation Law of the State of Delaware.

SPX's Certificate of Incorporation provides that directors and officers of SPX and those serving at the request of SPX as a director, officer, employee or agent of another corporation or entity will be indemnified by SPX to the fullest extent authorized by Delaware law. The indemnification right includes the right to be paid by SPX the expenses incurred in defending any proceeding in advance of its final disposition. The indemnification rights conferred by SPX's Certificate of Incorporation are not exclusive of any other right to which persons seeking indemnification may be entitled under any law, bylaw, agreement, vote of stockholders or disinterested directors or otherwise. SPX is authorized to purchase and maintain insurance on behalf of its directors and officers.

In the merger agreement, SPX has agreed that all rights to exculpation and indemnification for acts or omissions occurring at or prior to the effective time of the merger now existing in favor of the current or former directors or officers of VSI Holdings or any of its subsidiaries as provided in VSI Holdings' Articles of Incorporation or By-Laws or in any agreement will survive the merger and continue in full force and effect. SPX will indemnify those persons to the fullest extent permitted by applicable law and will use its reasonable best efforts to obtain and maintain in effect, or cause the surviving corporation to obtain and maintain in effect, VSI Holdings' current directors' and officers' liability insurance covering those persons who are currently covered by VSI Holdings' insurance policy, provided that SPX will not be required to pay an annual premium for that insurance in excess of 300% of the annual premiums currently paid by VSI Holdings, but in that case must purchase as much coverage as possible for that amount.

ITEM 21. EXHIBITS AND FINANCIAL STATEMENT SCHEDULES

(a) Exhibits.

The Exhibits to this registration statement are listed in the Index to Exhibits.

(b) Financial Statement Schedules

None.

ITEM 22. UNDERTAKINGS

The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the registration statement will be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time will be deemed to be the initial bona fide offering thereof.

The undersigned registrant hereby undertakes as follows: that prior to any public reoffering of the securities registered hereunder through use of a prospectus which is part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.

The registrant undertakes that every prospectus (i) that is filed pursuant to the paragraph immediately preceding, or (ii) that purports to meet the requirements of section 10(a)(3) of the Act and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment will be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time will be deemed to be the initial bona fide offering thereof.

Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

The undersigned registrant hereby undertakes to respond to requests for information that is incorporated by reference into the prospectus pursuant to Items 4, 10(b), 11 or 13 of this Form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in the documents filed subsequent to the effective date of the registration statement through the date of responding to the request.

The undersigned registrant hereby undertakes to supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the Registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Muskegon and State of Michigan on the 16th day of April, 2001.

SPX CORPORATION

By: /s/ PATRICK J. O'LEARY

Patrick J. O'Leary Vice President, Finance, Treasurer and Chief

Financial and Accounting Officer

POWER OF ATTORNEY

KNOW ALL PERSONS BY THESE PRESENTS, that each person whose signature appears below constitutes and appoints John B. Blystone, Christopher J. Kearney or Patrick J. O'Leary, and each of them, his true and lawful attorneys-in-fact and agents, with full power of substitution and resubstitution for him and in his name, place and stead, in any and all capacities, to sign, execute and file this Registration Statement and any amendments (including, without limitation, post-effective amendments) to this Registration Statement, and to file the same, with all exhibits thereto and all documents required to be filed with respect therewith, with the Securities and Exchange Commission or any regulatory authority, granting unto such attorneys-in-fact and agents full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection therewith and about the premises in order to effectuate the same as fully to all intents and purposes as he might or could do if personally present, hereby ratifying and confirming all that such attorneys-in-fact and agents or his or their substitute or substitutes may lawfully do or cause to be done.

Pursuant to the requirements of the Securities Act of 1933, as amended, this Registration Statement has been signed by the following persons in the capacities indicated on this 16th day of April, 2001.

/s/ JOHN B. BLYSTONE John B. Blystone	Chairman, President and Chief Executive Officer Director
/s/ PATRICK J. O'LEARY	Vice President, Finance, Treasurer and Chief
Patrick J. O'Leary	Financial Officer and Accounting Officer
/s/ J. KERMIT CAMPBELL	Director
J. Kermit Campbell	
/s/ SARAH R. COFFIN	Director
Sarah R. Coffin	
/s/ FRANK A. EHMANN	Director
Frank A. Ehmann	

/s/ EMERSON U. FULLWOOD	Director
Emerson U. Fullwood	
	Director
Charles E. Johnson II	- -
/s/ DAVID P. WILLIAMS	Director
David P. Williams	· ·

II-4

INDEX TO EXHIBITS

Agreement and Plan of Merger by and between SPX Corporation and VSI Holdings, Inc. dated as of March 24, 2001, together with a list of exhibits and schedules thereto (included as Appendix A to this proxy statement-prospectus).* 2.2 Indemnity and Restriction Agreement by and among SPX Corporation, Steve Toth, Jr., and the other parties thereto, dated as of March 24, 2001, together with a list of exhibits and schedules thereto (included as Appendix B to this proxy statement-prospectus).* 2.3 Asset Purchase Agreement by and among PII Ventures, L.L.C., Steve Toth, Jr. and SPX Corporation dated as of March 24, 2001, together with a list of exhibits and schedules thereto. 2.4 Form of Agreement and Plan of Merger by and between Advanced Animations, Inc. and PII Ventures, L.L.C. 4.1 Copies of the instruments with respect to the company's long-tern debt are available from the Securities and Exchange Commission upon request. 4.2 Rights Agreement, dated as of June 25, 1996 between the company and the Bank of New York, as Rights Agent, relating to Rights Agreement, by reference from the company's Registration Statement on Form 8-A filed on June 26, 1996. 4.3 Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon - Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Registration Rights Agreement, Depth of Peruary 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Fo	EXHIBIT NO.	DESCRIPTION
Appendix A to this proxy statement-prospectus).* Indemnity and Restriction Agreement by and among SPX Corporation, Steve Toth, Jr., and the other parties thereto, dated as of March 24, 2001, together with a list of exhibits and schedules thereto (included as Appendix B to this proxy statement-prospectus).* 2.3 Asset Purchase Agreement by and among PII Ventures, L.L.C., Steve Toth, Jr. and SPX Corporation dated as of March 24, 2001, together with a list of exhibits and schedules thereto.* 2.4 Form of Agreement and Plan of Merger by and between Advanced Animations, Inc. and PII Ventures, L.L.C. 4.1 Copies of the instruments with respect to the company's long-term debt are available from the Securities and Exchange Commission upon request. 4.2 Rights Agreement, dated as of June 25, 1996 between the company and The Bank of New York, as Rights Agent, relating to Rights to purchase preferred stock under certain circumstances, incorporated herein by reference from the company's Registration Statement on Form 8-A filed on June 26, 1996. 4.3 Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. ** 1 Tax Opi	2.1	and VSI Holdings, Inc. dated as of March 24, 2001, together
Asset Purchase Agreement by and among PII Ventures, L.L.C., Steve Toth, Jr. and SPX Corporation dated as of March 24, 2001, together with a list of exhibits and schedules thereto.* 2.4 Form of Agreement and Plan of Merger by and between Advanced Animations, Inc. and PII Ventures, L.L.C. 4.1 Copies of the instruments with respect to the company's long-term debt are available from the Securities and Exchange Commission upon request. 4.2 Rights Agreement, dated as of June 25, 1996 between the company and The Bank of New York, as Rights Agent, relating to Rights to purchase preferred stock under certain circumstances, incorporated herein by reference from the company's Registration Statement on Form 8-A filed on June 26, 1996. 4.3 Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX Corporation and The Bank of New York, 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon - Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.2 Tax Opinion of Gardner, Carton & Douglas. 8.3 Tax Opinion of Gardner, Carton & Douglas. 8.4 Tax Opinion of Gardner, Carton & Douglas. 8.5 Form 10-0 OF VSI Holdings, Inc. for the quarter ended December 31, 2000. 8.6 Consent of KPM6 LLP, independent public accountants for VSI Holdings, Inc. 8.7 Consent of KPM6 LLP, independent public accountants for VSI Holdings, Inc. 8.8 Consent of Michael Carton & Douglas (included in Exhibit	2.2	Appendix A to this proxy statement-prospectus).* Indemnity and Restriction Agreement by and among SPX Corporation, Steve Toth, Jr., and the other parties thereto, dated as of March 24, 2001, together with a list of exhibits and schedules thereto (included as Appendix B to this proxy
2.4 Form of Agreement and Plan of Merger by and between Advanced Animations, Inc. and PII Ventures, L.L.C. 4.1 Copies of the instruments with respect to the company's long-term debt are available from the Securities and Exchange Commission upon request. 4.2 Rights Agreement, dated as of June 25, 1996 between the company and The Bank of New York, as Rights Agent, relating to Rights to purchase preferred stock under certain circumstances, incorporated herein by reference from the company's Registration Statement on Form 8-A filed on June 26, 1996. 4.3 Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** Tax Opinion of Gardner, Carton & Douglas. 8.2 Tax Opinion of Gardner, Carton & Douglas. 8.3 Form 10-Q of VSI Holdings, Inc. for the year ended September 30, 2000. 8.3 Consent of FMPG LLP, independent public accountants for SPX Corporation. 8.3 Consent of FMPG LLP, independent public accountants for United Dominion Industries Limited. 8.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 8.5 Consent of Miller, Canfield, Pa	2.3	Asset Purchase Agreement by and among PII Ventures, L.L.C., Steve Toth, Jr. and SPX Corporation dated as of March 24, 2001, together with a list of exhibits and schedules
4.1 Copies of the instruments with respect to the company's long-term debt are available from the Securities and Exchange Commission upon request. 4.2 Rights Agreement, dated as of June 25, 1996 between the company and The Bank of New York, as Rights Agent, relating to Rights to purchase preferred stock under certain circumstances, incorporated herein by reference from the company's Registration Statement on Form 8-A filed on June 26, 1996. 4.3 Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Gardner, Carton & Douglas.** 13.1 Form 10-K of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.2 Consent of Flante & Moran, LlP, independent public accountants for SPX Corporation. 23.3 Consent of Flante & Moran, LlP, independent public accountants for VSI Holdings, Inc. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of McDonald Investments Inc. 24.6 Powers of Attorney (included on signature page). 25.7 VSI Holdings Proxy Card. 26.9 Powers of Attorney (included on signature page). 27.9 Powers of Attorney (included on sig	2.4	Form of Agreement and Plan of Merger by and between Advanced
4.2 Rights Agreement, dated as of June 25, 1996 between the company and The Bank of New York, as Rights Agent, relating to Rights to purchase preferred stock under certain circumstances, incorporated herein by reference from the company's Registration Statement on Form 8-A filed on June 26, 1996. 4.3 Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** 7ax Opinion of Gardner, Carton & Douglas.** 13.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 23.2 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.3 Consent of FMELP, independent public accountants for VSI Holdings, Inc. 23.4 Consent of FMPM LLP, independent public accountants for SPX Corporation. 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 25.5 VSI Holdings Proxy Card. 26.6 Form of Election and Letter of Transmittal. 27.6 Opinion of McDonald Investments, financial advisor to V	4.1	Copies of the instruments with respect to the company's long-term debt are available from the Securities and
Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's Registration Statement on Form 8-A filed on January 9, 1998. 4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** Tax Opinion of Gardner, Carton & Douglas.** 13.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 13.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Flante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for SPX Corporation. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 25.2 Form of Election and Letter of Transmittal. 26.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	4.2	Rights Agreement, dated as of June 25, 1996 between the company and The Bank of New York, as Rights Agent, relating to Rights to purchase preferred stock under certain circumstances, incorporated herein by reference from the company's Registration Statement on Form 8-A filed on June
4.4 Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** 8.2 Tax Opinion of Gardner, Carton & Douglas. 7.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 3.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of FMG LLP, independent public accountants for VSI Holdings, Inc. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	4.3	Amendment No. 1 to Rights Agreement, effective October 22, 1997, between SPX Corporation and The Bank of New York, incorporated herein by reference from the Company's
4.5 Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 4.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** 8.2 Tax Opinion of Gardner, Carton & Douglas. ** 13.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 13.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consent of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibits 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	4.4	Indenture between SPX and The Chase Manhattan Bank, dated as of February 6, 2001, incorporated herein by reference from the company's Registration Statement on Form S-3 (No.
A.6 Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364) filed on February 28, 2001. 5.1 Legal Opinion of Gardner, Carton & Douglas. 8.1 Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** 8.2 Tax Opinion of Gardner, Carton & Douglas.** 13.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 13.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	4.5	Form of Liquid Yield Option Note due 2021 (Zero Coupon Senior) incorporated herein by reference from the company's Registration Statement on Form S-3 (No. 333-56364)
Legal Opinion of Gardner, Carton & Douglas. Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** Tax Opinion of Gardner, Carton & Douglas.** Tax Opinion of Gardner, Carton & Douglas.** Torm 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). Consent of McDonald Investments Inc. Powers of Attorney (included on signature page). VSI Holdings Proxy Card. Porm of Election and Letter of Transmittal. Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	4.6	Registration Rights Agreement, dated as of February 6, 2001, by and between SPX Corporation and Merrill, Lynch, Pierce Fenner & Smith Incorporated incorporated herein by reference from the company's Registration Statement on Form S-3 (No.
8.2 Tax Opinion of Gardner, Carton & Douglas.** 13.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 13.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy		Legal Opinion of Gardner, Carton & Douglas.
13.1 Form 10-K of VSI Holdings, Inc. for the year ended September 30, 2000. 13.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy		Tax Opinion of Miller, Canfield, Paddock and Stone, P.L.C.** Tax Opinion of Gardner, Carton & Douglas **
13.2 Form 10-Q of VSI Holdings, Inc. for the quarter ended December 31, 2000. 23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy		Form 10-K of VSI Holdings, Inc. for the year ended September
23.1 Consent of Arthur Andersen LLP, independent public accountants for SPX Corporation. 23.2 Consent of Plante & Moran, LLP, independent public accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	13.2	Form 10-Q of VSI Holdings, Inc. for the quarter ended
accountants for VSI Holdings, Inc. 23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	23.1	Consent of Arthur Andersen LLP, independent public
23.3 Consent of KPMG LLP, independent public accountants for United Dominion Industries Limited. 23.4 Consents of Gardner, Carton & Douglas (included in Exhibits 5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	23.2	
5.1 and 8.2). 23.5 Consent of Miller, Canfield, Paddock and Stone, P.L.C. (included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	23.3	Consent of KPMG LLP, independent public accountants for
(included in Exhibit 8.1). 23.6 Consent of McDonald Investments Inc. 24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	23.4	5.1 and 8.2).
24.1 Powers of Attorney (included on signature page). 99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy	23.5	
99.1 VSI Holdings Proxy Card. 99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy		
99.2 Form of Election and Letter of Transmittal. 99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy		
99.3 Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy		· · · · · · · · · · · · · · · · · · ·
		Opinion of McDonald Investments, financial advisor to VSI Holdings (included as Appendix C to this proxy

* The exhibits and schedules are not filed, but SPX undertakes to furnish a copy of any exhibit or schedule to the Securities and Exchange Commission upon request.

^{**} To be filed by amendment.

1 Exhibit 2.3

ASSET PURCHASE AGREEMENT
DATED AS OF MARCH 24, 2001

BY AND AMONG

SPX CORPORATION,

PII VENTURES, L.L.C.

AND

STEVE TOTH, JR.

ASSET PURCHASE AGREEMENT

This ASSET PURCHASE AGREEMENT (the "Agreement") is made and entered into as of the 24th day of March, 2001, by and among SPX CORPORATION, a Delaware corporation ("SPX" or "Seller"), PII VENTURES, L.L.C., a Michigan limited liability company, by and through its Managing Member, STEVE TOTH, JR. ("Buyer"), and STEVE TOTH, JR., an individual and resident of the State of Michigan ("Toth").

RECITALS

WHEREAS, pursuant to the terms of that certain Agreement and Plan of Merger by and among VSI Holdings, Inc., a Georgia Corporation ("VSI Holdings") and SPX dated as of the date hereof (the "Overall Merger Agreement"), VSI Holdings shall be merged with and into SPX (the "Overall Merger");

WHEREAS, VSI Holdings, either directly or through its wholly-owned subsidiary, Visual Services, Inc. ("VSI Sub"), currently owns the Acquired Assets (defined below);

WHEREAS, Toth, as significant shareholder of VSI Holdings and as its President and Chief Executive Officer, is familiar with the Acquired Assets and desires to, upon completion of the Overall Merger, acquire from the Seller, by and through the Buyer, the Acquired Assets upon the terms and conditions set forth herein; it being understood as more fully expressed below, that certain assets will be purchased directly by the Buyer from the Seller and, in addition, Advanced Animations, Inc., a Georgia corporation ("Advanced Animations") will merge into the Buyer; and

WHEREAS, Buyer or its Affiliates' execution of an Indemnity and Restriction Agreement among Toth, SPX and the other parties set forth therein dated as of the date hereof (the "Indemnity and Restriction Agreement") and this Agreement was a material inducement to SPX to enter into the Overall Merger Agreement and to agree to consummate the Overall Merger.

NOW, THEREFORE, in consideration of the premises and the representations, warranties, covenants and agreements hereinafter set forth, and intending to be legally bound hereby, the parties hereto agree as follows:

ARTICLE I.

1.1. Definitions. For purposes of this Agreement and any attachments the terms defined in this Agreement when capitalized and used in this Agreement or any attachments shall have the respective meanings specified in this Agreement. In addition, the following terms when capitalized and used in this Agreement or any attachments shall have the meanings set forth below unless the context or use requires another or different meaning:

- (a) the Agency of Record Marketing between The Oz Entertainment Company, a Delaware corporation ("Oz Entertainment") and VSI Sub;
 - (b) all stock of OZ Entertainment;
- (c) all of the Membership Interests in Corporate Eagle V, L.L.C., a Michigan limited liability company;
- (d) all of the partnership interests in K.C. Investors, L.P., a Kansas limited partnership ("K.C. Investors");

- (e) all rights to receive dividends, distributions or other cash payments related to the Acquired Assets, including the pro rata portion of any annual cash payments made after the Closing with respect to any fiscal year including the year of the Closing;
- (f) all current and deferred federal, state and local income (including Michigan Single Business) Tax assets and refunds (including interest) related to the Acquired Assets; and

but excluding therefrom the Excluded Assets (as defined below).

"Advanced Animations" shall mean Advanced Animations, Inc., a Georgia corporation.

"Advanced Animations Merger" shall mean the merger of Advanced Animations into Buyer as described in Section 2.1 hereof and the conveyance, by operation of law, of all of the assets (except Excluded Assets) and liabilities (except Excluded Liabilities) of Advanced Animations to Buyer, including the rights and obligations under that certain Services Agreement dated September 21, 1998 by and between Advanced Animations and Oz Entertainment, the real property owned by Advanced Animations and located in Stockbridge, Vermont (the "Real Estate"), and all of the partnership interests in K.C. Investors owned by Advanced Animations:

"Advanced Animations Merger Agreement" shall mean that certain Agreement and Plan of Merger by and between Buyer and Advanced Animations, in the form attached hereto as Exhibit A.

"Affiliate" shall mean, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person, whether such control is through voting securities, contract or otherwise.

"Agreement" shall have the meaning set forth in the preamble.

"Ancillary Agreements" shall have the meaning set forth in Section 4.2. $\,$

"Assumed Liabilities" means all liabilities and obligations of Advanced Animations that will become the responsibility of the Buyer pursuant to the Advanced Animations Merger described in Section 2.1 hereof, and any other liabilities which relate to, arise or have arisen out of the remaining Acquired Assets at, prior to or after the Effective Time and are not Excluded Liabilities, including:

- (a) The liabilities and obligations under orders, contracts, leases and other commitments existing in Advanced Animations and/or related to the other Acquired Assets, regardless of when such liabilities and obligations arose;
- (b) All liabilities and obligations arising out of Buyer's obligations under this Agreement;
- (c) Any Taxes which are not federal, state, local, or foreign income Taxes; provided, however, Taxes with regard to the Advanced Animations Merger, as described in Section 2.1 hereof, shall be assumed and paid for by Seller; it being understood that the Advanced Animations Merger does not constitute a reorganization within the meaning of Section 368 of the Code but is, instead, treated for tax purposes as a deemed sale of Advanced Animations' assets to Buyer followed by a constructive liquidation, under Section 332 of the Code, of the Cash Consideration (as defined in the Advanced Animations Merger Agreement) into its sole stockholder; and
- (d) All other liabilities and obligations related to the Acquired Assets and Advanced Animations including liabilities and obligations related to a Release, Hazardous Materials and violations of Environmental Laws.

"Authorization" shall mean all franchises, licenses, permits, easements, rights, applications, filings, registrations and other authorizations necessary for the ownership and use of the Acquired Assets.

"Buyer" shall have the meaning set forth in the preamble.

"Buyer Group" shall have the meaning set forth in Section 14.1.

"Claim Notice" shall have the meaning set forth in Section 14.4.

"Closing" shall have the meaning set forth in Section 3.1.

"Closing Date" shall have the meaning set forth in Section 3.1.

"Code" shall mean the United States Internal Revenue Code of 1986, as amended.

"Effective Time" shall have the meaning set forth in Section 3.1.

"Environmental Laws" means the Toxic Substances Control Act, 15 USC 2601 et seq.; Coastal Zone Management Act of 1972, 16 USC 1451 et seq.; Clean Water Act, 33 USC 1251 et seq.; Resource Conversation and Recovery Act of 1976 ("RCRA"), 42 USC 6901 et seq.; Clean Air Act, 42 USC 7401 et seq.; Comprehensive Environmental Response, Compensation and Liability Act ("CERCLA"), 42 USC 9601 et seq.; Hazardous Materials Transportation Act, 49 USC 1801 et seq.; Safe Drinking Water Act, 42 USC 300f et seq.; Emergency Planning and Community Right-to-Know Act of 1986, 42 USC 11001 et seq.; Federal Insecticide Fungicide, and Rodenticide Act, 7 USA 136 et seq.; Occupational Safety and Health Act, 29 USC 651 et seq.; and all other federal, state, county, municipal and local, foreign and other statutes, laws, regulations, orders and ordinances that relate to or deal with the regulation or protection of human health or the environment, or to emissions, discharges, releases of pollutants, contaminants or toxic hazardous wastes into the environment all as currently in effect.

"Excluded Assets", notwithstanding any other provision of this Agreement, means the following rights, properties and assets of Seller as the same shall exist as of the Effective Time:

- (a) All assets of the Seller not specifically included in the Acquired Assets; and $\,$
- (b) All rights of recovery relating to the Excluded Liabilities and Excluded Assets.

"Excluded Liabilities" means the following liabilities and obligations as the same shall exist as of the Effective Time:

- (a) All liabilities and obligations arising out of Seller's obligations under this Agreement;
- (b) All liabilities for current and deferred federal, state, local and foreign income Taxes, including interest, penalties, charges, fees, imposts, duties or other assessments with respect thereto, arising out of the ownership of the Acquired Assets or the operations of Advanced Animations for periods ending on or prior to the Effective Time; including Tax liabilities arising out of the Advanced Animations Merger except for transfer Taxes set forth in Section 3.4;
- (c) All liabilities and obligations related to or arising out of Excluded Assets; and $\,$
- (d) All indebtedness of Advanced Animations to Seller. It is acknowledged that indebtedness of Advanced Animations to VSI Holdings shall be canceled and contributed to the capital of Advanced Animations prior to the Effective Time.

"Governmental Approval" means the required filings, notice to, consent or approval of, any Governmental Authority other than any actions, consents, approvals or filings otherwise expressly referred to in this Agreement.

"Governmental Authority" means any United States federal, state or local or any foreign government, governmental, regulatory or administrative authority, instrumentality, agency or commission or any court, tribunal, judicial or arbitral body.

"Hazardous Material" means any (i) substance which currently is defined as a "hazardous waste," "hazardous substance," pollutant or contaminant under any applicable Environmental Law; or (ii) petroleum, including crude oil or any fraction thereof.

"Liens" shall have the meaning set forth in Section 4.3.

"Overall Merger" shall mean the Merger of VSI Holdings into SPX.

"Notice Period" shall have the meaning set forth in Section 14.4.

"Parties" shall mean collectively Seller and Buyer.

"Person" shall mean any individual, sole proprietorship, partnership, limited liability company, joint venture, trust, unincorporated association, corporation, entity or government (whether federal, foreign, state, county, city or otherwise, including any instrumentality, division, agency or department thereof).

"Promptly" means within five (5) business days.

"Purchase Price" shall have the meaning set forth in Section 3.2.

"Release" means any release, spill, emission, leaking, pumping, injection, deposit, discharge or migration of a Hazardous Material into the environment, including the movement of such materials through or in the air, soil, surface water or ground water.

"Seller" shall have the meaning set forth in the preamble.

"Seller Group" shall have the meaning set forth in Section 14.2.

"Taxes" means any federal, state, local or foreign taxes, assessments, interest, penalties, deficiencies, fees and other governmental charges or impositions (including all income tax, unemployment compensation, social security, payroll, sales and use, excise, privilege, property, ad valorem, franchise, license, school and any other tax or similar governmental charge or imposition under laws of the United States or any state or municipal or political subdivision thereof or any foreign country or political subdivision thereof), customs and duties.

1.2. Interpretations. When reference is made in this Agreement to an Article, Section, Exhibit or Schedule, such reference is to an Article or Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated. The headings contained in this Agreement are for reference purposes only and shall not affect the interpretation of this Agreement. Whenever the words "include", "includes", or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation". Whenever the words "hereof", "herein", "hereunder" and words of similar import are used in this Agreement, they refer to this Agreement as a whole and not to any particular provision of this Agreement. All terms defined in this Agreement shall have the defined meanings given therein when used in any certificate or other document made or delivered pursuant to this Agreement unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to in this Agreement means such agreement, instrument or statute as from time to time amended, modified or supplemented, including in the case of statutes by

succession of comparable successor statutes. All references to statutes in this Agreement shall be deemed to include the applicable regulations promulgated under such statutes. All dollar amounts referred to in the this Agreement are in United States Dollars.

ARTICLE II. MERGER OF ADVANCED ANIMATIONS INTO BUYER (PII VENTURES, L.L.C.) AND ACQUISITION BY BUYER OF OTHER "ACQUIRED ASSETS"

- 2.1 Merger. At the Closing, and on the terms set forth in the Advanced Animations Merger Agreement, the parties shall execute, and shall cause any Affiliates necessary to execute, the Advanced Animations Merger Agreement in substantially the form attached hereto. The Advanced Animations Merger Agreement and Georgia Articles of Merger will be filed with the Georgia Secretary of State pursuant to Section 14-2-1109 of the Georgia Business Corporation Code and the Advanced Animations Merger Agreement and a Michigan Merger Certificate will be filed with the Michigan Department of Consumer and Industry Services pursuant to MCLA Section 450.4705a(3), and upon such filings and their acceptance, Advanced Animations shall be merged into the Buyer. The parties recognize that the laws of the States of Georgia and Michigan allow a Georgia corporation to merge into a Michigan limited liability company; however, the parties further recognize that the Merger is not a tax free merger under the Code. For tax purposes, the Advanced Animations Merger will, instead, be a deemed sale by the Seller to the Buyer of all of Advanced Animations' assets followed by a constructive liquidation of the Cash Consideration into its sole stockholder under Section 332 of the Code. SPX shall assume and pay all Taxes associated with the Advanced Animations Merger.
- 2.2. Purchase and Sale of Remaining Acquired Assets. At the Closing, and on the terms and subject to the conditions of this Agreement, Buyer agrees to purchase from Seller, and Seller agrees to sell, transfer and assign to Buyer, all of the Acquired Assets. To the extent any of the Acquired Assets are held by an Affiliate of Seller, Seller shall cause such Affiliate to sell, transfer and assign such asset to Buyer at Closing. Notwithstanding any other provision of this Agreement, Seller is not selling or transferring to Buyer any rights in or to the Excluded Assets.
- 2.3. Assumption of Liabilities. At the Closing, and on the terms and subject to the conditions of this Agreement, Buyer shall assume the Assumed Liabilities, and Buyer shall pay, perform and discharge the Assumed Liabilities as and when due.
- 2.4. Excluded Liabilities. Buyer shall not assume, or otherwise be responsible for, any Excluded Liabilities. Seller shall retain, and be responsible for, all Excluded Liabilities.

ARTICLE III. CLOSING AND PURCHASE PRICE

3.1. Closing. Subject to the conditions contained in this Agreement, the closing of the purchase and sale of the Acquired Assets, the Advanced Animations Merger, and other transactions contemplated by this Agreement (the "Closing") shall take place at the offices of Miller, Canfield, Paddock and Stone, P.L.C., in Troy, Michigan, at 9:00 a.m., local time, on the first business day following the Closing Date (as defined in the Overall Merger Agreement) of the Merger, or at such other place, date and time as the Parties may agree provided that in all events the Closing shall occur immediately after the Closing Date (as defined in the Overall Merger Agreement) of the Overall Merger. The date on which the Closing occurs is referred to herein as the "Closing Date." The effective time of the Closing shall be 11:58 p.m., local time, on the Closing Date (the "Effective Time"). If for any reason the conditions to Seller's obligations contained in Articles VIII and IX or the conditions to Buyer's obligations contained in Articles VIII and X are not satisfied on or before the scheduled date for the Closing specified above, Seller (in the case of the failure of its conditions to be satisfied) or Buyer (in the case of the failure of its conditions to be satisfied) may, from time to time thereafter, by written notice to the other Party, schedule a later date for the Closing; provided, however, that (i) such Party shall not be entitled to so schedule a later date for the Closing if any failure of a condition it is relying upon to schedule a later date for the Closing is due to a breach by such Party of its obligations under this Agreement, and (ii) no such later scheduled date for the Closing shall affect the right of any Party to terminate this Agreement pursuant to Article XIII. The parties recognize that the Advanced Animations

Merger documentation (including the Advanced Animations Merger Agreement, and the Georgia Articles of Merger and Michigan Merger Certificate filings) shall be filed in the States of Georgia and Michigan on the Closing Date described in this Section 3.1.

- 3.2. Purchase Price; Payment. The purchase price for the Acquired Assets and other obligations of Seller under this Agreement will be \$3,690,000 (the "Purchase Price"), payable on the Closing Date by wire transfer of immediately available funds to such bank account as Seller designates to Buyer in writing on the Closing Date.
- 3.3. Tax Allocation. Buyer and Seller shall allocate the Purchase Price, the aggregate dollar amount of the Assumed Liabilities and the Cash Consideration delivered, and the liabilities assumed under the Advanced Animations Merger Agreement. Buyer and Seller shall file all tax returns and statements, forms and schedules in connection therewith consistent with such allocation and shall take no position contrary thereto unless required by law. Within thirty (30) days after the Closing Date, the parties shall agree to a good faith allocation. If the parties cannot agree to a good faith allocation of the items described in this Section 3.3, then, in such case, the matter shall be subject to binding arbitration pursuant to Section 15.12 hereof. The Seller and Buyer shall file Form 8594, Asset Acquisition Statement, with the Internal Revenue Service, and shall follow the allocation described in this Section 3.3 pursuant to the provisions of Section 1060 of the Code.
- 3.4. Transfer Taxes. Except for Excluded Liabilities, notwithstanding any other provision herein, any documentary and transfer Taxes and any sales, use or other Taxes imposed by reason of the transfers of Acquired Assets provided hereunder and the Advanced Animations Merger as provided under the Advanced Animations Merger Agreement, and any deficiency, interest or penalty asserted with respect thereto shall be borne by Buyer.

ARTICLE IV. [RESERVED]

ARTICLE V. REPRESENTATIONS AND WARRANTIES OF BUYER

Buver represents and warrants to SPX as follows:

- 5.1. Enforceability. This Agreement has been duly executed and delivered by Buyer and constitutes the legal, valid and binding obligation of Buyer enforceable against Buyer in accordance with its terms.
- 5.2 Limited Liability Company Existence. The Buyer is a limited liability company duly organized, validly existing, and in good standing under the laws of the State of Michigan.
- 5.3 Limited Liability Company Authorization; Enforceability. The Buyer has the requisite limited liability company power and authority to execute, deliver and perform its obligations under this Agreement and the transactions and other Agreements and instruments contemplated by this Agreement. The execution, delivery and performance of this Agreement by the Buyer has been duly authorized by membership action of the Buyer. This Agreement has been, and other agreements (a) documents and instruments required or anticipated to be delivered by Buyer in accordance with the provisions hereof (the "Buyer Ancillary Agreements") have been or upon execution thereof will be duly executed and delivered on behalf of Buyer, as appropriate, by duly authorized members of the Buyer and this Agreement constitutes, and the Buyer Ancillary Agreements when executed and delivered, will constitute, the legal, valid and binding obligations of the Buyer enforceable against the Buyer in accordance with their respective terms.
- 5.4. No Conflict. The execution, delivery and performance of this Agreement by Buyer does not and will not violate, conflict with or result in the breach of any term, condition or provision of, or require the consent of any other Party to, (a) any existing law, ordinance, or governmental rule or regulation to which Buyer is subject, (b) any judgment, order, writ, injunction, decree or award of any court, arbitrator or governmental or regulatory official,

body or authority which is applicable to Buyer, (c) the Articles of Organization or Operating Agreement of, or any securities issued by, Buyer, (d) any mortgage, indenture, agreement, contract, commitment, lease, plan or other instrument, document or understanding, oral or written, to which Buyer is a Party or by which Buyer is otherwise bound, or (e) to Buyer's knowledge, any contract or agreement relating to any of the Acquired Assets. Except as aforesaid, and except for merger articles, certificates and agreements that are to be filed in the States of Georgia and Michigan, after the Closing Date, with regard to the Advanced Animations Merger, no authorization, approval or consent of, and no registration or filing with, any Governmental Authority is required in connection with the execution, delivery and performance of this Agreement by Buyer.

5.5. Full Disclosure. Buyer does not know of any breaches of the representations or warranties made by SPX and contained in Article VI of this Agreement as modified by the Schedules hereto, or any other document delivered in connection herewith.

ARTICLE VI. REPRESENTATIONS AND WARRANTIES OF SPX

SPX represents and warrants to Buyer as follows:

- 6.1. Corporate Existence. SPX is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- 6.2. Corporate Authorization; Enforceability. SPX has the requisite corporate power and authority to execute, deliver and perform its obligations under this Agreement and the transactions and other agreements and instruments contemplated by this Agreement. The execution, delivery and performance of this Agreement by SPX has been duly authorized by all necessary corporate action. This Agreement has been, and the other agreements, documents and instruments required or anticipated to be delivered by SPX in accordance with the provisions hereof (the "SPX Ancillary Agreements") have been or upon execution thereof will be, duly executed and delivered on behalf of SPX, as appropriate, by duly authorized officers, and this Agreement constitutes, and the SPX Ancillary Agreements when executed and delivered will constitute, the legal, valid and binding obligations of SPX, enforceable against SPX in accordance with their respective terms.
- 6.3. No Conflict. The execution, delivery and performance of this Agreement and the SPX Ancillary Agreements by SPX does not and will not (i) violate, conflict with, or constitute a default under SPX's Certificate of Incorporation or By-laws, or result in the acceleration of any obligation under, any contract, commitment, license, purchase order, security agreement, mortgage, indenture, note, deed, lien, lease, agreement, instrument, document or understanding (written or oral) or any order, judgment or decree by which SPX is bound or is affected, which could have a material adverse effect on any of the Acquired Assets, (ii) result in the creation or imposition of any Lien upon any of the Acquired Assets, (iii) violate any law, statute, judgment, order, writ, injunction, decree or award of any Governmental Authority which is applicable to SPX, or (iv) constitute an event which, after notice or lapse of time or both, would result in any violation, conflict, default, acceleration, or creation or imposition of any Liens.
- 6.4. Brokers. No broker, finder, investment banker or other third party is entitled to any brokerage, finder's or other fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of SPX.
- 6.5. Acquired Assets. SPX quitclaims, assigns and conveys the Acquired Assets to Buyer hereunder on an "AS IS", "WHERE IS", "WITH ALL FAULTS" basis, and without any warranties, representations or guarantees, either express or implied, of any kind, nature, or type whatsoever, including, but not limited to, any warranty as to the fitness for a particular purpose or merchantability of the personal property. The provisions of this Section 6.5 shall also apply to the assets to be acquired by the Buyer pursuant to the Advanced Animations Merger.

ARTICLE VII. ADDITIONAL COVENANTS

- 7.1. Public Announcements. So long as this Agreement is in effect, Buyer shall not, directly or indirectly, (a) issue or cause the publication of any press release or any other announcement or communication with respect to the transactions contemplated hereby or the Overall Merger, without the written consent of SPX, or (b) discuss with the press or the media this Agreement, the Overall Merger Agreement, the Overall Merger or the transactions contemplated hereby (and will refer any and all questions and inquiries to SPX), except in any case under (a) or (b) where such release or announcement is required by applicable Law or pursuant to any applicable listing agreement with, or rules or regulations of, American Stock Exchange, in which case Buyer, prior to making such announcement, will consult with SPX regarding the same. The parties hereto recognize, however, that the Advanced Animations Merger shall be noticed to the public and published in the Fulton County Daily Report, as is required under OCGA Section 14.2.1105.1, being Georgia Business Corporation Code Section 22-1105.1, entitled "Notice of Merger or Share Exchange."
- 7.2. Consents to Assignment. Buyer shall use reasonable commercial efforts (which shall not include any obligation by Buyer to make any payment) to obtain the consents or approvals (or effective waivers thereof) of assignment from those Persons whose consents or approvals are required for the assignment of Seller's rights under the Acquired Assets.
- 7.3. Insurance. Coverage of the Real Estate under all insurance policies of Seller and its Affiliates shall cease as of the Closing Date, provided, however, that Seller shall not cancel, or cause to be cancelled, coverage of the Real Estate under any insurance policies maintained by, and in the sole name of, Advanced Animations. From and after the Closing Date, Buyer will be responsible for obtaining and maintaining all insurance coverages with respect to the Real Estate. Buyer shall have no rights with respect to any insurance policies of Seller.
- 7.4. Third Party Consents. To the extent that Seller's rights under any agreement, contract, commitment, lease, Authorization or other Acquired Asset to be assigned to Buyer hereunder may not be assigned without the consent of another Person which has not been obtained, this Agreement shall not constitute an agreement to assign the same if an attempted assignment would constitute a breach thereof or be unlawful, and Seller and Buyer shall use reasonable commercial efforts to obtain any such required consent as soon as reasonably possible. If any such consent shall not be obtained or if any attempted assignment would be ineffective or would impair Buyer's rights under the Acquired Asset in question so that Buyer would not in effect acquire the benefit of substantially all such rights, Seller, to the maximum extent permitted by law, shall, if Buyer so requests, cooperate with Buyer in any reasonable arrangement designed to provide such benefits thereunder to Buyer. With respect to, and to the extent of, any transfer, subcontract or assignment of the contracts, agreements or commitments made to Buyer by Seller hereunder, Buyer hereby agrees to assume, perform, discharge when due, and indemnify Seller from and against, all obligations and liabilities of Seller with respect to the applicable underlying contract, agreement or commitment.
- 7.5. Further Assurances. Seller, from time to time after the Closing, at Buyer's request, will execute, acknowledge and deliver to Buyer such other instruments of conveyance and transfer and will take such other actions and execute and deliver such other documents, certifications and further assurances as Buyer may reasonably require in order to vest more effectively in Buyer, or to put Buyer more fully in possession of, any of the Acquired Assets, including the vesting of all assets of Advanced Animations pursuant to the Advanced Animations Merger and the filing of all necessary documentation with the State of Georgia and the State of Michigan, or to better enable Buyer to complete, perform or discharge any of the Assumed Liabilities at the Closing. Each of the Parties will cooperate with the other and execute and deliver to the other Parties such other instruments and documents and take such other actions as may be reasonably requested from time to time by any other Party as necessary to carry out, evidence and confirm the intended purposes of this Agreement.
- 7.6 Approvals for Advanced Animations Merger. The Seller or VSI Sub shall obtain and present to the Buyer at the Closing, all appropriate and required Shareholder and Director approval of the Advanced Animations Merger. Buyer shall obtain and present to Seller at the Closing, all appropriate and required member or limited liability company approval of the Advanced Animations Merger.

ARTICLE VIII. CONDITIONS PRECEDENT TO THE OBLIGATIONS OF EACH PARTY TO CLOSE

The respective obligations of each Party to consummate the transactions contemplated herein shall be subject to the fulfillment or written waiver, to the extent permitted by law, at or prior to the Closing of the following conditions:

- 8.1. Litigation; Proceedings. No litigation or other proceeding by any Governmental Authority or other third Person shall have been commenced that challenges the validity or legality of, or materially restricts the benefits to either Party of, the transactions contemplated hereby. No judgment order, injunction or decree issued by any Governmental Authority or other legal restraint or prohibition preventing the consummation of the Closing shall be in effect. No law, statute, rule, regulation, order, injunction or decree shall have been enacted, entered, promulgated or enforced by any Governmental Authority that prohibits, materially restricts or makes illegal the consummation of the Closing.
- 8.2. Overall Merger Agreement. The merger contemplated by the Overall Merger Agreement shall have become effective.

ARTICLE IX. CONDITIONS PRECEDENT TO OBLIGATIONS OF SELLER TO CLOSE

The obligations of Seller to complete the transactions contemplated thereby shall be subject to the fulfillment or written waiver at or prior to the Closing of the following conditions precedent:

- 9.1. Representations and Warranties. The representations and warranties of Buyer set forth in this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date (as though made on and as of the Closing Date except (i) to the extent such representations and warranties are by their expressed provisions made as of a specified date and (ii) for the effect of transactions contemplated by this Agreement).
- 9.2. Performance of Obligations. Buyer shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date.

ARTICLE X. CONDITIONS PRECEDENT TO OBLIGATIONS OF BUYER TO CLOSE

The obligations of Buyer to complete the transactions contemplated thereby shall be subject to fulfillment or written waiver at or prior to the Closing of the following conditions precedent:

- 10.1. Representations and Warranties. The representations and warranties of SPX set forth in this Agreement shall be true and correct in all material respects as of the date of this Agreement and as of the Closing Date (as though made on and as of the Closing Date except (i) to the extent such representations and warranties are by their expressed provisions made as of a specified date, in which case such representation and warranty shall be true in all material respects as of such specified date, and (ii) for the effect of transactions contemplated by this Agreement).
- 10.2. Performance of Obligations. SPX shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date.

ARTICLE XI.
DELIVERIES AT THE CLOSING BY SELLER

At the Closing, and subject to the terms and conditions herein contained, Seller shall deliver to Buyer the following:

- 11.1. Assignment and Assumption Agreement. The assignment and assumption agreement in the form and substance mutually agreed to by the Parties and their respective counsels (the "Assignment and Assumption Agreement").
- 11.2 Advanced Animations Merger Agreement. All documentation in connection with the Advanced Animations Merger including the Plan and Agreement of Merger, the Georgia Articles of Merger and Michigan Merger Certificate, and any other documents necessary to carry out the filings and purposes of the Advanced Animations Merger.
- 11.3. Stock Certificates. Original stock certificates, if applicable, evidencing ownership of the number of shares of those Acquired Assets (all stock of OZ Entertainment), duly endorsed for transfer in blank or accompanied by duly executed original assignments separate from such certificates (the "Stock Certificates"). Pursuant to the Advanced Animations Merger, any outstanding stock certificates for Advanced Animations, whether held by VSI Sub or SPX or one of its affiliates, shall be delivered to Buyer on the Closing Date.
- 11.4. Instruments of Transfer. Such other bills of sale, deeds, endorsements, consents and instruments of transfer or assignment in the form and substance mutually agreed to by the Parties and approved by their respective counsels as shall be reasonably requested by Buyer to confirm and vest in Buyer all of Seller's and its Affiliates title to the Acquired Assets as acquired pursuant to the Overall Merger Agreement, and the right to assume and perform the Assumed Liabilities.
- 11.5 Assignment of Agency of Record Marketing Services Agreement. An assignment of the Agency of Record Marketing Services Agreement between OZ Entertainment and VSI Sub in the form and substance mutually agreed to by the Parties and their respective counsels.
- 11.6 Assignments of Membership Interests and Partnership Interests. An Assignment and Assumption of Membership Interests assigning to Buyer all of Seller's membership interests in Corporate Eagle V, L.L.C., and an Assignment and Assumption of Partnership Interest assigning to Buyer all of Seller's limited partnership interests in K.C. Investors, in the form and substance mutually agreed to by the Parties and their respective counsels.
- 11.7. Officers' Certificate. A certificate executed by the Vice President or similar executive officer of the Seller or an Affiliate of Seller in the form and substance mutually agreed to by the Parties and approved by their respective counsels.
- 11.8. Secretary's Certificate. A certificate executed by the corporate Secretary of the Seller in the form and substance mutually agreed to by the Parties and approved by their respective counsels.
- 11.9. Other Documents. Such other documents, instruments or assignments as shall be reasonably requested by Buyer and its counsel or required to be delivered by Seller pursuant to this Agreement or in any other instrument or document delivered by any of the Parties pursuant to this Agreement.
- 11.10. Books, records, etc. All of the books, records, papers, files and other data belonging to Seller and used exclusively with respect to the Acquired Assets.

ARTICLE XII. DELIVERIES AT THE CLOSING BY BUYER

At the Closing, and subject to the terms and conditions herein contained, Buyer shall deliver to Seller the following:

12.1. Purchase Price. The Purchase Price in accordance with Section 3.2.

- 12.2. Assignment and Assumption Agreement. The Assignment and Assumption Agreement.
- 12.3. Manager's Certificate. A certificate executed by the President and Chief Executive Officer, Managing Member, or similar executive officer of the Buyer in the form and substance mutually agreed to by the Parties and approved by their respective counsels.
- 12.4. Other Documents. Such other documents, instruments or assignments as shall be reasonably requested by Seller and its counsel or required to be delivered by Buyer pursuant to this Agreement or in any other instrument or document delivered by any of the Parties pursuant to this Agreement.

ARTICLE XIII. TERMINATION AND ABANDONMENT

- 13.1. Events of Termination. This Agreement may be terminated at any time before the Closing: (i) by unanimous consent of Buyer and SPX; (ii) by Seller if there is a breach of or failure by Buyer to perform in any material respect any of the representations, warranties, commitments, covenants or conditions under this Agreement or any of the conditions precedent set forth in Articles VIII and IX of this Agreement, which breach or failure is not cured within ten (10) days after written notice thereof is given to Buyer; or (iii) by Buyer if there is a breach of or failure by Seller to perform in any material respect any of the representations, warranties, commitments, covenants or conditions under this Agreement or any of the conditions precedent set forth in Articles VIII and X of this Agreement, which breach or failure is not cured within ten (10) days after written notice thereof is given to Seller; or (iv) by either Buyer or SPX if the Closing does not occur on or before ten (10) days after the Closing Deadline (as defined in the Overall Merger Agreement), unless the failure of the Closing to occur by such date shall be due to the action or failure to act of the party seeking to terminate this Agreement, which action or failure to act constitutes a breach of this Agreement.
- 13.2. Effect of Termination. In the event of termination and abandonment by any Party as above provided in clauses (ii), (iii), or (iv) of Section 13.1, written notice shall Promptly be given to the other Party, which notice shall clearly specify the condition precedent or breach or failure of such notified party to perform or satisfy its obligations under this Agreement, and this Agreement shall become null and void and of no further force or effect, and no Party (or any of its Affiliates, directors, officers, agents or representatives) shall have any liability or obligation hereunder (except for any liability of any Party then in breach); provided, however, that the provisions of Section 15.2 (Expenses) shall survive any such termination.

ARTICLE XIV. INDEMNIFICATION

- 14.1. Indemnification by Seller. From and after the Closing, Seller will reimburse, indemnify and hold harmless Buyer and its employees, agents, successors and assigns ("Buyer Group") from and against and in respect of any damages, losses, deficiencies, liabilities, costs and expenses (a "Loss") incurred or suffered by any member of the Buyer Group that result from, relate to or arise out of: (i) Seller's failure fully to pay or satisfy any of the Excluded Liabilities in accordance with their terms; (ii) Seller's failure to pay and report the Taxes associated with the Advanced Animations Merger which the Seller recognizes is not a tax free merger under the Code; and (iii) any breach or violation of any representation or warranty or nonfulfillment of any agreement or covenant on the part of Seller under this Agreement.
- 14.2. Indemnification by Buyer. From and after the Closing, Buyer and Toth, jointly and severally, will reimburse, indemnify and hold harmless Seller and its officers, directors, employees, agents, successors or assigns ("Seller Group") from and against and in respect of any Loss incurred or suffered by any member of the Seller Group that result from, relate to or arise out of: (i) Buyer's failure to pay or satisfy any of the Assumed Liabilities; or (ii) any breach or violation of any representation or warranty or non-fulfilment of any agreement or covenant on the part of Buyer under this Agreement.
- 14.3. Survival of Representations and Warranties. All covenants and all representations and warranties made by the Parties in this Agreement or in any Schedule furnished hereunder shall survive the Closing and remain

in effect indefinitely. Notwithstanding any investigation or audit conducted before or after the Closing Date or the decision of any Party to complete the Closing, each Party shall be entitled to rely upon the representations and warranties set forth herein and therein.

- 14.4 Method of Asserting Claims, Etc. In the event that any claim or demand for which Seller would be liable to the Buyer Group is asserted against or sought to be collected from the Buyer Group by a third party, the Buyer Group shall Promptly notify Seller of such claim or demand, specifying the nature of such claim or demand and the amount or the estimated amount thereof to the extent then feasible (which estimate shall not be conclusive of the final amount of such claim and demand) (the "Claim Notice"). Seller shall have twenty days from the personal delivery or mailing of the Claim Notice (the "Notice Period") to notify the Buyer Group, (A) whether Seller disputes its liability to the Buyer Group with respect to such claim or demand and (B) notwithstanding any such dispute, whether it desires, at its sole cost and expense, to defend the Buyer Group against such claim or demand.
 - (a) In the event that Seller notifies the Buyer Group within the Notice Period that it desires to defend the Buyer Group against such claim or demand then, except as hereinafter provided, Seller shall have the right to defend the Buyer Group by appropriate proceedings, in which proceedings Seller shall use its reasonable best efforts to settle or prosecute by them to a final conclusion in such a manner as to avoid any risk, to the extent reasonably possible, of the Buyer Group becoming subject to liability for any other matter; provided, however, Seller shall not, without the prior written consent of the Buyer, consent to the entry of any judgment against the Buyer Group or enter into any settlement or compromise which does not include, as an unconditional term thereof, the giving by the claimant or plaintiff to the Buyer Group of a release, in form and substance reasonably satisfactory to the Buyer, as the case may be, from all liability in respect of such claim or litigation. If any of the Buyer Group desires to participate in, but not control, any such defense or settlement, it may do so at its sole cost and expense. If, in the reasonable opinion of the Buyer, any such claim or demand or the litigation or resolution of any such claim or demand involves an issue or matter which could have a materially adverse effect on the business, operations, assets or properties of the Buyer Group, then the Buyer Group shall have the right to control the defense or settlement of any such claim or demand and its reasonable costs and expenses shall be included as part of the indemnification obligation of Seller hereunder; provided, however, that the Buyer Group shall not settle any such claim or demand without the prior written consent of Seller which consent shall not be unreasonably withheld. If the Buyer Group should elect to exercise such right, Seller shall have the right to participate in, but not control, the defense or settlement of such claim or demand at its sole cost and expense.
 - (b) If Seller elects not to defend the Buyer Group against such claim or demand, whether by not giving Buyer timely notice as provided above or otherwise, then the amount of any such claim or demand, or if the same be defended by Seller or by the Buyer Group (but none of the Buyer Group shall have any obligation to defend any such claim or demand), then that portion thereof as to which such defense is unsuccessful, in each case shall be conclusively deemed to be a liability of Seller hereunder, unless Seller shall have disputed its liability to the Buyer Group hereunder.
 - (c) In the event the Buyer Group should have a claim against Seller hereunder that does not involve a claim or demand being asserted against or sought to be collected from it by a third Party, Buyer shall Promptly send a Claim Notice with respect to such claim to Seller; provided that the failure to provide such Claim Notice will not affect the indemnification obligations of Seller unless Seller is prejudiced thereby. If Seller does not notify the Buyer within the Notice Period that it disputes such claim, the amount of such claim shall be conclusively deemed a liability of Seller hereunder.
 - (d) All claims for indemnification by Seller Group under this Agreement shall be asserted and resolved under the procedures set forth above substituting in the appropriate place "Seller Group" for "Buyer Group" and "Buyer" for "Seller".
- 14.5. Payment. Upon the determination of liability under this Article XIV, the appropriate Party shall pay to the other, as the case may be, within ten (10) days after such determination, the amount of any claim for indemnification made hereunder.

- - (a) Seller shall have no obligation to indemnify the Buyer Group or otherwise have liability to Buyer Group and neither Buyer nor Toth shall have any obligation to indemnify Seller Group or otherwise have any liability to Seller Group for consequential damages, special damages, incidental damages, indirect damages, punitive damages, lost profits or similar items.
 - (b) The Party entitled to indemnification shall take all reasonable steps to mitigate indemnifiable liabilities and damages upon and after becoming aware of any event which could reasonably be expected to give rise to any liabilities and damages that are indemnifiable hereunder.
 - (c) To the extent that Seller discharges any claim for indemnification hereunder, Seller shall be subrogated to all rights of the Buyer Group against third parties.
- 14.7. Exclusive Remedy. The Parties acknowledge that their sole remedy after the Closing for any breach of any representation, warranty or covenant contained in this Agreement (but specifically excluding the Overall Merger Agreement and the Indemnity and Restriction Agreement) shall be the indemnification provisions set forth in this Article XIV. In addition, each Party acknowledges that all of its obligations in this Agreement to indemnify any other Party are subject to and shall be in accordance with this Article XIV.

ARTICLE XV. MISCELLANEOUS

- 15.1. Brokers' and Finders' Fees. Each Party shall be responsible for the payment of any fees to brokers or finders engaged by such Party in connection herewith.
- 15.2. Expenses. Except as otherwise provided in this Agreement, each Party shall pay its own expenses incidental to the preparation of this Agreement, the carrying out of the provisions of this Agreement and the consummation of the transactions contemplated hereby, except that Buyer shall pay all federal, state and local sales, documentary, stamp, registration and other transfer taxes, if any, due as a result of the purchase, sale or transfer of the Acquired Assets in accordance herewith (but for Excluded Liabilities and Taxes (except transfer taxes set forth in Section 3.4) associated with the Advanced Animations Merger and the fact that such Merger is not a tax free merger under the Code).
- 15.3. Entire Agreement. This Agreement contains the entire agreement between the parties hereto with respect to the transactions contemplated herein and supersedes all other prior agreements, understandings and letters related hereto except for the terms and conditions of the Overall Merger Agreement, the Advanced Animations Merger Agreement, and the Indemnity and Restriction Agreement, which will continue in full force and effect.
- 15.4. Waiver. Any term or provision of this Agreement may be waived at any time by the Party entitled to the benefit thereof by a written instrument duly executed by such Party. The waiver by either Party of a breach of any provision of this Agreement shall not operate or be construed as a waiver of any subsequent breach.
- 15.5. Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly given when delivered in person, by facsimile, receipt confirmed, or on the next business day when sent by overnight courier or on the second succeeding business day when sent by registered or certified mail (postage prepaid, return receipt requested) to the respective parties at the following addresses (or at such other address for a party as shall be specified by like notice):

If to Buyer, to:

PII Ventures, L.L.C. 41000 Woodward Avenue Bloomfield Hills, MI 48304 Attention: Steve Toth, Jr. Phone: (248) 554-6511 Fax: (248) 554-6634 With a simultaneous copy to:

Gutman & Bigelman, P.C. 3000 Town Center Building

Suite 2450

Southfield, MI 48034 Attn: Stephen L. Gutman Telephone: (248) 352-2300 Telecopy Number: (248) 352-3366

If to SPX, to:

SPX Corporation 28635 Mound Road

Warren, Michigan 48092

Attention: Fabrizio Rasetti, Esq. Telephone: (810) 578-7429 Telecopy Number: (810) 578-7470

With a simultaneous copy to:

SPX Corporation

700 Terrace Point Drive Muskegon, Michigan 49443-3301 Attention: General Counsel Telecopy Number: (231) 724-5824

With a simultaneous copy to:

Gardner, Carton & Douglas 321 North Clark Street

Suite 3400

Chicago, Illinois 60610

Attention: Stephen A. Tsoris, Esq. Telecopy Number: (312) 644-3381

or to such other address as the addressee may have specified in a notice duly given to the sender as provided herein, which notice will be effective only upon receipt.

15.6. Amendment. This Agreement may be amended, modified or supplemented only by a written agreement among SPX and Buyer.

15.7. Governing Law. This Agreement shall be deemed to be made in, and shall be interpreted, construed and governed by and in accordance with the internal laws of, the Laws of the State of Michigan. Each of SPX and Buyer hereby irrevocably and unconditionally consents to submit to the jurisdiction of the federal and state courts located in Michigan for any litigation arising out of or relating to this Agreement and the transactions contemplated hereby (and agrees not to commence any litigation relating thereto except in such courts), waives any objection to the laying of venue of any such litigation in such courts and agrees not to plead or claim in any such court that such litigation brought therein has been brought in an inconvenient forum.

15.8. Binding Effect; No Third Parties; Assignment. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, and they shall not be construed as conferring any rights on any other Persons. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto prior to the Effective Time without the prior written consent of the other parties hereto, provided that Buyer may assign all or part of his rights or obligations hereunder to one or more entities owned in whole or in part by Buyer; provided further, that (i) any such assignment will not relieve Buyer of any of its obligations hereunder and (ii) as a condition to such assignment, Buyer will deliver to SPX ten (10) business days' prior written notice thereof and Buyer's unconditional guarantee of each such assignee's performance of this Agreement. All of the terms and provisions of

this Agreement shall be binding upon and inure to the benefit of and be enforceable by the successors and assigns of SPX and Buyer.

- 15.9. Invalidity. In the event that any one or more of the provisions contained in this Agreement shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, then to the maximum extent permitted by law, such invalidity, illegality or unenforceability shall not affect any other provision of this Agreement.
- 15.10. Counterparts; Facsimile. This Agreement may be executed in any number of counterparts, each of which shall be an original, but such counterparts shall together constitute one and the same instrument. This Agreement may be executed and delivered by facsimile transmissions, and a facsimile signature of any Party shall be effective as an original signature.
- 15.11. Construction. The Parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumptions or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of this Agreement.
- 15.12. Binding Arbitration for Allocation of Purchase Price. Any dispute, controversy, or claim arising out of or relating to the allocation of the Purchase Price under Section 3.3 of this Agreement shall at the request of any party be resolved in binding arbitration, subject to the following:
 - (a) any arbitration shall proceed in accordance with Title 9 of the United States Code, as it may be amended or recodified from time to time ("Title 9"), and the current Commercial Arbitration Rules (the "Arbitration Rules") of the American Arbitration Association ("AAA") to the extent that Title 9 and the Arbitration Rules do not conflict with any provision of this Section 15.12;
 - (b) no provision of or the exercise of any rights under this Section 15.12 shall limit the right of any party to seek and obtain provisional or ancillary remedies (such as injunctive relief, attachment, or the appointment of a receiver) from any court having jurisdiction before, during, or after the pendency of an arbitration proceeding under this Section. The institution and maintenance of any such action or proceeding shall not constitute a waiver of the right of any party (including the party taking the action or instituting the proceeding) to submit a dispute, controversy, or claim to arbitration under this Section;
 - (c) any award, order, or judgment made pursuant to arbitration shall be deemed final and may be entered in any court having jurisdiction over the enforcement of the award, order, or judgment. Each party agrees to submit to the jurisdiction of any court for purposes of the enforcement of the award, order, or judgment;
 - (d) the arbitration shall be held before one (1) arbitrator knowledgeable in the general subject matter of the dispute, controversy, or claim and selected by AAA in accordance with the Arbitration Rules;
 - (e) the arbitration shall be held at the office of AAA located in Southfield, Michigan (as the same may be from time to time relocated), or at another place the parties agree on;
 - (f) in any arbitration proceeding under this Section, subject to the award of the arbitrator, each party shall pay all its own expenses, an equal share of the fees and expenses of the arbitrator. The arbitrator shall have the power to award recovery of costs and fees (including reasonable attorney fees, administrative and AAA fees, and arbitrator's fees) among the parties as the arbitrator determines to be equitable under the circumstances; and
 - (g) the interpretation and construction of this Section, including, but not limited to, its validity and enforceability, shall be governed by Title 9 of the U.S. Code, notwithstanding the choice of law

set forth in Section 15.7 of this Agreement.

15.13. Guaranty of Payment. All payments made by Toth hereunder shall be deemed to be a guaranty of payment as opposed to a guaranty of collection.

IN WITNESS WHEREOF, the Parties have duly executed this Asset Purchase Agreement as of the date first written.

SPX CORPORATION, a Delaware Corporation

By: /s/ Thomas J. Riordan

Name: Thomas J. Riordan Title: Vice President

"SPX" AND "SELLER"

PII VENTURES, L.L.C., a Michigan limited liability company

By: /s/ Steve Toth, Jr.

Name: Steve Toth, Jr. Title: Managing Member

"BUYER"

/s/ Steve Toth, Jr. STEVE TOTH, JR.

"TOTH"

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER, is dated this ____ day of ____, 2001, by and between PII VENTURES, L.L.C., a Michigan limited liability company, having its registered office at 41000 Woodward, Bloomfield Hills, Michigan 48304 (hereinafter sometimes referred to as the "SURVIVING ENTITY"), and ADVANCED ANIMATIONS, INC., a Georgia corporation, having its registered office at 717 Channing Drive, N.W., Atlanta, Georgia 30318 (hereinafter sometimes referred to as the "MERGED CORPORATION"), with the Surviving Entity and the Merged Corporation hereinafter sometimes referred to collectively as "CONSTITUENT ENTITIES."

WITNESSETH:

WHEREAS, the Surviving Entity is a limited liability company duly organized and existing under the laws of the State of Michigan, whose Articles of Organization were filed in the State of Michigan on March 19, 2001, (Michigan LLC I.D. #B86318), with 100% of its Membership Interests issued to the persons listed in Paragraph E below, all of whom are entitled to vote on this Merger; and

WHEREAS, the Merged Corporation is a corporation duly organized and existing under the laws of the State of Georgia, incorporated on January 10, 1997 (Georgia Control #K703407), with authorized capital stock of 10,000 common shares, its only authorized class of securities, of which one (1) share is issued and outstanding, and is owned entirely by SPX Corporation, a Delaware corporation ("SELLER"), which outstanding share is entitled to vote on this Merger; and

WHEREAS, the Members of the Surviving Entity, and the Board of Directors and sole shareholder of the Merged Corporation, have by consent resolutions without meetings approved this Agreement and Plan of Merger, and have declared it advisable for the best interests of the Constituent Entities that the Merged Corporation merge with and into the Surviving Entity in the manner and upon the terms and conditions hereinafter set forth and with the effect provided by and pursuant to the Michigan Limited Liability Company Act (MCLA Section 450.4705a [the "MICHIGAN ACT"]), and the OCGA Section 14-2-1109 (being Section 22-1109 of the Georgia Business Corporation Code [the "GEORGIA ACT"]), which respective state statutes permit the Merger herein contemplated;

NOW, THEREFORE, in consideration of the premises and the mutual agreements, covenants and provisions hereinafter contained, it is agreed that the Merged Corporation shall be merged with and into the Surviving Entity and that the terms and conditions of such Merger and the mode of carrying the same into effect and the manner of surrendering the shares of the Merged Corporation to the Surviving Entity shall be as follows:

- A. MERGER. The Constituent Entities shall, on the Effective Date, be merged into a single entity by the Merged Corporation merging into Surviving Entity. The separate existence of the Merged Corporation shall cease on the Effective Date and the existence of the Surviving Entity shall continue unaffected and unimpaired by the merger with all of the rights, privileges, immunities and powers and subject to all the duties and liabilities of a limited liability company organized under the laws of the State of Michigan.
- B. NAME OF SURVIVING ENTITY. The name of the Surviving Entity is PII Ventures, L.L.C., a Michigan limited liability company.

- C. ARTICLES OF ORGANIZATION. The Articles of Organization of the Surviving Entity shall continue to be its Articles of Organization following the Effective Date of the Merger until the same shall be altered or amended as therein provided or as provided by law. Such Articles of Organization are not restated herein, and shall remain as in existence prior to this Merger. The Articles of Organization of the Surviving Entity are set forth in its entirety and attached hereto as Exhibit "A", and all the terms and provisions thereof are hereby incorporated in this Agreement and made a part hereof with the same force and effect as if herein set forth in full; and, from and after the Effective Date of the Merger and until further amended as provided by law, said Exhibit "A", separate and apart from this Agreement and Plan of Merger, shall be, and may be separately certified as, the Articles of Organization of the Surviving Entity.
- D. OPERATING AGREEMENT. The Operating Agreement of the Surviving Entity shall be and remain the Operating Agreement of the Surviving Entity, until altered or amended as therein provided or as provided by law.
- E. MEMBERS AND OFFICERS. The Members of the Surviving Entity, prior to, and from and after the Effective Date of the Merger, shall be the following:

CLT & ASSOCIATES, L.P., a Michigan limited partnership

MEMBER

STEVE TOTH, JR. (or his revocable living trust)

MEMBER

Such Members of the Surviving Entity shall continue in office until their respective successors shall be elected or appointed and qualified.

F. METHOD OF MERGER. The mode of carrying the Merger into effect and the manner of surrendering the outstanding share of the Merged Corporation to the Surviving Entity shall be as follows:

- 1. SURRENDER OF MERGED CORPORATION SHARE. The outstanding share of common stock of the Merged Corporation, and all rights in respect thereof, shall be converted into the right to receive \$16,750,000 (the "Cash Consideration"), and such share shall forthwith be surrendered and delivered to the Surviving Entity by the sole shareholder -- parent corporation of the Merged Corporation, and shall not be converted into membership interests into the Surviving Entity; it being understood that (i) the Merger described herein of a corporation into a limited liability company is authorized under the Georgia Act and the Michigan Act, however, (ii) it is not a tax free reorganization pursuant to Section 368 of the Internal Revenue Code of 1986, as amended ("CODE"), and (iii) is, in fact, for federal income tax purposes, a deemed sale of the assets of the Merged Corporation to the Surviving Entity followed by a constructive liquidation of the Cash Consideration into the Merged Corporation's sole stockholder -- parent corporation under Section 332 of the Code, and that (iv) Seller will only receive the Cash Consideration and shall not receive, in exchange, any membership interests in the Surviving Entity, nor any uncertificated equity interests in the Surviving Entity, nor anything else.
- 2. EFFECTIVE DATE; PAYMENT OF CASH CONSIDERATION. This Merger shall be effective, for all purposes including, but not limited to, accounting purposes and as between the parties hereto, as of the close of business on ______, 2001 ("EFFECTIVE DATE"). On the Effective Date, Surviving Entity shall pay Seller the Cash Consideration by wire transfer of immediately available funds to such bank account as Seller designates in writing on or before the Effective Date.
- G. PROPERTY RIGHTS, ETC., VESTED IN SURVIVING ENTITY. Upon the Merger becoming effective, all the property, rights, privileges, franchises, patents, trademarks, licenses,

registrations, and other assets of every kind and description of the Merged Corporation, including the Services Agreement, dated September 21, 1998, by and between Advanced Animations and The Oz Entertainment Company, shall be transferred to, vested in and devolve upon the Surviving Entity without further act or deed, and all property, rights, and every other interest of the Merged Corporation in its assets shall be, without exception, the property of the Surviving Entity as previously held by the Merged Corporation. The Merged Corporation hereby agrees from time to time, as and when requested by the Surviving Entity or by its successors or assigns, to execute and deliver or cause to be executed and delivered all such deeds and instruments and to take or cause to be taken such further or other action as the Surviving Entity may deem necessary or desirable in order to vest in and confirm to the Surviving Entity title to and possession of any property of the Merged Corporation acquired or to be acquired by reason of or as a result of the Merger herein provided for and otherwise to carry out the intent and purposes hereof and the proper officers and members of the Surviving Entity are fully authorized in the name of the Merged Corporation and the Surviving Entity to take any and all such action.

- H. SERVICE OF PROCESS. The Surviving Entity may be served with process in the State of Michigan in any proceeding for enforcement of any obligation of the Merged Corporation as well as for enforcement of any obligation of the Surviving Entity arising from the Merger. The address to which a copy of such process shall be mailed is 41000 Woodward, Bloomfield Hills, Michigan 48304.
- I. FURTHER ACTIONS. Each of the Constituent Entities shall take or cause to be taken all action, or do or cause to be done all things necessary including all necessary filings in

Georgia, for purposes of the Merged Corporation, and in Michigan, for purposes of the Surviving Entity and the Constituent Entities, that are proper and advisable under the laws of the States of Michigan and Georgia in order to make effective the Merger.

THE ABOVE AGREEMENT AND PLAN OF MERGER, having been executed on behalf of the Surviving Entity, and having been adopted by its Members, and having been executed on behalf of the Merged Corporation, and having been adopted by its sole shareholder and Board of Directors, in accordance with the provisions of the above named Georgia Act, and the Michigan Act, the duly authorized Officers of the Constituent Entities do hereby execute the said Agreement and Plan of Merger as the respective act, deed and agreement of said Constituent Entities on the day and year first above written.

WITNESSES:	PII VENTURES, L.L.C., a Michigan limited liability company
	By: STEVE TOTH, JR. Its: MANAGING MEMBER
	"SURVIVING ENTITY"

ADVANCED ANIMATIONS, INC., a Georgia corporation
 By: Its:
 "MERGED CORPORATION"

[Letterhead of Gardner, Carton & Douglas]

April 16, 2001

SPX Corporation 700 Terrace Point Drive Muskegon, Michigan 49443-3301

Re: Registration Statement on Form S-4

Ladies and Gentlemen:

We have acted as special counsel for SPX Corporation, a Delaware corporation (the "Company"), in connection with the proposed issuance by the Company of up to 1,443,286 shares of Common Stock, \$10.00 par value per share, of the Company (the "Shares") upon consummation of the proposed merger of VSI Holdings, Inc., a Georgia corporation, with and into the Company, as described in the Registration Statement filed with the Securities and Exchange Commission on Form S-4 (the "Registration Statement"). We have examined the Certificate of Incorporation of the Company, as amended, and such other documents as we have deemed necessary for the purposes of this opinion.

Based upon the foregoing, we are of the opinion that the Shares covered by the Registration Statement have been duly authorized, and, when issued, delivered and paid for as contemplated by the Registration Statement, will be validly issued, fully paid and non-assessable.

We consent to the reference to our firm in the Prospectus constituting a part of the Registration Statement and to the filing of this opinion as an Exhibit to the Registration Statement.

Very truly yours,

/s/ GARDNER, CARTON & DOUGLAS

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-K

- [X] ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 FOR THE FISCAL YEAR ENDED SEPTEMBER 30, 2000
- [] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

COMMISSION FILE NUMBER: 1-12942

VSI HOLDINGS, INC.
(EXACT NAME OF REGISTRANT AS SPECIFIED IN ITS CHARTER)

GEORGIA

22-2135522

(STATE OR OTHER JURISDICTION OF INCORPORATION OR ORGANIZATION)

(I.R.S. EMPLOYER IDENTIFICATION NO.)

41000 WOODWARD AVENUE
BLOOMFIELD HILLS, MICHIGAN 48304-2263
(ADDRESS OF PRINCIPAL EXECUTIVE OFFICES) (ZIP CODE)

REGISTRANT'S TELEPHONE NUMBER, INCLUDING AREA CODE: (248) 644-0500

SECURITIES REGISTERED PURSUANT TO SECTION 12(B) OF THE ACT: COMMON STOCK, \$.01 PAR VALUE; AMERICAN STOCK EXCHANGE

SECURITIES REGISTERED PURSUANT TO SECTION 12(G) OF THE ACT: NONE

Indicate by check mark whether Registrant has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that Registrant was required to file such reports), and has been subject to such filing requirements for the past 90 days. Yes [X] No []

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of Registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. []

The aggregate market value of the voting stock held by non-affiliates of the Registrant (3,149,643 shares), as of December 1, 2000, was approximately \$9,055,224 (at closing price of \$2.875).

The Registrant had 33,179,825 shares of Common Stock, \$.01 par value, outstanding on December 1, 2000, excluding 400,250 treasury shares.

DOCUMENTS INCORPORATED BY REFERENCE -- NONE

PART I.

ITEM 1. BUSINESS.

VSI Holdings helps corporations improve their performance.

That deceptively simple statement is the easiest description of what we do because it covers the vast array of services we offer.

Over the past two decades the explosion of new media and information sources has greatly complicated the marketing process. In the not-too-distant past a company could hire an advertising agency, develop an advertising plan, and make strategic media buys, and have a reasonable assumption that their message would get through to their potential customers.

In today's media marketplace a consumer has a choice of hundreds of cable television channels, digital broadcasting, satellite television (and radio), and the ever-increasing choices on the Internet. This has forced many companies to re-evaluate their marketing strategies, not just in terms of what external avenues to use to reach customers, but also to find ways to coordinate their efforts within their organization so that their messages are consistent, effective and impactful.

VSI Holdings (hereinafter referred to as the "company," "we," "us," "our") helps companies do just that. We employ approximately 1,000 individuals in all operations, none of which are covered by an employment contract, nor are represented by a union. To date, we believe we have been successful in our efforts to recruit and retain qualified employees, but there is no assurance that we will continue to be as successful in the future. We believe relations with our employees are good.

We operate our business through two separate business segments: Marketing Services segment and Entertainment/Edutainment segment. See Note 13 of Notes to Consolidated Financial Statements for detailed segment information.

MARKETING SERVICES SEGMENT (Visual Services, Inc., Vispac, Inc., PSG International Inc., eCity Studios, Inc.)

INTEGRATED MARKETING SOLUTIONS

Our products and services encompass a vast range of capabilities that can be customized and assembled to meet the specific needs of a client. Several examples of this integrated approach from the past year are listed below.

General Motors Auto Show in Motion -- In less than 90 days, we orchestrated a large product experience, aimed at non-GM-owning households. The traveling event, held within a one million square-foot site, featured six test tracks with the opportunity to drive up to 130 GM cars and trucks, along with selected competitive vehicles, in a neutral environment. From Internet registration to interactive product information displays that allowed participants to choose the information they wanted, consumers were met at every turn with convenient technology and entertaining product experiences. We produced the event on a "turnkey" basis, providing GM with:

- Development and management of competitive customer databases
- Mailed invitations to consumers
- Event registration over the Internet and other methods
- Product information presentations and displays
- Site management
- Travel and logistics
- Consumer research

In the first nine months of 2000, the event logged more than 115,000 test drives in 5 markets with more than 37,000 participants. More than 94 percent of the participants said they were likely to consider a GM product for their next vehicle purchase or lease, and 95 percent said they would likely visit a GM dealership the next time they were in the market for a new vehicle.

Ford Blue Oval Certified -- In the past year Ford Motor Company changed its pricing structure with dealers by requiring dealers to meet a long list of customer satisfaction criteria in order to qualify for a price discount from the company. We communicated this program to dealers in a series of regional meetings. In addition, we created and continue to manage a Web site designed to keep dealers informed of their status and performance in the program.

Chevrolet Silverado Heavy Duty Launch -- Chevrolet selected us to launch their new line of Heavy Duty Silverado pickups to its internal audience of dealerships and sales consultants. We produced a series of informational "ride-and-drive" meetings where dealership personnel not only learn about the trucks, but drive them in a trackside environment. We prepared the curriculum, conducted the events and handled all travel and logistics, as well as producing a software which matched the right truck for a buyer's specific needs.

Visteon Mobile Communications Center -- Visteon is the newly independent automotive supplier spun off from Ford Motor Company. As part of our strategy to help Visteon take its message to Wall Street and customers, we created a "Mobile Communications Center," a high-tech, high-touch mobile venue to create awareness and communicate Visteon capabilities. The unit includes:

- Tractor-trailer and generator
- Trailer that expands to allow a 22 x 40-foot presentation area
- 33" Plasma display screen with looping welcome video
- Card-scan registration with bar code to track visitors for follow-up
- "Safety buck," which emulates front-end vehicle crash to showcase Visteon systems at work
- A "technoscope" that uses two- and three-dimensional animation and video to explore Visteon technologies "under the skin" of a vehicle
- "Vistelink" touch-screen computer monitor that combines video, audio, graphics and animation to allow participants to interactively "discover" more than 50 separate Visteon technologies and systems

The Mobile Communications Center debuted in June 2000 to various Wall Street audiences in conjunction with Visteon's debut as an independent, publicly-traded company. Since then, the unit has made numerous appearances at manufacturing sites and other venues in the U.S. and Mexico, and is scheduled for additional sites well into 2001.

Ford Motor Company E-Commerce

We are the manager of the Ford iCollection Web site. The development and operation of this online store is one of a suite of services we have developed to address Internet marketing needs for our clients. The iCollection offers consumers a convenient and secure online retail store to purchase Ford, Lincoln, Mercury and Mustang apparel, vehicle accessories and product literature. In addition to these Ford consumer stores, we also manage Rotunda Online, an exclusive iCollection Internet retailer for dealers that offer equipment and materials.

Team Ford Racing

Seeking to capitalize on the growing interest in motorsports, Ford Motor Company chose us to set up a Web site and "virtual" connection to fans of its race vehicles and drivers. The subscription-funded site

offers exclusive information and updates for members, along with special opportunities to purchase souvenirs and racing paraphernalia. We developed the Internet site and continue to manage it.

Bristol-Myers Squibb

We handle over 900,000 phone calls a year with Bristol-Myers Squibb customers requesting more information on pharmaceutical products. Our specially trained phone operators provide interface with consumers, answering questions, sending literature within the 48 hours required by law and making referrals to healthcare professionals at Bristol-Myers Squibb. We also help maintain Bristol-Myers Squibb's consumer database by making daily information transfers to their database vendor.

Ford XL2000: The Next Generation

Sometimes an organization must go beyond information or skill-based training and change its collective attitude and build a new culture. We help Ford effect this process by developing and executing sessions that force participants to re-examine fundamental beliefs and conduct. We have begun pilot implementation of the follow-up to the Excellence in Leadership (XL2000) training and culture change initiative for Ford Motor Company managers and dealership management teams. The "next generation" will build on the XL2000 learning initiatives to enable participants to achieve new, higher levels of customer care and understanding. The three-and-a-half day curriculum was developed by our instructional design team and is conducted by professional trainers. The course is held at the XL Center in Allen Park, Michigan, a 42,000-square-foot conference center with a training capacity of 150 participants, which we designed and built. After the course is attended by the participant, we have satellite broadcasts on a regular basis which aim to sustain the changes people initiate at the XL Center. Since its inception in 1995, more than 18,000 Ford, Lincoln, Mercury and Jaguar dealers and managers, as well as Ford Motor Company representatives, have participated.

OUR PRODUCTS AND SERVICES

Our specific products and services can be divided into four broad categories:

Education and Training (within the Marketing Services segment) -- We help companies improve the performance of their people, offering training, from the traditional classroom to satellites and the Internet.

Back-End Services (within the Marketing Services segment) -- We help companies develop and enhance ongoing dialogue with their customers, and help deliver e-commerce and other fulfillment services.

Marketing Services (within the Marketing Services segment) -- Using a variety of traditional and high-technology solutions, we help companies market and sell their products and develop and maintain relationships with their customers.

Edutainment/Entertainment (within the Entertainment/Edutainment segment) -- We make science and technology come alive for the public with animatronic (animated/robotic) characters and display properties that make learning an entertaining experience. We also make displays that are fun and entertaining for amusement parks and other general audience venues.

Education and Training

People are the key to success in any organization, and corporations realize they won't be successful unless their people have the training to make them effective.

As a full-service provider of training solutions, we work closely with clients to:

- Identify specific needs
- Set objectives based on real-world applications
- Develop curriculum

- Produce courses
- Conduct the actual training
- Evaluate results
- Modify where appropriate

Our instructional designers and writers are involved during this entire process, ensuring that clearly-defined objectives are set and met.

We also develop and execute certification programs that aim to maximize training effectiveness in an organization. These programs:

- Set clearly-defined expectations of students (representing a variety of job titles)
- Ensure that the training material improves on-the-job performance
- Develop tests that measure learning retention
- Execute the testing process
- Tabulate and report results
- Reward students
- Develop and maintain a participant database

We develop and execute a variety of training through traditional and high-tech media.

Types of Training

Product Information -- Before salespeople can sell a product, they must know features and benefits and how to effectively present those attributes to prospects. Our product information training translates what is often highly technical information into practical benefits that a salesperson can readily understand and use with customers. Above all, our product training focuses on the real-world environment of the sales consultants and their customers.

Skills Training -- Product training concentrates on the "what" part of the sales equation. Our skills training concentrates on the "how" -- how the product information is presented. We offer salespeople the opportunity to improve their selling, presenting and negotiating skills. We also help supervisors improve their management skills.

In-Dealership Consulting -- We also do extensive in-dealership work, helping automotive retailers improve their operations by assisting with organizational issues and processes, and by working directly with dealership managers and employees. This consulting includes one-on-one sessions with the dealer principal, small-group sessions with managers, as well as direct training of dealership staff.

Process, Cultural and Behavior Change -- Sometimes an organization must go beyond information or skill-based training and change its collective attitude and build a new culture. We help the organization effect this process by developing and executing sessions that force participants to re-examine fundamental beliefs and conduct. Our XL2000 session, described previously, is a perfect example.

Training Media

Internet -- The growth of the Internet is impacting how people learn. We see the Internet being used increasingly for education and training. We are developing interactive Web-based training courses that literally deliver the information to the employee's workstation.

Interactive Distance Learning (IDL) -- We are a developer of curriculum for this satellite-based teaching medium. IDL connects an instructor to a remote classroom through a live signal transmitted over satellite. While the teachers cannot see their students, they can talk with them; and the students not only

talk with the instructors, but also see them (as well as visual instructional materials -- graphics, charts, videos, etc.). IDL has proven to be a highly cost-effective way to train large groups of people spread over a wide area. For example, employees at an automobile dealership can take a course at their desks or workstations, instead of having to drive 200-300 miles for a training session. VSI Holdings not only develops the courses, but also trains and manages subject matter experts who can deliver this kind of training.

CD-ROM -- Computer disc technology is also helping to make interactive training more accessible to organizations and corporations. We develop training that individuals can take at their own work stations at their own time. This makes the training extremely flexible in terms of scheduling for the individual and the organization.

Classroom Training -- Live, instructor-led classroom training remains an effective way to teach, especially when the material is complicated, and when discussion and small-group exercises are important. Our instructional designers and writers develop these courses, and our instructors deliver the actual sessions.

Learning Centers/Seminars -- We also develop and execute these live training "events" that may spread over the course of several hours or several days. These training sessions may be held in a remote hotel or conference center, or in our own "Vision Center," a 38,000-square-foot training center with theater, break-out rooms, cafeteria and showroom/display area, or in our XL2000 Center (previously mentioned).

Ride-and-Drive -- In training a sales consultant in how to sell a particular vehicle, there is no substitute for getting that person behind the wheel of the vehicle. A ride-and-drive event does just that, for hundreds of sales people. They get a chance to drive the vehicles on a controlled track with various simulations (hills, braking, obstacle courses, acceleration, wet pavement, etc.).

Video -- "Seeing is believing." That's why video remains such an important role in training. While "stand alone" video is still used effectively to train various audiences, the trend is to incorporate video into other media -- Web, CD-ROM, IDL, classroom, etc. We produce more than 120 training videos a year.

Print -- Printed material remains an important support element for virtually every kind of training. Instructor and participant manuals, meeting leader guides, product documentation, reference materials, brochures, flyers -- all serve to reinforce the training message, and all are developed and produced by us.

Marketing Services

Marketing and selling products has never been easy. Today, with the explosion of the Internet leading to an increasingly complicated media marketplace, the competition seems to focus as much on gaining "share of mind" with customers, as it does with actual share of market.

Our marketing services encompasses a broad range of disciplines, including writing, art direction, print and graphic production to effectively support a variety of marketing efforts:

Communications Services -- We create plans for standardizing and synchronizing product messages. We develop core materials and apply them across the board to everything from launch events to public relations.

New Product Launch Events -- These events can be targeted to consumers and/or key internal audiences, such as suppliers, field personnel and retailers. We conduct the events in a variety of ways -- satellite/videoconference broadcasts, seminars, ride-and-drive events, demonstrations, clinics, point-of-sale materials and displays, shows/meetings, or combinations of these media.

Shows/Meetings -- Business theater continues to be an effective way to reach large corporate audiences. Our event producers make use of a variety of media and technologies to stage entertaining and

often elaborate theatrical productions to reach key audiences with various messages. We handle all elements of the show production, including:

- Overall planning
- Theme development and conceptualization
- Creative (writing, multi-media production)
- Show production and execution
- Logistics
- Travel

Ride-and-Drive Events -- While most ride-and-drive events concentrate on training salespeople, we have also used the concept to introduce consumers to cars and trucks, as described above.

Customer Relationship Management -- We help establish and maintain relationships with consumers through a variety of interactive tools, including:

- Internet applications
- Personalized direct mail
- In-bound call center applications (consumers call a toll-free number for information)

Experience-Based Marketing -- We help companies meet customers where they are. We develop and execute experiential events that allow customers to see, touch, feel and use products and services. These events can help companies launch products, reach new audiences or build brand and product awareness with existing customers. We use a variety of resources and skills to develop these "experiences":

- Point-of-sale displays
- Interactive kiosks
- Touring exhibits
- On-site promotional events
- Product demonstration programs
- Video
- Publications/literature

Internal Campaigns -- In addition to providing comprehensive training for company employees, we also develop informational campaigns that reach key support personnel within an organization so that they understand products, services and processes essential to ensure effective sales and operations.

Back-End Services

A great Web site or, for that matter, a great marketing campaign is only a start. Business does not operate on image alone. The need for behind-the-scenes "fulfillment" has always been important for companies; with the explosion of Web-based communications, that traditional fulfillment role has taken on even greater significance. We have expanded our long-standing traditional fulfillment capabilities into highly integrated delivery solutions. We offer client companies comprehensive support for material production, fulfillment and distribution. This year we fulfilled and shipped more than 80,000 package requests per week and printed more than 100 million impressions.

Our comprehensive back-end services include:

- Administrative support
- Online Internet ordering and processing

- Database processing
- Complete binding
- High volume black-and-white and color-capable printing
- Electronic and press printing
- Just-in-time fulfillment
- Warehousing and distribution

Our fulfillment and distribution services keep our clients in touch with every aspect of their project, using various communication resources:

- Client terminal-to-computer tie-ins provide clients direct links to specific program information and data stored on our servers
- Electronic file transfers to digital print on-demand center
- Custom-designed data reports
- Toll-free hotlines
- Interactive data systems

Call Center

Our Call Center provides services that support a client's overall marketing effort. It is not a "boiler room," the kind of outbound telephone soliciting that annoys customers and gives telemarketing a bad name. Rather, it is an extension of our clients' operations. When our specialists communicate with a client's customers, these specialists are the voice of the brand. More than 90 percent of all calls to the Call Center are from consumers seeking product information or purchase assistance.

Examples of One-to-One Marketing Programs at the Call Center:

- Direct sales (sales in response to mail offer, Web site offers and transactions occurring at automotive dealerships)
- Catalog sales and order processing and fulfillment
- Web-based communications
- Vehicle service reminders and recall campaigns
- Owner retention programs
- Lead generation
- Market research
- Automotive retail operations support

The Center is located in Livonia, Michigan in a 45,000-square-foot center with over 400 workstations.

Information Technology

Managing information is an important function for any business, but it can be a cumbersome task, especially for large organizations. Our Information Technologies (IT) operations help fulfill this critical component for many companies. Our IT services allow companies to focus on core business competencies.

Our comprehensive set of tools and services for building analytical solutions includes:

- 24-hour operation with in-house technical support staff, disaster recovery and security services
- Open system environment can transfer data from multiple client environments, accepting various data in various formats and/or platforms
- Development and management of intranet and Internet sites
- Daily communication with clients that uses the Internet to transfer secure information, File Transfer Protocol that allows various file formats to be exchanged, or connectivity using T-1 lines
- Secure client links to data and servers through the use of firewalls
- Design of customized programs, databases and reports/query tools to help clients collect and analyze data
- Development and management of client Internet-based ordering systems for online processing and business-to-business applications

Entertainment/Edutainment

VSI Holdings' subsidiary, Advanced Animations, Inc., is a worldwide supplier of entertaining and educational animatronic (animatronics uses hydraulic and pneumatic compliant motion technology to create lifelike creatures with realistic facial and body movements) displays for the theme park, corporate, museum, casino and retail industries. We design and fabricate entertainment products, including animatronics, touring shows and exhibits, kiosks, sets and scenery, and much more.

Examples of our work include:

- Universal Studio's Theme Parks (Florida, California, Japan) T2: 3D Cyborg attraction and the Men In Black attraction in the Florida park
- Actors from the Atlantis Fountain Show at the Forum Mall in Las Vegas
- MGM Grand Theatre's "Morgana" The Fire Breathing Dragon in Las Vegas
- Elvis Presley at Toussaud's Museum in Las Vegas
- The Signature Clock Tower at certain FAO Schwarz retail stores

We also specialize in event marketing, touring shows, museum exhibitions and other edutainment-based attractions.

We are heavily involved in expanding the edutainment industry -- a new form of amusement, which takes the value of education and adds a healthy dose of entertainment to create memorable, interactive learning experiences.

Edutainment-based attractions produced by us include:

Grossology: The (Impolite) Science of the Human Body -- an interactive exhibition based on a popular series of books by science teacher Sylvia Branzei. The 5,000-square-foot exhibition promotes scientific discovery of the human body based on the theory that the best way to get kids interested in science is to present it in terms they find most appealing. Grossology premiered January 2000 at ScienceWorld British Columbia and is touring North America on a five-year commitment visiting major museums, science centers, theme parks and other similar venues. Bookings for our two Grossology exhibits cover the next four years.

The Drunk Driving Simulator -- a traveling road show with three tours that puts students behind the wheel of an interactive vehicle programmed to simulate driving under the influence of alcohol. The tours travel to high schools throughout the United States and have made more than 3,250 appearances in 200

cities. Nearly one million students have participated with the simulator, and three million people have observed the tour since 1988.

CUSTOMERS

Both VSI and Vispac receive over 90% of their revenue in the automotive industry. The various divisions of Ford, General Motors, and Nissan accounted for 74%, 65%, and 78% of our revenues for fiscal years 2000, 1999, and 1998 respectively. The loss of any of these customers would have a material adverse effect on us. While these companies have historically been steady customers, there can be no assurance that such relationships will continue.

Future revenues are dependent on such factors as new product introductions, the automotive industry's attention to process improvement within the dealerships, the industry's focus on customer satisfaction and global training, and the speed at which our clients embrace electronic commerce as an alternative. Recent developments among our automotive clients suggest that the migration to electronic commerce will be rapid.

In addition, we have a small percentage of our business that is based on long-term contracts with our customers. For the nearly four decades of our core marketing service business, we have needed to constantly develop and sell new programs.

INDUSTRY AND COMPETITION

The number of suppliers to the automotive industry is dwindling. Manufacturers are demanding capital investments in systems and technology from their suppliers. This trend is causing suppliers to consolidate and streamline their operations. As a turnkey business communication provider to the automotive industry, management believes it is well positioned to continue to compete in this highly competitive environment.

Manufacturers are focusing on their core business processes, which generally involves product development, assembly, distribution and financing. We expect to continue to benefit from the practice of turning other essential but non-core business processes such as marketing, communication, and training functions over to third party vendors.

As a consolidator of business product training and technique, we believe that the use of online and distance learning technology will continue to grow. More and more corporations are using Web and distance learning training as an alternative to costly trips to attend or conduct training sessions.

We provide a broad range of services and products that compete primarily with a variety of agencies, incentive and distribution fulfillment, and marketing services firms. These firms generally provide a limited range of services that compete only with a portion of our services. Management believes that no single company or small group of companies dominates the marketing services industry.

We are aggressively implementing new technologies to help our clients communicate with consumers and train their employees. We offer Web-based solutions, satellite-based training, CD-ROMs and a variety of other methods described in this report. The accelerating rate of change in technology makes it more difficult to predict the future of the technologies our business will need to use in order for us to continue to be successful.

Competition in the Marketing Services segment is intense, and we expect that competition will increase. We compete on the basis of service quality, creativity and price. We compete against many companies, some of which have significantly greater financial resources than we have. There can be no assurance that we will have the financial resources, technical expertise or marketing, distribution and support capabilities to compete in the future. Competitive pressures could reduce the demand for our products and services, cause us to reduce prices, increase expenses or cause delays or cancellations of customer orders, any of which could adversely affect our business and results of operations.

In October 2000, the Marketing Services segment had backlogs of approximately \$92 million, all of which is expected to be completed by the coming fiscal year. This is an 8% increase over the approximately \$85 million in October 1999. We consider backlog to represent firmly committed business. The firmness of such orders is subject to future events (e.g., cancellation of a purchase order by a client due to strike, budget constraints or other client internal matters) which may cause the amount of the backlog actually fulfilled to change. The principal markets for our services are the North American operations of automobile manufacturers and their distribution networks, including automotive dealerships. Services are sold directly to the manufacturer and/or dealer. We use a variety of distribution methods to deliver services, including the U.S. Postal Service, commercial freight services, the Internet and satellite transmissions.

There are currently no products in development that would require significant additional financial investment to bring to market. Materials have historically been widely available from numerous sources. We have no material patents, trademarks, licenses, franchises, or concessions. There is no significant seasonality in the Marketing Services segment. The Marketing Services segment competes primarily in the automotive industry, where slow payment of invoices is a standard practice. Bank lines of credit are used to finance receivables.

THE ENTERTAINMENT/EDUTAINMENT SEGMENT Advanced Animations, Inc. (AAI)

Advanced Animations' products are sold on a custom, made-to-order basis directly to companies worldwide.

Materials have historically been widely available from numerous sources. We have no material patents, trademarks, franchises or concessions. There is no significant seasonality in the Entertainment/ Edutainment segment. We receive payments as work progresses on custom projects, so there are no significant working capital issues. AAI has no customers who constitute more than 10% of consolidated revenue.

In October 2000 the Entertainment/Edutainment segment had backlogs of approximately \$2 million, all of which is expected to be completed in the coming fiscal year. This compares to approximately \$6 million in October 1999. The firmness of such orders is subject to future events which may cause the amount of the backlog actually fulfilled to change.

Competition in the Entertainment/Edutainment segment is intense, and we expect that competition will increase. We compete on the basis of product quality, creativity and price. There can be no assurance that we will have the financial resources, technical expertise or marketing, distribution and support capabilities to compete in the future. Competitive pressures could reduce the demand for our products and services, cause us to reduce prices, increase expenses or cause delays or cancellations of customer orders, any of which could adversely affect the business and results of operations.

HISTORY OF COMPANY

As of October 1, 1996, Martin S. Suchik, was the President of The Banker's Note, Inc. ("TBN"), a public company incorporated in April 1981. Prior to 1997, Suchik's uncle -- Steve Toth, Jr., the President and Chief Executive Officer of the Company and its affiliates ("Toth") held a 33% interest in TBN as the result of the provision of financing which had started in 1991. In February 1997, TBN acquired Advanced Animations, Inc. ("AAI"), a company controlled by Toth, for 7,563,077 shares of TBN stock and Toth became TBN's controlling shareholder. In April 1997, the Company reincorporated in Georgia from Texas, changed its corporate name to "VSI Holdings, Inc." (VSIH) from "The Banker's Note, Inc.", and placed its retailing business in an operating subsidiary, BKNT Retail Stores, Inc. ("RSI"). As of July 1 and September 30, 1997, the Company acquired Vispac, Inc. ("Vispac") for 6,200,000 shares of VSIH stock and Visual Services, Inc. ("VSI") for a net of 14,285,715 shares of VSIH stock. Both Vispac and VSI were also controlled by Toth. As of September 30, 1998, we sold subsidiaries constituting the former retail segment to Mr. Suchik and related entities. See Item 13 Contingent Liability for Discontinued Operations.

ITEM 2. PROPERTIES.

Our headquarters are located in Bloomfield Hills, Michigan and consists of 103,000 square feet. The current lease continues until February 2003 and has a five-year renewal option. This facility is also the base for subsidiary VSI, which also leases a training facility/office of 38,000 square feet in Rochester Hills, Michigan, a training facility/office of 42,000 square feet in Allen Park, Michigan, a 23,000 square foot storage facility in Livonia, Michigan and a 12,000 square foot sales office in Cypress, California. VSI owns a 45,000 square foot office building in Livonia, Michigan, which houses its dialogue center and has a collateralized mortgage of \$2,349,000.

Vispac is based in a 149,000 square foot office/warehouse building in Livonia, Michigan that it leases from a partnership that is owned by a Toth family investment partnership. Vispac owns two warehouse facilities in Livonia, Michigan of 92,000 and 93,000 square feet that collateralize mortgages of \$2,553,000 and \$2,352,000.

The Entertainment/Edutainment segment is based in a 26,900 square foot office/manufacturing plant in Stockbridge, Vermont. The property is owned by AAI.

We generally consider the productive capacity of each facility operated by each of our industry segments to be adequate and suitable for our requirements. We are currently undertaking an extensive study of our organizational real estate needs with particular focus on the alternatives available for the headquarters' operations after that lease expires in 2003.

ITEM 3. LEGAL PROCEEDINGS.

The Company is periodically involved in routine proceedings. There are no legal matters, existing, pending, or threatened, which management presently believes could result in a material loss to the Company.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS.

No matters were submitted to a vote of security holders during the fourth quarter of fiscal 2000.

PART II

ITEM 5. MARKET FOR REGISTRANT'S COMMON EQUITY AND RELATED STOCKHOLDER MATTERS.

The Common Stock trades on the American Stock Exchange under the "VIS" symbol. The table sets forth the trading prices for the Common Stock by quarter as reported for the last two years. The range of prices for the Common Stock, as reported by Reuters Limited, follows:

FISCAL QUARTERS	HIGH	LOW
First Quarter, 1999	\$6.00	\$3.19
Second Quarter, 1999	7.25	4.75
Third Quarter, 1999	5.88	4.13
Fourth Quarter, 1999	4.69	3.13
First Quarter, 2000	\$4.44	\$3.00
Second Quarter, 2000	4.00	2.75
Third Quarter, 2000	3.38	2.50
Fourth Quarter, 2000	3.81	2.13

Based on past requests for proxy materials, we believe that we have substantially more beneficial holders of our Common Stock than the approximately 375 "of record" holders on December 1, 2000. We currently reinvest all earnings rather than paying cash dividends and, for the foreseeable future, intend to

continue that policy. Our loan agreements prohibit the payment of cash dividends without prior bank approval. No dividends were declared in the two year period ended September 30, 2000.

ITEM 6. SELECTED FINANCIAL DATA.

BALANCE SHEET DATA

			AS OF		
	SEP. 30 2000	SEP. 30 1999	SEP. 30 1998	SEP. 30 1997	SEP. 30 1996
		(IN THOUSANDS	5)	
Working Capital	\$ 10,231	\$ 9,074	\$ 9,121	(\$ 1,969)	\$ 9,105
Total Assets	116,129	98,302	89,555	77,069	66,856
Long-Term Debt	19,429	19,296	17,506	5,281	1,900
Redeemable Common Stock	. 0	. 0	1,960	. 0	. 0
Total Liabilities	88,578	75,983	69,413	65,402	36,784
Stockholders' Equity	27,551	22,319	20, 142	11,667	28,072
See footnotes below	, 1	, 1	, 1	2 & 3	, 2

OPERATING DATA

			YEAR ENDED		
	SEP. 30 2000	SEP. 30 1999	SEP. 30 1998	SEP. 30 1997	SEP. 30 1996
		(IN THOUSA	NDS, EXCEPT	PER SHARE)	
Net Sales Income (Loss) from Continuing	\$187,255	\$143,360	\$163,158	\$130,526	\$128,213
Operations	5,550	(498)	9,211	9,658	9,141
Operations Per Share	0.17	(0.02)	0.28	0.20	0.18
Number of SharesSee footnotes below	33,070 1 & 5	32,816 1 & 5	32,851 1	32,784 2 & 4	32,767 2 & 4

The Company paid no dividends on its Common Stock during any of the periods presented, and has no current plans to change that policy.

- 1. September 30, 2000, 1999 and 1998 balance sheet data and operating data exclude BKNT Retail Stores, Inc. accounts.
- 2. September 30, 1997, and 1996 Balance Sheet data include the accounts of BKNT Retail Stores, Inc. The operating data excludes the operations of BKNT Retail Stores, Inc.
- 3. See Item 13 Declared Distributions to Stockholders regarding distributions of previously taxed undistributed earnings to the stockholders of acquired subsidiaries. This reduced working capital and stockholders' equity.
- 4. 1997 and 1996 operating data and earnings per share information are pro forma amounts and have been calculated as if the subsidiaries had been consolidated for both years.
- 5. The loss from continuing operations in 1999 includes charges to earnings of \$421,000 and \$2,154,000 related to the impairment of long-lived assets and a reduction in the value of goodwill, respectively. The income from continuing operations in 2000 includes a charge to earnings of \$516,000 related to the impairment of long-lived assets. See Notes 2 and 4 of Notes to Consolidated Financial Statements.

ITEM 7. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS.

The following discussion and analysis should be read in conjunction with the consolidated financial statements and the notes thereto included elsewhere in this report. The following discussion contains certain forward-looking statements relating to our anticipated future financial conditions and operating results and our current business plans. In the future, our financial condition and operating results could differ materially from those discussed herein and our current business plans could be altered in response to market conditions and other factors beyond our control. Important factors that could cause or contribute to such difference or changes include those discussed elsewhere in this report (see the disclosures under "Cautionary Statement for the Purpose of the "Safe Harbor" Provisions of the Private Securities Litigation Reform Act of 1995").

Future Accounting Changes -- In December 1999, the Securities and Exchange Commission issued Staff Accounting Bulletin 101. The staff accounting bulletin summarizes certain of the staff's views in applying generally accepted accounting principles to revenue recognition in financial statements. Any changes that may be required as a result of this bulletin are required to be implemented no later than the last fiscal quarter of the fiscal year beginning October 1, 2000. Management is finalizing its assessment of the impact this bulletin will have, if any, on the Company's financial statements.

OPERATING RESULTS

MARKETING SERVICES SEGMENT:

Year Ended September 30, 2000

Revenues from the Marketing Services segment increased 36% to \$182,403,000 for the year ended September 30, 2000 from \$134,295,000 last year. This increase was due to various projects, including those projects mentioned in Item 1. The largest new projects this year were vehicle launch activities, test drive programs, and dealership training. During the year, a continuing contract with annual revenue of approximately \$15 million was lost. We do not expect the see the entire impact of this in the next year, as it is being phased out. Management will seek to replace the business or make cost cutting measures as necessary.

Cost of revenue as a percentage of sales increased approximately four percentage points from last year. The large value of the jobs mentioned in the revenue paragraph above had a higher cost of revenue proportion than our other revenues.

Income from Operations from the Marketing Services segment increased from \$486,000, to \$17,524,000 in the current year. The rate of increase in operating expenses was less than the rate of increase in revenues because we were able to better utilize our core skilled staff. Thus, we were able to experience significant revenue growth without having to increase our staff as much as we would have needed to otherwise. The increase in operating expenses is primarily personnel expense increases.

Year Ended September 30, 1999

Revenues from the Marketing Services segment decreased 14% to \$134,295,000 for the year ended September 30, 1999 from \$155,681,000 in the previous year. This decrease was due to approximately \$10,000,000 in various projects which were completed and not replaced with similar work during the fiscal year, and another \$11,000,000 was lost due to project scope reduction and schedule slippage.

Cost of revenue as a percentage of sales decreased approximately four percentage points from the previous year. The jobs mentioned in the revenue paragraph above had a higher cost of revenue proportion than our other revenues, and accounted for the decrease.

Income from Operations from the Marketing Services segment decreased from \$14,834,000, to \$486,000. Our operating expenses decreased slightly in 1999. The rate of decrease in operating expenses was less than the rate of decrease in revenues. While there was a decrease in operating expenses, this decrease was not sufficient to offset the unexpected decline in revenues. Although sales decreased and

personnel reductions were made, management chose to maintain a core skilled staff as we anticipated future sales growth opportunities. Another factor which negatively impacted operating expenses was depreciation expense which rose \$800,000 primarily due to a full year's depreciation of a call center building and related equipment, purchased in fiscal year 1998, and its contents. \$2,154,000 of Goodwill in PSG was written off because it was considered to be impaired (see Note 2 of Notes to the Consolidated Financial Statements).

Year Ended September 30, 1998

Revenues from the Marketing Services segment increased 26% to \$155,681,000 for the year ended September 30, 1998 from \$124,020,000 in 1997. This increase was due in large part to the pilot phase of a ride and drive program for an automotive manufacturer. VSI drew from the full range of its in-house capabilities to create, produce and field the program within a 90-day timeframe. VSI also completed the launch of a new truck product for an automotive manufacturer and was immediately awarded the assignment to launch two new 1999 model year trucks.

Income from Operations from the Marketing Services segment increased 27% for fiscal 1998, to \$14,834,000, up from \$11,674,000 the year before. The increase is attributable to an increase in sales and the efficient utilization of overhead.

ENTERTAINMENT/EDUTAINMENT SECTOR:

Year Ended September 30, 2000

Revenues from the Entertainment/Edutainment segment decreased 47% to \$4,852,000 for the year from \$9,065,000 last year. During the year we were told by our potential customers that their plans were on hold. Our Grossology touring exhibits and the Drunk Driving Simulator, as mentioned in Item 1, were our primary revenue for the year. The Drunk Driving Simulator program completed in December, 2000. Unless and until another sponsor can be arranged, revenues for this segment will be approximately \$2 million lower than it otherwise would be.

Income from operations from the Entertainment/Edutainment segment decreased to a loss of \$3,240,000 from an operating income of \$2,352,000 for the prior year. The decrease was due primarily to the lack of revenue mentioned above. Although revenues decreased, operating expenses rose as we shifted production efforts to the construction of three Grossology touring exhibits. Upon completion of these exhibits, layoffs and other cost cutting measures have been implemented. The market for our products appears to remain soft. While we do not expect to have this kind of loss next year, it is uncertain whether or not this sector will be profitable. It is difficult to anticipate sales and income trends.

Year Ended September 30, 1999

Revenues from the Entertainment/Edutainment segment increased 21% to \$9,065,000 for the year ended September 30, 1999 from \$7,477,000 for the prior year. The increase was due primarily to new animation work received from Universal Studios Florida Men in Black exhibit and for the E.T. Adventure attraction coming to Universal Studios -- Japan.

Income from operations from the Entertainment/Edutainment segment increased 48% to \$2,352,000 from \$1,584,000 for the prior year. The increase was due primarily to efficiencies achieved from utilization of existing tooling and technologies on repetitive projects. It is difficult to anticipate trends.

Year Ended September 30, 1998

Revenues from the Entertainment/Edutainment segment increased 15% to \$7,477,000 for the year ended September 30, 1998 from \$6,506,000 for the prior year. The increase was due primarily to new animation work received from Universal Studios Hollywood for T2:3D "Battle Across Time" attraction and for the E.T. Adventure attraction coming to Universal Studios -- Japan.

Income from operations from the Entertainment/Edutainment segment increased 58% to \$1,584,000 from \$1,002,000 for the prior year. The increase was due primarily to efficiencies achieved from utilization of existing tooling and technologies on repetitive projects.

LIQUIDITY AND CAPITAL RESOURCES:

OPERATING ACTIVITIES.

Our two market segments have contrasting operating capital needs. The Marketing Services segment requires considerable operating capital to support its accounts receivable and maintain its marketing infrastructure; while billing is periodic with little risk of non-payment, client payment is typically slow. It also experiences a relatively steady cash flow; there is a historically increased need for working capital in the late summer and early autumn as new vehicle communications services are provided. This results in greater bank borrowings during that time of year; the lines of credit are designed to meet any such need we may have. This year, our increased sales led to significant increases in our accounts receivable, accounts payable and notes payable balances at year end. Certain amounts received from clients in advance had been restricted and held in escrow until costs related to a specific job are incurred by the Company. During the year ended September 30, 2000, the escrow arrangement was terminated, resulting in reduced balances for cash in escrow and advances from clients.

The Entertainment/Edutainment segment requires relatively little capital for operating purposes, because its clients typically pay sizable deposits before projects begin, and make progress payments during project fabrication. Sales represent discretionary spending on the part of its customers and their customers. Because of this, projects are sometimes delayed; conversely several different projects can be awarded in a short period of time. The revenue and earnings of the Entertainment/Edutainment segment are highly variable, and will probably continue to be so.

Generally, all obligations are met out of cash flow generated from operations and borrowings. Based on current business conditions, we expect to meet our current obligations with cash flow generated from operations and borrowings.

INVESTING ACTIVITIES.

Capital expenditures for 2000 were approximately \$4.1 million higher than 1999. This is primarily attributed to the construction of three Grossology touring exhibits in our Entertainment/Edutainment segment, and to investment in Interactive Voice Response and other systems for our call center in our Marketing Services segment.

Venture investments include Navidec, Oz Entertainment Company (OEC), and eCollege.com.

Navidec (NVDC -- Nasdaq) is a developer of web sites and information services. Oz Entertainment Company is developing a theme park, both physically and virtually, based on the Wizard of Oz movie and 40 additional books in the Oz series; the park is scheduled for completion in 2004. eCollege.com is a provider of online learning solutions for colleges and universities, as well as advanced web-based corporate training. We view these strategic investments in equity positions as opportunities to leverage our core competencies to expand our business lines and enter new markets. Bank covenants prohibit us from making significant additional investments in these companies.

Navidec is a developer of web sites and web based complete automotive purchase transaction and information services for prospective customers. Their product, referred to as Driveoff.com, was sold during the year to CarPoint, which is majority-owned by Microsoft. At January 9, 2001, the 431,525 shares, acquired for \$2,450,000, had a fair market value of \$1,510,000.

During the year, we invested an additional \$500,000 in Oz Entertainment Company (OEC), bringing our total investment to \$4.5 million, of which \$4 million is invested in K.C. Investors, LP (as a limited partner). This partnership was formed as an investment partnership to fund the continuing development of OEC, a corporation in which the partnership holds a majority interest. OEC will develop the Wonderful

World Of Oz theme park. We recognized losses of \$658,000 and \$530,000 for the years ended September 30, 2000, and 1999, respectively, related to this investment. We expect to see continued losses until the opening of the park, currently scheduled for 2004. The park is planned to be constructed on 9,000 acres of land owned by the federal government. It is necessary to receive title to the land before construction can begin. In order to receive title to the land, approval must be received from two governmental authorities. The first authority tabled consideration of this until February, 2001. It is anticipated that title will be received during the next year. The success of the park as an investment is dependent upon receiving title to the land, as well as certain infrastructure improvements being completed by or paid by governmental agencies, financing arranged through governmental agencies, as well as additional public or private financing. If the park does not open, the entire investment, currently valued at \$3,312,000, is at risk. The projections provided and prepared by the management of K.C. Investors, LP and OEC forecast that the park will be profitable upon opening. See Item 13 for a discussion of related party investments in K.C. Investors, LP and in OEC.

During fiscal year 1999, we invested \$3.5 million in convertible preferred stock in a private placement offering of eCollege.com, a company engaged in providing technology and services that enable colleges and universities to offer an online environment for distance and off-campus virtual learning. They currently have contracts with more than 190 education partners. eCollege.com sold 4.5 million shares of its common stock in an initial public offering which took place during our fiscal year 2000. As part of the offering, our investment converted into 468,808 shares of common stock. During the year, we also invested an additional \$50,000 to acquire 4,500 shares of eCollege.com. At January 9, 2001 our investment in eCollege.com (ECLG -- Nasdaq) had a fair market value of \$2,544,000.

Purchase of the Performance Systems Group Inc.

In February 1998, we acquired the assets of The Performance Systems Group for approximately \$4.5 million, consisting of 280,000 shares of our common stock and \$2.6 million in cash. Performance Systems Group provides in-field consulting and change process sustainment services primarily to automobile dealerships in Canada, Australia, and New Zealand. The acquisition was accounted for under the purchase method. See Notes 1 and 2 of Notes to Consolidated Financial Statements.

See the discussion in Item 13 regarding related party notes receivable and payable and advances, and declared distributions to stockholders.

FINANCING ACTIVITIES.

At September 30, 2000, we had lines of credit totaling \$37,000,000; interest on these lines were at London Inter-Bank Offered Rate ("LIBOR") plus 1.5%; at September 30, 2000 the interest rate was 8.12% and the outstanding balances were \$35,214,000. These lines of credit mature in March, 2001 and March, 2002. These lines of credit have covenants restricting us from borrowing elsewhere, loaning or guaranteeing a loan of another company without the prior written consent of the bank; transferring assets except in the ordinary course of business; and declaring dividends. Other covenants mandate certain levels of net worth and working capital, and that the ratio of total liabilities to net worth, debt service ratio and current ratio do not exceed certain amounts.

We have had a long term relationship with the bank. Through the years, it has provided financing and lines of credit for us. There can, however, be no assurances that the lines of credit will be renewed when they mature. If we are unable to renew these lines of credit, other sources of financing would be sought, primarily a line of credit from another banking institution.

Since we are a net borrower of funds, minimal cash balances are kept on hand. At any point in time, we may have more money in checks outstanding than the cash balance. When checks are presented for payment, the bank notifies us, and we borrow on our lines of credit to cover the checks.

During the year, we borrowed \$4.5 million from related parties at a 7.00 interest rate, due on demand. At September 30, 2000, the balance was \$3,420,000.

At September 30, 2000, our long-term debt consists of mortgages totaling \$7,254,000 and a term note relating to building improvements of \$390,000. The mortgages bear interest at fixed rates ranging from 6.30% to 8.17% and at prime (9.50% at September 30, 2000). The mortgages include a balloon payment for any unpaid balances at the end of the term of the notes. The balloon payments are due beginning June, 2004 through October, 2005. No other long-term debt financing for facilities is expected in fiscal year 2001. We anticipate selling one of the warehouses we own in Livonia, Michigan; the sale price is expected to exceed the net book value, and would generate cash in excess of the mortgage balance.

On a long-term basis, increased financing may be necessary to fund any large project awarded to us, or any acquisitions we may make. While there are no current plans to conduct an offering of stock, in the long term, that cannot be ruled out.

During fiscal year 2000, we acquired \$321,000 of treasury stock.

CAPITAL ACTIVITIES.

Employees exercised stock options for 440,000, 2,000 and 64,000 shares in fiscal year 2000, 1999 and 1998 respectively. We granted stock options at a weighted average price of \$3.66, \$6.81 and \$6.28 for 168,000, 153,000 and 586,000 in fiscal year 2000, 1999 and 1998 respectively. The options are exercisable two and three years from the date of grant in two equal parts, and expire five years after the date of grant.

In December 1997, we implemented a restricted stock plan for 500,000 shares. During 1999, the number of shares authorized for the plan was increased to 1,000,000. Awards of 51,000, 102,298 shares and 462,375 shares were granted under the restricted stock plan during the fiscal years ended September 30, 2000, 1999 and 1998, respectively. The shares vest one, two and three years from the date of grant in three equal parts; in December 2000, 1999 and 1998, we issued 163,000, 175,000 and 154,000 shares respectively.

We do not expect the exercise of stock options, or purchase of shares, by employees and directors to be a material source of capital in fiscal year 2001. We believe that cash flows from operations along with borrowings will be sufficient to finance our activities in 2001. We have no current plans to conduct an offering of our shares to the public in fiscal year 2001.

OUTLOOK

Although the past few months have been challenging ones for the stock market, the U.S. economy's continued control of inflationary factors and modest interest rates, added to low unemployment, argues for a relatively stable automotive market. Truck sales continue to fuel the auto industry's growth, with most automotive industry analysts predicting sales of approximately 15,000,000 to 16,000,000 new vehicles (automobiles and trucks) in the United States in 2001. While this is lower than sales in 2000, the market has been relatively consistent over the past few years. Approximately 90% of our revenues are directly associated with the automotive industry. We cannot determine what impact a reduction in automobile sales may have on us.

In the Marketing Services segment, we continue to align ourselves with people, products and processes that recognize and take advantage of how the world goes about conducting business and accessing information. Our ties to the auto industry includes customer contact activities. The emphasis on globalization, combined with budgets needed to reach a worldwide audience, is placing new emphasis on training opportunities on the World Wide Web. Our substantial investments in this area make us poised to take advantage of the opportunities as they arise.

Customer Relationship Management initiatives are placing increasingly critical importance on e-business within the auto industry as clients look for answers and assistance from turnkey solution providers such as us. We help clients identify their customer base, create and execute strategic tactical marketing, distribute and fulfill and ultimately evaluate results to justify the expenditure. As manufacturers continue to outsource, we have an opportunity for growth by providing additional services to these manufacturers, advertisers, and other marketers.

We believe that experiential events are becoming increasingly popular among clients as they continue to realize the impressive long-term results from this area of customer relationship building. We deliver all-inclusive experiential events that encompass the wide array of products and services that we can offer our clients, from Internet online registration to product procurement and exit interview research studies.

"CAUTIONARY STATEMENT FOR THE PURPOSES OF THE "SAFE HARBOR" PROVISIONS OF THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995"

Certain statements in Management Discussion and Analysis of Financial Condition and Results of Operations and certain other sections of this Annual Report are forward-looking. These may be identified by the use of forward-looking words or phrases such as "believe," "expect," "anticipate," "should," "planned," "estimated," and "potential," among others. These forward-looking statements are based on our reasonable current expectations. The Private Securities Litigation Reform Act of 1995 provides a "safe harbor" for such forward-looking statements. In order to comply with the terms of the safe harbor, we note that a variety of factors could cause our actual results or experience to differ materially from the anticipated results or other expectations expressed in such forward-looking statements. The risks and uncertainties that may affect the operations, performance, development and our results include but are not limited to: (1) the complexity and uncertainty regarding the development and customer acceptance of new products and services; (2) the loss of market share through competition; (3) the introduction of competing products or service technologies by other companies; (4) pricing pressures from competitors and/or customers; (5) our inability to protect proprietary information and technology; (6) market acceptance of our touring exhibits in our Entertainment/Edutainment segment; (7) the loss of key employees and/or customers; (8) our customers continued reliance on outsourcing; (9) changes in our capital structure and cost of capital, and ability to borrow sufficient funds at reasonable rates (10) inability of the developers of the "Wonderful World of Oz" theme park to obtain final transfer of the property, to complete the timely construction of the park, and to operate it profitably once the park opens; (11) uncertainties relating to business and economic conditions; (12) management evaluation of staffing levels.

ITEM 7A. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK.

Interest Rate Risk. Our earnings are affected by changes in short-term interest rates as a result of our revolving credit agreements, which bear interest at a floating rate. We do not use derivative or other financial instruments to mitigate the interest rate risk or for trading purposes. Risk can be estimated by measuring the impact of a near-term adverse movement of 100 basis points in short-term market interest rates. If short-term market interest rates average 100 basis points more in the next 12 months, the adverse impact on our results of operations would be approximately \$194,000, net of income tax benefit. We do not anticipate any material near-term future earnings or cash flow expenses from changes in interest rates related to our long-term debt obligations as almost all of our long-term debt obligations have fixed rates.

Foreign Currency Risk. Although we conduct business in foreign countries, principally Canada and Australia, foreign currency translation gains and losses are not material to our consolidated financial position, results of operation or cash flows. Accordingly, we are not currently subject to material foreign currency exchange rate risks from the effects that exchange rate movements of foreign currencies would have on our future costs or on future cash flows we would receive from our foreign investment. To date, we have not entered into any foreign currency forward exchange contracts or other derivative financial instruments for trading purposes or to hedge the effects of adverse fluctuations in foreign currency exchange rates.

Investment Risk. We invest in equity instruments of privately-held companies in the Internet information technology and entertainment areas for business and strategic purposes. These investments are included in long-term assets, and are accounted for under the cost method or the equity method. For these non-quoted investments, our policy is to regularly review the assumptions underlying the operating performance and cash flow forecasts in assessing the carrying values. We identify and record impairment losses on these investments when events and circumstances indicate that such assets are permanently impaired. To date, no such impairment has been recorded.

We are also exposed to equity price risk on our investments in publicly traded companies. Our available-for-sale securities at September 30, 2000 include our equity positions in Navidec, Inc. and eCollege.com, which have experienced significant volatility in their stock prices. We do not attempt to reduce or eliminate our market exposure on these securities. A 20% adverse change in equity price would result in an approximate \$810,000 decrease in fair value in our available-for-sale securities, based upon January 9, 2001 closing market prices for Navidec and eCollege.com.

ITEM 8. FINANCIAL STATEMENTS AND SUPPLEMENTARY DATA.

See Consolidated Financial Statements and Notes to Consolidated Financial Statements attached for financial statements and supplementary data.

ITEM 9. CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE.

None.

PART III

ITEM 10. DIRECTORS AND EXECUTIVE OFFICERS OF THE REGISTRANT.

Steve Toth, Jr., age 76, became President and Chief Executive Officer of the Company in April 1997 and has been a Board member since March 1994. Toth has served as President of subsidiary Visual Services, Inc. since 1962. He also serves as President of subsidiaries Vispac, Inc. and Advanced Animations, Inc.

Martin S. Suchik, age 55, is a director of the Company since 1997. Suchik is the President at the former subsidiary BKNT Retail Stores, Inc., where he has served in that capacity since its 1981 founding. Suchik is the nephew of Toth.

Thomas W. Marquis, age 56, has been the Chief Financial Officer of subsidiary Visual Services, Inc. since 1988. He became Treasurer and Chief Financial and Accounting Officer of the Company in April 1997, became Secretary of the Company in June 1998, and has been a Board member since March 1994. Marquis serves as Senior Vice President, Secretary and Treasurer of subsidiaries Visual Services, Inc., Vispac, Inc. and Advanced Animations, Inc.

Harold A. Poling, age 75, has been a Board member since May, 2000. He was Chairman of the Board and Chief Executive Officer of Ford Motor Company. He retired January 1, 1994. Recently, he was appointed Chairman of Eclipse Aviation Corporation. He is a Director of Shell Oil Company, Flint Ink Corporation, Kellogg Company, Meritor Automotive, Inc. and Thermadyne Holdings Corporation. He is also a board consultant for The LTV Corporation, and is on the Investment Banking Advisory Board of Donaldson, Lufkin & Jenrette.

Ralph Armijo, age 47, has been a Board member since May, 2000. He has served as the President, Chief Executive Officer and as a Director of Navidec since its inception in 1993. He serves as Chairman of the Board of DriveOff.com, Inc., and is a Director for the Internet Automobile Dealers Marketing Association.

William James, age 66, has been a Board member since May, 2000. He has been President and Chief Executive Officer of James Cable Partners, L.P., a privately held Delaware limited partnership and a provider of cable television in nine states, since its formation in 1988. Prior to that, he was an Executive Vice President with Capital Cities Communications (which became Capital Cities/ABC, Inc.).

Robert Sui, age 47, became a Company director in 1998. He is a Senior Vice President with Merrill Lynch's Private Client Group, where he has served for the last 22 years.

Steve Schultz, age 44, is Executive Vice President of Sales for subsidiary Visual Services, Inc., where he has served since 1994.

Terry Davis, age 61, is Executive Vice President of Strategic Planning for subsidiary Visual Services, Inc., where he has served since 1982.

In the last year, the Board held three meetings, which all directors attended. Toth, Marquis, and Sui comprise the Executive Committee; James, Sui, and Poling comprise the Audit Committee; Poling, Armijo, and Marquis comprise the Compensation Committee. None of the Committees met last year.

Each non-executive officer director receives a \$15,000 annual fee, plus a \$5,000 meeting fee for meetings in excess of three per year, with no additional payment for membership on or meetings of any committees, except pursuant to the Independent Director Stock Option Plan (see stock ownership table). Except for Toth and Suchik, no officer or director is related to another by blood, marriage or adoption, not more remote than first cousin. In the last year, Forms 4 were filed by Toth (6), and by Suchik (5). All Forms 4 were filed on a timely basis. Except as noted, all directors are believed to have filed this year's annual Forms 5 on a substantially timely basis; James filed several days late, and we have been unable to ascertain whether Suchik has filed his Form 5.

Officers serve at the discretion of the Board of Directors. All Directors are elected at each annual meeting of shareholders.

ITEM 11. EXECUTIVE COMPENSATION.

		ANNUAL	COMPENSATION	N	LONG TERM COMPENSATION AWARDS
(A)	(B)	(C)	(D)	(E)	(F)
NAME AND PRINCIPAL POSITION	YEAR	SALARY (\$)	BONUS (\$)	OTHER ANNUAL COMPENSATION (\$)	RESTRICTED STOCK AWARDS (\$)
Steve Toth Jr	2000	312,000	Θ	0	
CEO	1999	312,000	0	1,920	
323	1998	642,000	Õ	950	
Thomas W. Marquis Treasurer,	2000	170,000	0	0	
Secretary	1999	150,000	Θ	1,800	
	1998	150,000	0	885	
Steve Schultz	2000	220,000	0	0	
Exec V P Sales	1999	206,660	0	1,920	
	1998	206,660	0	950	\$566,234
Terry Davis	2000	300,000	0	0	
Exec V P	1999	300,000	0	1,920	
	1998	300,000	0	950	

None of the executive officers have contractual compensation agreements. The "other" amounts listed are the Company's 401(k) contributions. Other than noted, none of the above officers received long term compensation payouts or awards covered by the table above.

In fiscal year 1998, Steve Schultz was awarded 88,821 shares of restricted stock, of which 29,607 shares were issued in December, 1998, 1999 and 2000. Officers may participate in the Company's 401(k) and stock option plans. Such stock option plans provide that the price of any common stock issued to officers, directors, employees and their affiliates pursuant to any stock grant or exercise of any stock option shall be no less than the fair market value of the Common Stock on the date of the stock or option grant.

SEC rules require a performance graph comparing, over a five-year period, the performance of our common stock against a broad market index and against either a published industry or line-of-business index or a group of peer issuers or issuers with similar market capitalization. We chose the Wilshire Small Cap Index as the broad market index, and the Russell 2000 Index because the companies covered by that index include many of the companies in the Marketing Services industry. The performance graph assumes an initial investment of \$100 and reinvestment of all dividends.

[PERFORMANCE GRAPH]

CHART POINTS	09/30/95	09/30/96	09/30/97	09/30/98	09/30/99	09/30/00	
VSI Holdings, Inc.	100.00	77.78	1,055.56	911.11	788.89	655.56	
Wilshire Small Cap	100.00	111.45	157.62	123.72	162.39	216.42	
Russell 2000	100.00	111.60	146.21	117.14	137.67	167.98	

ITEM 12. SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT.

Voting rights are held by the owners of the Common Stock, of which each share is entitled to one vote on each matter coming before the shareholders. Only one class of Common Stock is authorized; none of the authorized Preferred Stock has been issued. On December 1, 2000, the Company had 33,179,825 shares of Common Stock (net of 400,250 treasury shares) outstanding. Such shares, and shares issuable under options exercisable within 60 days, were owned by:

NAME AND ADDRESS	SHARES OWNED	% OWNERSHIP
Steve Toth, Jr. and family members(1)	27,905,755	84.10%
Bloomfield Hills, Michigan 48304		
Martin S. Suchik(2)	504,025	1.52%
Atlanta, Georgia 30318		
Thomas W. Marquis(3)	514,431	1.55%
41000 Woodward Avenue		
Bloomfield Hills, Michigan 48304		

NAME AND ADDRESS	SHARES OWNED	% OWNERSHIF
Steve Schultz41000 Woodward Avenue	97,192	. 29%
Bloomfield Hills, Michigan 48304 Terry Davis41000 Woodward Avenue	918,906	2.77%
Bloomfield Hills, Michigan 48304 Harold Poling(4)	309,600	0.93%
Dearborn, Michigan 48126 Ralph Armijo(4)	2,100	. 01%
Greenwood Village, Colorado 80111 Robert F. Sui (4)	4,600	. 01%
Bloomfield Hills, Michigan 48304 William James(4)	7,100	. 02%
Bloomfield Hills, Michigan 48304 All directors and officers as a group (9 persons) (1-4)	30,263,709	90.57%

- -----

- (1) Toth is trustee of trusts benefiting him that own 400,000 and 11,826,323 shares. Toth's spouse is the trustee of trusts benefiting Toth's daughter that owns 11,712,199 shares, and of another trust benefiting her that owns 1,010,797 shares. Toth disclaims beneficial interest in the shares held by trusts benefiting his daughter and spouse, but such shares are included with his holdings. CLT, a Michigan partnership affiliated with Toth ("CLT"), holds 2,055,586 shares. 900,850 shares are held by a charitable foundation, of which Toth is executive director.
- (2) Suchik owns 422,392 shares, while his IRA owns another 11,113 shares. Suchik is fully vested in two options for 40,000 and 30,520 shares under the 1986 Incentive and Non-Qualified Plans exercisable at \$.55 and \$.50 per share, respectively; these options expire on January 15, 2001.
- (3) Marquis is trustee of a trust benefiting him that owns 261,567 shares. His spouse is trustee of a trust benefiting her that owns 252,864 shares. Marquis disclaims beneficial ownership of his spouse's shares.
- (4) Poling owns 250,000 shares and was granted 15,000 non-employee options exercisable at \$6.20 per share in November, 1997, and 85,000 non-employee options in November, 1998, with 50,000 exercisable at \$5.75 per share and 35,000 exercisable at \$8.20 per share. James was granted 5,000 non-employee options exercisable at \$6.20 per share in December, 1997. Under the Independent Director Stock Option Plan, Poling, James, Armijo and Sui were granted new 30,000 share options in May, 2000 exercisable for \$3.25 per share; such options vest at the rate of 2,100 shares for each Board meeting attended, with a minimum of 7,500 annually. Sui's appointment as a director was confirmed by the shareholders on April 8, 1998 at which time he received a 10,000 share option exercisable for \$6.375 per share; of which 2,500 have vested, and 7,500 have been canceled. Poling, James, Armijo and Sui are vested in 59,600, 7,100, 2,100 and 4,600 shares respectively.

ITEM 13. CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS.

Steve Toth, Jr. (Toth), President, with his family, directly and indirectly owns 84.10% of the Company as of December 1, 2000.

Loans to and from Toth Family. At September 30, 2000 and September 30, 1999, we had advanced \$804,000 and \$686,000, respectively, to Toth, which amounts are non-interest bearing and unsecured. In addition, a demand note in the amount of \$500,000 which bears interest at 7.00% and is unsecured, was borrowed from Toth, and is outstanding at September 30, 2000. No restrictive covenants were imposed.

Note Payable to CLT Associates, L.P. (CLT). CLT is a partnership which is controlled by Toth. During the year, we borrowed \$4 million from CLT at 7.00%. The note is unsecured, and imposes no restrictive covenants. The balance at September 30, 2000 was \$2,920,000.

CLT Stock Option. The 1993 reorganization plan of the former Retail segment provided that Toth be issued stock options for 1,600,000 shares. Toth assigned an 825,000 share option to CLT, which exercised the last 425,000 shares in November, 1999 at \$.15625 per share.

CLT has invested \$7.63 million in Oz Entertainment Corporation and KC Investors, LP. These are the organizations which are developing the Wonderful World of Oz theme park which is scheduled to open in 2004. Our investment, combined with CLT totaled was \$12.13 million at September 30, 2000. Together, these investments account for direct and indirect ownership of approximately 30% of Oz Entertainment Corporation. See Note 3 of Notes to Consolidated Financial Statements.

Declared Distributions to Stockholders. Prior to their acquisition, in February, July and September 1997, the income of each subsidiary (Advanced Animations, Inc., Vispac, Inc. and Visual Services, Inc.) was taxed to their respective stockholders and members. A portion of such income was distributed to pay such taxes. At September 30, 1997, we had declared, and owed, distributions of \$20,659,000 to such stockholders and members. Such distributions relate not only to income earned by such subsidiaries from October 1, 1996 until their acquisition, but also include income previously retained by such subsidiaries as necessary working capital. \$11,494,000 of the \$20,659,000 was converted into a subordinated note payable from us to these former Stockholders, bearing interest at 7.00%, due December, 2002, and subordinated to all bank debt. \$7,553,000 of this subordinated note belongs to Toth. Interest of \$529,000 owing to Toth was paid during the fiscal year.

Lease of Real Estate. Toth directly and indirectly owns the entire interest in a partnership (CLT) that owns the building in which Vispac is based. The lease is for \$551,000 per year until November, 2001 and \$638,000 thereafter until it expires in September 2006. The rental charges do not exceed those ordinarily and customarily paid in the community.

Contingent liability for discontinued operations. We are the guarantor of several store leases being used by the BKNT Retail Stores, Inc. At September 30, 2000 the lease obligations amounted to \$435,000. This amount will reduce to \$113,000 in the fiscal year ending 2001; \$25,000 in the fiscal year ending 2002. This contingent liability is offset by the pledging of 216,250 Company shares by Martin Suchik.

VSI HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED FINANCIAL REPORT SEPTEMBER 30, 2000

CONTENTS

REPORT LETTER	1
CONSOLIDATED FINANCIAL STATEMENTS	
Balance Sheet	2
Statement of Operations	3-4
Statement of Changes in Stockholders' Equity	5
Statement of Cash Flows	6
Notes to Consolidated Financial Statements	7-29

INDEPENDENT AUDITOR'S REPORT

TO THE BOARD OF DIRECTORS AND STOCKHOLDERS VSI HOLDINGS, INC. AND SUBSIDIARIES

We have audited the accompanying consolidated balance sheet of VSI Holdings, Inc. and subsidiaries as of September 30, 2000 and 1999 and the related consolidated statements of operations, changes in stockholders' equity and cash flows for each year in the three-year period ended September 30, 2000. These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits.

We conducted our audits in accordance with generally accepted auditing standards. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the consolidated financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the consolidated financial statements referred to above present fairly, in all material respects, the consolidated financial position of VSI Holdings, Inc. and subsidiaries at September 30, 2000 and 1999 and the consolidated results of their operations and their cash flows for each year in the three-year period ended September 30, 2000, in conformity with generally accepted accounting principles.

PLANTE & MORAN, LLP

Ann Arbor, Michigan December 22, 2000

VSI HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEET

	SEPTEME	
	2000	1999
ASSETS CURRENT ASSETS		
Cash	\$ 905,000	\$ 552,000 2,930,000
Billed	53,907,000 18,525,000 201,000 438,000 3,744,000 899,000 761,000	39,820,000 9,548,000 92,000 410,000 4,883,000 207,000 1,451,000 5,259,000 609,000
Total current assets NOTES RECEIVABLE Officers PROPERTY, PLANT AND EQUIPMENT Net (Note 4) DEFERRED TAX ASSET (Note 8) GOODWILL INVESTMENT IN AVAILABLE-FOR-SALE SECURITIES (Note 3) INVESTMENTS (Note 3)	79,380,000 921,000 22,394,000 830,000 1,283,000 7,131,000 4,190,000	65,761,000 777,000 21,575,000 715,000 1,710,000 7,764,000
Total assets	\$116,129,000 =======	\$98,302,000 ======
LIABILITIES AND STOCKHOLDERS' EQUIT		
CURRENT LIABILITIES Current portion of long-term debt (Note 6)	\$ 552,000 3,420,000 23,479,000 35,214,000 3,970,000 2,139,000 375,000	\$ 516,000 18,452,000 28,426,000 3,318,000 5,975,000
Total current liabilities	69,149,000 12,337,000 7,092,000	56,687,000 11,636,000 7,660,000
shares authorized, no shares issued	336,000 8,071,000 20,270,000 730,000 (1,856,000)	330,000 6,949,000 14,720,000 1,855,000 (1,535,000)
Total stockholders' equity	27,551,000	22,319,000
Total liabilities and stockholders' equity	\$116,129,000 =======	\$98,302,000 ======
		

See Accompanying Summary of Accounting Policies and Notes to Consolidated Financial Statements 25

VSI HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF OPERATIONS

YEAR ENDED SEPTEMBER 30 2000 1999 1998 -----REVENUE..... \$187,255,000 \$143,360,000 \$163,158,000 **EXPENSES** Cost of revenue..... 82,471,000 57,105,000 71,861,000 Operating expenses..... 90,737,000 82,901,000 74,810,000 Total expenses..... 173,208,000 140,006,000 146,671,000 3,354,000 OPERATING INCOME..... 16,487,000 14,047,000 OTHER INCOME (EXPENSES) Equity in losses of unconsolidated investee (Note 3)..... (1,515,000)(400,000)(516,000) Interest and other income (expense)..... 69,000 (37,000)168,000 Gain (loss) on sale of assets..... (32,000)(3,005,000) Interest expense..... (3,708,000)(2,320,000)Total other expenses..... (4,986,000)(3,921,000)(2,389,000)INCOME (LOSS) -- Before income taxes..... 9,061,000 (567,000)14,098,000 PROVISION FOR (BENEFIT FROM) INCOME TAXES (Note 8)..... 3,511,000 (69,000) 4,887,000 INCOME (LOSS) FROM CONTINUING OPERATIONS..... 5,550,000 (498,000) 9,211,000 DISCONTINUED OPERATIONS (Note 14) Loss from discontinued operations -- Net of income tax benefit of \$220,000..... (428,000)Gain on disposal of subsidiary -- Net of income tax of \$140,000..... 271,000 Total discontinued operations..... (157,000) 5,550,000 (498,000) NET INCOME (LOSS)..... 9,054,000 OTHER COMPREHENSIVE INCOME (EXPENSE) -- Net of tax Foreign currency translation adjustment..... (17,000)24,000 (23,000)(1,108,000)Unrealized gain (loss) on securities (Note 3).... 1,854,000 --Total other comprehensive income (expense)...... 1,878,000 (1,125,000)(23,000) COMPREHENSIVE INCOME..... \$ 1,380,000 \$ 4,425,000 \$ 9,031,000 ========= ========= ========= EARNINGS PER SHARE Basic: Income (loss) from continuing operations..... 0.17 (0.02)0.28 Loss from discontinued operations..... (0.01)Gain on disposal of subsidiary..... 0.01 \$ 0.28 Basic net income (loss) per share..... \$ 0.17 \$ (0.02) ========= ========= ========= Fully diluted: Income (loss) from continuing operations..... 0.17 (0.02)0.27 Loss from discontinued operations..... (0.01)Gain on disposal of subsidiary..... 0.01 \$ 0.17 Fully diluted net income (loss) per share..... \$ (0.02) \$ 0.27 ========= WEIGHTED AVERAGE SHARES OUTSTANDING 33,070,000 Basic..... 32,816,000 32,851,000 Effect of stock options..... 262,000 682,000

See Accompanying Summary of Accounting Policies and Notes to Consolidated Financial Statements

Fully diluted.....

33,332,000

=========

32,816,000

=========

33.533.000

=========

VSI HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CHANGES IN STOCKHOLDERS' EQUITY

	COMMON STOCK ACCUMULATED OTHER COMPREHEN			COMPREHENSIVE I	NSIVE INCOME		
	SHARES	AMOUNT	ADDITIONAL PAID-IN CAPITAL	RETAINED EARNINGS	TREASURY STOCK	STOCK SUBSCRIPTIONS RECEIVABLE	
BALANCE October 1, 1997	40,371,000	\$404,000	\$ 7,917,000	\$ 6,253,000	\$(2,907,000)	\$	
Net income Exercise of stock options (Note				9,054,000			
12) Issuance of stock options (Note	64,000		45,000				
12)Acquisition of stock for			83,000				
treasury				 (89,000)	(736,000)		
Distributions to stockholders Foreign currency translation				(89,000)			
adjustmentsStock issued in acquisition (Note							
1) Issuance of stock	280,000	3,000	162 000			(25, 000)	
	26,000		163,000			(25,000) 	
BALANCE September 30, 1998 Net loss	40,741,000	407,000 	8,208,000	15,218,000 (498,000)	(3,643,000)	(25,000) 	
Exercise of stock options (Note 12)	2,000		1,000				
Issuance of restricted stock (Note	,		•				
11) Acquisition of stock for	154,000	2,000	985,000				
treasury Foreign currency translation					(798,000)		
adjustments							
Issuance of stock Retirement of treasury stock	87,000 (8,024,000)	1,000 (80,000)	417,000 (2,826,000)		2,906,000		
Repayment of stock subscriptions	(8,024,000)	(80,000)	(2,020,000)		2,900,000	25,000	
Tax benefit of stock options						20,000	
exercised Issuance of stock options (Note			92,000				
12)			72,000				
Unrealized gain on investments							
BALANCE September 30, 1999	32,960,000	330,000	6,949,000	14,720,000	(1,535,000)		
Net income Exercise of stock options (Note		,		5,550,000			
12)Issuance of restricted stock (Note	440,000	4,000	70,000				
11)Acquisition of stock for	175,000	2,000	1,037,000				
treasury Foreign currency translation					(321,000)		
adjustments							
Issuance of stock	5,000		15,000				
Unrealized loss on investments							
BALANCE September 30, 2000	33,580,000	\$336,000	\$ 8,071,000 ======	\$20,270,000	\$(1,856,000) =======	\$ =======	

ACCUMULATED OTHER COMPREHENSIVE INCOME

	FOREIGN CURRENCY UNREALIZED GAIN TRANSLATION (LOSS) ON ADJUSTMENTS INVESTMENTS		TOTAL STOCKHOLDERS' EQUITY
BALANCE October 1, 1997	\$	\$	\$11,667,000 9,054,000
Exercise of stock options (Note 12)			45,000
12)			83,000
treasury			(736,000)
Distributions to stockholders Foreign currency translation			(89,000)
adjustmentsStock issued in acquisition (Note	(23,000)		(23,000)
1)			3,000
Issuance of stock			138,000
BALANCE September 30, 1998	(23,000)		20,142,000
Net loss Exercise of stock options (Note	'		(498,000)
12)Issuance of restricted stock (Note			1,000

11)			987,000
Acquisition of stock for			
treasury			(798,000)
Foreign currency translation			
adjustments	24,000		24,000
Issuance of stock			418,000
Retirement of treasury stock			
Repayment of stock subscriptions			25,000
Tax benefit of stock options			,
exercised			92,000
Issuance of stock options (Note			,
12)			72,000
Unrealized gain on investments		1,854,000	1,854,000
om carred garn on investments in in			
BALANCE September 30, 1999	1,000	1,854,000	22,319,000
Net income			5,550,000
Exercise of stock options (Note			0,000,000
12)			74,000
Issuance of restricted stock (Note			14,000
and the second s			1,039,000
11) Acquisition of stock for			1,039,000
•			(221 000)
treasury			(321,000)
Foreign currency translation	(47 000)		(47,000)
adjustments	(17,000)		(17,000)
Issuance of stock			15,000
Unrealized loss on investments		(1,108,000)	(1,108,000)
BALANCE September 30, 2000	\$(16,000)	\$ 746,000	\$27,551,000
	======	========	========

See Accompanying Summary of Accounting Policies and Notes to Consolidated Financial Statements $\,$

VSI HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENT OF CASH FLOWS

	YEAR		
	2000	1999	1998
CASH FLOWS FROM OPERATING ACTIVITIES			
Net income (loss)	\$ 5,550,000	\$ (498,000)	\$ 9,054,000
cash from operating activities: Depreciation and amortization	5,817,000	5,669,000	4,865,000
Write-off of goodwill Bad Debt expense	526,000	2,154,000	
Equity in losses of unconsolidated investee Noncash retirement plan contribution	1,515,000 	400,000 229,000	295,000
Noncash proceeds from sale of subsidiary Deferred income taxes	(237,000)	(347,000)	(736,000) 244,000
(Gain) Loss on sale of property, plant and equipment Issuance of stock options	(168,000)	 72,000	32,000 83,000
Write-off of assets	516,000	421,000	
Trade accounts receivable	(24,090,000) (28,000)	198,000 (1,000)	(12,881,000) 381,000
Accumulated costs of uncompleted programs Other current assets	1,139,000 (152,000)	(1,663,000) 556,000	(555,000) 2,361,000
Prepaid/accrued federal income tax Trade accounts payable	3,590,000 5,027,000	(5,921,000) 6,526,000	4,562,000 2,000,000
Accrued liabilities Advances from customers for uncompleted	1,691,000	488,000	1,607,000
projects	(2,670,000)	800,000	1,177,000
Net cash provided by (used in) operating activities	(1,974,000)	9,083,000	12,489,000
Additions to property, plant and equipment Proceeds from sale of property, plant and	(7,165,000)	(3,061,000)	(13,506,000)
equipment Additions to notes receivable and advances	608,000 (253,000)	(146,000)	2,000 (1,138,000)
Payments on notes receivable and advances Investments in affiliates		1,200,000	`9,708,000´ (2,925,000)
Investment in available-for-sale securities Purchase of other investments	(50,000) (941,000)	(2,450,000) (7,150,000)	(307,000)
Net cash used in investing activities CASH FLOWS FROM FINANCING ACTIVITIES	(7,801,000)	(11,607,000)	(8,166,000)
Principal payments on long-term debt	(532,000)	(918,000) 2,621,000	(164,000) 3,580,000
Principal payments on related party debt Proceeds from related party debt	(1,080,000) 5,201,000	(18,000) 160,000	11,055,000
Net borrowings on notes payable Proceeds from issuance of stock	6,788,000 89,000	3,287,000 190,000	1,994,000 211,000
Distributions to stockholders Acquisition of treasury stock	(321,000)	(798,000)	(20,748,000)
Proceeds from payment of stock subscription Redemption of common stock		25,000 (1,960,000)	
Net cash provided by (used in) financing			
activities EFFECT OF EXCHANGE RATE CHANGES ON CASH	10,145,000 (17,000)	2,589,000 24,000	(4,072,000) (23,000)
NET INCREASE IN CASH	353,000 552,000	89,000 463,000	228,000 235,000

CASH -- End of year.....

\$ 905,000

==========

\$ 463,000

=========

\$ 552,000

==========

VSI HOLDINGS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS SEPTEMBER 30, 2000 AND 1999

NOTE 1 -- ORGANIZATION OF THE COMPANY AND BASIS OF PRESENTATION

The accompanying consolidated financial statements include the accounts of VSI Holdings, Inc. (the "Company"), its wholly owned subsidiaries, consisting of Advanced Animations, Inc., Vispac, Inc., Visual Services, Inc., PSG International, Inc. and its majority owned subsidiary, eCity Studios, Inc. Intercompany balances and transactions have been eliminated in consolidation.

Advanced Animations, Inc. designs and manufactures product simulators and animatronic displays. Customers are primarily from the retail and entertainment industry throughout the world.

Vispac, Inc. provides administrative and promotional services, warehousing and packaging operations and call center operations predominantly for North American automobile manufacturers.

Visual Services, Inc. is a broad-based provider of educational curriculums and product training, interactive technology-based Distance Learning Systems, product launches, web site development, direct response and site-based marketing, and change process and cultural change consulting. Customers are primarily North American automobile manufacturers.

During the year ended September 30, 2000, the Company, in connection with Oz Entertainment Co. (see Note 3), formed eCity Studios, Inc., a Corporation in which VSI Holdings, Inc. acquired a 70% interest in the stock. eCityStudios, Inc. has been included in the consolidated financial statements. eCity Studios, Inc. provides web-site development consulting to customers throughout North America.

PSG International, Inc. provides curriculum development and product training to automobile manufacturers at the dealership level throughout Canada, Australia, New Zealand and Taiwan. PSG International, Inc. was formed during the year ended September 30, 1998 when the Company completed the purchase of the assets of an unrelated company. The total purchase price was \$4,543,625 and resulted in goodwill of \$4,484,000. The Company issued 280,000 shares of common stock valued at \$7.00 per share, with the balance of \$2,583,625 paid in cash.

As part of the PSG International, Inc. purchase agreement, the shares issued to the seller were subject to a "put option" in which the seller had the option to sell the shares back to the Company in April 1999 for \$2,100,000 (\$7.50 per share). During the year ended September 30, 1999, the "put option" was exercised and the shares were subsequently retired to treasury stock.

NOTE 2 -- SIGNIFICANT ACCOUNTING POLICIES

REVENUE RECOGNITION -- Visual Services, Inc., Vispac, Inc., PSG International, Inc., and eCity Studios, Inc. recognize revenue over the period of the contract as individually identifiable phases of contracts are completed. Amounts recognized are accumulated in unbilled accounts receivable until billed in accordance with contract terms.

Advanced Animations, Inc. records revenue on display contracts of varying duration on the basis of the Company's estimates of the percentage of completion of individual contracts. A percentage of the contract price, determined by the ratio of incurred costs to total estimated costs, is included in revenue and the incurred costs are charged against this revenue. Revisions in cost and profit estimates during the course of the work are reflected in the accounting period in which the facts that require the revision become known. Billings are made in accordance with contract terms. At the time a loss on a contract becomes known, the entire amount of the estimated loss is accrued.

CASH IN ESCROW -- Certain amounts received from clients in advance had been restricted and held in escrow until costs related to a specific job are incurred by the Company. During the year ended September 30, 2000, the escrow arrangement was terminated.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

INVENTORY -- Inventory, which consists of raw materials and supplies, is recorded at the lower of cost, determined on the specific unit basis, or market.

ACCUMULATED COSTS OF UNCOMPLETED PROGRAMS -- Accumulated costs of uncompleted programs consist of costs accumulated on various service-related contracts of Visual Services, Inc. The accumulated costs are included as a current asset, as they consist of ongoing job costs, which will be recorded as cost of revenue against the appropriate revenue recognized within the next fiscal year. At September 30, 2000 and 1999, indirect contract costs amounting to \$733,000 and \$720,000, respectively, were capitalized and included in accumulated costs.

PROPERTY, PLANT AND EQUIPMENT -- Property, plant and equipment are recorded at cost. Depreciation is computed using accelerated and straight-line methods over the estimated useful lives of the assets. Amortization of leasehold improvements is computed on the straight-line method over the estimated useful lives of the assets or the lease term. Costs of maintenance and repairs are charged to expense when incurred. Production costs, representing the direct materials, labor, overhead costs and interest associated with self-constructed assets, are capitalized during construction. Once completed and placed in service, the assets are reclassified to equipment and depreciated over the appropriate depreciable lives.

GOODWILL -- Goodwill represents the excess cost of acquiring PSG International, Inc. over the fair value of its net assets at the date of acquisition. The carrying value of goodwill is evaluated annually based on the projected discounted cash flows of PSG International, Inc. To the extent the carrying value of goodwill significantly exceeds the projected discounted cash flows, a write-down of goodwill is recorded. During the year ended September 30, 1999, as a result of a reduction in the estimated future business generated by PSG International, Inc., goodwill of \$2,154,000 relating to the acquisition was written off because it was considered to have no continuing value. In addition, the Company reduced the life over which the remaining goodwill was being amortized from 15 to 6 years. This change in the estimated life of the goodwill had the effect of decreasing net income for the year ended September 30, 1999 by \$260,000 (\$0.01 per share). At September 30, 2000 and 1999, accumulated amortization amounted to \$1,047,000 and \$620,000, respectively.

Total amortization expense amounted to \$427,000, \$422,000 and \$198,000 for the years ended September 30, 2000, 1999 and 1998, respectively.

INCOME TAXES -- Deferred tax assets and liabilities are recognized based on the difference between financial statement carrying amounts and income tax bases of assets and liabilities using currently enacted income tax rates. Valuation allowances are established when necessary to reduce deferred tax assets to the amount expected to be realized. Income tax expense is the tax payable or refundable for the period plus or minus the change during the year in deferred tax assets and liabilities.

RETIREMENT PLAN -- The Company has a voluntary retirement savings plan designed in accordance with Section 401(k) of the Internal Revenue Code that covers all eligible employees. Employer contributions are discretionary and determined annually by management. There was no contribution made for the year ended September 30, 2000. Employer contributions amounted to \$229,000 of the Company's stock and \$212,000 of cash for the years ended September 30, 1999 and 1998, respectively.

EARNINGS PER SHARE -- Earnings per share is computed using the weighted average number of shares outstanding. Basic net income per share is computed by dividing net income by the weighted average number of common shares outstanding during the year, without regard to stock options and restricted stock awards outstanding. In the computation of fully diluted earnings per share, the treasury stock method of determining weighted average shares is required, which assumes the exercise of existing stock options and awards and the repurchase of shares with the proceeds, if such a computation has a dilutive effect.

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

STOCK OPTIONS -- The Company has several stock option plans (see Note 12). Options granted to nonemployees are accounted for at fair value. Options granted to employees and directors are accounted for using the intrinsic value method, under which compensation expense is recorded at the amount by which the market price of the underlying stock at the grant date exceeds the exercise price of an option. Under the Company's plans, the exercise price on all options granted equals or exceeds the fair value of the stock at the grant date. Accordingly, no compensation cost is recorded as a result of stock option awards to employees under the plans.

IMPAIRMENT OF LONG-LIVED ASSETS -- The Company reviews long-lived assets for impairment when circumstances indicate the carrying amount of an asset may not be recoverable. An impairment is recognized when the sum of undiscounted estimated future cash flows expected to result from the use of the asset is less than the carrying value.

INVESTMENTS -- The Company's securities investments that are bought and held principally for the purpose of selling them in the near term are classified as trading securities. Trading securities are recorded at fair value on the balance sheet in current assets, with the change in fair value during the period included in earnings.

Debt securities that the Company has the positive intent and ability to hold to maturity are classified as held-to-maturity securities and recorded at amortized cost in investments. Securities investments not classified as either held-to-maturity or trading securities are classified as available-for-sale securities. Available-for-sale securities are recorded at fair value in investments on the balance sheet, with the change in fair value during the period excluded from earnings and recorded net of tax as a component of other comprehensive income.

The Company's investment in nonmarketable securities is stated at cost.

INVESTMENT IN PARTNERSHIPS -- Investments in partnerships are accounted for using the equity method.

COMPREHENSIVE INCOME -- Accounting principles generally require that recognized revenue, expenses, gains and losses be included in net income. Certain changes in assets and liabilities, however, such as unrealized gains and losses on available-for-sale securities and foreign currency translation adjustments, are reported as a direct adjustment to the equity section of the balance sheet. Such items, along with net income, are considered components of comprehensive income.

FOREIGN CURRENCY TRANSLATION -- The assets and liabilities denominated in a foreign currency are translated into U.S. dollars at the current rate of exchange on the last day of the reporting period, revenue and expenses are translated at the average monthly exchange rates, and all other equity transactions are translated using the actual rate on the date of the transaction.

USE OF ESTIMATES -- The preparation of financial statements in conformity with generally accepted accounting principles requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of revenue and expenses during the reporting period. For example, the Company recognizes certain revenue using the percentage of completion method, which requires the use of significant estimates. Actual results could differ from those estimates.

FUTURE ACCOUNTING CHANGES -- In December 1999, the Securities and Exchange Commission issued Staff Accounting Bulletin 101. The staff accounting bulletin summarizes certain of the staff's views in applying generally accepted accounting principles to revenue recognition in financial statements. Any changes that may be required as a result of this bulletin are required to be implemented no later than the last fiscal quarter of the fiscal year beginning October 1, 2000. Management is finalizing its assessment of the impact this bulletin will have, if any, on the Company's financial statements.

VSI HOLDINGS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

RECLASSIFICATIONS -- The Company has made certain reclassifications to the statement of operations. These reclassifications had no impact on net income, earnings per share or stockholders' equity.

NOTE 3 -- INVESTMENTS

The available-for-sale securities consist of the following:

	2000	1999
431,525 shares of Navidec, Inc. common stock (representing a less than 5 percent ownership interest)	\$3,344,000	\$5,259,000
less than 5 percent ownership interest)	3,787,000	
Total investment in available-for-sale securities	\$7,131,000 ======	\$5,259,000 ======

During the year ended September 30, 2000, upon a public offering by eCollege.com, the Company converted its nonmarketable investment in eCollege.com into 468,808 shares of common stock. Upon the public offering, the Company purchased an additional 4,500 shares of common stock.

Unrealized gains on the available-for-sale securities are included as a separate component of stockholders' equity, net of any related tax effect. At September 30, 2000 and 1999, gross unrealized holding gains on available-for-sale securities amounted to \$1,131,000, and \$2,809,000, respectively. As of December 22, 2000, the market value of the available-for-sale securities was approximately \$2,737,000, and the gross unrealized holding loss was approximately \$3,273,000.

The Company's investment in partnerships and nonmarketable securities consists of the following:

	2000	1999
KC Investors, L.P	\$2,812,000	\$3,470,000
Corporate Eagle Five, L.L.C	540,000	762,000
Visual Learning Systems, L.L.C	(124,000)	32,000
Performance Planning, L.L.C	462,000	
Total investment in partnerships	3,690,000	4,264,000
Nonmarketable equity securities	500,000	3,500,000
Total investment in partnership and nonmarketable		
securities	\$4,190,000	\$7,764,000
	========	========

The investment in partnerships consists of several partnership investments that have been accounted for under the equity method. The investment in KC Investors, L.P. (KCI) represents an approximate 15 percent interest in the partnership. KCI was formed as an investment partnership to fund the continuing development of the Oz Entertainment Company, a corporation in which KCI holds a majority interest. The Oz Entertainment Company is a development stage company engaged in the early stages of development of a theme park and resort in Kansas based on the 1939 MGM film, "The Wizard of Oz".

In addition to the cash invested, the Company has provided web-site development and promotional services totaling \$2,247,000 to the Oz Entertainment Company. At September 30, 1999, the Company had receivables of approximately \$1,410,000 and accumulated costs on uncompleted programs of approximately \$147,000 relating to KCI and the Oz Entertainment Company. The amounts were collected subsequent to September 30, 2000 from additional funds invested by CLT Associates, L.P.

VSI HOLDINGS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

CLT Associates, L.P. (CLT), a partnership controlled by the majority stockholder of the Company, owns an additional 6 percent interest in KC Investors, L.P. and a direct interest in preferred stock of the Oz Entertainment Company.

The investment in nonmarketable securities at September 30, 2000 consists of 1,342 shares of senior preferred stock in Oz Entertainment Company, a corporation in which KCI (a partnership investment of the Company) holds a majority interest. The investment in nonmarketable securities at September 30, 1999 consisted of 100,014 shares of convertible preferred stock in eCollege.com.

NOTE 4 -- PROPERTY, PLANT AND EQUIPMENT

Property, plant and equipment consist of the following:

	2000	1999	DEPRECIABLE LIFE - YEARS
Land and land improvements	\$ 1,286,000	\$ 1,286,000	
Building	8,171,000	8,171,000	18-39
Furniture, fixtures and equipment	38,547,000	35,539,000	5-10
Leasehold improvements	3,810,000	3,808,000	18-39
Production costs		1,250,000	
Vehicles	960,000	2,113,000	5
Total Less accumulated depreciation and	52,774,000	52,167,000	
amortization	30,380,000	30,592,000	
Net carrying amount	\$22,394,000	\$21,575,000	
	========	========	

Depreciation expense amounted to \$5,390,000, \$5,668,000 and \$4,666,000 for the years ended September 30, 2000, 1999 and 1998, respectively.

During the year ended September 30, 2000, a touring exhibit was deemed to be impaired and written down to fair value. Fair value was determined by estimating future cash inflows from the exhibit. The Company estimated that there would be no future net cash flows and recorded an impairment loss of \$516,000 (included in operating expenses) for the year ended September 30, 2000.

During the year ended September 30, 1999, several customized trailers with electronic equipment, along with specialized tents used for specific jobs, were deemed to be impaired and written down to their fair value. Fair value, which was determined by estimating the future cash inflows of such assets, exceeded their carrying value by \$421,000. An impairment loss of that amount (included in operating expenses) has been charged to operations for the year ended September 30, 1999.

NOTE 5 -- NOTES PAYABLE

The Company has several lines of credit, as follows:

	_000	
Bank line of credit permitting borrowings up to \$32,000,000 at .50 percent below the bank's prime rate (9.50 percent at September 30, 2000). Borrowings equal to or greater than \$500,000 can be made for fixed periods of time at a fixed rate equal to LIBOR plus 1.50 percent (LIBOR at September 30, 2000 was 6.62 percent). Collateralized by all assets of the Company and expiring in March 2001	\$24,434,000	\$22,413,000

2000

1999

	2000	1999
Bank line of credit permitting borrowings up to \$5,000,000 at .50 percent below the bank's prime rate (9.50 percent at September 30, 2000). Borrowings equal to or greater than \$500,000 can be made for fixed periods of time at a fixed rate equal to LIBOR plus 1.50 percent (LIBOR at September 30, 2000 was 6.62 percent) Collateralized by		
all assets of the Company and expiring in March 2002 Checks written but not yet presented to the bank. Upon presentation to the bank, additional borrowings will be made on the line of credit. The Company policy is to reflect these checks as additional amounts payable to the	4,900,000	3,750,000
bank	5,880,000	2,263,000
Total notes payable to bank	\$35,214,000 ======	\$28,426,000

The loan agreements contain certain covenants requiring that, among other things, the Company maintain certain levels of net worth and working capital and that the ratio of total liabilities to net worth, debt service ratio and current ratio do not exceed certain amounts.

Notes payable to related parties consists of the following:

	2000	1999
Notes payable to stockholders of VSI Holdings, Inc., unsecured, bearing interest at 7.00 percent and due December 31, 2002. The notes are subordinated to all bank debt	\$12,337,000	\$11,636,000
Note payable to a partnership in which the Company's controlling stockholder owns 100 percent through direct and indirect ownership. The unsecured note is due on		
demand and bears interest at 7.00 percent Note payable to a stockholder, unsecured, bearing interest	2,920,000	
at 7.00 percent and due on demand	500,000	
Total Less current portion	15,757,000 3,420,000	11,636,000
Long-term portion	\$12,337,000 ======	\$11,636,000 ======

The weighted-average interest rate on short-term notes payable was 8.70 percent and 6.90 percent as of September 30, 2000 and 1999, respectively.

NOTE 6 -- LONG-TERM DEBT

Long-term debt consists of the following:

	2000	1999
Mortgage payable to bank, bearing interest at the bank's prime rate (9.50 percent at September 30, 2000), due in monthly installments of \$21,765 including interest, with a final principal payment due on September 1, 2004. The mortgage is collateralized by the related land and building with a net book value of \$2,768,000 at September		
30, 2000	2,352,000	2,400,000

	2000	1999
Mortgage payable to bank, bearing interest at 6.30 percent, due in monthly installments of \$18,199 including interest, with a final principal payment due on October 8, 2005. The mortgage is collateralized by the related land and building with a net book value of \$3,355,000 at September 30, 2000	2,349,000	2,415,000
with the final payment due October 8, 2001. The loan is collateralized by all assets of the Company	390,000	756,000
TotalLess current portion	7,644,000 552,000	8,176,000 516,000
Long-term portion	\$7,092,000 ======	\$7,660,000 ======

Estimated principal payments due on the long-term debt are as follows:

YEARS ENDING SEPTEMBER 30	AMOUNT
2001	
2002	246,000
2003	
2004	4,586,000
2005	
2006 and thereafter	1,938,000
Total	\$7,644,000
	========

NOTE 7 -- COMMITMENTS AND CONTINGENCIES

LEASE COMMITMENTS -- The Company utilizes operating leases for equipment, warehouses and operating facilities. For most locations, the Company pays taxes, insurance and maintenance costs. Lease terms generally range from one to six years with renewal options for additional three- to five-year periods.

The Company leases one of its primary operating facilities from a partnership, of which the Company's controlling stockholder owns 100 percent through direct and indirect ownership.

The minimum lease payments for the remaining years under the above leases are as follows:

YEARS ENDING SEPTEMBER 30	RELATED PARTY	OTHER	TOTAL
2001	\$ 551,000	\$4,102,000	\$ 4,653,000
2002	623,000	3,791,000	4,414,000
2003	638,000	1,759,000	2,397,000
2004	638,000	121,000	759,000
2005	638,000	30,000	668,000
2006 and thereafter	797,000		797,000
Total	\$3,885,000	\$9,803,000	\$13,688,000
	========	========	========

VSI HOLDINGS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

The Company remains the guarantor of the operating leases of BKNT Retail Stores, Inc. (see Note 14). The sole stockholder of BKNT Retail Stores, Inc. has pledged 216,750 shares of VSI Holdings, Inc. as collateral relating to the lease commitments. The minimum lease payments for the remaining years under these leases are as follows:

YEARS ENDING SEPTEMBER 30	AMOUNT
2001	\$322,000
2002	88,000
2003	25,000
Total	\$435,000
	=======

Rent expense was as follows for the years ended September 30:

	2000	1999	1998
Related party	\$ 551,000	\$ 551,000	\$ 551,000
Other	3,133,000	3,169,000	4,176,000
Total	\$3,684,000	\$3,720,000	\$4,727,000
	========	========	========

NOTE 8 -- INCOME TAXES

The following is a summary of the deferred tax assets and liabilities at September 30, 2000 and 1999:

	2000	1999
Deferred tax assets:		
Depreciation and amortization	\$1,050,000	\$ 991,000
Net operating loss carryforwards		36,000
Accrued expenses not deductible for tax purposes	900,000	670,000
Losses from partnership investments Equity method	405,000	180,000
Total deferred tax assets Deferred tax liabilities Unrealized gain from	2,355,000	1,877,000
investments	(385,000)	(955,000)
Net deferred tax asset before valuation allowance Valuation allowance	1,970,000 (241,000)	922,000
Net deferred tax asset	\$1,729,000 ======	\$ 922,000 =====

The provision for income taxes consists of the following:

	2000	1999	1998
Current Deferred	. , ,	\$278,000 (347,000)	\$4,643,000 244,000
Total provision for (benefit from) income taxes	\$3,511,000 ======	\$(69,000) ======	\$4,887,000

A reconciliation of taxes on income from continuing operations based on the statutory federal income tax rate to the provision for income taxes is as follows:

	2000	1999	1998
Tax computed at statutory federal income tax			
rate	\$3,080,000	\$(193,000)	\$4,793,000
Nondeductible expenses	196,000	164,000	92,000
Change in valuation allowance	241,000		
Adjustments to prior year taxes and other	(6,000)	(40,000)	2,000
Total provision for (benefit from) income			
taxes	\$3,511,000	\$ (69,000)	\$4,887,000
	========	=======	========

A valuation allowance has been established due to the uncertainty of the Company being able to realize certain losses from partnerships for tax purposes.

NOTE 9 -- CASH FLOWS

Cash paid during the years ended September 30, 2000, 1999 and 1998 for interest amounted to \$3,589,000, \$3,013,000 and \$2,123,000, respectively. Cash paid for income taxes in those years was \$158,000, \$6,199,000 and \$6,000, respectively.

The Company had the following noncash transactions:

- During 2000, the Company received 1,342 shares of senior preferred stock in Oz Entertainment Company in exchange for a reduction of an accounts receivable due from Oz Entertainment Company of \$500,000.
- During the years ended September 30, 2000 and 1999, the Company issued 174,713 and 154,126 shares of stock under its restricted stock compensation plan, respectively (see Note 11). As a result, the Company reduced the accrual for stock compensation and recorded additional stock and paid-in capital totaling \$1,039,000 and \$987,000, respectively.
- During 1998, the Company purchased PSG International, Inc. As part of the purchase price, the Company issued 280,000 shares of common stock with a value at the date of purchase of \$1,960,000 (see Note 1).
- During 1998, the Company received 144,000 shares of its own common stock valued at \$736,000 as proceeds on the sale of the discontinued operations.

NOTE 10 -- SELF-INSURANCE PLAN

The Company is substantially self-insured for employee medical and dental claims. The policy year of the plan is October 1 to September 30. The Company has purchased stop-loss insurance for individual claims that exceed \$75,000 annually. The Company has estimated the amount of incurred but not reported claims (IBNR) based on historical claims reported after year end and has recorded an accrual to the extent that the estimated IBNR exceeds any pre-funded amounts.

NOTE 11 -- STOCK COMPENSATION

In December 1997, the Board of Directors approved a restricted stock compensation plan for certain key employees. Under the plan, key employees are allocated the right to receive stock, subject to forfeiture if employment terminates prior to the end of prescribed periods ranging from one to three years. During the years ended September 30, 2000, 1999 and 1998, the Company awarded key employees the right to receive shares. The market value of these shares will be recognized and charged as compensation expense

as earned over the future service period. The following is a summary of the activity in the restricted stock compensation plan:

	2	000	1999 199		1998	
	UNISSUED SHARES	UNRECOGNIZED COMPENSATION	UNISSUED SHARES	UNRECOGNIZED COMPENSATION	UNISSUED SHARES	UNRECOGNIZED COMPENSATION
Share rights outstanding: Balance						
Beginning of yearRights awarded	381,525	\$ 663,000	462,375			\$
during year Rights forfeited		149,000	102,298	463,000	462,375	2,961,000
during year Shares issued	(26,805)	(161,000)	(29,022)	(188,000)		
during year Compensation	(174,713)		(154, 126)			
recognized		(468,000)		(1,216,000)		(1,357,000)
Balance End of year	231,007 =====	\$ 183,000 ======	381,525 ======	\$ 663,000 ======	462,375 =====	\$ 1,604,000 ======
Vested, unissued shares End of year						

NOTE 12 -- STOCK OPTIONS

The Company issued options for the Company's common stock in the following arrangements:

Mr. Toth's Options

In 1993, the Company granted its majority stockholder options to purchase 825,000 shares of common stock at \$.15625 per share for providing assistance with financing in accordance with a Plan of Reorganization. The final 425,000 of options as of September 30, 1999 were exercised during the year ended September 30, 2000.

Other Options

During the years ended September 30, 1999 and 1998, the Company issued stock options for 85,000 and 58,000 common shares, respectively, to certain consultants of the Company. The exercise prices for the years ended September 30, 1999 and 1998 ranged from \$5.75 to \$8.20 and \$6.20 to \$8.20, respectively. The fair value of the options amounted to \$89,000 and \$83,000, respectively, which was charged to expense. Generally, the options are exercisable between two and three years from the date of issuance. No options have been exercised. Options for 143,000 shares remain outstanding at September 30, 2000.

The Company has stock options outstanding or issuable for the benefit of employees and directors under the following plans:

1986 INCENTIVE STOCK OPTION PLAN -- Options under this plan were granted to officers and key employees at prices not less than the market price at date of grant. Options are generally exercisable one-third annually commencing 12 months after the date granted and expire at the end of six years. This plan terminated in March 1996 and no new options will be granted. There were no options exercised during the

years ended September 30, 2000 and 1999. Options for 22,000 shares were exercised during the year ended September 30, 1998. Options for 40,000 shares remain outstanding at September 30, 2000.

1986 NONQUALIFIED STOCK OPTION PLAN -- Options under this plan were granted to officers and employees at prices not less than the market price at date of grant. Options are generally exercisable one-third annually commencing 12 months after the date granted and expire at the end of 10 years. This plan terminated in March 1996 and no new options will be granted under the plan. Options for 15,000, 2,090 and 42,290 shares were exercised during the years ended September 30, 2000, 1999 and 1998, respectively. Options for 33,620 shares remain outstanding at September 30, 2000.

1997 NONQUALIFIED STOCK OPTION PLAN AND 1997 INCENTIVE STOCK OPTION PLAN -- These plans were established during 1997 to issue options to officers and employees at prices not less than the market price at date of grant. Each plan is authorized to issue options for 500,000 shares of the Company's common stock. During the year ended September 30, 1999, the Company approved the addition of 1,000,000 options to the 1997 Incentive Stock Option Plan. Generally, the options vest over a three-year period with 50 percent vesting after two years and the remaining vesting after three years from the date of grant. During the years ended September 30, 2000, 1999 and 1998, 48,000, 67,666, and 518,000 incentive stock options were granted, respectively. Options for 64,666 and 51,000 were cancelled during the years ended September 30, 2000 and 1999, respectively. No options have been exercised under these plans. Options for 518,000 shares remain outstanding at September 30, 2000.

INDEPENDENT DIRECTOR STOCK OPTION PLAN -- Options under this plan are granted to independent directors who are neither employees nor beneficial owners of 5 percent or more of the Company's common stock at prices equal to the market price of the Company's common stock at date of grant. During the years ended September 30, 2000 and 1998, 120,000 and 10,000 options were granted, respectively. No options were granted during the year ended September 30, 1999. During the year ended September 30, 2000, options for 27,500 were cancelled. No options were exercised during the years ended September 2000, 1999 and 1998. Options granted are usually exercisable 30 days from date of grant as determined by vesting schedules in the plan. Options for 122,500 shares remain outstanding at September 30, 2000.

The Company has adopted the disclosure-only provisions of Statement of Financial Accounting Standards No. 123, Accounting for Stock-Based Compensation. Accordingly, no compensation cost has been recognized for the stock option plans. Had compensation cost for the Company's stock option plans been determined based on the fair value at the grant date, the Company's income from continuing operations for 2000, 1999 and 1998 would have been reduced by \$386,000 (\$.01 per share), \$236,000 (\$.01 per share), and \$100,000 (no effect on earnings per share), respectively.

Information regarding these fixed-price option plans for the years ended September 30, 2000, 1999 and 1998 are as follows:

	2000		1999	1999 199		
	SHARES	WEIGHTED AVERAGE EXERCISE PRICE	SHARES	WEIGHTED AVERAGE EXERCISE PRICE	SHARES	WEIGHTED AVERAGE EXERCISE PRICE
Options outstanding Beginning of year		\$3.91	, ,		600,000	\$0.38
Canceled Granted	(92,166) 168,000		(51,000)	7.19 6.81	 586,000	6.28
Exercised	(440,000)		(2,090)		(64,290)	0.71
Options outstanding						
End of year	857,120	5.57	1,221,286	3.91	1,121,710	3.64
Option price range						
End of year	\$.50 to \$8.70		\$.15625 to \$8.70		\$.15625 to \$8.70	
Option price range for						
exercised shares	\$.15625 to \$.50		\$.50		\$.50 to \$.75	
Options available for future grants End of year Weighted average fair value of	1,249,000		1,417,000		570,000	
options granted during the year	\$0.57		\$1.22		\$1.13	

The following table summarizes information about fixed price stock options outstanding at September 30, 2000:

	OPTIONS	OUTSTANDING	OUTSTANDING OPTIONS EXERCISABLE		
RANGE OF EXERCISE PRICES	NUMBER OUTSTANDING AT SEPTEMBER 30, 2000	WEIGHTED AVERAGE REMAINING CONTRACTUAL LIFE	WEIGHTED AVERAGE EXERCISE	NUMBER EXERCISABLE AT SEPTEMBER 30, 2000	WEIGHTED AVERAGE EXERCISE PRICE
\$.50 to .55 3.25 to 4.00 5.20 to 6.70 7.90 to 8.70	73,620 149,000 521,000 113,500	1 year 5 years 2 years 2 years	.53 3.40 6.28 8.42	73,620 8,400 247,500 63,000	.53 3.25 6.38 8.60
.50 to 8.70	857,120 ======		5.57	392,520 ======	5.57

NOTE 13 -- SEGMENT INFORMATION AND MAJOR CUSTOMERS

The Company's operations are classified into two major business segments: marketing services and entertainment. The marketing services segment performs administrative and data management services, creates, prints and prepares promotional materials and performs other marketing services. In addition, the marketing services segment provides product and leadership training and creates and produces video training products, industrial theater and meetings.

The entertainment segment designs and manufactures animated displays for the retail and entertainment industry throughout the world. In addition, the entertainment segment operates and administers touring animated educational displays.

The Company had sales from foreign activities of \$5,422,000, \$5,269,000 and \$5,375,000 for the years ended September 30, 2000, 1999 and 1998, respectively.

Summarized financial information by each of the Company's two industry segments for the three-year period ended September 30, 2000 is as follows:

	2000	1999	1998
Revenue: Marketing services sector	\$182,403,000	\$134,295,000	\$155,681,000
Entertainment sector	4,852,000	9,065,000	7,477,000
Consolidated total	\$187,255,000	\$143,360,000	\$163,158,000
	======	======	======
Income (loss) from operations: Marketing services sector Entertainment sector Equity in earnings of unconsolidated	\$ 17,524,000	\$ 486,000	\$ 14,834,000
	(3,240,000)	2,352,000	1,584,000
investee Interest expense	(1,515,000) (3,708,000)	(400,000) (3,005,000)	(2,320,000)
Consolidated income (loss) from operations before income taxes	\$ 9,061,000	\$ (567,000)	\$ 14,098,000
	======	======	=======
Identifiable assets: Marketing services sector Entertainment sector	\$110,952,000	\$ 91,175,000	\$ 83,848,000
	5,177,000	7,127,000	5,707,000
Consolidated total	\$116,129,000	\$ 98,302,000	\$ 89,555,000
	======	======	======
Depreciation and amortization: Marketing services sector Entertainment sector	\$ 4,173,000	\$ 7,408,000	\$ 4,564,000
	1,644,000	415,000	301,000
Consolidated total	\$ 5,817,000	\$ 7,823,000	\$ 4,865,000
	======	=======	======
Capital expenditures: Marketing services sector Entertainment sector	\$ 4,658,000	\$ 2,736,000	\$ 11,848,000
	2,507,000	325,000	1,658,000
Consolidated total	\$ 7,165,000	\$ 3,061,000	\$ 13,506,000
	=======	======	======

The following companies are considered major customers of the marketing service sector and comprise 10 percent or greater of the Company's net sales: $\frac{1}{2} \left(\frac{1}{2} \right) \left(\frac{1}{2$

	2000	1999	1998
Customer A Customer B Customer C	39% 32 3	32% 23 10	25% 44 9
Total	74% ==	 65% ==	78% ==

NOTE 14 -- DISCONTINUED OPERATIONS

On June 30, 1998, the Company adopted a formal plan to sell the stock of its wholly owned subsidiaries, BKNT Retail Stores, Inc., J.D. Dash, Inc. and BKNT, Inc. to Mr. Martin Suchik, a former officer in VSI Holdings, Inc., in exchange for stock in VSI Holdings, Inc. The Company completed the sale on September 30, 1998. The VSI Holdings, Inc. stock exchanged had a total value of \$736,000 on the date the sale was completed.

Operating results of BKNT Retail Stores, Inc., J.D. Dash, Inc. and BKNT, Inc. for the nine months ended June 30, 1998 are shown separately as discontinued operations in the accompanying statement of operations.

Net sales of BKNT Retail Stores, Inc., J.D. Dash, Inc. and BKNT, Inc. for the year ended September 30, 1998 was approximately \$12,190,000 and has not been included in net sales in the accompanying statement of operations.

NOTE 15 -- QUARTERLY RESULTS (UNAUDITED)

The Company believes that the following information reflects all normal recurring adjustments necessary for a fair presentation of the information for the periods presented. The following tables contain selected unaudited consolidated statements of income for each quarter for fiscal 2000 and 1999. The operating results for any quarter are not necessarily indicative of results for any future period.

FISCAL	YEAR	2000
--------	------	------

	4TH QUARTER	3RD QUARTER	2ND QUARTER	1ST QUARTER
	(TN TU	OUSANDS, EXCEPT	TOD DED SUADE	DATA)
	(IN IU	OUSANDS, EXCEPT	FUR PER SHARE	DATA)
Revenue	\$58,830	\$45,636	\$44,689	\$38,100
Operating Income	4,766	5,022	3,897	2,130
Net Income (Loss)	1,048	2,195	1,451	856
Net Income per common share:				
Basic	0.03	0.07	0.04	0.03
Fully diluted	0.03	0.07	0.04	0.03
Weighted average shares outstanding (a):				
Basic	33,172	33,198	33,261	32,761
Fully diluted	33,383	33,397	33,430	32,972

FISCAL YEAR 1999

	4TH QUARTER	3RD QUARTER	2ND QUARTER FOR PER SHARE	1ST QUARTER
Revenue Operating Income (Loss) Net Income (Loss) Net Income (loss) per common share:	\$39,071	\$28,439	\$40,513	\$35,337
	173	(2,747)	3,139	2,789
	(1,125)	(2,327)	1,444	1,510
BasicFully diluted	(0.03)	(0.07)	0.04	0.05
	(0.03)	(0.07)	0.04	0.05
Basic	32,631	32,623	32,916	32,822
Fully diluted	32,631	32,623	33,602	33,448

⁽a) Earnings per common share are computed independently for each of the quarters presented. Therefore, the sum of the quarterly per common share information may not equal the annual earnings per common share.

NOTE 16 -- FAIR VALUES OF FINANCIAL INSTRUMENTS

A summary of the fair values of financial instruments, as well as the methods and significant assumptions used to estimate fair values, is as follows:

SHORT-TERM FINANCIAL INSTRUMENTS -- The fair value of short-term financial instruments, including cash, trade accounts receivable and payable, accrued liabilities and advances from customers, approximate

VSI HOLDINGS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS -- (CONTINUED)

the carrying amounts in the accompanying consolidated financial statements due to the short maturity of such instruments.

MARKETABLE SECURITIES -- The carrying amount of investments in available-for-sale securities is equal to their fair values based on quoted market prices.

INVESTMENTS -- Investments consist of investments in a privately held corporation and partnership interests. There is no market for the Company's investment in the privately held corporation or its investment in partnerships. It was impracticable to estimate the fair values of those investments. The following is a summary of combined pertinent information about the investment in Oz Entertainment Company and the Company's investment in K.C. Investors, L.P.:

	2000	1999
Total assets	\$26,017,000	\$23,718,000
Total equity	11,085,000	15,660,000
Total revenue		
Net loss	(6,189,000)	(4,993,000)

NOTES RECEIVABLE AND ADVANCES, NOTES PAYABLE TO BANK AND LONG-TERM DEBT -- The fair values approximate the carrying amounts since the note rates approximate rates currently available to the Company for notes with similar terms and maturities.

NOTES PAYABLE TO RELATED PARTIES -- The estimated fair value of the note payable to related parties at September 30, 2000 and 1999 was approximately \$15,064,000 and \$11,636,000, respectively. The estimated fair value was determined using rates currently available to the Company.

NOTE 17 -- LITIGATION

As of September 30, 2000 the Company has pending litigation with a former employee and stockholder who is seeking damages for wrongful discharge and denial of the fair value for Company stock options. Management feels the case is without merit and plans to vigorously defend the lawsuit. At this time, the case is in preliminary stages and the outcome is not determinable, although management believes the outcome will not have a material effect on the Company's financial position.

PART IV

- ITEM 14. EXHIBITS, FINANCIAL STATEMENT SCHEDULES, AND REPORTS ON FORM 8-K.
- (A) FINANCIAL STATEMENTS, FINANCIAL STATEMENT SCHEDULES AND EXHIBITS.
- 1. Consolidated Financial Statements of VSI Holdings, Inc. and subsidiaries and Independent Auditors' Report are filed herewith as a separate section of this report.
- 2. All schedules for which provision is made in the applicable accounting regulations of the Securities and Exchange Commission have been omitted because such schedules are not required under the related instructions or are inapplicable or because the information required is included in the Consolidated Financial Statements or notes thereon.

 - *3.1 Articles of Incorporation of the Registrant dated April 21, 1997, together with Articles of Merger of Registrant and The Banker's Note, Inc. dated April 21, 1997, filed as Exhibit 3.1 to Form 10-K for fiscal year ended September 30, 1997.
 - *3.2 By-Laws of the Registrant, amended and effective on September 12, 1997, filed as Exhibit 3.2 to Form 10-K for fiscal year ended September 30, 1997.
 - *4.1 VSI Holdings, Inc. 1997 Incentive Stock Option Plan as approved at the Annual Shareholders' Meeting held on April 21, 1997, filed as Exhibit 4.1 to Form 10-K for fiscal year ended September 30, 1997.
 *4.2 VSI Holdings, Inc. 1997 Non-Qualified Stock Option Plan as
 - *4.2 VSI Holdings, Inc. 1997 Non-Qualified Stock Option Plan as approved at the Annual Shareholders' Meeting held on April 21, 1997, filed as Exhibit 4.2 to Form 10-K for fiscal year ended September 30, 1997.
 - *4.3 The Banker's Note, Inc. Independent Director Stock Option Plan as approved at the Annual Shareholders' Meeting held on June 23, 1989, filed as Exhibit 4.3 to Form 10-K for fiscal year ended September 30, 1997.
 - year ended September 30, 1997.

 *4.4 The Banker's Note, Inc. 1986 Incentive Stock Option Plan as approved at the Annual Shareholders' Meeting held on June 16, 1986, filed as Exhibit 4.1 to Form 10-K for fiscal year ended September 30, 1996.
 - *4.5 The Banker's Note, Inc. 1986 Non-Qualified Stock Option Plan as approved at the Annual Shareholders' Meeting held on June 16, 1986, filed as Exhibit 4.2 to Form 10-K for fiscal year ended September 30, 1996.
 - *4.6 VSI Holdings, Inc. Restricted Stock Plan December 1, 1997 as approved at the annual shareholders meeting held on April 8, 1998, filed as Exhibit 4.6 to Form 10-K for fiscal year ended September 30, 1998.
 - *4.7 VSI Holdings, Inc. Employee Stock Purchase Plan October 7, 1997 as approved at the annual shareholders meeting held on April 8, 1998, filed as Exhibit 4.7 to Form 10-K for fiscal year ended September 30, 1998.
 - *4.8 Ádvanced Animations, Inc. Agreement and Plan of Merger dated February 7, 1997, filed as Exhibit 4.8 to Form 10-K for fiscal year ended September 30, 1998.
 - *4.9 VISPAC, Inc. Agreement and Plan of Merger dated June 13, 1997, filed as Exhibit 4.9 to Form 10-K for fiscal year ended September 30, 1998.

- *4.10 Visual Services, Inc. Agreement and Plan of Merger dated September 24, 1997, filed as Exhibit 4.10 to Form 10-K for fiscal year ended September 30, 1998. +21.1 List of Subsidiaries of the Registrant. +23.1 Consent of Plante & Moran LLP, Independent Auditors.

- (b) REPORTS ON FORM 8-K: NONE

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

VSI Holdings, Inc.

(Registrant)

By: /s/ STEVE TOTH, JR.

Steve Toth, Jr., President and Chief Executive Officer

Date: January 12, 2001

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the Registrant and in the capacities and on the dates indicated.

/s/ STEVE TOTH, JR. Steve Toth, Jr.	Director, President and Chief Executive Officer	January	12,	2001
/s/ MARTIN S. SUCHIK	Director	January	12,	2001
Martin S. Suchik				
/s/ THOMAS W. MARQUIS	Director, Treasurer, Secretary, Chief Accounting Officer, and	January	12,	2001
Thomas W. Marquis	Principal Financial Officer			
/s/ ROBERT SUI	Director	January	12,	2001
Robert Sui				
		January	12,	2001
Harold Poling				
		January	12,	2001
Ralph Armijo				
		January	12,	2001
William James				

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-0

[X] QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE QUARTERLY PERIOD ENDED DECEMBER 31, 2000

[] TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE TRANSITION PERIOD FROM

TO

COMMISSION FILE NO. 1-12942

GEORGIA (STATE OR OTHER JURISDICTION OF INCORPORATION OR ORGANIZATION) 22-2135522 (I.R.S. EMPLOYER IDENTIFICATION NO.)

41000 WOODWARD AVENUE BLOOMFIELD HILLS, MI 48304-2263 (ADDRESS OF PRINCIPAL EXECUTIVE OFFICES)

(248) 644-0500 (REGISTRANT'S TELEPHONE NUMBER, INCLUDING AREA CODE)

FOR INFORMATION REGARDING THIS FILING, CONTACT: PEGGY TOTH

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months, and (2) has been subject to such filing requirements for the past 90 days.

Yes [X] No []

There were 33,742,410 shares of Common Stock, par value \$.01 per share, outstanding at December 31, 2000. The Company held 400,250 of these shares as treasury stock.

PART 1. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

VSI HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEET

	DECEMBER 31, 2000	SEPTEMBER 30, 2000
	(UNAUDITED)	(AUDITED)
ASSETS		
CURRENT ASSETS Cash Trade accounts receivable:	\$ 3,250,000	\$ 905,000
Billed	41,698,000 25,378,000 171,000 184,000 358,000 5,476,000 378,000 233,000	53,907,000 18,525,000 201,000 438,000 3,744,000 899,000 761,000
Total Current Assets LONG-TERM PORTION OF NOTES RECEIVABLE Related Parties PROPERTY, PLANT AND EQUIPMENT (NET) DEFERRED TAX ASSET Investment in Available-for-Sale Securities INVESTMENTS GOODWILL NET	77,126,000 1,001,000 21,654,000 2,243,000 2,975,000 4,190,000 1,176,000	79,380,000 921,000 22,394,000 830,000 7,131,000 4,190,000 1,283,000
Total Assets	\$110,365,000	\$116,129,000
LIABILITIES AND STOCKHOLDERS' EQUI	======= ITY	========
CURRENT LIABILITIES		
Current portion of long-term debt Notes payable Related party Trade accounts payable Notes payable to bank Accrued liabilities Federal income tax payable Advances from customers for uncompleted projects	\$ 439,000 1,988,000 16,861,000 42,101,000 2,214,000 (76,000) 726,000	\$ 552,000 3,420,000 23,479,000 35,214,000 3,970,000 2,139,000 375,000
Total Current Liabilities	64,253,000	69,149,000
LONG-TERM LIABILITIES Notes payable Related parties Long-term debt Other	11,854,000 7,066,000	12,337,000 7,092,000
Total Long-Term LiabilitiesSTOCKHOLDERS' EQUITY	18,920,000	19,429,000
Preferred stock \$1.00 par value per share, 2,000,000 shares authorized, no shares issued		
and 33,580,000 at September 30, 2000	337,000	336,000
2000, and September 30, 2000	(1,856,000) 9,017,000 (2,014,000) 21,708,000	(1,856,000) 8,071,000 730,000 20,270,000
Total Stockholders' Equity Total Liabilities and Stockholders' Equity	27,192,000 \$110,365,000 =======	27,551,000 \$116,129,000 =======

See Notes to Consolidated Financial Statements

VSI HOLDINGS, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENT OF INCOME

	THREE MONTHS ENDED		
	DECEMBER 31, 2000	DECEMBER 31, 1999	
	(UNAUDITED)	(UNAUDITED)	
REVENUEEXPENSES	\$43,329,000	\$38,100,000	
Cost of revenue	15,643,000 24,158,000	13,705,000 22,265,000	
Total Expenses	39,801,000 3,528,000	35,970,000 2,130,000	
Interest income and other income and expense	(327,000) (964,000)	(41,000) (791,000)	
Total Other Expenses	(1,291,000) 2,237,000 799,000	(832,000) 1,298,000 442,000	
NET INCOME	\$ 1,438,000 =======	\$ 856,000 ======	
OTHER COMPREHENSIVE INCOME (LOSS) Foreign Currency Translation Adjustment Unrealized loss on Securities, Net of Tax Benefit of	(1,000)	(66,000)	
\$1,413,000	(2,743,000)	1,021,000	
TOTAL OTHER COMPREHENSIVE INCOME (LOSS)		\$ 955,000 \$ 1,811,000	
Basic: Fully Diluted: Weighted Average Shares Basic. Dilutive.	\$0.04 \$0.04 33,180,000 33,279,000	\$0.03 \$0.03 32,761,000 32,972,000	

See Notes to Consolidated Financial Statements $\ensuremath{\mathbf 2}$

VSI HOLDINGS, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENT OF CASH FLOWS

	THREE MONTHS ENDED		
	DECEMBER 31, 2000	DECEMBER 31, 1999	
	(UNAUDITED)		
Cash Flows from Operating Activities Net Income	\$ 1,438,000	\$ 856,000	
Depreciation and amortization	1,273,000 254,000 521,000	1,462,000 203,000 207,000	
Trade accounts receivable	3,956,000 80,000 528,000 (1,732,000)	1,807,000 (40,000) 234,000 (508,000)	
Trade accounts payable	(6,618,000) (3,029,000) 351,000	(4,251,000) (538,000) (282,000)	
Net cash used in operating activities	(2,978,000) 30,000 (264,000) (426,000)	(850,000) (80,000) (143,000) (410,000)	
Investment in unconsolidated investments	(254,000)	(523,000)	
Net cash used in investing activities	(914,000)	(1,156,000)	
Changes Long Term Debt	(139,000) (515,000) 6,887,000	(131,000) 42,000 2,013,000 65,000	
Proceeds from issuance of stock	5,000	4 000 000	
Net cash provided by financing activities Effect of Exchange Rate Changes on Cash Net Increase (Decrease) in Cash Cash Beginning of Period	6,238,000 (1,000) 2,345,000 905,000	1,989,000 (66,000) (83,000) 552,000	
Cash End of Period		\$ 469,000 ======	

See Notes to Consolidated Financial Statements

VSI HOLDINGS, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. The consolidated financial statements included herein have been prepared by the Company without audit pursuant to the rules of the Securities and Exchange Commission. Preparing financial statements requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues, and expenses. Examples include provisions for bad debts and the length of product life cycles and buildings' lives. Actual results may differ from these estimates. Certain information and footnote disclosures normally included in financial statements prepared in accordance with generally accepted accounting principles have been condensed or omitted pursuant to such rules and regulations.

In the opinion of management, the accompanying consolidated balance sheet and consolidated statements of income and cash flows include all adjustments (consisting only of normal recurring items) necessary for a fair presentation of the results for the interim period, in conformity with generally accepted accounting principles.

- 2. The interim financial information presented herein should be read in conjunction with Management's Discussion and Analysis and financial statements and related notes included in the Registrant's Annual Report on Form 10-K for the year ended September 30, 2000. Results for interim periods should not be considered indicative of the results that may be expected for the year ended September 30, 2001.
- 3. Certain amounts for prior periods were reclassified to conform with present period presentation.
- 4. We evaluate the carrying value of long-lived assets for potential impairment on an ongoing basis. Such evaluations consider management's plans for future operations, recent operating results, undiscounted annual cash flows and other economic factors related to the operation to which the asset applies.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis should be read in conjunction with the consolidated financial statements and the notes thereto included elsewhere in this report. The following discussion contains certain forward-looking statements relating to our anticipated future financial conditions and operating results and our current business plans. In the future, our financial condition and operating results could differ materially from those discussed herein and our current business plans could be altered in response to market conditions and other factors beyond our control. Important factors that could cause or contribute to such difference or changes include those discussed elsewhere in this report (see the disclosures under "Cautionary Statement for the Purpose of the "Safe Harbor" Provisions of the Private Securities Litigation Reform Act of 1995").

BUSINESS DESCRIPTION:

VSI Holdings, Inc. (the "Company", "we", "our", "us") presently consists of wholly-owned subsidiaries in the Marketing Services and Entertainment business sectors under the following trade names:

Visual Services, Inc., a broad-based provider of educational curriculums and product training; interactive technology-based distance learning systems; product launches; Web site development, internet, intranet, and extranet solutions; direct-response and site-based marketing; change process and cultural change consulting.

Vispac, Inc., integrated logistics and call center operations.

Performance Systems Group; in-field consulting and change process sustainment services.

eCity Studios, Inc., a web site development company.

Advanced Animations, Inc., a manufacturer of product simulators, animatronic figures and displays for theme parks, casinos, and retail.

We are attempting to position ourselves to take advantage of opportunities created by changes in technology. One of our practices has been the usage of a wide variety of technologies, without overdependence on any one technology. This allows us to meet client needs with whatever technology is most appropriate.

We serve our global customers from our Bloomfield Hills, Michigan headquarters and other offices in Michigan, California, Vermont, and Canada. We employ more than 1,000 professionals.

The accompanying consolidated financial statements include the accounts of the Company and its wholly-owned subsidiaries, consisting of Advanced Animations, Inc., Vispac, Inc., Visual Services, Inc., eCity Studios, Inc. and PSG International, Inc. Inter-company balances and transactions have been eliminated in consolidation.

OPERATING RESULTS

Revenues were \$ 43,329,000 for the three months ended December 31, 2000, compared to \$38,100,000 for the same period last year. This increase of 14% for the three months is attributable to additional business with our primary clients. Revenues for the quarter included work done on a large scale ride-and-drive (the opportunity for consumers to drive and compare competitive vehicles in a neutral environment), certification programs for automotive dealerships and other work. In the prior year, a continuing contract with annual revenue of approximately \$15 million was lost. We do not expect the see the entire impact of this in the current year, as it is being phased out. This business was largely phased out by the end of December, 2000. Unless management is able to replace the business or make cost cutting measures in a timely manner, this could lead to an adverse impact on our operating results.

We compete in very competitive and volatile markets. The Marketing Services segment is subject to intense competition, as well as delays in project fulfillment due to matters beyond its control, such as delayed product launches, strikes at clients, and other factors. The Entertainment segment's sales represent discretionary spending on the part of its customers, and their customers. In either sector, if projects end up being deferred from this fiscal year to next fiscal year or canceled, it could have an adverse effect on operating results. Such factors make it difficult to project full year financial results.

Operating Expense. Our operating expenses have grown to \$24,158,000 for the three months ended December 31, 2000 from \$22,265,000 in the three months ended December 31, 1999. This increase of 9% is mainly attributable to the following factors: (1) additional personnel needs due to incremental business; (2) wage escalations for computer-industry and other professionals; (3) Michigan's extremely tight and competitive contract labor supply; and (4) increased dependence on contract labor to staff additional business, resulting in higher labor costs.

Our future operating results will depend in part on management's ability to manage any future growth and control expenses. We intend to pursue the continued growth of our business, however, there can be no assurance that such growth will be achieved. A decline in revenues, without a corresponding and timely reduction in staffing and other expenses, or a staffing increase that is not accompanied by a corresponding increase in revenues, could have a material adverse effect on our operating results.

Accounts Payable was down by approximately \$6.6 million while Notes Payable -- Bank were up approximately the same amount. Timing of payments to suppliers caused these changes. A large check was received from a client at the end of the quarter, but had not yet cleared our bank; this caused our cash balance to rise and our Accounts Receivable balance to decline. While Accounts Receivable has declined overall, the unbilled component has risen due to a delay in receiving purchase orders on some projects, and delays in billing some other projects; this amount is expected to decrease in the remainder of the year.

LIQUIDITY AND CAPITAL RESOURCES

We have various bank lines of credit totaling \$42,000,000, which mature in March, 2001 and March, 2002. At December 31, 2000, we had borrowed \$38,851,000 (including outstanding checks, less cash balances) against these lines. Interest on these lines is primarily based on LIBOR (London Inter-Bank Offered Rate) plus 1.5%. Our borrowing rate at December 31, 2000 was 8.20%.

We have had a long-term relationship with our current bank. Through the years, it has provided financing and lines of credit for us. There can, however, be no assurance that the lines of credit will be renewed when they mature. If we are unable to renew the lines of credit, other sources of financing would be sought, primarily lines of credit from another banking institution.

We have the rights to design worldwide touring and permanent exhibitions based on the series of Grossology-themed books authored by science teacher Sylvia Branzei. The first touring exhibition debuted in February, 2000 in Vancouver, British Columbia. The exhibition promotes the scientific discovery of the human body based on the theory that the best way to get kids interested in science is to present it with a dose of entertainment and in terms that they find most appealing. Thus far, the exhibition has been successful in museums, science centers, and theme parks. Bookings for these exhibits cover the next four years. We have made an investment of approximately \$2.5 million. We do not anticipate any additional material investment on these existing exhibits.

Since we are a net borrower of funds, minimal cash balances are kept on hand. At any point in time, we may have more money in checks outstanding than the cash balance. When checks are presented for payment, the bank notifies us. We borrow on our lines of credit to cover the checks.

We believe that cash flows from operations, along with borrowings, will be sufficient to finance our activities in 2001. On a long-term basis, increased financing may be necessary to fund any large project awarded to us, or any acquisitions we may make. We have no current plans to conduct an offering of our shares to the public in fiscal year 2001.

Stock and Stock Options Granted

This year, we granted 20,000 shares of restricted stock to certain key employees. The shares vest one, two, and three years from the date of grant in three equal parts. We do not expect the exercise of stock options, or purchase of shares, by employees to be a material source of capital in fiscal year 2001. During the quarter, 160,000 shares vested and were issued, the rights to which had been granted in prior years. This issuance caused a reduction in Accrued Liabilities and an increase in Stockholders' Equity for approximately \$900,000.

INVESTMENTS

At September 30, 2000, we had invested \$4.5 million in Oz Entertainment Company and in a limited partnership (as a limited partner) which will develop a theme park, located in Kansas, based on the story "The Wizard of Oz". We have recognized losses totalling \$1,188,000 prior to this fiscal year, and an additional \$150,000 in the first quarter of this year. We expect to see continued losses until the opening of the park, currently scheduled for 2004. The park is planned to be constructed on 9,000 acres of land owned by the federal government. It is necessary to receive title to the land before construction can begin. In order to receive title to the land, approval must be received from two governmental authorities. The first authority tabled consideration of this in November, 2000 until late February, 2001. It is anticipated that title will be received during the next year. The success of the park as an investment is dependent upon receiving title to the land, as well as certain infrastructure improvements being completed by or paid by governmental agencies, financing arranged through governmental agencies, as well as additional public or private financing. If the park does not open, the entire investment, currently valued at \$3,162,000, is at risk. The projections provided and prepared by the management of K.C. Investors, LP and OEC forecast that the park will be profitable upon opening. During the quarter, \$1.4 million of Accounts Receivable from OEC was collected from additional funds invested by a related party.

Prior to the current fiscal year, we invested \$3.5 million in convertible preferred stock in a private placement offering of eCollege.com (NASDAQ -- ECLG), a company engaged in developing Internet-based education for colleges and universities. Through relationships with its educational partners, it develops, manages and markets on-line courses and degree programs. Upon completion of their initial public offering of common stock in December, 1999, our investment was converted to 468,808 shares of common stock. In addition, we invested \$49,500 to acquire 4,500 shares of their stock during their initial public offering. At February 7, 2001, our investment had a market value of \$2,574,000. The unrealized loss on this investment is recorded as part of Comprehensive Income.

Prior to the current fiscal year, we exercised options to purchase 431,525 shares of Navidec, Inc. (NASDAQ -- NVDC) for \$2,450,000. Navidec is a developer of web sites and web based complete automotive purchase transaction and information services for prospective customers. Their product, referred to as Driveoff.com, was sold during 2000 to CarPoint, which is majority-owned by Microsoft. At February 7, 2001, these shares had a market value of \$1,483,000.

We have committed to a joint venture with Learning Byte, International (LBI). LBI, based in Minneapolis, Minnesota, is a web based provider of training content and curriculum. As part of our commitment, subsequent to December 31, 2000, we have advance funded approximately \$400,000, with a total commitment of \$1,000,000. Our plan is to offset these amounts with future usage of their services. If we do not have sufficient business activity with LBI, the funding amounts will be written off.

At December 31, 2000, we had invested approximately \$650,000 in a joint venture in PerforMaps, L.L.C. PerforMaps, based in Minneapolis, Minnesota, is a provider of goods and services related to specific projects associated with cultural change initiatives.

While we have no current plans for further investments, management expects to make future investment in promising companies in their early development which are in a related line to business.

"CAUTIONARY STATEMENT FOR THE PURPOSES OF THE "SAFE HARBOR" PROVISIONS OF THE PRIVATE SECURITIES LITIGATION REFORM ACT OF 1995"

Certain statements in Management Discussion and Analysis of Financial Condition and Results of Operations and certain other sections of this Annual Report are forward-looking. These may be identified by the use of forward-looking words or phrases such as "believe," "expect," "anticipate," "should," "planned," "estimated," and "potential," among others. These forward-looking statements are based on our reasonable current expectations. The Private Securities Litigation Reform Act of 1995 provides a "safe harbor" such forward-looking statements. In order to comply with the terms of the safe harbor, we note that a variety of factors could cause our actual results or experience to differ materially from the anticipated results or other expectations expressed in such forward-looking statements. The risks and uncertainties that may affect the operations, performance, development and our results include but are not limited to: (1) the complexity and uncertainty regarding the development and customer acceptance of new products and services; (2) the loss of market share through competition; (3) the introduction of competing products or service technologies by other companies; (4) pricing pressures from competitors and/or customers; (5) our inability to protect proprietary information and technology; (6) market acceptance of our touring exhibits in our Entertainment/Edutainment segment; (7) the loss of key employees and/or customers; (8) our customers continued reliance on outsourcing; (9) changes in our capital structure and cost of capital, and ability to borrow sufficient funds at reasonable rates (10) inability of the developers of the "Wonderful World of Oz" theme park to obtain final transfer of the property, to complete the timely construction of the park, and to operate it profitably once the park opens; (11) uncertainties relating to business and economic conditions; (12) management evaluation of staffing levels; (13) future investments of the Company.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Interest Rate Risk. Our earnings are affected by changes in short-term interest rates as a result of our revolving credit agreements, which bear interest at a floating rate. We do not use derivative or other

financial instruments to mitigate the interest rate risk or for trading purposes. Risk can be estimated by measuring the impact of a near-term adverse movement of 100 basis points in short-term market interest rates. If short-term market interest rates average 100 basis points more in the next 12 months, the adverse impact on our results of operations would be approximately \$256,000, net of income tax benefit. We do not anticipate any material near-term future earnings or cash flow expenses from changes in interest rates related to our long-term debt obligations as all of our long-term debt obligations have fixed rates

Foreign Currency Risk. Although we conduct business in foreign countries, principally Canada and Australia, foreign currency translation gains and losses are not material to our consolidated financial position, results of operation or cash flows. Accordingly, we are not currently subject to material foreign currency exchange rate risks from the effects that exchange rate movements of foreign currencies would have on our future costs or on future cash flows we would receive from our foreign investment. To date, we have not entered into any foreign currency forward exchange contracts or other derivative financial instruments for trading purposes or to hedge the effects of adverse fluctuations in foreign currency exchange rates.

Investment Risk for Privately Held Companies. We invest in equity instruments of privately-held companies in the internet information technology and entertainment areas for business and strategic purposes. These investments are included in long-term assets, and are accounted for under the cost method or the equity method. For these non-quoted investments, our policy is to regularly review the assumptions underlying the operating performance and cash flow forecasts in assessing the carrying values. We identify and record impairment losses on these investments when events and circumstances indicate that such assets are permanently impaired. To date, no such impairment has been recorded.

Investment Risk for Publicly Traded Companies. We are also exposed to equity price risk on our investments in publicly traded companies. Our available-for-sale securities include our equity positions in Navidec, Inc., and eCollege.com, both of which have experienced significant volatility in their stock prices since going public. We do not attempt to reduce or eliminate our market exposure on these securities. A 20% adverse change in equity price would result in an approximate \$811,000 decrease in fair value in our available-for-sale securities, based upon February 7, 2001 closing market prices for Navidec and eCollege.com.

PART II.

OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We are periodically involved in routine proceedings. There are no legal matters, existing, pending, or threatened, which management presently believes could result in a material loss to us.

ITEM 2. CHANGES IN SECURITIES AND USE OF PROCEEDS

None

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

None

ITEM 5. OTHER INFORMATION

None

10

TEM 6	EXHIBITS	ΔNID	REPORTS	ΟN	FORM	8-	. K

a. Exhibits

None

b. Reports on Form 8-K

Pursuant to the requirement of the Securities and Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

> VSI HOLDINGS, INC. Registrant

February 14, 2001

/s/ STEVE TOTH, JR.

Steve Toth, Jr., Director, President and Chief Executive Officer

February 14, 2001

/s/ THOMAS W. MARQUIS

Thomas W. Marquis, Director, Treasurer, Chief Accounting and Financial Officer

CONSENT OF INDEPENDENT PUBLIC ACCOUNTANTS

As independent public accountants, we hereby consent to the incorporation by reference in this registration statement on Form S-4 of our report dated February 9, 2001, on the Company's consolidated financial statements as of December 31, 2000 and 1999 and for each of the three years in the period ending on December 31, 2000 included in the Company's Form 10-K/A for the year ended December 31, 2000 and to all references to our Firm included in this registration statement.

/s/ ARTHUR ANDERSEN LLP

Arthur Andersen LLP

Chicago, Illinois April 13, 2001 1

Exhibit 23.2

CONSENT OF INDEPENDENT AUDITORS

We consent to the reference to our firm under the caption "Experts" in the Registration Statement on Form S-4 of SPX Corporation registering 1,443,286 shares of its common stock and to the incorporation by reference therein of our report dated December 22, 2000, with respect to the consolidated financial statements of VSI Holdings, Inc. included in its Annual Report on Form 10-K for the year ended September 30, 2000, filed with the Securities and Exchange Commission.

Ann Arbor, Michigan April 16, 2001 /S/ PLANTE & MORAN, LLP

EXHIBIT 23.3

CONSENT OF INDEPENDENT AUDITORS

To Board of Directors United Dominion Industries Limited

We consent to the incorporation by reference in the Registration Statement on Form S-4 dated April 16, 2001, of SPX Corporation of our report dated January 25, 2001, except as to note 14 which is as of March 11, 2001, with respect to the consolidated statements of financial position of United Dominion Industries Limited as at December 31, 2000 and 1999 and the related consolidated statements of income, cash flows and changes in shareholders' equity for each of the years in the three-year period ended December 31, 2000, which report is included in the December 31, 2000 Annual Report to Shareholders of United Dominion Industries Limited filed as Exhibit 23.1 to such annual report on Form 40-F of United Dominion Industries Limited, which report also appears in and is incorporated by reference in Form 8-K dated April 13, 2001, filed by SPX Corporation.

/s/ KPMG LLP Chartered Accountants

Toronto, Canada April 16, 2001

CONSENT OF McDONALD INVESTMENTS INC.

We hereby consent (i) to the use of our opinion letter to the Board of Directors of VSI Holdings, Inc. ("VSI Holdings") included as Appendix C to the proxy statement-prospectus relating to the transaction provided for by that certain Agreement and Plan of Merger by and between SPX Corporation and VSI Holdings, dated as of March 24, 2001, and (ii) to the references to our firm and such opinion in such proxy statement-prospectus. In giving such consent, we do not admit that we come within the category of persons whose consent is required under Section 7 of the Securities Act of 1933, as amended, or the rules and regulations of the Securities and Exchange Commission thereunder, nor do we thereby admit that we are experts with respect to any part of such Registration Statement within the meaning of the term "experts" as used in the Securities Act of 1933, as amended, or the rules and regulations of the Securities and Exchange Commission thereunder.

/s/ McDonald Investments Inc.

EXHIBIT 99.1

PROXY

VSI HOLDINGS, INC.

PROXY SOLICITED ON BEHALF OF YOUR BOARD OF DIRECTORS FOR THE SPECIAL MEETING OF SHAREHOLDERS TO BE HELD ON ,

The Shareholder executing this Proxy appoints Thomas W. Marquis and Robert J. Marshall, and each of them, each with full power to appoint his substitute, attorneys and proxies to represent the Shareholder and to vote and act with respect to all shares of common stock, \$.01 par value per share, of VSI Holdings, Inc. ("VSI Holdings") that the Shareholder would be entitled to vote on all matters which come before the Special Meeting of Shareholders of VSI Holdings referred to above (the "Special Meeting") and at any adjournment(s) or postponement(s) of the Special Meeting.

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS OF VSI HOLDINGS, INC. IF THIS PROXY IS PROPERLY EXECUTED, THE SHARES OF VSI HOLDINGS COMMON STOCK REPRESENTED BY THIS PROXY WILL BE VOTED AS SPECIFIED. IF NO SPECIFICATION IS MADE, SUCH SHARES WILL BE VOTED FOR THE APPROVAL AND ADOPTION OF THE AGREEMENT AND PLAN OF MERGER DATED AS OF MARCH 24, 2001, BETWEEN VSI HOLDINGS AND SPX CORPORATION ("SPX"), AND ALL OF THE TRANSACTIONS CONTEMPLATED BY THAT AGREEMENT (INCLUDING, WITHOUT LIMITATION, THE BUSINESS COMBINATION OF VSI HOLDINGS AND SPX). THE SHARES OF VSI HOLDINGS COMMON STOCK REPRESENTED BY THIS PROXY WILL BE VOTED IN THE DISCRETION OF THE PROXIES ON ANY OTHER MATTERS WHICH MAY COME BEFORE THE SPECIAL MEETING.

(Continued and to be signed on reverse side.)

 Approval and adoption of the Agreement and Plan of Merger dated as of March 24, 2001, between VSI Holdings and SPX, and all of the transactions contemplated by that agreement (including, without limitation, the business combination of VSI Holdings and SPX).

YOUR BOARD OF DIRECTORS RECOMMENDS THAT YOU VOTE FOR THIS PROPOSAL.

[] FOR [] AGAINST [] ABSTAIN

In their discretion, the proxies are authorized to vote upon such other business as may properly come before the meeting.

The undersigned Shareholder hereby (i) revokes any and all proxies previously executed with respect to the Special Meeting, and (ii) acknowledges receipt of the Notice and Proxy Statement-Prospectus for the Special Meeting.

Dated:	_
2001 Signature	,
Title	
Signature	-

Please sign exactly as name appears on this proxy. When shares are held by joint tenants, both should sign. When signing as attorney, executor, administrator, trustee or guardian, please give full title as such. If a corporation, please sign in full corporate name by an authorized officer. If a partnership, please sign in partnership name by authorized person.

FORM OF ELECTION AND LETTER OF TRANSMITTAL

TO ACCOMPANY CERTIFICATES REPRESENTING SHARES OF COMMON STOCK, \$.01 PAR VALUE PER SHARE, OF

VSI HOLDINGS, INC.

PLEASE READ AND FOLLOW THE ACCOMPANYING INSTRUCTIONS CAREFULLY

THE EXCHANGE AGENT IS: EQUISERVE TRUST COMPANY, N.A.

BY MAIL:	BY HAND:	BY OVERNIGHT COURIER:
[Insert appropriate address]	[Insert appropriate address]	[Insert appropriate address]

For Confirmation Telephone:

TO BE EFFECTIVE, THIS FORM OF ELECTION AND LETTER OF TRANSMITTAL MUST BE RECEIVED BY THE EXCHANGE AGENT BY 5:00 P.M. EASTERN DAYLIGHT SAVINGS TIME , 2001 (OR, IF THE MEETING OF VSI HOLDINGS SHAREHOLDERS TO APPROVE THE MERGER IS POSTPONED, TWO BUSINESS DAYS BEFORE THE DATE OF THE MEETING), TOGETHER WITH (1) THE CERTIFICATE(S) REPRESENTING ALL SHARES OF VSI HOLDINGS COMMON STOCK TO WHICH THIS FORM OF ELECTION AND LETTER OF TRANSMITTAL RELATES OR (2) A PROPERLY COMPLETED GUARANTEE OF DELIVERY WITH RESPECT TO THAT CERTIFICATE(S). DELIVERY OF SHARES OF VSI HOLDINGS COMMON STOCK MAY ALSO BE MADE BY BOOK-ENTRY TRANSFER TO THE EXCHANGE AGENT'S ACCOUNT AT THE DEPOSITORY TRUST COMPANY ("DTC"). SEE GENERAL INSTRUCTION 15.

DELIVERY OF THIS FORM OF ELECTION AND LETTER OF TRANSMITTAL TO AN ADDRESS OTHER THAN THAT OF THE EXCHANGE AGENT WILL NOT CONSTITUTE A VALID DELIVERY TO THE EXCHANGE AGENT.

DESCRIPTION OF SHARES OF VSI HOLDINGS COMMON STOCK	SURRENDERED	
NAME(S) AND ADDRESS(ES) OF		
RECORD HOLDER(S) (AS SHOWN ON THE CERTIFICATE(S))*	CERTIFICATES BEING SURRENDERED (ATTACH ADDITIONAL LIST IF NECESSARY)	
	NUMBER OF SHARES REPRESENTED BY EACH CERTIFICATE (OR	
	CERTIFICATE COVERED BY A NUMBER(S)** GUARANTEE OF DELIVERY)	
	TOTAL CHARGO	
	TOTAL SHARES	

- * For a delivery using the guarantee of delivery procedures, exactly as name(s) will appear on the certificate(s) when delivered.
- ** Certificate numbers are not required if certificates will be delivered using the guarantee of delivery procedures.

Pursuant to the Agreement and Plan of Merger, dated as of March 24, 2001, by and between SPX Corporation and VSI Holdings, Inc., the undersigned surrenders to the exchange agent the certificate(s) representing all of the shares of VSI Holdings common stock, \$.01 par value per share, which the undersigned owns of record and elects, in the manner indicated below, to have the VSI Holdings common stock evidenced by that certificate(s) converted into the right to receive, for all of the undersigned's shares of VSI common stock, either:

- (A) All stock consideration in the form of 0.043 shares of SPX common stock per share of VSI Holdings common stock; or
- (B) Combination consideration consisting of: (1) 0.043 shares of SPX common stock per share of VSI Holdings common stock for 55% of the VSI Holdings shares held by the undersigned and (2) \$4.35 per share of VSI Holdings common stock for 45% of the VSI Holdings shares held by the undersigned, subject to adjustment as provided in Section 1.5 of the Agreement and Plan of Merger.

IF THE UNDERSIGNED DOES NOT MARK ONE OF THE ELECTION BOXES BELOW, THE UNDERSIGNED WILL BE DEEMED TO HAVE INDICATED NO PREFERENCE AS TO THE RECEIPT OF ALL STOCK CONSIDERATION OR A COMBINATION OF STOCK CONSIDERATION AND CASH CONSIDERATION AND THE UNDERSIGNED WILL AUTOMATICALLY RECEIVE ONLY SPX COMMON STOCK IN EXCHANGE FOR THE UNDERSIGNED'S VSI HOLDINGS COMMON STOCK.

The undersigned understands that this election is subject to certain terms, conditions and limitations explained in the merger agreement and described in the accompanying proxy statement-prospectus. A copy of the merger agreement is attached to the proxy statement-prospectus as Appendix A. These terms, conditions and limitations include, but are not limited to, the fact that pursuant to the allocation procedures described in the proxy statement-prospectus and set forth in the merger agreement, holders may be subject to an adjustment process in which a holder who elects the combination consideration of 45% cash and 55% stock for the holder's shares of VSI Holdings common stock receives a larger proportion of stock and a smaller proportion of cash. All elections are subject to the allocation procedures

set forth in Section 1.5 of the merger agreement. The allocation procedures are described under the caption "Election of Form of Consideration; Conversion of Shares" in the proxy statement-prospectus. The undersigned is urged to read the merger agreement and the proxy statement-prospectus in their entirety before completing this form of election and letter of transmittal.

The undersigned understands that the definitive terms under which the merger will be effected in accordance with the merger agreement, including the amount and form of consideration to be received by holders of VSI Holdings common stock, the effect of this form of election and letter of transmittal, and certain conditions to the consummation of the merger, are summarized in the proxy statement-prospectus and all of those definitive terms and conditions are described in full in the merger agreement. The undersigned also understands that different tax consequences may be associated with each of the election options, and the undersigned is aware that those consequences are summarized in general terms in the proxy statement-prospectus section entitled "Certain Federal Income Tax Consequences." The undersigned hereby makes the following election for all of the undersigned's shares of VSI Holdings common stock owned of record and surrendered with this form of election and letter of transmittal:

ELECTION

Check one of the boxes below:

[] All Stock Election

[] Combination Election (Cash and Stock)

- (A) None of the above elections is made or more than one of the above elections is made;
- (B) The undersigned fails to follow the instructions on this form of election and letter of transmittal (including failure to submit share certificate(s), confirmation of a book-entry transfer of shares at DTC or a guarantee of delivery) or otherwise fails to properly make an election;
- (C) A completed form of election and letter of transmittal (including submission of the holder's share certificate(s), confirmation of a book-entry transfer of the shares at DTC or a guarantee of delivery) is not received by the exchange agent by 5:00 p.m. Eastern Daylight Savings time on (or, if the meeting of VSI Holdings shareholders to approve the merger is postponed, two business days before the date of the meeting);
- (D) The undersigned returns this form of election and letter of transmittal with a guarantee of delivery but does not deliver the share certificate(s) representing the shares in respect of which the election is being made or confirmation of a book-entry transfer of the shares at DTC within the time set forth on the guarantee of delivery.

In order to receive the merger consideration, this form of election and letter of transmittal must be (1) completed and signed in the space in the box labeled "Shareholder(s) Sign Here" and on the Substitute Form W-9 and (2) mailed or delivered with the holder's share certificate(s) or a guarantee of delivery to the exchange agent at the address (or the facsimile number solely with respect to a guarantee of delivery) set forth above. In order to properly make an election, the form of election and letter of transmittal and other required documents must be received by the exchange agent prior to the election deadline. SPX will notify the exchange agent of any extension of the election deadline by oral notice (promptly confirmed in writing) or written notice and will make a press release or other public announcement of that extension prior to 9:00 a.m., Eastern Daylight Savings time, on the next business day following the previously scheduled election deadline.

If the undersigned is acting in a representative or fiduciary capacity for a particular beneficial owner, the undersigned hereby certifies that this form of election and letter of transmittal covers all of the shares of VSI Holdings common stock owned of record by the undersigned in a representative or fiduciary capacity for that particular beneficial owner.

The undersigned hereby acknowledges receipt of the proxy statement-prospectus and agrees that all elections, instructions and orders in this form of election and letter of transmittal are subject to the terms and conditions of the merger agreement, the proxy statement-prospectus and the instructions applicable to this form of election and letter of transmittal. The undersigned represents and warrants that the undersigned is, as of the date of this form of election and letter of transmittal, and will be, as of the effective time of the merger, the record holder of the shares of VSI Holdings common stock represented by the certificate(s) being surrendered, with good title to those shares of common stock and full power and authority (1) to sell, assign and transfer those common shares free and clear of all liens, restrictions, charges and encumbrances, and not subject to any adverse claims and (2) to make the election indicated herein. The undersigned will, upon request, execute any additional documents necessary or desirable to complete the surrender and exchange of those shares of VSI Holdings common stock.

The undersigned hereby irrevocably appoints the exchange agent as agent of the undersigned, to effect the exchange as described in the merger agreement and the instructions in this form of election and letter of transmittal. The undersigned hereby authorizes and instructs the exchange agent to deliver the certificate(s) covered by this form of election and letter of transmittal, and to receive on the undersigned's

behalf, in exchange for the VSI Holdings common shares represented by that certificate(s), any check and/or any certificate(s) for SPX common stock issuable to the undersigned. Furthermore, the undersigned authorizes the exchange agent to follow any election and to rely upon all representations, certifications and instructions contained in this form of election and letter of transmittal. All authority conferred or agreed to be conferred in this form of election and letter of transmittal is binding upon the successors, assigns, heirs, executors, administrators and legal representatives of the undersigned and is not affected by, and survives, the death or incapacity of the undersigned.

Record holders of shares of VSI Holdings common stock who are nominees, trustees or who hold shares in other representative capacities may submit a separate form of election and letter of transmittal for each beneficial holder for whom that record holder is a nominee, trustee or representative; provided, however, that at the request of SPX, that record holder must certify to the satisfaction of SPX that the record holder holds those shares as nominee for the beneficial owner(s) thereof. Each beneficial owner for whom a form of election and letter of transmittal is submitted will be treated as a separate holder of shares of VSI Holdings common stock, subject to the provisions concerning joint elections.

Completing and returning this form of election and letter of transmittal does not have the effect of casting a vote with respect to approval of the merger agreement and approval of the related transactions at the special meeting of shareholders of VSI Holdings.

GUARANTEE OF DELIVERY (TO BE USED ONLY AS TO CERTIFICATES NOT TRANSMITTED WITH THIS FORM OF ELECTION AND LETTER OF TRANSMITTAL) (SEE INSTRUCTION 15)

The undersigned (check applicable box), [] a member of a registered national securities exchange, [] a member of the National Association of Securities Dealers, Inc., or [] a commercial bank or trust company in the United States, guarantees to deliver to the exchange agent either all of the certificate(s) for shares of VSI Holdings common stock to which this form of election and letter of transmittal relates, or such of those certificates as are identified below, duly endorsed in blank or otherwise in form acceptable for transfer on the books of VSI Holdings, no later than 5:00 p.m., Eastern Daylight Savings time, on, 2001 (or, if the meeting of VSI Holdings shareholders to approve the merger is postponed, two business days before the date of the meeting). SHARES REPRESENTED BY EACH CERTIFICATE CERTIFICATE NUMBER ______ - ---------------(FIRM -- PLEASE PRINT) (AUTHORIZED SIGNATURE) DATED: -----(ADDRESS) -----

(DAYTIME AREA CODE AND TELEPHONE NUMBER)

SPECIAL PAYMENT INSTRUCTIONS

(SEE INSTRUCTIONS 4 & 6)

To be completed ONLY if the check is to be made payable to and/or the certificate(s) for SPX common stock is to be issued in the name of someone other than the record holder(s) of the shares of VSI Holdings common stock or the name of the record holder(s) needs to be corrected or changed.

Issue: [] Certificate [] Check to:
(PLEASE PRINT) Name:
Address:
(INCLUDE ZIP CODE)
TAX IDENTIFICATION NUMBER OR SOCIAL SECURITY NUMBER:

SPECIAL DELIVERY INSTRUCTIONS

(SEE INSTRUCTIONS 4 & 6)

Complete ONLY if the check and/or the certificate(s) for SPX common stock are to be issued in the name of the record holder(s) of the shares of VSI Holdings common stock but are to be sent to another person or to an address other than as set forth beneath the record holder's signature on this form of election and letter of transmittal.

Check or certificate(s) for SPX common stock to be delivered to:*

(PLEASE PEN Name:	RINT)					
Address:						
(INCLUDE 2		 	 	 	 	

*PLEASE ATTACH ADDITIONAL SHEETS IF NECESSARY.

SHAREHOLDER(S) SIGN HERE (ALSO COMPLETE SUBSTITUTE FORM W-9 BELOW)

Please sign exactly as your name(s) appear(s) on your certificate(s). If this is a joint election, each person covered by this form of election and letter of transmittal must sign personally.

letter of transmittal must sign personally.
A check(s) or certificate(s) for SPX common stock will be issued only in the name of the person(s) submitting this form of election and letter of transmittal and will be mailed to the address indicated beneath the person's signature unless the special delivery or special payment instructions are completed.
(SIGNATURE(S) OF OWNER(S) SEE INSTRUCTION 5)
Dated: , 2001
,
Social Security or Other Tax Identification Number(s):
If signature is by a person(s) other than the record holder(s) and in the capacity of trustee, executor, administrator, guardian, attorney-in-fact, officer of a corporation or any other persons(s) acting in a fiduciary or representative capacity, please provide the following information. See Instruction 5.
(PLEASE PRINT)
Name:
Capacity:
Address:
/TNCLUDE ZTD CODE\
(INCLUDE ZIP CODE)
Daytime Telephone Number:
SIGNATURE GUARANTEE (IF REQUIRED BY INSTRUCTION 4 OR 6) APPLY SIGNATURE GUARANTEE MEDALLION BELOW
The undersigned hereby guarantees the signature(s) which appear(s) on this form of election and letter of transmittal.
Dated:, 2001
NAME OF ELIGIBLE INSTITUTION ISSUING GUARANTEE
NOTE: THE SURRENDERED CERTIFICATES DO NOT REQUIRE AN ENDORSEMENT OR A

NOTE: THE SURRENDERED CERTIFICATES DO NOT REQUIRE AN ENDORSEMENT OR A GUARANTEE OF SIGNATURE IF THE CHECK AND/OR CERTIFICATE REPRESENTING SPX COMMON STOCK IS TO BE ISSUED IN EXACTLY THE NAME OF THE RECORD HOLDER AS INSCRIBED ON THE SURRENDERED CERTIFICATES OF VSI HOLDINGS COMMON STOCK.

SUBSTITUTE FORM W-9							
PAYER:	NAME (IF JOINT NAMES, LIST FIRST AND CIRCLE THE NAME OF THE PERSON OR ENTITY WHOSE NUMBER YOU ENTER IN PART 1 BELOW): ADDRESS: CITY, STATE AND ZIP CODE: LIST ACCOUNT NUMBER(S) HERE (OPTIONAL):						
	PART 1 Please provide your Taxpayer						
	Identification Number (TIN) in the box at right and certify by signing and dating below.	Social Security Number or Employer Identification Number (If awaiting TIN write "Applied for")					
DEPARTMENT OF THE TREASURY INTERNAL REVENUE SERVICE	PART 2 Check the box if you are subject to backup withholding because: (1) You have been notified by the Internal Revenue Service that you are subject to backup withholding as a result of failure to report all interest or dividends [], or (2) The Internal Revenue Service has not notified you that you are no longer subject to						
	backup withholding []						
PAYOR'S REQUEST FOR TAXPAYER IDENTIFICATION NUMBER ("TIN")	ON CONTRACTOR OF THE PROPERTY						
	Certification Under the penalties of perjury, I certify that the information provided on this form is true, correct and complete.						
	SIGNATURE:	DATE:					

NOTE: FAILURE TO COMPLETE AND RETURN THIS FORM MAY RESULT IN BACKUP U.S. FEDERAL INCOME TAX WITHHOLDING OF 31% OF ANY PAYMENTS MADE TO YOU PURSUANT TO THE MERGER AGREEMENT. BACKUP WITHHOLDING IS NOT AN ADDITIONAL TAX BUT MERELY AN ADVANCE PAYMENT, WHICH MAY BE REFUNDED TO THE EXTENT IT RESULTS IN OVERPAYMENT OF TAX. CERTAIN PERSONS ARE EXEMPT FROM BACKUP WITHHOLDING. PLEASE REVIEW THE ATTACHED "GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9" FOR ADDITIONAL DETAILS.

INSTRUCTIONS

This form of election and letter of transmittal should be completed and submitted to the exchange agent prior to the election deadline by those holders of shares of VSI Holdings common stock desiring to make an election. Holders of VSI Holdings common stock who do not complete and submit this form of election and letter of transmittal prior to the election deadline cannot make an election. They will be deemed to have made a non-election and will automatically receive SPX common stock in exchange for their VSI Holdings common stock.

Until a record holder's share certificate(s) or confirmation of a book-entry transfer of the shares at DTC is received by the exchange agent at the address (or the facsimile number solely with respect to a guarantee of delivery) set forth on the front of this form of election and letter of transmittal, together with any other documents the exchange agent may require, and until these documents are processed for exchange by the exchange agent, the holder will not receive any certificate representing the stock consideration and any check representing any cash consideration or the check representing cash in lieu of fractional shares (if any) in exchange for the holder's share certificate(s).

VSI Holdings shareholders will not receive interest on any cash consideration they receive. No dividend or distribution with respect to shares of SPX common stock will be payable on or with respect to any fractional share, and fractional share interests will not entitle their owners to vote or to any other rights of a SPX stockholder. In lieu of any fractional share of SPX common stock, SPX will pay to each former VSI Holdings shareholder who otherwise would be entitled to receive a fractional share of SPX common stock an amount in cash equal to \$101.17 multiplied by the fractional share of SPX common stock to which the holder would otherwise be entitled. SPX will not pay declared dividends or distributions to persons entitled to receive certificates for shares of SPX common stock until they surrender their certificates for SPX common stock, at which time SPX will pay the dividends or distributions. In no event will the persons entitled to receive these dividends be entitled to interest on them.

1. TIME IN WHICH TO MAKE AN ELECTION

For an election to be validly made, the exchange agent must receive, at the address set forth on the front of this form of election and letter of transmittal, prior to the election deadline, this form of election and letter of transmittal, properly completed and executed, and accompanied by all of the certificates representing the shares of VSI Holdings common stock owned by that holder, confirmation of a book-entry transfer of the shares at DTC or by a guarantee of delivery. Any shareholder whose form of election and letter of transmittal and certificates (or confirmation of a book-entry transfer of the shares at DTC or guarantee of delivery) are not so received will be deemed to have made a non-election. In the event shares of VSI Holdings common stock covered by a guarantee of delivery are not received (or are not the subject of a confirmation of a book-entry transfer of the shares at DTC) by the date set forth in the guarantee of delivery, unless that deadline has been extended in accordance with the terms of the merger agreement, the holder will be deemed to have made a non-election.

2. CHANGE OR REVOCATION OF ELECTION

Any holder of VSI Holdings common stock who has made an election by submitting a form of election and letter of transmittal to the exchange agent may at any time prior to the election deadline change that holder's election by submitting to the exchange agent a revised form of election and letter of transmittal, properly completed and signed, that is received by the exchange agent prior to the election deadline. Any holder of VSI Holdings common stock may, at any time prior to the election deadline, revoke his election by written notice received by the exchange agent prior to the election deadline or by withdrawal prior to the election deadline of his certificates for VSI Holdings common stock, or of the guarantee of delivery of such certificates, previously deposited with the exchange agent. All elections will be revoked automatically if the exchange agent is notified in writing by SPX or VSI Holdings that the merger agreement has been terminated. Any holder of VSI Holdings common stock who had deposited certificates for VSI Holdings common stock with the exchange agent will have the right to withdraw the

certificates for VSI Holdings common stock by written notice received by the exchange agent. Elections as to the desired form of consideration may not be changed after the merger is effective.

NOMINEES

Record holders of VSI Holdings common stock who are nominees, trustees or who hold shares of VSI Holdings common stock in other representative capacities may submit a separate form of election and letter of transmittal for each beneficial owner for whom the record holder is a nominee; provided, however, that the record holder certifies that each form of election and letter of transmittal covers all VSI Holdings stock held by the record holder for a particular beneficial owner. Each beneficial owner for which a form of election and letter of transmittal is submitted will be treated as a separate holder of VSI Holdings common stock.

4. GUARANTEE OF SIGNATURES

No signature guarantee is required on this form of election and letter of transmittal if this form of election and letter of transmittal is signed by the record holder(s) of the VSI Holdings common stock tendered with this form of election and letter of transmittal, and the certificate representing SPX common stock and/or the check, if applicable, are to be issued to that record holder(s) without any correction or change in the name of the record holder(s). IN ALL OTHER CASES, ALL SIGNATURES ON THIS FORM OF ELECTION AND LETTER OF TRANSMITTAL MUST BE GUARANTEED. All signatures required to be guaranteed in accordance with these instructions must be guaranteed by a bank, broker or other institution that is a member of a Medallion Signature Guaranty Program. Public notaries cannot execute acceptable guarantees of signatures.

- 5. SIGNATURES ON FORM OF ELECTION AND LETTER OF TRANSMITTAL, STOCK POWERS AND ENDORSEMENTS
- (A) If this form of election and letter of transmittal is signed by the record holder(s) of the certificate(s) being tendered without any alteration, variation, correction or change in the name of the record holder(s), the signature(s) must correspond exactly with the name(s) as written on the face of the certificate(s) without any change whatsoever. In the event the name of the record holder(s) needs to be corrected or has changed (by marriage or otherwise), see Instruction 6.
- (B) If any shares of VSI Holdings common stock being tendered are held of record by two or more joint holders, each of the joint holders must sign this form of election and letter of transmittal.
- (C) If this form of election and letter of transmittal is signed by the record holder(s) of the share certificate(s) listed and transmitted by this form of election and letter of transmittal, no endorsements of the certificate(s) or separate stock powers are required.
- (D) If any surrendered shares of VSI Holdings common stock are registered in different names on several share certificates, it will be necessary to complete, sign and submit as many separate forms of election and letters of transmittal as there are different registrations of share certificates.
- (E) If this form of election and letter of transmittal is signed by a person(s) other than the record holder(s) of the certificate(s) listed, the certificate(s) must be endorsed or accompanied by appropriate stock powers, in either case signed exactly as the name of the record holder(s) appears on the certificate(s). Signatures on the certificate or stock powers must be guaranteed. See Instruction 4.
- (F) If this form of election and letter of transmittal or any share certificate(s) or stock power(s) is signed by a person(s) other than the record holder(s) of the share certificate(s) listed and the signer(s) is acting in the capacity of trustee, executor, administrator, guardian, attorney-in-fact, officer of a corporation or any other person(s) acting in a fiduciary or representative capacity, that person(s) must so indicate when signing and must submit proper evidence satisfactory to the exchange agent of authority to so act.

6. SPECIAL PAYMENT AND DELIVERY INSTRUCTIONS

Unless instructions to the contrary are given in the box entitled "Special Payment Instructions" or the box entitled "Special Delivery Instructions, certificate representing SPX common stock and/or the check to be distributed upon the surrender of the shares of VSI Holdings common stock pursuant to this form of election and letter of transmittal will be issued in the name and mailed to the address of the record holder(s) indicated in the box entitled "Description of Shares of VSI Holdings Common Stock Surrendered." If the certificate and/or check are to be issued in the name of a person(s) other than the record holder(s) or if the name of the record holder(s) needs to be corrected or changed (by marriage or otherwise), the box entitled "Special Payment Instructions" must be completed. If the certificate and/or check are to be sent to a person(s) other than the record holder(s) or to the record holder(s) at an address other than that shown in the box entitled "Description of VSI Holdings Common Shares Surrendered," the box entitled "Special Delivery Instructions" must be completed. If the box entitled "Special Payment Instructions" is completed, or the box entitled "Special Delivery Instructions" is completed other than for the sole purpose of changing the address of the record holder(s), the signature(s) of the person(s) signing this form of election and letter of transmittal must be guaranteed. See Instruction 4.

7. IMPORTANT INFORMATION REGARDING 31% BACKUP WITHHOLDING

Under U.S. federal income tax law, the holder of VSI Holdings common stock must report and certify his or her social security or other taxpayer identification number and further certify that the holder is not subject to backup withholding due to notified underreporting. Failure to complete the Substitute Form W-9 above could result in certain penalties as well as backup withholding of 31% of payments due to the holder. See the attached "Guidelines for Certification of Taxpayer Identification Number on Substitute Form W-9" for additional instructions.

INADEQUATE SPACE

If there is inadequate space to complete any box or to sign this form of election and letter of transmittal, the information or signatures required to be provided must be set forth on additional sheets substantially in the form of the corresponding portion of this form of election and letter of transmittal and attached to this form of election and letter of transmittal.

9. INDICATION OF CERTIFICATE NUMBERS AND SHARES

This form of election and letter of transmittal should indicate the certificate number(s) of the certificate(s) representing the shares of VSI Holdings common stock covered by this form of election and letter of transmittal and the number of shares represented by each certificate.

10. METHOD OF DELIVERY

The method of delivery of all documents is at the option and risk of the holder of VSI Holdings common stock. If delivery is by mail, the use of registered mail, with return receipt requested, properly insured, is strongly recommended. A return envelope is enclosed. It is suggested that this form of election and letter of transmittal be hand delivered or mailed to the exchange agent as soon as possible. Delivery of the documents will be deemed effective, and risk of loss and title with respect thereto will pass, only when materials are actually received by the exchange agent.

11. PAYMENT WILL BE MADE BY A SINGLE CHECK OR CERTIFICATE

Normally, a single check and/or a single certificate representing SPX common stock will be issued; however, if for tax purposes or otherwise a holder wishes to have the certificates issued in particular denominations, explicit written instructions should be provided to the exchange agent. Holders participating in a joint election will receive a single check or share certificate for the holders' shares of VSI Holdings common stock.

12. LOST CERTIFICATES

If any certificate representing VSI Holdings common stock has been lost, stolen or destroyed, the holder should notify the exchange agent in writing and await instructions as to how to proceed.

13. NON-CONSUMMATION OF MERGER

Consummation of the merger is subject to the required approval of the shareholders of VSI Holdings and to the satisfaction of certain other conditions. No payments related to any surrender of the certificate(s) will be made prior to the consummation of the merger, and no payments will be made to shareholders if the merger agreement is terminated. If the merger agreement is terminated, all elections will be void and of no effect and certificates submitted to the exchange agent will be returned as soon as practicable to the persons submitting them.

14. VOTING RIGHTS AND DIVIDENDS

Holders of VSI Holdings common stock will continue to have the right to vote and to receive any dividends paid on all VSI Holdings common stock deposited by them with the exchange agent until the merger becomes effective.

15. GUARANTEE OF DELIVERY

Holders of VSI Holdings common stock whose certificates are not immediately available or who cannot deliver their certificates and all other required documents to the exchange agent or cannot complete the procedure for delivery of VSI Holdings common stock by book-entry transfer into the exchange agent's account at DTC prior to the election deadline, may deliver their VSI Holdings common stock by properly completing and duly executing a guarantee of delivery if (1) the guarantee of delivery is made by or through a member of a registered national securities exchange, a member of the National Association of Securities Dealers, Inc. or by a commercial bank or trust company in the United States, (2) prior to the election deadline, the exchange agent receives a properly completed and duly executed guarantee of delivery, as provided herein, together with a properly completed and duly executed form of election and letter of transmittal and any other documents required by this form of election and letter of transmittal; and (3) the certificates for all the VSI Holdings common stock covered by the guarantee of delivery, in proper form for transfer (or confirmation of a book-entry transfer of such VSI Holdings common stock into the exchange agent's account at DTC), are received by the exchange agent within the time set forth on the guarantee of delivery. If the above requirements are not satisfied in a timely manner, the holder will be deemed to have made a non-election.

16. CONSTRUCTION

All elections will be considered in accordance with the terms and conditions of the merger agreement.

All questions with respect to the form of election and letter of transmittal (including, without limitation, questions relating to the timeliness, effectiveness or revocation of any election) will be resolved by SPX in its sole discretion and such resolution will be final and binding.

With the consent of SPX, the exchange agent may (but is not required to) waive any immaterial defects or variances in the manner in which the form of election and letter of transmittal has been completed and submitted so long as the intent of the holder of shares of VSI Holdings common stock submitting the form of election and letter of transmittal is reasonably clear. The exchange agent is under no obligation to provide notification of any defects in the deposit and surrender of any certificate(s) formerly representing VSI Holdings common stock.

17. MISCELLANEOUS

No fraction of a share of SPX common stock will be issued upon the surrender for exchange of a certificate(s) for VSI Holdings common stock. Instead, the undersigned will receive a cash payment equal to the amount derived by multiplying the fractional share by \$101.17.

COMPLETING AND RETURNING THIS FORM OF ELECTION AND LETTER OF TRANSMITTAL DOES NOT HAVE THE EFFECT OF CASTING A VOTE WITH RESPECT TO APPROVAL OF THE MERGER AGREEMENT AND APPROVAL OF THE RELATED TRANSACTIONS AT THE SPECIAL MEETING OF SHAREHOLDERS OF VSI HOLDINGS).

18. QUESTIONS AND REQUESTS FOR INFORMATION

Questions and requests for information or assistance relating to this form of election and letter of transmittal should be directed to EquiServe Trust Company, N.A., as exchange agent, at the following numbers: (1) for banks or brokers: and (2) for all other stockholder requests: . Additional copies of the proxy statement and this form of election and letter of transmittal may be obtained from EquiServe Trust Company, N.A. by calling the telephone number set forth in the preceding sentence.

GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9

GUIDELINES FOR DETERMINING THE PROPER IDENTIFICATION NUMBER TO GIVE THE PAYER.--Social Security numbers have nine digits separated by two hyphens: i.e., 000-00-0000. Employer identification numbers have nine digits separated by only one hyphen: i.e., 00-0000000. The table below will help determine the number to give to the payer.

FOR THIS TYPE OF ACCOUNT: SOCIAL SECURITY

GIVE THE NUMBER OF - -

An individual's account

Two or more individuals (joint

account)

Husband and wife (joint account)

4. Custodian account of a minor (Uniform Gift to Minors Act)

Adult and minor (joint account)

6. Account in the name of guardian or committee for a designated ward,

minor, or incompetent person 7. a. The usual revocable savings trust account (grantor is also

trustee) b. So-called trust account that is The actual owner(1)

not a legal or valid trust under the State law

The individual The actual owner of the account or, if combined funds, any one of the other

individuals(1) The actual owner of the account or, if joint funds, either person(1) The minor(2)

The adult or, if the minor is the only contributor, the minor(1) The ward, minor, or incompetent person(3) The grantortrustee(1)

FOR THIS TYPE OF ACCOUNT: IDENTIFICATION

GIVE THE EMPLOYER NUMBER OF--

- ------

8. Sole proprietorship account

A valid trust, estate, or pension

The Owner(4) The legal entity (Do not provide the identification number of the personal representative or trustee unless the legal entity itself is not designated in the account title)(5) The corporation The organization

10. Corporate account 11.

Religious, charitable, or educational organization account

Partnership account

13. Association, club or other taxexempt organization

14. A broker or registered nominee

15. Account with the Department of Agriculture in the name of a public entity (such as State or local government, school district, or prison) that receives agricultural program payments

The partnership The organization

The broker or nominee The public entity

(1) List first and circle the name of the person whose number you provide.

(2) Circle the minor's name and furnish the minor's social security number.

(3) Circle the ward's, minor's or incompetent person's name and provide that person's social security number.

(4) Show the name of the owner.

(5) List first and circle the name of the legal trust, estate, or pension trust.

NOTE: If no name is circled when there is more than one name, the number will be considered to be that of the first name listed.

GUIDELINES FOR CERTIFICATION OF TAXPAYER IDENTIFICATION NUMBER ON SUBSTITUTE FORM W-9

OBTAINING A NUMBER

If you do not have a taxpayer identification number or you do not know your number, you must file Form SS-5, Application for a Social Security Number Card, or Form SS-4, Application for an Employer Identification Number. These forms may be obtained at your local office of the Social Security Administration or the Internal Revenue Service. You may also obtain Form SS-5 on the Internet at http://www.ssa.gov or Form SS-4 at http://www.irs.gov.

PAYEES EXEMPT FROM BACKUP WITHHOLDING

Payees specifically exempted from backup withholding on ALL payments include the following:

- A corporation.
- A financial institution.
- An organization exempt from tax under Section 501(a) of the Internal Revenue Code of 1986, as amended, or an individual retirement plan.
- The United States or any agency or instrumentality thereof. A State, the District of Columbia, a possession of the United States, or any subdivision or instrumentality thereof.
- A foreign government, a political subdivision of a foreign government, or any agency or instrumentality thereof.
- An international organization or any agency or instrumentality thereof.
- A registered dealer in securities or commodities registered in the U.S. or a possession of the U.S.
- A real estate investment trust.
- A common trust fund operated by a bank under Section 584(a) of the Code.
- An exempt charitable remainder trust, or a non-exempt trust described in section 4947(a)(1) of the Code.
- An entity registered at all times under the Investment Company Act of 1940.
- A foreign central bank of issue.

Payments of dividends and patronage dividends not generally subject to backup withholding include the following:

- Payments to nonresident aliens subject to withholding under section 1441 of the Code.
- Payments to partnerships not engaged in a trade or business in the United States and which have at least one nonresident partner.
- Payments of patronage dividends not paid in money.
- Payments made by certain foreign organizations.

Payments of interest not generally subject to backup withholding include:

- Payments of interest on obligations issued by individuals.
 Payments of tax-exempt interest (including exempt-interest dividends under section 852) of the Code.
- Payments described in section 6049(b)(5) of the Code to non-resident aliens.
- Payments on tax-free covenant bonds under section 1451 of the Code.
- Payments made by certain foreign organizations. Payments made to a nominee.

NOTE: You may be subject to backup withholding if this interest is \$600 or more and is paid in the course of the payer's trade or business and you have not provided your correct taxpayer identification number to the payer.

Exempt payees described above must still complete the Substitute Form W-9 $\,$ enclosed with this form of election and letter of transmittal to avoid possible erroneous backup withholding. FILE SUBSTITUTE FORM W-9 WITH THE PAYER, REMEMBERING TO CERTIFY YOUR TAXPAYER IDENTIFICATION NUMBER ON PART III OF THE FORM, WRITE "EXEMPT" ON THE FACE OF THE FORM, AND SIGN AND DATE THE FORM AND RETURN IT TO THE PAYER. IF YOU ARE A NON-RESIDENT ALIEN OR A FOREIGN ENTITY NOT SUBJECT TO BACKUP WITHHOLDING GIVE THE PAYER A COMPLETED FORM W-8 CERTIFICATE OF FOREIGN STATUS.

Payments that are not subject to information reporting are also not subject to backup withholding. For details, see sections 6041, 6041A, 6042, 6044, 6045, 6049, 6050A, and 6050N of the Code and their regulations.

PRIVACY ACT NOTICE -- Section 6109 of the Code requires most recipients of dividends, interest, or other payments to give taxpayer identification numbers to payers who must report the payments to the IRS. The IRS uses the numbers for identification purposes and to help verify the accuracy of your tax return. Payers must be given the numbers whether or not recipients are required to file a tax return. The IRS may also provide this information to the Department of Justice and state and local authorities for tax administration and enforcement purposes. Payers must generally withhold 31% of taxable interest, dividends, and certain other payments to a payee who does not furnish a taxpayer identification number to a payer. Certain penalties may also apply.

PENALTIES

- (1) PENALTY FOR FAILURE TO FURNISH TAXPAYER IDENTIFICATION NUMBER -- If you fail to furnish your taxpayer identification number to a payer, you are subject to a penalty of \$50 for each such failure unless your failure is due to reasonable cause and not to willful neglect.
- (2) CIVIL PENALTY FOR FALSE INFORMATION WITH RESPECT TO WITHHOLDING -- If you make a false statement with no reasonable basis that results in no imposition of backup withholding, you are subject to a penalty of \$500.

(3) CRIMINAL PENALTY FOR FALSIFYING INFORMATION -- Falsifying certifications or affirmations may subject you to criminal penalties including fines and/or imprisonment.

FOR ADDITIONAL INFORMATION CONTACT YOUR TAX CONSULTANT OR THE INTERNAL REVENUE SERVICE.